Copyright 2011 Michael Dillon Brennan

INTEGRATED PROJECT DELIVERY: A NORMATIVE MODEL FOR VALUE CREATION IN COMPLEX MILITARY MEDICAL PROJECTS

 $\mathbf{B}\mathbf{Y}$

MICHAEL DILLON BRENNAN

DISSERTATION

Submitted in partial fulfillment of the requirements for the degree of Doctor of Philosophy in Architecture in the Graduate College of the University of Illinois at Urbana-Champaign, 2011

Urbana, Illinois

Doctoral Committee:

Professor Michael K. Kim, Chair Professor Gaines B. Hall Associate Professor Abbas Aminmansour Associate Professor Liang Y. Liu

ABSTRACT

Integrated Project Delivery (IPD) is a collaborative project delivery process that reduces waste and maximizes value creation by harnessing the expertise and experience of all relevant stakeholders throughout the design and construction process of a given project. The owner, designer, builder, and user integrate people, systems, business structures and practices into a project team to optimize collective value creation while maintaining balance of individual benefits. The research utilizes a Delphi survey method to (1) select and assess the relative importance of critical success factors for value-creation, and (2) use the selected factors as a basis for evaluating the effectiveness of three project delivery methods: Design-Bid-Build, Design-Build, and Integrated Project Delivery. Key concepts for the theoretical framework of the research include: a macro-viewpoint understanding of project delivery must be adopted to define and measure value-creation; a significant relationship exists between project goals and value-creation; perceptions of relative importance and success differ within the stakeholder groups, and Integrated Project Delivery logically provides a normative model to optimize the collaborative processes to account for the different stakeholder perceptions and provide maximum benefit of all. The research and Delphi panel of experts is centered on the U.S. Army's Integrated Design-Bid-Build (IDBB) pilot-projects which for the first time test Integrated Project Delivery on a large scale within Military Construction (MILCON) program.

ii

To Richard Ray Dickason Jr., M.D., Ph.D. ...until we meet again, May God hold you in the palm of His hand. Vita, Dulcedo, Spes.

ACKNOWLEDGEMENTS

Professor Kim has been a wonderful mentor who planted the seed of knowledge which started this academic journey. I am grateful for his dedication and the countless hours of face-toface counsel he has provided in the development of this dissertation. Professors Aminmansour, Hall and Liu have been instrumental is shaping my research and providing encouragement to pursue new ideas. The committee has provided the motivation and focus necessary to complete this daunting task in the short time allotted to me by the Army.

I greatly appreciate all the dedicated professionals who participated in the Delphi survey for taking the time out of their busy schedules to assist in this research study. Your expertise and vast experience contributed greatly to this richness of this research project.

I would like to thank the U.S. Army Surgeon General and the U.S. Army Medical Command for sponsoring my doctoral research, and Colonel Rick Bond for his endorsement as Commander of the U.S. Army Health Facility Planning Agency to attend this program. Colonel Thom Kurmel, D.Des. and Colonel Dale Brown, PhD, my role models for critical thinking in the context of military leadership... I thank them for many years of leadership, sage guidance, and friendship.

Professors emeriti Walt Lewis and Dick Betts for providing much needed encouragement and friendship. I also thank Doug Sturgeon, a brother-in-arms and valued colleague during my studies at the University of Illinois for his friendship and support.

My dear departed friend Rich, for whom this dissertation is dedicated, for convincing me not to abandon my academic dream before it even started. In 1987 he recognized the potential I could not see at the time, and I am forever grateful.

iv

Most importantly I thank my family for love, support and encouragement. Patrick (Dad) Brennan and Uncle John-Paul Brennan provide much needed proof reading and constructive criticism along the way. I thank my mother, Katie Brennan, for the comforting over-watch, and wish you were here to share this milestone in my life. To my wonderful children Olivia, Máire, Séamus, and Ruari, I love you and appreciate your patience throughout this research project. To my beautiful wife Mia, I could not have done this without your unconditional love, support and encouragement. I love you and look forward to wherever our journey leads us together.

Finally, to my brothers and sisters in arms, the service members and veterans of the United States military, I am honored to be counted among you. I thank my veteran friends at the University of Illinois and in Champaign for their service, and greatly appreciate their comradery.

TABLE OF CONTENTS

JST OF ABBREVIATIONS	xi
CHAPTER 1. INTRODUCTION	1
1.1. Problem Statement	1
1.2. Research Objectives	5
1.3. Organization of Dissertation	6
CHAPTER 2. INTEGRATED PROJECT DELIVERY	7
2.1. IPD Overview	7
2.2. Military Construction and Integrated Project Delivery	12
2.2.1. Public Sector versus Private Sector	12
2.2.2. Department of Defense Medical Military Construction (MED MILCON)	14
2.2.3. Integrated Design-Bid-Build (IDBB)	15
2.3. Multi-party IPD Contract	20
2.3.1. Financial Incentive Structure	21
2.3.2. Increased Revenue Opportunity	23
2.3.3. Contingency Structure	23
2.3.4. Specification Philosophy	24
2.3.5. Contractual Incentives for Builder to Participate in IPD	24
2.3.6. Overcoming Resistance to Change	26
2.4. Significance	27
2.4.1. Equitable Risk Allocation & Dispute Resolution	28
2.4.2. Maximize Collaboration and Open Communication	31
2.4.3. Promote Sustainable-Building Design, Construction and Operations	31
2.5. Arguments for IPD	36
2.5.1. The Owner	38
2.5.2. The Designer	42
2.5.3. The Builder	42
2.5.4. The User	43
2.6. Need for Study	43

CHAPTER 3. LITERATURE REVIEW	46
3.1. Historical Overview	46
3.2. Literature Specific to Integrated Project Delivery	46
3.3. Conclusions from Literature	50
CHAPTER 4. THEORETICAL FRAMEWORK	52
4.1. Introduction	52
4.2. Relationship between Goals and Value-Creation	52
4.3. Goals Define Value-Creation	57
4.4. Macro Viewpoint	61
4.5. Research Concept:	64
 4.6. IPD: Logical Normative Model	69 69 70 71 71 72
CHAPTER 5. IDBB IN PRACTICE	73
5.1. Introduction	73
5.2. Time performance	74
5.3. Quality Assessment: Fort Belvoir Community Hospital	75
5.4. NGA Campus East Lessons Learned	76
5.5. Fort Belvoir Community Hospital Lessons Learned	76
5.6. Improving on IDBB	77
CHAPTER 6. RESEARCH METHOD	80
6.1. Introduction	80
6.2. Survey Problem Statement	80

6.3. Survey Model:	Modified-Delphi	
6.4. Procedures		
6.5. Survey Overview	W	
6.6. Survey Instrume 6.6.1. Validity 6.6.2. Confiden 6.6.3. Institution	ents Testing tiality nal Review Board	
CHAPTER 7. ROUND	1 SURVEY REPORT	
7.1. Introduction		
7.2. Round 1 Sample	e	
7.3. Participant Valio	dation	
7.4. Selection of Imp	portant Critical Success Factors	
7.5. Overall: Data &	č Analysis	
7.6. Design Category	y: Data & Analysis	
7.7. Construction Ca	ategory: Data & Analysis	
7.8. Process Categor	ry: Data & Analysis	
7.9. Impact Category	y: Data & Analysis	
7.10. Top 25% Sum	mary	
7.11. Round 1 Findin	ngs	
CHAPTER 8. ROUND	2 SURVEY REPORT	
8.1. Introduction		
8.2. Round 2 Sample	e	
 8.3. Evaluation of C: 8.3.1. Overall 8.3.2. Owners 8.3.3. Designers 8.3.4. Builders 8.3.5. Users 	ritical Success Factorss	

8.4. Round 2 Inter-Rater Agreement	
8.5. Reliability Test	164
8.6. Round 2 Correlations	
8.7. Relative Importance by Rank-Orders	
8.8. Top 25% Summary	
8.9. Difference Analysis	
8.10. Round 2 Results Comparison of Rounds 1 & 2	
8.11. Findings & Conclusions	
CHAPTER 9. ROUND 3 REPORT	190
9.1. Introduction	190
9.2. Familiarity with Project Delivery Methods	
9.3. Efficacy of Project Delivery Methods	199
9.4. Design-Bid-Build: Data & Analysis	201
9.5. Design-Build (DB):	
9.6. Round 3 Integrated Project Delivery (IPD)	
9.7. Round 3 Overall Data (DBB, DB, & IPD)	
 9.8. Significance Testing of Relative Effectiveness	217 218 220 221 221 227 233 234
9.8.7. Comparison of Median Weighted Averages 9.8.8. Implementability of IPD	
9.9. Round 3 Findings	
CHAPTER 10. CONCLUSIONS AND FUTURE RESEARCH	
10.1. Summary of Work	

10.2. Contribution to Knowledge	
10.3. Extensions and Future Research	
APPENDIX A: INTEGRATED DESIGN-BID-BUILD PILOT-PROJECTS	
APPENDIX B: INSTITUTIONAL REVIEW BOARD (IRB)	
APPENDIX C: ROUND 1 SURVEY INSTRUMENT	
APPENDIX D: ROUND 2 SURVEY INSTRUMENT	
APPENDIX E: ROUND 3 SURVEY INSTRUMENT	
APPENDIX F: ROUND 3 WEIGHTED SUM CALCULATIONS	
BIBLIOGRAPHY	
AUTHOR'S BIOGRAPHY	

LIST OF ABBREVIATIONS

%Rank	Fractional rank-order	
A/E	Architect / Engineer: abbreviation for design team	
AGC	The Associated General Contractors of America	
AHP	Analytical Hierarch Process	
AIA	American Institute of Architects	
ASD-HA	Assistant Secretary of Defense for Health Affairs	
BIM	Building Information Modeling	
BRAC	Base Realignment and Closure	
COE	(U.S. Army) Corps of Engineers	
CSF	Critical Success Factor	
DART	Dispute Avoidance and Resolution	
DB	Design-Build	
DBB	Design-Bid-Build	
DBIA	The Design-Build Institute of America	
DoD	Department of Defense	
DOI	Degree of Interaction	
DPW	Department of Public Works	
ECI	Early Contractor Involvement	
FAR	Federal Acquisition Regulation	
IDBB	Integrated Design-Bid-Build	
IPD	Integrated Project Delivery	
IRP	Institutional Review Board	
KPI	Key Performance Indicator	
LEED	Leadership in Energy and Environmental Design	
LEED AP	Leadership in Energy and Environmental Design Accredited Professional	
MED MILCON Medical Military Construction		
MILCON	Military Construction	
NAVFAC	Naval Facility Engineering Command	
NGA	National Geospatial Intelligence Agency	
NTP	Notice to Proceed	
RFP	Request for Proposal	
TMA	Tricare Management Activity	
UFC	Unified Facilities Criteria	
USACE	U.S. Army Corps of Engineers	
USAHFPA	U.S. Army Health Facility Planning Agency	
USGBC	U.S. Green Building Council	
W	Kendall's Coefficient of Concordance	

CHAPTER 1.

INTRODUCTION

1.1. Problem Statement

The traditional project delivery method is a fragmented linear process, where each fragment¹ is dependent on the substantial completion of precedent activities before the start of subsequent activity. The project delivery process is similar to the chain reaction of falling dominos. For example, construction start requires completion of a design. If the design domino does not fall, then the chain reaction is halted. Even when the design domino topples towards the construction domino, the chain reaction between design and construction is fragmented by the acquisition process: advertisement, bid, selection and award. These fragmentations, or gaps, that occur during the process timeline from initial concept until project completion, are problematic to the alignment and continuity of project goals between project delivery stakeholder groups.

The definition and interpretation of goals are relative to each stakeholder's personal interests for a given fragment throughout the project delivery process. Even if there is some measure of continuity of major project goals, there is no incentive for any party to be overly concerned about goals outside the scope of their separate project involvement. Collective rationality is not a burden imposed on a group's members. Traditional project delivery methods have evolved to maximize stakeholder utility within the framework of the fragmented process.

¹ Note: These "fragments" of the project delivery process are later defined in Chapter 4 as *micro-viewpoint* aspects of the project delivery process. Correspondingly, a *macro-viewpoint* encompasses all the micro-viewpoint phases such as concept development, design, construction, and operations by which overall project success may be measured.

The designer works to optimize design and the builder works to optimize construction. Each industry over time has naturally optimized itself within the context of the traditional project delivery framework. As long as the project delivery process is fragmented linearly, integration efforts such as constructability reviews and partnering are limited in their impact because they merely attempt to improve upon the existing sub-optimal process. Furthermore, sub-optimal is an Integrated Project Delivery process which does not achieve group utility.

As Weirich states, "A group may fail to maximize collective utility although each member maximizes utility individually". ² For project delivery to successfully reach its potential for maximizing value-creation, the sum of each stakeholder's utility must be superadditive.³ As such there is no more critical step than carefully establishing a project delivery team fully invested in a collaborative spirit. This is a major shortcoming of the traditional Design-Bid-Build project delivery process, where each side sees the other as a potential obstacle to successful project delivery. For this reason, the traditional method is sometimes derogatorily referred to as the 'Over the Wall' method to highlight separation of design and construction processes. The architect completes the design process, and then throws the completed design (figuratively) over the wall to an unknown construction contractor to build. The 'wall' is an inhibitor, a barrier, to collaboration and a primary cause of adversarial relationships between the design and construction professionals who have no formal relationship. The result is increased

² Paul Weirich, *Collective Rationality: Equilibrium in Cooperative Games* (New York: Oxford University Press, 2010), 60.

³ Imma Curiel, *Cooperative Game Theory and Applications: Cooperative Game Arising from Combinatorial Optimization Problems* (Boston, MA: Klumer Academic Publishers, 1997), 2.

project cost and reduced building quality.⁴ This inherent flaw in the traditional Design-Bid-Build process inhibits value-creation.

Another problem with the linear process of traditional Design-Bid-Build is you get so far down the road before a major conflict is identified too late to backtrack to the proper or optimal solution or fix. Invariably, compromises are made which may degrade, or at least make it more difficult to progress toward project goals. In isolation, each compromise may seem small and the best course of action to achieve the project goals at that time, but there certainly will be unknown consequence to the compromise. For example, a compromise made for the sake of ease of design may have an opposite effect on construction. Achievement of the project goals is what creates value, thus the necessity of compromises, however small, chip away at the ultimate value-creation of the project. In a linear process, the effects of each compromise are additively compounded, leading to a cumulative degradation to value-creation.

It is easy for each party along the linear, fragmented process to say they share the same project goals, but in fact, the actual separation of timelines, functions, and actions gives each stakeholder a unique and different perspective of the process in relation to their individual goals. For example, the contractor may be burdened by economic, political, meteorological, and other unforeseen realities during construction which the designer could not forecast during design. Decisions made early, in good faith, by the Designer to uphold the project goals become binding to subsequent players and in some cases become major barriers to attaining project goals. Maybe the builder has a better way, or the owner/user has a different expected outcome. Depending on the rigidity of the design and specification, the builder may have a narrow scope of variance in which to adopt means and methods to ensure true continuity of project goals. Thus, the Builder's

⁴ George Elvin, *Integrated Practice in Architecture: Mastering Design-Build, Fast-Track, and Building Information Modeling* (Hoboken, New Jersey: John Wiley & Sons, 2007). 234.

decisions can compromise both the design goals and/or the Owner's and User's goals diminishing overall value-creation.

Architects design space and place great value on the quality of space. To a construction contractor the space is free and it is the building components, features and systems which define the space and drive cost. This disparity is demonstrated by the typical difference between how architects and construction contractors measure net and gross building space. Architects typically measure based on factors of floor space while construction contractors measure from exterior limits of the building envelope. This may appear to be a minor discrepancy, but when applied to the order of magnitude of a large scale project the "minor discrepancy" may translate into multiple millions of dollars of difference between the designer's and constructor's unit cost estimates. This is only one example of the importance of having an integrated team where all stakeholders are on the same sheet of music from project inception.

Designers must try hard to predict the effects of the proposed design as much and as well as they can with the understanding that redesign or upkeep must follow.⁵ A value of the IPD process is that the designer does not necessarily need to predict an effect of the proposed design. By having the construction contractor involved in early aspects of concept design, or even as early as programming, the designer has access, in many instances, to real-time feedback of what the effects will be. Designers are notoriously weak in establishing accurate cost estimates. Decisions made at the earliest phases of design based on the designer's best prediction of cost can send a project down a design development path from which there is no return. After reaching this point, there is no return and the best redesign and upkeep efforts become mitigation measures for a self-created and incurable problem. A key concept of an IPD process is to limit

⁵ Michael Kyong-il Kim, "Countermodeling as a Strategy for Decision Making: Epistemological Problems in Design." (Ph.D. diss., University of California, Berkley, 1980), 13.

the designer's predictive variables for the purpose of maintaining continuity and alignment of project goals.

The purpose of any project delivery process is to attain the desired goals for which the project exists. Successful attainment of the project goals is intended to create value for the Owner and User, but also directly and indirectly creates value for other project stakeholders as well. In simple terms value-creation is the performance of activities to provide increased benefit. Value is a function of the output resulting from input of resources. In practice there is no simple definition of value, or value-creation, because assessment of benefit is sensitive to the perceptions of each beneficiary. Value is relative to the perception of each stakeholder in the project delivery process. Although there are many stakeholders involved in large, complex projects, this research focuses on four primary stakeholder groups: Owner, Designer, Builder, and User. It is logical to assume each of these four stakeholder groups have differing perceptions of relative importance of project goals and value-creation based on each group's unique, independent assessment of perceived benefit.

Hypotheses: Integrated Project Delivery (IPD) is more effective than traditional project delivery for achieving expected goals throughout the project delivery process, therefore providing a process which is more advantageous to value-creation.

1.2. Research Objectives

The objectives of this research are: (1) to define the relationship between project goals and value-creation; (2) to indentify critical success factors which key project stakeholders believe are important to project success and value-creation; (3) to establish the existence of significant differences between each stakeholder's perception of project success relative to self-

5

interest and project-interest perspectives; (4) to establish the existence of significant differences between what critical success factors each project stakeholder group perceives as most important for value-creation; (5) establish a macro-view assessment of project delivery as a basis for evaluating project success; (6) demonstrate that Integrated Project Delivery is implementable by the project stakeholders; and (7) determine a normative model for project delivery of large complex military medical projects by comparing the effectiveness of design-bid-build, designbuild, and Integrated Project Delivery methods to achieve the most important critical success factors.

1.3. Organization of Dissertation

Chapter 1 of the dissertation has provided an introduction and brief overview of this research. Chapter 2 defines Integrated Project Delivery and reviews its origin and evolution in an analytical and historical context of project delivery. Also introduced is the Military Construction context which provides focus for the research project, and the multi-party contractual relationships of IPD. Further explanations of IPD include significance, desirability, and the need for further study. Chapter 3 reviews the literature. Chapter 4 provides the theoretical framework of the research based on the relationship between value-creation and project goals, stakeholder perceptions and viewpoints from which to assess value-creation. Chapter 5 provides cross analysis between the qualitative survey research and case study field assessments. Chapter 6 provides the research method and describes in detail the modified-Delphi survey model used to categorize, collect and analyze the data. Three rounds of survey research were performed and Chapters 7, 8, and 9 report the results of each survey. Finally, Chapter 10 provides summaries of the work, contributions, and potential future research topics. All references are provided in footnotes, and a bibliography is provided after the appendices.

6

CHAPTER 2.

INTEGRATED PROJECT DELIVERY

2.1. IPD Overview

The design and construction industries struggle to define the term Integrated Project Delivery (IPD). The definition and interpretation of IPD varies among different people, organizations, or industries. To some, IPD is associated solely with the Design-Build project delivery method. Design-Build is one item in the toolbox for achieving an IPD process but not the only one. Interpretations of defining IPD are influenced by the "What's in it for me?" mindset. Or, what is the value of pursuing an integrated process? From each point of view (owner, designer, constructor, and other stakeholders) the relative value may in fact, or in perception, be vastly different.



Figure 2.1. Design / Construction process relationship

The American Institute of Architects defines Integrated Project Delivery (IPD) as a project delivery approach that integrates people, systems, business structures and practices into a process that collaboratively harnesses the talents and insights of all the participants to reduce waste and optimized efficiency through all phases of design, fabrication and construction.⁶

⁶ The American Institute of Architects, *Integrated Project Delivery: A Guide* (AIA National & AIA California Council, 2007).

Historically, the master-builder naturally, and logically, integrated all aspects of project delivery through his single, comprehensive mind. Through millennia of hard earned lessons of trial and error, each generation of master builders passed on the art of building which resulted in the construction of great edifices and wonders of the world: the Great Pyramids, Greek temples, Roman aqueducts, and medieval cathedrals. These master builders conceptualized and designed their projects based on their comprehensive knowledge of materials and construction, and built them under their close/direct supervision. Since every aspect of the project, including both the design and the construction, was carefully orchestrated in a single head, the comprehensive approach of the master builder must have been the key to successful project delivery.

Why over the course of time has the process of project delivery disintegrated into separate professions of design and construction? This is an important question because one must understand the reasons for the disintegration of the project delivery process to form an argument for re-integration. There are a multitude of technological, social, political, and cultural factors which have contributed to how projects are delivered today. The early master-builders were the products of the *evolutionary* development of the art of building using common materials, tools, equipment, and labor practices. The explosion of science and technology in more recent times has resulted in *revolutionary* changes as to how projects are designed and constructed. The industrial revolution brought about unprecedented complexity and professional specialization to tackle the new challenges. New advancements in material science, engineering, and construction technology add to a body of knowledge that is seemingly too great for any one person to master. Likewise, the complexities of modern life demand that our designers and builders produce buildings with requirements increasingly more complex than ever before. Today's hospitals, airports, industrial plants, and all manner of high-technology facilities, built in a more strict

8

regulatory and environmentally conscious landscape, provide unprecedented technical challenges to today's designers and builders. Thus, the project delivery process has fragmented, or disintegrated, into numerous sub-professions, not only between design and construction but also within each. These separate design and construction professions, where art and science are often at odds with each other, establish degrees of separation posing significant communication challenges to an integrated process.

The inter-relationship between the design and construction professions, as well as within the intra-relationships, are at best sub-optimal and at worst dysfunctional. Efficient communication between the designers and builders to orchestrate the myriad of project activities is paramount to successful project execution. Unfortunately, the process has evolved such that the communication moves primarily in one direction from designer to constructor. The designer attempts to communicate their design intent through drawings and specifications. At the point when the construction contract is awarded, the builder must interpret those static (and dated) drawings and specifications and apply them within the context of a dynamic environment which the designer may or may not have taken correctly into account. The process is inherently incremental and linear where decisions made early must rely on predictions of conditions that will exists during the later phases of the process.

Until very recently, the communication tools available to the design and construction professions have changed very little since the times of the master builders while the substance of communication became far more complicated. Projects relied on two-dimensional drawings and detailed written instructions (specifications) to communicate design intent. The laborious process of producing truckloads of construction documents created a project bureaucracy which further contributed to the complexity and complication of the project delivery process. Feedback

9

of improvements or changes from the construction side of the process was limited by the ability of the designers to rapidly assess the comprehensive impact of a proposed design modification relative to the stage of design completion.⁷

New three and four-dimensional computer modeling and Building Information Modeling (BIM) applications make it possible for the first time for the designer and constructor to communicate in real-time throughout the design and construction process.⁸ Finally there are tools available which afford the opportunity to adopt a non-linear method to integrate the project delivery process. As these tools are further refined, and even more advanced tools are invented, the capability of a new virtual-mind to comprehensively process of all aspects of design and construction for even the most complex projects becomes possible. Perhaps the time has come for the return of the concept of the master builder. Today's master builder is not an individual, but an integrated multi-party team of owner, designer, builder, and other related professions. Together these professionals form a master-team working together to provide a truly comprehensive project delivery process.

Typically design is a predecessor to construction. In isolation every task or activity requires design to translate an idea to a physical end state. At this fundamental level design always comes first. In a perfect world a perfect design would result in a perfect translation of the idea/design into a physical manifestation of the designer's intent. Of course, our world is not perfect, and many variables are encountered during the construction phase that requires change to the design. A few such variables are design errors or emissions; unforeseen site conditions; adverse weather conditions; social/political/legal/regulatory issues; market and economy

⁷ Michael Kyong-il Kim, ARCH 599 Report Feedback, Mkkim1@illinois.edu (University of Illinois at Urbana-Champaign, July 01 2009).

⁸ Elvin, Integrated Practice in Architecture: Mastering Design-Build, Fast-Track, and Building Information Modeling.

fluctuations; and availability of labor, equipment, and materials. In general the concept of an integrated process is to integrate early collaborative involvement of the project builder into design development activities.

The complex nature of healthcare design and construction provides a great working laboratory to study the impact different project delivery methods have on successfully achieving project goals and creating value. Large, complex projects provide more opportunity for errors, omissions, and miscommunication than small, simple projects. Complexity magnifies sensitivity to the delivery method's impact on project outcomes where small errors have big consequences. Complex projects require a greater degree of effort and accuracy in execution of both design and construction activities. Complex medical projects provide a good platform to assess how different delivery methods impact the relationship between design and construction in different ways.

Only recently has the health care industry shifted from the long-held conviction that Design-Bid-Build (DBB) was the only appropriate method for executing a large, complex, medical project. A "perfect" 100% design informed by all manner of consultants, experts, and specialists was the only way to ensure the value of the design met the owner's intent. A problem with the DBB solution is the length of time required to deliver the project. Military hospital projects would regularly take seven-plus years to progress from concept to operations.⁹ At that point a "new" hospital was already obsolete by nearly a decade!

⁹ U.S. Department of Defense, *Unified Facilities Criteria (UFC)* 4-510-1 - Design: Medical Military Facilities (Department of Defense, 19 November, 2009), 24.

2.2. Military Construction and Integrated Project Delivery

The research focuses on military construction projects exclusively. This section provides background for understanding the context of military construction (MILCON) in relation to private projects, and to how Integrated Project Delivery is approached in the MILCON program.

2.2.1. Public Sector versus Private Sector

It is important to highlight the differences between public and private project delivery because this study will focus on case studies in only one particular federal public sector -- U.S. Military Construction. An important distinction between public and private sector definition and implementation of an Integrated Project Delivery process is the level of contractual freedom and creativity the owner possesses in assembling the desired integrated team and project delivery process. Private project owners, through the contractual framework of offer, acceptance, and considerations usually have no external limitations on how they assemble their integrated project team. Private owners are at liberty to negotiate all aspects of the project delivery process with designers, builders, or any other contractors required for establishing their vision of an Integrated Project Delivery team. All aspects of schedule, quality, cost, and risk allocation are negotiable. Private owners are limited only by creativity, availability of private resources and qualified bidders.¹⁰

Unlike the private sector, the public sector is encumbered with a myriad of local, State, and/or Federal acquisition regulations and statutes enacted to help ensure proper stewardship of the public's resources. The public owner must negotiate contracts within the parameters of statutes and political policies. For example, Federal projects such as the case study projects are subject to requirements for providing for employment opportunities to disadvantaged minorities,

¹⁰ Justin Sweet and Mark M. Schneier, *Legal Aspects of Architecture, Engineering, and the Construction Process*, 1970, 8th ed. (Stamford, CT: Cengage Learning, 2008), 340.

women, people with disabilities, and disabled veterans; setting aside awards for small or disadvantaged business; ensuring workers are paid at local prevailing wages (Davis-Bacon Act)¹¹, and protection of American manufacturers (Buy American Act)¹² among many numerous other requirements which do not burden private projects.¹³ Additionally, a public owner's project delivery process often is exposed to significantly more external stakeholders¹⁴ as part of mandated public/community participation in the project development and vetting processes. Public projects are inherently political and exposed to bureaucratic risk often outside the control of the owners or end users of the projects. The public owner must work within the budgetary framework required to plan, program and publically fund all project costs. The typical bureaucracies of public sector projects make them less nimble than private sector projects. Multiple layers of authority distributed among multiple public agencies add transactional time and cost to public projects.

A benefit of studying public sector projects is that there are established project parameters which provide a level of normalization when comparing projects. It is difficult to make direct comparisons of projects because there are so many project variables such as site, location, time, and other aspects of planning, design, and construction. The uniform set of acquisition rules between projects allows sensible comparison between projects. The Integrated-Design-Bid-Build project delivery method, further discussed below, is a good example of one such public project delivery process.

¹¹ The Davis-Bacon Act, as Amended *United States Code* § 40 USC Sec. 3141-3148 (1931, Amended 2002)

¹² The Buy American Act United States Code § 41 USC Sec. 10a-10d (2009)

¹³ Sweet and Schneier, *Legal Aspects of Architecture, Engineering, and the Construction Process*, 390.

¹⁴ John A. White, Marvin H. Agee, and Kenneth E. Case, *Principles of Engineering Economic Analysis, Third Edition* (New York: Wiley Publishing, Inc., 1989), 332.

In theory an optimal Integrated Project Delivery process has all parties (Owner, Designer, Builder, and User) collaborating from project inception through completion of construction. In practice for public sector projects, it is not feasible, due to rules and regulations, to fully integrate the design and construction process. For example Federal rules require separate procurement processes for design and construction services. The Brooks Act¹⁵ requires that Architect/Engineer services be awarded based on qualifications rather than on lowest-bid. The rationale being that in regard to design it is not in the public's best interest to award design work to the lowest bidder. But, on the other hand, the Federal Acquisition Regulations require that construction services be awarded based on price or best value. It is, however, possible to work within the parameters of the existing public sector rules and regulations to establish a project delivery process as close as possible to the theoretical optimum for Integrated Project Delivery -simultaneous start of both construction and design services.

2.2.2. Department of Defense Medical Military Construction (MED MILCON)

This study examines case study projects primarily from the Medical Military Construction Program (Medical MILCON) of the United States Department of Defense. The Medical MILCON program is a portion of the Congressional, Title X, military construction program (10 United States Code, Chapter 169 - Military Construction and Military Family Housing)¹⁶ managed by the office of the Assistant Secretary of Defense for Health Affairs (ASD-HA). ASD-HA provides budgetary and policy oversight to the medical departments of the three

¹⁵ The Brooks Act: Federal Government Selection of Architects and Engineers *Public Law 92-582* (1972) (92nd Congress, H.R. 12807, October 27, 1972).

¹⁶ Military Construction and Military Family Housing *Title 10 - Armed Forces United States Code* § Title 10 USC Chapter 169 (2009)

military services (Army, Navy, and Air Force).¹⁷ Each of the services has a different approach as to how it executes planning and delivery of projects. However, they all link back to ASD-HA through their respective Surgeons General. By law the U.S. Army Corps of Engineers (USACE) is the contracting agent for the design and construction of projects in Army and Air Force areas of operations, and the Naval Facility Engineering Command (NAVFAC) is the contracting agent for projects in Navy area of operations. MILCON projects are all subject to the same Federal acquisition statutes and regulations. Each service's Surgeon General oversees an agency, bureau, or office which acts as owner's and users' representatives in the programming, planning, design, and construction of medical facility projects. There are easily four or five layers of corporate "owner-stakeholders" before adding the operational level (medical facility users) stakeholders, design team, and construction team to the overall project delivery team.

2.2.3. Integrated Design-Bid-Build (IDBB)

To overcome the limitations and challenges of Design-Bid-Build and Design-Build, and to adopt more integrated practices, the U.S. Army Corps of Engineers developed a hybrid project delivery method. This hybrid method executes an Integrated Project Delivery process within the established statues and policies governing the MILCON program. Integrated-Design-Bid-Build (IDBB), Figure 2.2, is a delivery process where separate contracts for A/E¹⁸ and construction services are awarded concurrently to allow collaborative simultaneous design and construction activities to be executed. Two key aspects of the IDBB concept are early contractor involvement (ECI) and extension of A/E Title II Services.

¹⁷ U.S. Department of Defense, *Unified Facilities Criteria (UFC)* 4-510-1 - Design: Medical Military Facilities.

¹⁸ Architecture and Engineering Design Service Provider



Figure 2.2. Integrated Design-Bid-Build (IDBB) concept

2.2.3.A. Early Contractor Involvement

Within the parameters of the Federal Acquisition Regulations a construction contractor is contracted for pre-construction services (non-Brooks Act services)¹⁹ with an option for construction services. The purpose of the pre-construction services and early contractor involvement is to allow the construction contractor the opportunity to provide collaborative input to early stages of Title I A/E Services (plans, specifications, and design) and to eliminate the need to have a 100% completed design before awarding the construction contract. Because some level of minimal design must be completed in order to competitively bid construction services, it is impossible to simultaneously award the separate design and construction contracts. To provide enough project data in a Request For Proposal (RFP), a minimal level of comprehensive design must be completed to provide sufficient project data upon which construction contractors may bid. The COE sequences initial A/E design services to include RFP development. The A/E must immediately provide a minimal schematic concept design plus estimates for design data

¹⁹ The Brooks Act: Federal Government Selection of Architects and Engineers *Public Law 92-582* (1972) (92nd Congress, H.R. 12807, October 27, 1972)

which normally would not be produced until further design development (See Figure 2.3) to allow construction services to be procured as early as possible.



Figure 2.3. IDBB minimal design for Construction Bid

2.2.3.B. Extended A/E Services

In a traditional Design-Bid-Build process the construction contract is not awarded until after 100% completion of the design (Title I A/E services), after which the A/E may or may not provide Title II Services (inspection and/or observation) of construction operations, depending on the contract. Because IDBB overlaps the design and construction processes, Title II A/E services provide real-time feedback from the construction of early design packages which can be used to optimize or improve the quality of subsequent design packages.

The IDBB process differs from Design-Build (DB). The DB design process is linear with a break between the RFP concept design and continuation with the DB contractor's in-house design team after award of the DB contract. This arrangement is inherently inefficient, especially for complex projects, because there is no continuity between the separate RFP and DB design teams. Large, complex Medical MILCON projects have been executed almost



Figure 2.4. Project Delivery Process Timeline Comparison

exclusively using the traditional Design-Bid-Build project delivery method. USACE and NAVFAC were early adopters of the Design-Build project delivery method, but DB has historically played a minor role in the Medical MILCON program, reserved for smaller, less-complex projects such as renovations or smaller clinical facilities. IDBB is a multi-party contractual arrangement where the same A/E develops both the request for proposal (RFP) conceptual design, and is retained to complete all remaining design development in collaboration with the successfully bidding construction contractor. In a typical MILCON DB project a separate A/E is contracted to develop a concept design, typically to the 35% design-completion stage, as part of the RFP package (Figure 2.4).

2.2.3.C. IDBB Pilot-Projects

The IDBB process has been tested successfully on small scale projects by the Kansas City - District of the US Army Corps of Engineers. The early successes of the test projects resulted in Department of Army approval for a Pilot Program to utilize the IDBB process in support of Base Realignment and Closure (BRAC) projects to meet extremely aggressive project timelines for large complex project types. Under the BRAC Act of 2005, BRAC funded projects must be completed no later than 15 September 2011, and the Department of Defense recognized that the traditional project delivery methods would not be adequate to meet the statutory deadline for four projects of particularly large scope and complexity. The four projects selected for the pilot program include two each located at Fort Belvoir in the National Capital Area of Northern Virginia, and two located at Fort Sam Houston in San Antonio, Texas:

- The National Geo-Spatial Intelligence Agency's East Campus, Fort Belvoir, Virginia
- Fort Belvoir Community Hospital, Fort Belvoir, Virginia
- San Antonio Military Medical Center, Fort Sam Houston, San Antonio, Texas
- Battle Field Health Trauma Laboratory, Fort Sam Houston, San Antonio, Texas

Additional project information may be found in Appendix A: IDBB Pilot-Projects. This study primarily utilizes the large hospital projects at Fort Belvoir, Virginia and San Antonio, Texas as case studies to evaluate the integrated project teams and integrated processes role in enhancing value creation. Each of the hospital projects is highly complex with project scopes near or exceeding one million square feet and total project costs approaching or exceeding \$1 billion each. Although the National Geospatial Intelligence Agency (NGA) project is not a medical facility, its scope and program is equally large and complex as the hospital projects. Members of the NGA project delivery team participated in the research survey.

19

2.3. Multi-party IPD Contract

Between 2007 and 2008 private industry had begun to adopt standard form contracts for multi-party Integrated Project Delivery. The American Institute of Architects (AIA) in 2007 published the first of such standard form contracts: C195TM–2008 Standard Form of Single Purpose Entity Agreement for Integrated Project Delivery. The terms of the contract allow equitable sharing of risk and reward. Owner, designer, and builder collaboratively work together from the beginning to deliver the project based upon mutually agreed upon project goals and target costs.²⁰ In 2007 The Associated General Contractors of America (AGC) for the first time failed to adopt the AIA's standard from contracts, and established a competing standard form ConsensusDocs300 - Standard Form of Tri-Party Agreement for Collaborative Project Delivery.²¹ Not surprisingly, the main difference between AIA and AGC's approach to multiparty IPD centers on which industry should dominate the process. The AIA document has the architect acting as the owner's agent throughout the IPD process from planning to construction, while the AGC document does not provide a significant role for the architect during the construction phase.

The private owner is at liberty to adopt and/modify any contract model he or she pleases and negotiate terms satisfactory to all parties. Public owners do not share the contractual freedom of their private counterparts. Public owners must work within the established policies, regulations, and laws which govern their capital investment programs. For the Military Construction (MILCON) case studies the U.S. Army Corps of Engineers (USACE) has adopted

²⁰ The American Institute of Architects, *Integrated Project Delivery: A Guide* (AIA National & AIA California Council, 2007).

²¹ Phillip G. Bernstein and Martin Hague. "Integrated Project Delivery (IPD): Why Owners Choose Multi-Party." Webinar. Washington, DC: AGC of America (The Associated General Contractors of America), 2009.

an early contractor involvement (ECI) concept to work within the federal acquisition regulations.²² Integrated Design-Bid-Build is a hybrid-variation of a very mature and time-tested contractual method. For some design and construction agents the bureaucracy compared to private sector work can be a strong disincentive to bid on public projects such the case-studies. For designers and builders with significant experience in the public sector, familiarity with the system can be an incentive to go along with the financial certainty of federally funded work.

2.3.1. Financial Incentive Structure

The primary contractual features of multi-party IPD are collaboration and equitable, shared risk/reward allocation. The rewards come in the form of financial incentives based on overall project cost performance. The intent of the incentive is to promote creativity and innovation by offering significant motivation to overcome the additional risk exposure which results from pursuing unproven design and/or construction means and methods. Private sector projects may be negotiated in any way agreed upon by the parties in regards to risk, reward, and financial incentives. The public case-study projects are governed under the Federal Acquisition Regulation (FAR)²³ which does allow limited incentives, but only under narrow specifications. FAR Clause 16.403 Fixed Price Incentive Contracts²⁴ and FAR Clause 52.216-17(d)(2)

16.403-2 Fixed-price incentive (successive targets) contracts.

 ²² U.S. Army Corps of Engineers. "Early Contractor Involvement (EIC)." In *Industry Workshop*.
 7 March 2010. New Orleans, LA: U.S. Army Corps of Engineers, 28 January, 2009.

 ²³ Department of Defense General Services Administration, and National Aeronautic and Space Administration, Cong., Federal Acquisition Regulation (FAR) VOLUME II -- Parts 52, 53, & Index (TITLE 48—FEDERAL ACQUISITION REGULATIONS SYSTEM 2005)
 ²⁴ 16.403 Fixed-price incentive contracts.

⁽a) Description. A fixed-price incentive contract is a fixed-price contract that provides for adjusting profit and establishing the final contract price by application of a formula based on the relationship of total final negotiated cost to total target cost. The final price is subject to a price ceiling, negotiated at the outset. The two forms of fixed-price incentive contracts, firm target and successive targets, are further described in 16.403-1 and 16.403-2 below.

⁽a) Description.

Application of the Incentive: *Establishing firm fixed price or final profit adjustment formula*²⁵ define the allowable limits for incentives.

The IDBB contract method allows for a series of successive targets where the contract is definitized at completion of major project packages or phases. The allocation of incentive payment is depicted in Figure 2.5. The incentive pool is created by the project delivery team's ability to deliver the project below the target price. The FAR provides an incentive range between minimum of 1% and maximum of 7.6% of the total initial cost. For a \$500+ million project the financial incentive is quite significant (\$5 million - \$38 million range), but due to the regulatory limitations, the incentives are not as generous as what could be offered by a private owner.

⁽¹⁾ A fixed-price incentive (successive targets) contract specifies the following elements, all of which are negotiated at the outset:

⁽i) An initial target cost.

⁽ii) An initial target profit.

⁽iii) An initial profit adjustment formula to be used for establishing the firm target profit, including a ceiling and floor for the firm target profit. (This formula normally provides for a lesser degree of contractor cost responsibility than would a formula for establishing final profit and price.)

⁽iv) The production point at which the firm target cost and firm target profit will be negotiated (usually before delivery or shop completion of the first item).

⁽v) A ceiling price that is the maximum that may be paid to the contractor, except for any adjustment under other contract clauses providing for equitable adjustment or other revision of the contract price under stated circumstances.

⁽²⁾ When the production point specified in the contract is reached, the parties negotiate the firm target cost, giving consideration to cost experience under the contract and other pertinent factors. The firm target profit is established by the formula.

²⁵ Application of the Incentive *Establishing firm fixed price or final profit adjustment formula* Excerpt from FAR Clause 52.216-17(d)(2) "If the total firm target cost is more than the total initial target cost, the total initial target profit shall be decreased. If the total firm target cost is less than the total initial target cost, the total initial target profit shall be increased. The initial target profit shall be increased or decreased by *TBN* percent of the difference between the total initial target cost and the total firm target cost. The resulting amount shall be the total firm target profit; provided, that in no event shall the total firm target profit be less than *1.0%* or more than 7.6% of the total initial cost." (*TBN*: To be determined by negotiation)



Figure 2.5. Incentive Profit Adjustment (USACE)

2.3.2. Increased Revenue Opportunity

Provisions for pre-construction services provides the builder with an a new, almost-zerorisk revenue source to improve the cash flow bottom line. The fee-based services allow for the construction agent to collaborate with the Owner and Designer at the earliest phases of project development. Payment is not tied to actual construction placement. Payment for preconstruction services start as soon as provision of the pre-construction service is authorized and notice to proceed is issued (NTP).

2.3.3. Contingency Structure

The owner provides contingency cost reserves for both design and construction services. Due to the risk of proceeding with limited design information it is important for the owner to carry appropriate levels of contingency funding as part of equitable risk allocation. Contractors may more competitively bid the work if they don't have to build in hidden contingencies into their price proposals.

2.3.4. Specification Philosophy

The Corps of Engineers, for one, is known for having voluminous stacks of very detailed, and often onerous, prescriptive specifications. Prescriptive specifications stifle creativity and innovation, and add cost to project delivery. A collaborative project delivery process is best served by maximum use of descriptive performance specifications. By communicating the owner's project requirements in as general terms as possible, the project team can creatively pursue innovative alternatives which may be unknown to the owner at the time the project requirements were developed. The best qualified, innovative designers and builders have little incentive to sign on to a project in which their creativity is limited by onerous, excessive detail of overly prescriptive specifications.

2.3.5. Contractual Incentives for Builder to Participate in IPD

The following will discuss key contract terms which provide incentive for a construction contractor to engage in IPD. Typically the terms below will be found in either the general or supplemental section of the contract.

2.3.5.A. Roles and Responsibilities

Clearly stating the collaborative roles and responsibilities of each party up-front, sets the tone for successful project execution. In IPD the owner concedes a level of control to his or her collaborative partners. The process requires more participation and associated resources from each party, but also affords the contractor more input and control during early design phases which could limit or restrict efficient employment of means and methods during later construction phases.

24
2.3.5.B. Risk Allocation

The more equitable allocation of risk in IPD is a great incentive for the construction contractor over more traditional methods where the owner directs as much risk as possible to the builder. The risk allocation is tied to each party's share of the incentive pool creating a situation where each party benefits more when the whole benefits. Risk is transparent and controllable.

2.3.5.C. Change Management

Theoretically an IPD project should have no changes do to the collaborative nature of the project delivery. Eliminating or minimizing changes allows more efficient execution of construction, and lowers transactional cost of administrative processing. The change order process is minimal. When there are changes they can occur quickly and smoothly.

2.3.5.D. Schedule/Time

Construction scheduling determines design schedule. The process adapts to optimize constructability. Role and responsibilities clearly state the timeline for processing any project documents which are required to facilitate construction placement.

2.3.5.E. Dispute Resolution:

Many projects adopt a Zero-Litigation contract clause as part of a multi-step dispute and resolution ladder with escalation up to binding arbitration or some other level short of litigation.

2.3.5.F. Building Information Modeling (BIM)

Unlike in traditional project delivery, the construction contractor and sub-contractors have the ability to participate in early model design rather than use BIM as a tool merely to identify and resolve conflicts and clashes after the fact. By designing in BIM, the contractor is able to complete rapid, accurate cost estimating.

25

2.3.5.G. Control & Influence

For contractors who always find fault with the design, they now have a seat at the decision table during the design process.

2.3.6. Overcoming Resistance to Change

Some aspects of IPD will serve as disincentives for some potential construction contractors to compete for projects. IPD requires considerable administrative involvement at all levels of project management. Financial control and cost reporting are paramount to successfully managing an integrated project. Earned Value Management is administratively burdensome and requires significant reporting from the field. The process can be very chaotic and greatly relies on strong communication and public relations skills. Personality conflicts and adversarial relationships have the potential to derail the collaborative process. Human resources planning and training is an important, but often overlooked aspect of collaboration.

Some smaller contractors and sub-contractors have resisted adopting BIM practices, because the technology required to collaborate may be out of reach for some otherwise highly qualified contractors and/or sub-contractors.

And finally some contractors will refuse to change practices. Although increasing in popularity and use, IPD is relatively new and few designers or contractors have experience participating in a fully integrated multi-party process. There is a significant learning curve to adapt to the administrative scope and speed required to keep communication open and transparent. Some of the common arguments against IPD are that the multi-party contracts have not been legally tested, or that the process creates contractual problems for the stakeholders.

The Design-Build Institute of America (DBIA) placed a multi-page advertisement in ENR magazine which railed against IPD for the reasons stated above, and added that DB is already an integrated delivery method. Obviously, DBIA has a vested interest in protecting

26

market share for its membership, and has no interest in promoting a competing delivery method. It is true that DB is a more integrated approach than traditional DBB, but the exact same argument could have been made about the DB method 15-20 years ago when it was first entering the industry. Today Design-Build is highly successful and has established itself as the most used method for project delivery. At one time DB was untested and had to go through the same growing pains that IPD certainly will as the process is tested and refined with each pioneering project. There is always risk when applying new methods and technologies, but that does not excuse the need for seeking an improved process. While DB certainly may be the optimal project delivery for some projects, there are also projects which don't fit the DB model. The emergence of Integrated Project Delivery is based on the industry's response to demand for improved project delivery to overcome increasingly complex and challenging requirements.²⁶

2.4. Significance

Even with a highly successful project there are significant efficiencies and improvements which may have been achieved through process improvement. Relying on post-occupancyevolutions to assess project success and gain lessons learned does nothing to improve the project just completed and offers marginal value to the "next" project which develops with its own unique context, environment, and issues. Developing a normative model provides a proactive process for contemporaneous process improvement feed-back throughout the Integrated Project Delivery process. In today's economic climate it is imperative to find better ways to maximize value creation in our buildings. There is great potential for IPD to provide such an advantage.

²⁶ Greg Howell and Will Lichtig, "Special Report: LCI Response to the May 5th, 2010 ENR Article, "Integrated-Project-Delivery Boosters Ignore Many Flashing Red Lights,"" 21 May 2010, Lean Construction Institute, 25 May 2010 http://www.leanconstruction.org/pdf/SpecialReportENR.pdf>.

However, the IPD process must be studied to validate that the intuitive, and often anecdotal benefits, are actual and justify the additional costs and effort of implementing IPD practices and process.

The goal of this research is process improvement. Because the Integrated Project Delivery process spans many disciplines, significantly design and construction, the focus of the research is how collaboration between these disciplines enhances value-creation. Where does the research fit in the discipline? It fits precisely at the intersection of design and construction where traditionally a designer's idea is translated by a builder into a physical object. The normally linear relationship will be studied from a non-linear perspective where the builder enhances design, and the designer enhances constructability thereby expanding the limits of the intersection between design and construction. A Win-Win-Win scenario, where the owner, designer, and builder all will be better off, lies in the contractual framework which binds the parties to the integrated, collaborative process. The terms of a well negotiated contract provide a roadmap to success for all parties. Further discussed in this section are aspects of the research topic which are significant to the field: risk allocation and dispute resolution; collaboration and communication; sustainable building practices; and motivation and incentive for value-creation. Aspects significant specifically to each party (Owner, Designer, Builder, and User) will be addressed separately.

2.4.1. Equitable Risk Allocation & Dispute Resolution

Sweet & Martin state the ingredients for dispute are the "Eternal Triangle" of Owner, Designer, and Builder.²⁷ The contractual relationship of the "Eternal Triangle" determines the risk profile of each project stakeholder. Equitable risk allocation is the single most important

²⁷ Sweet and Schneier, *Legal Aspects of Architecture, Engineering, and the Construction Process*, 85.

issue to optimizing collaboration between the multi-party entities of an Integrated Project Delivery (IPD) project. The time to address risk sharing is from the very beginning phases of the project delivery process.²⁸ Risk sharing may appear counter intuitive to an owner who is highly interested in protecting his or her investment. Unlike traditional Design-Bid-Build (DBB) where typically the owner attempts to transfer as much risk as possible to the contractor, the collaborative nature of IDBB is in alignment with Vega's principle of equitable risk allocation: "The general guiding principle of risk allocation should be that the different parties involved should seek a multi-beneficial distribution of risk. A dominant party that off-loads all project risks onto others is unlikely to enhance the chances for a successful outcome".²⁹ A project in which risks are distributed more justly also sets the stage for greater communication and interaction among the parties, resulting in more honest and productive negotiations when unforeseen conditions become apparent. Sharing project risk among a larger pool of highly experienced and knowledgeable team members results in a lowering of the contingency costs each member carries as overhead. By better balancing the risk allocation between the Owner, Designer, and Builder, the parties can more efficiently manage uncertainties.³⁰

Equitable risk allocation also promotes innovation to the benefit of all parties. Risk analysis is critical to the decision process by which a designer and/or builder develop the plans, means, and methods to deliver the owner's building requirements. Each entity's business decisions are influenced by perception and analysis of risk when contemplating break-even

²⁸ Feniosky Pena-Mora, Carlos E. Sosa, and D. Sean McCone, *Introduction to Construction Dispute Resolution* (Upper Saddle River, NJ: Prentice Hall, 2003), 47.

²⁹ Arturo Olvera Vega, "Risk Allocation in Infrastructure Financing," *Journal of Project Finance* 3, no. 2 (Summer 1997): 38-42.

³⁰ Pena-Mora, Sosa, and McCone, *Introduction to Construction Dispute Resolution*, 47.

points for individual financial considerations.³¹ Lowering the risk profile affords more opportunity to introduce acceptable risk through innovative or unproven means/methods, thereby allowing more creative opportunities to be generated by the integrated collaboration.

Dispute Resolution and Avoidance Techniques (DART) are critical to the overall measure of success each entity achieves.³² Issues or disputes which remain unresolved until after the project is completed result in claims.³³ If the contract does not address dispute resolution and claims management, it is likely the problem will not be settled internally between the project entities and litigation will ensue. Litigation risk adds significant cost to large, complex, or ill-defined programs. Parties to the contract will price their services to cover the costs of perceived risk. Not only does failure to address dispute resolution increase overhead costs, it also becomes a barrier to collaboration as each party must allocated resources to protecting themselves. The tremendous cost of litigation can easily outweigh all benefits received in an otherwise successfully managed project. Often the severe cost of litigation leaves even the adjudicated winner a financial loser.³⁴ As the saying goes, "in litigation everyone loses except the lawyers"! The best way to mitigate the costs of dispute and ensure success is to avoid disputes altogether.

The IPD process is designed to optimize collaboration, equitably share risk, and collectively address dispute resolution and avoidance. Bridging the design-build gap of traditional DBB project delivery overcomes a significant barrier to equitable risk allocation and

³¹ John A. White, Marvin H. Agee, and Kenneth E. Case, *Principles of Engineering Economic Analysis, Third Edition* (New York: Wiley Publishing, Inc., 1989), 372.

³² Pena-Mora, Sosa, and McCone, *Introduction to Construction Dispute Resolution*, 45.

³³ Barbara J. Jackson, *Construction Management Jump Start* (Indianapolis, IN: Wiley Publishing, Inc., 2004), 186.

³⁴ Sweet and Schneier, *Legal Aspects of Architecture, Engineering, and the Construction Process*, 15.

dispute resolution/avoidance.³⁵ By eliminating the 'over-the-wall' barrier, IPD promotes trust and teamwork through better risk management and dispute resolution.³⁶

2.4.2. Maximize Collaboration and Open Communication

Maximum collaboration and open communication is essential to all aspects of project delivery. Having all stakeholders on the same sheet of music throughout the project delivery process creates an overarching continuity of project goals and objectives. Each entity benefits from the open collaborative communication. The Owner is required to be more involved in the process than they normally would be in a traditional project delivery process. The Owner shares development of the program and vision with the Designer and Builder, and is able to receive continuous design validation and optimization throughout the process. Because theoretically there are zero non-discretionary change orders and a faster project completion, the Owner benefits from lower final project costs. An additional benefit to the Owner is the ability to re-invest incentive reward shares back into the project effectively buying more building for the same dollar. The re-investment of the owners' share of savings provides additional revenue to the Designer and Builder -- again a Win-Win-Win scenario.

2.4.3. Promote Sustainable-Building Design, Construction and Operations

Integrated Project Delivery provides the ideal framework for incorporating sustainable concepts throughout the lifecycle of the building from conceptual planning through user operations. The U.S. Green Building Council (USGBC) is the driving force in defining principles and characteristics of sustainable buildings. USGBC defines characteristics of sustainable building centered on a balance between environmental, social, and economic

 ³⁵ Pena-Mora, Sosa, and McCone, Introduction to Construction Dispute Resolution, 76.
³⁶ Elvin, Integrated Practice in Architecture: Mastering Design-Build, Fast-Track, and Building Information Modeling, 56.

prosperity.³⁷ USGBC's Leadership in Energy and Environmental Design (LEED) certification program defines sustainable characteristic of both buildings and the building process. Key areas include: Sustainable Sites, Water Efficiency, Energy Efficiency, Materials and Resources, Indoor Environmental Quality and Innovation in Design.

This study focuses on public projects, specifically Department of Defense Medical Military Construction (MILCON) projects. Public projects have significantly longer life-cycles than private projects. Since public buildings are expected to operate for fifty to one hundred years or more, they have much to gain from the benefits of sustainable building practices that provide a significant long-term value to the owner.

The Federal Government has mandated its own principle characteristics of sustainable buildings, based in large part on the USGBC's voluntary programs.³⁸ Executive Order 13423 signed by President Bush in 2006 directed that all federal agencies adopt sustainable design and construction practices for all new federal buildings and for major renovations of existing facilities.³⁹ The Interagency Sustainability Working Group comprised of leadership from all applicable federal agencies developed a uniform set of goals, objectives and principles for sustainable buildings. The federal goals include: reduce the total ownership cost of facilities; improve energy efficiency and water conservation; provide safe, healthy, and productive built environments; and, promote sustainable environmental stewardship.

³⁷ U.S. Green Building Council, *Guiding Principles*. 2006, U.S. Green Building Council, 14 April 2010 http://communicate.usgbc.org/usgbc/2006/08.15.06_guiding_principles/ guidingPrinciples/>.

³⁸ In Collaboration with the Interagency Sustainability Working Group, Federal Energy Management Program, U.S. Department of Energy, *The Business Case for Sustainable Design in Federal Facilities* (2003)

³⁹ Title 3 - The President, Presidential Documents, Federal Register, *Executive Order 13423 of January 24, 2007 - Strengthening Federal Environmental, Energy, and Transportation Management* 3979, at 3979-23 (2007)

To support the federal goals the following five characteristics of high performance and sustainable buildings were published⁴⁰:

1. Employ Integrated Design Principles.

Federal leadership recognized that the use of collaborative, integrated planning and design is essential to realizing sustainable building goals. In traditional project delivery methods the focus is on production of specific products: a design and a building. There is no continuity of vision. Integrated practice, on the other hand, is a continuous collaboration from concept through completion of construction, and on into operations. Integrated practices offer the best opportunity to realize successful adoption and implementation of sustainable building principles.⁴¹

- 2. Optimize Energy Performance
- **3. Protect and Conserve Water**
- 4. Enhance Indoor Environmental Quality
- 5. Reduce Environmental Impact of Materials.

The U.S. Army Corps of Engineers implemented an additional policy that all new construction projects will meet, at a minimum, all requirements for LEED Silver certification, and all design and construction teams under USACE contract must have LEED Accredited Professionals (AP) assigned to the project team.⁴²

From an operational standpoint the concept of sustainable building goes beyond the physical and performance aspects of the building and site. The IDBB pilot-projects provide

good examples of the need for sustainability of functional operations. The functionality and

⁴⁰ The Office of the Federal Environmental Executive, Interagency Sustainability Working Group, *Federal Leadership in High Performance and Sustainable Buildings Memorandum of* Understanding (2006)

⁴¹ Elvin, Integrated Practice in Architecture: Mastering Design-Build, Fast-Track, and Building Information Modeling, 213.

⁴² Annette Stumpf, U.S. Army Corps of Engineers, "Army Sustainability Policy," in *Infrastructure Systems Conference: Building National Technical Competency*, April 16, 2010 (Cleveland, OH, July 21 2009).

operations of health care facilities are very sensitive to emergent technology and innovative medical practices. The building must be highly adaptable to sustain state-of-the-art medical care and implement best medical practices. Often this requires frequent reconfiguring of entire clinics and departments, or expanding to accommodate new equipment and protocols. Only a comprehensive integrated approach to design and construction will ensure a sustainable vision well into the future use of the building.

Sustainability in context of the building's surroundings is also important. Dr. Kim has developed the concept of a Meronic goal,⁴³ where the building must fit as a part of the whole surroundings. An integrated project team is best suited to partner with local communities and governments to ensure building sustainability in the context of community does not create less-sustainable conditions for others outside the boundaries of the construction site.

The proper time for the Owner to incorporate sustainable building principles is at first conceptualization of the project requirements, if not earlier. It is important for the Owner to integrate sustainability into his/her functional philosophy or concept of operations.⁴⁴

Because many aspects of sustainable buildings are related to construction means and methods to reduce waste and improve efficiency throughout construction activities, it is critical to have the builder involved as early as possible. Contractor involvement at the earliest phases of design is a perfect means to ensure sustainable features in the design are not at odds with reality during construction. An example of this involvement is specifying a certain renewable/sustainable flooring material during design to gain two LEED points, then finding out

 ⁴³ Michael Kyong-il Kim, 2011. File: Manuscript in Progress. Integrative Design of Buildings: Principles and Strategies. University of Illinois at Urbana-Champaign, Unpublished, Chapter 3.
⁴⁴ In Collaboration with the Interagency Sustainability Working Group, Federal Energy Management Program, U.S. Department of Energy, *The Business Case for Sustainable Design in Federal Facilities Resource Document*, at 2-17 (2003)

during construction that use of the specified "renewable" flooring material will actually net a loss of 3 LEED points because the means and methods required to install it penalize other aspects of sustainability (non-local material, requires adhesives which reduce air quality, etc...).

It is the responsibility of the owner to ensure that the terms of the contracts for both the designer and the builder include specific requirements for sustainable practices. A good example is the USACE contract requirement for LEED AP certification for project team members and that the completed building must meet minimum requirements for LEED Silver certification. The general conditions of the contract must specify the roles and responsibilities of each party in achieving the desired sustainable building goals.

The owner also must be careful that the requirements are not so specified in such a prescriptive manner as to prevent the project team from adopting creative design features, new sustainable materials, or innovative construction means/methods. Descriptive, performance specifications are best for communicating project requirements to the designer and builder in a manner which promotes creativity and innovation.

There are some cases where the owner may wish to specifically prescribe certain sustainable building features for symbolic or marketing purposes. Most aspects of sustainable buildings involve building systems hidden in walls and ceilings such as mechanical, electrical, and plumbing systems. Some sustainable features can be designed as signature architectural elements with high sustainable function. Examples include green roofs, water features which collect and utilize rainwater, or other features with which users and visitors may interact. In medical facilities it is becoming common to feature 'healing gardens' which are integrated into storm water management systems.

35

How to afford implementing sustainable design? In most cases intelligent design decisions can implement sustainable building principles at little additional project cost. In some cases sustainable practices result in lower first costs for projects. The significant selling point of sustainable buildings is the long-term economic benefit to all parties involved. An Integrated Project Delivery process provides a collaborative bridge to ensure continuity of sustainable goals throughout the project delivery process.

2.5. Arguments for IPD

There are two philosophical approaches an Owner can adopt to influence the Designer and Builders contract performance: the Carrot or the Stick.⁴⁵ The Stick method, common in traditional DBB projects and especially in public sector DBB projects, involves assessing penalties for non-performance. An example is liquidated damages (LD) assessed against a construction contractor (i.e. \$1,000 dollars per day) for failure to meet a schedule deadline. The terms of the LD penalty is negotiated into the contract and becomes a risk (not necessarily equitably allocated) which the Builder assumes.⁴⁶ Ironically, despite the owner believing he or she is motivating the contractors' performance, in practice hurt their own self-interest because contractors will cover the risk at additional cost to the owner and pocket any and all associated savings when the project succeeds.

The philosophical approach most aligned with the collaborative nature of IPD is the "Carrot", or incentive. Collaboration may be greatly enhanced by negotiating contract provisions for shared incentives. The goal of the incentives is to motivate the contractors to maximize efficiency and increase productivity. Innovation requires contractors to assume the

⁴⁵ Jackson, Construction Management Jump Start, 181.

⁴⁶ Sweet and Schneier, *Legal Aspects of Architecture, Engineering, and the Construction Process*, 577.

risk of using untested means and methods. Financial incentives help promote creativity by rewarding contractors who innovate.

Incentives allow the owner, designer, and contractor to all share savings created by successful project execution. Each entity is positively motivated to collaborate to the fullest to maximize profit for Designer and Builder, while at the same time providing the Owner with savings. A perfect Win-Win scenario for the project team.

Integrated Project Delivery instills collaborative behavior through the employment of intelligently-crafted incentives. By diminishing the natural tendency of each entity to protect self-interest at the expense of the whole, and reinforcing project-centric behavior through incentives, IPD provides significant motivation for collaboration. As stated in a 2009 AGC publication, O'Connor writes, "While teamwork is built on trust, the IPD community is not altruistic. Incentives must be crafted so as to provide the real prospect of economic benefit for high performance".⁴⁷ Incentives may not guarantee success, but establishment of significant, attainable incentives greatly increases the probability of financial success for high performing contractors. When all parties succeed, all parties benefit.

This research benefits from the application of each profession's individual arts and sciences towards collaborative value creation. Due to the high costs of design and construction, even small improvements in the project delivery process have the potential to yield significant savings in time, money, and quality. This research is relevant not only to the design and construction industries but also to owners or organizations with significant capital investment programs. These savings provide benefit not only to the owner but to all parties in the project delivery

⁴⁷ Jr. O'Connor, Patrick J., *Integrated Project Delivery: Collaboration Through New Contract Forms*, Faegre & Benson, LLP (Minneapolis, MN: The Associated General Contractors of America, 2009), 19.

process. Therefore, in addition to significant academic contributions to the field, this study is also significant to professional practice on the project site.

2.5.1. The Owner

An important part of selling an Integrated Project Delivery (IPD) method is identifying which owners and projects types are viable candidates to fully realize the advantages and benefits which IPD has to offer. An IPD process will be highly advantageous to many, but not all, owners. It is important that the study clearly defines which owners and project types most appropriate for IPD.

Integrated Design-Bid-Build (IDBB) project delivery is unique to the Department of Defense's Military Construction (MILCON) program and has been developed as an IPD process within the parameters of the established MILCON (public sector) rules, regulations and statutes. IDBB, or enhanced-IDBB can serve all federal public owners well, and other state and local public owners where local acquisition regulations permit IPD. Although this study is focused on public sector projects (MILCON), the methodology is certainly applicable to private sector owners as well. Multi-party Integrated Project Delivery methods, such as IPD, require intense owner involvement which may not be ideal for all owners or project types.

Owners unsophisticated in the project delivery process may not be able to fully add value to a collaborative process for which they have little understanding. It is possible for such owners to hire outside consultants to act as the owner's representative for participation in the IPD process. Some small, non-complex (simple), or 'boiler-plate' projects will not warrant the investment of time and resources required of intensive owner participation in multi-party IPD. Owners unable or unwilling to fully commit to a collaborative multi-party IPD method may find

38

the Design-Bid-Build (DBB) or Design-Build (DB) project delivery methods more appropriate for their situation.

Owners who have the most to gain from a multi-party IPD method are those who are: frustrated with traditional project delivery, and desire to actively participate in all aspects of the project delivery. These aspects are typically complex "one of a kind projects" with no standard design, require fast-track delivery, and are unclear on program and requirements. "They will know it when they see it", and are interested in maximizing value-creation.⁴⁸ These are the owners most likely to be influenced to adopt IPD based on the outcome of this study.

The content of this study, assuming the hypothesis is true, will sell the method to owners by demonstrating that IPD provides better value-creation than traditional project delivery methods for complex hospital projects. There are many critical success factors (CSFs) which contribute to overall project success and subsequent creation of value.⁴⁹ The importance of CSFs is weighted relative to the perspectives of the stakeholders (Owner, Designer, Builder, & User). This study will identify and analyze CSFs which project stakeholders use to define and assess project success. The CSFs will provide a framework by which to evaluate the relative effectiveness of different project delivery methods to successfully attain the most important project goals. Critical success factors an owner is most likely to be interested in include: owner satisfaction, cost performance, quality performance, time performance, and Innovation/ Improvement.⁵⁰ The results and statistical significance will be shown for all CSF.

 ⁴⁸ U.S. Army Corps of Engineers, "Early Contractor Involvement (EIC)," in *Industry Workshop*,
7 March 2010 (New Orleans, LA: U.S. Army Corps of Engineers, 28 January 2009).
⁴⁹ D.K.H Chua, Y.C. Kog, and P.K. Loh, "Critical Success Factors for Different Project

Objectives," *Journal of Construction Engineering and Management* 125, no. 3 (May/June 1999). ⁵⁰ John F. Y. Yeung, Albert P. C. Chan, and Daniel W. M. Chan, "Developing a Performance Index for Relationship-Based Construction Projects in Australia: Delphi Study," *Journal of Management in Engineering* 25, no. 2 (April 2009).

Demonstrating IPD yields higher measured effectiveness for value-creation compared to other traditional project delivery methods, especially for the CSF's the owner is most interested in, will make a strong case for owners to adopt an IPD method.

The MacLeamy Curve⁵¹ provides a graphic representation demonstrating significant design advantages of IPD. Figure 2.6 –The MacLeamy Curve shows the inverse relationship between cost of design changes and ability to impact project outcome (cost & function) over the duration of the project delivery process. The thin-line represents the "good idea cut-off" point along the project timeline. As the project progresses, the ability to implement "good ideas" to improve the design, correct errors, or otherwise enhance value-creation becomes limited, while at the same time the cost of such changes, represented by the dotted-line, increases. The bold-



Figure 2.6. Depiction of the MacLeamy Curve

⁵¹The "MacLeamy Curve" illustrates the advantages of Integrated Project Delivery. Introduced in the Construction Users Roundtable's "Collaboration, Integrated Information, and Project Lifecycle in Building Design and Construction and Operation" (WP-1202, August 2004)"

line represents design activity for an IPD process compared to design activity for traditional project delivery methods represented by the bold-dashed-line. By frontloading collaborative knowledge and coordination the IPD process shifts the design curve left keeping all design activities under the "good idea cut-off" line and reducing the impact costs of design enhancements or changes over the course of the project delivery duration. ⁵² Conceptually the owner is able to reduce cost and increase quality of the design compared to traditional project delivery. The MacLeamy Curve advantage is especially significant for owners with complex, innovative projects with fast-track requirements, or for owners who don't have clearly defined program and/or requirements. By introducing the builder to the conceptual phase of early design the builder can collaborate with the designer to adopt the most efficient means and methods.

Finally, IPD puts the owner in control of the entire collaborative process. With the multiparty contractual arrangement of IPD the Owner maintains privity⁵³ with both the design and construction agents. By contrast, with a Design-Build (DB) method, one contract, the Owner may not be able to custom-build the optimal pairing of designer and builder to suit their project requirements. Although DB is a more integrated approach than traditional DBB, the owner does not have the same control of the design process as in the IPD method. Depending on the weighting and statistical significance the research observes, the results of the study may be interpreted by owners as demonstrating IPD as superior to both DBB and DB.

⁵² Stuart Eckblad, Zigmund Rabel, and Jim Bedrick, The American Institute of Architects, "Integrated Project Delivery: Putting It All Together," in *AIA 150 Convention 2007: Growing Beyond Green*, Integrated Project Delivery: What, Why, and How (San Francisco, CA: The American Institute of Architects - California Council, 2 May 2007), 21.

⁵³ Contractual Relationship

The implications of a value-creation approach of this study, although valuable to the design and construction industries, is even more so to the owner. The Owners and Users, unlike the other stakeholders, realize the impact of the success, or lack thereof, of the project delivery process long after final payments and contract closeouts with the design and construction agents have occurred. The Designer and Builder walk away with lessons learned to do better on the next project, but the Owner/User is tied to the lifecycle operation of the completed facility. Even if IPD produces only small improvements in value-creation for the Owner during the short-term of the project delivery process, the impact is magnified greatly when considering the macroview, long-term impact additional value provides over the life of the facility.

2.5.2. The Designer

The Designer benefits from reduced predictive variables because the Builder provides instantaneous feedback of actual means and methods that result in fewer design errors and omissions. Improved design accuracy reduces Designer out-of-pocket design losses by eliminating the need to correct errors and omissions. The MacLeamy Curve discussed above is significant to the Owner, and the Designer as well. According to the Association of General Contractors the increased efficiency and accuracy resulting from IPD practices reduces the Designer's liability by up to 85%. The Designer benefits greatly from the resulting reduction in risk and also from better operational practices. An additional benefit to the Designer is the upside of increased fee potential due to extension of services provided.⁵⁴

2.5.3. The Builder

The Builder typically assumes the greatest risk exposure when entering into a construction contract. Therefore, there must be significant incentive for a construction contractor

⁵⁴ Bernstein and Hague, "Integrated Project Delivery (IPD): Why Owners Choose Multi-Party."

to enter into any given project delivery method. Due to the complex nature of a multi-party Integrated Project Delivery (IPD) process, it is even more necessary to clearly delineate each party's roles, responsibilities, and requirements. The conceptual framework of the contract must have quality terms which provide a strong incentive for the design and construction agents to agree to the multi-party collaboration.

2.5.4. The User

Typically the Users are consulted in the planning and programming stages and actively participate in design reviews. Often the User requirements get lost along the way, or the requirements change during the project delivery process. In the healthcare industry, state of the art technology and practices rapidly evolve and change. The integrated process requires more in term of User participation because the linear and methodical design review process of traditional project delivery is replaced by a dynamic non-linear integrated design process. By having the User fully on board, the project team ensures long-term operational goals are part of the decision cycle as the project progresses. The User has a better means of ensuring that an operationally optimal facility is provided.

2.6. Need for Study

There has been little academic research published regarding Integrated Project Delivery. Where significant academic research has been conducted the focus has been on the tools which facilitate an integrated approach to project delivery such as project information processing, communication and sharing; technology advancements which support collaborative design such as Building Information Modeling (BIM); or the administrative complexities of multi-party

43

contracts for collaboration.⁵⁵ Intuitively IPD appears to be superior to more traditional methods of project delivery, but few research studies focus on the value, or value-creation, of the IPD process on whole. There is much anecdotal evidence that IPD provides an owner with more value, but more substantive research is needed to explain how and why the IPD process creates value on a macro-level.

IPD increasingly is becoming a more recognized and popular method of project delivery across the design and construction industries. Although integration is not an entirely new concept, it is only in recent years that industry-wide attention has been given to the benefits of multi-party IPD. The American Institute of Architects was the first professional organization to attempt to address IPD multi-party contracts in 2007.⁵⁶ The Associated General Contractors of America (AGC) has recognized IPD as important to the future of the construction industry, but failed to endorse the AIA's new multi-party standard form contracts and has developed its own competing set of multi-party contract standard forms referred to as ConcensusDocs to address the IPD process.⁵⁷ The Design-Build Institute of America (DBIA), despite being one of the first professional organizations to promote an "integrated" approach to project delivery, has chosen to defend its single contract Design-Build process as superior to IPD. It is evident that there is not a common framework and understanding of the IPD process across the design and construction industries. Research which validates value creation concepts, goals, or practices has the potential to unify the industry and facilitate advancement of the industry's value-creation ability. The emergent nature of IPD across the design and construction industries calls for more research to validate and shape the evolution of IPD concepts to create equitable best value for all parties to

⁵⁵ Elvin, Integrated Practice in Architecture: Mastering Design-Build, Fast-Track, and Building Information Modeling.

⁵⁶ The American Institute of Architects, *Integrated Project Delivery: A Guide*.

⁵⁷ Bernstein and Hague, "Integrated Project Delivery (IPD): Why Owners Choose Multi-Party".

the IPD process. The U.S. military's IDBB pilot-projects are a pioneering effort in the use of Integrated Project Delivery practices for public sector projects. Careful examination of the experimental public foray into Integrated Project Delivery is essential for identification and justification for process improvements leading to better value creation and better stewardship of public resources. Although this study's focus is on military medical construction projects, the findings and applications are relevant to other project types both public and private.

CHAPTER 3.

LITERATURE REVIEW

3.1. Historical Overview

The history of the classic master-builder has been well documented by historians and even by their own writing. The earliest treatises and texts on architecture provide insight into both ancient traditions of the master-builder's comprehensive command of the entire project delivery process,⁵⁸ and the Architect's gravitation towards pursuit of the art of design separated from the art of construction.⁵⁹ As the schism between the designer and the builder has steadily widened over the centuries, literature mostly has ignored the gap between the two professions and little study has been devoted to bridging or closing the gap. Research and literature have focused primarily on the refinement of the art and science of project delivery within the paradigm of what has been known as the traditional project delivery method, Design-Bid-Build, or its derivatives.

3.2. Literature Specific to Integrated Project Delivery

At this time very little substantial literature specific to Integrated Project Delivery has been published. Due the emergent nature of the subject, the majority of published material has been limited to professional journals and conference papers. Both the design and construction industries have started forums for the exploration and development of Integrated Project Delivery and have published initial working-concepts of IPD which have sparked animated

⁵⁸ Vitruvius, *The Ten Books on Architecture, 1914, translated by Morris Hickey Morgan (New York: Dover Publications, 1st Century B.C.).*

⁵⁹ Leon Battista Alberti, *On the Art of Building in Ten Books*, translated by Josheph Rykwert, Neil Leach and Robert Travernor (Cambridge, MA: MIT Press, 1402-1472), 442.

discussion within the respective industries and hopefully will help plant the seeds for serious academic research.

George Elvin's book, "Integrated Practice in Architecture: Mastering Design-Build, Fast-Track, and Building Information Modeling" published in 2007 is the first book devoted entirely to Integrated Practice. Elvin provides a good overview of integrated practice but mostly in the context of Design-Build project delivery or boutique-firms specializing in integrated practices. He identifies practices and tools which enhance integration and provide a framework for IPD.⁶⁰

In 2007 the American Institute of Architects published "Integrated Project Delivery: A Guide". This document represents not only the design industry's recognition of the importance of IPD but served as a catalyst for dialogue and concept development within both the architectural community and the construction industry. The guide clearly states that it is a "working" document and intended to provide a forum for establishing and adopting common definitions and practices. An important change to the traditional DBB project delivery process and the more integrated DB process is the establishment of multi-party contract language, a more optimal project delivery process around the concept of IPD.⁶¹

James Pocock, in his doctoral dissertation, focuses on the degree of interaction (DOI) between stakeholders in the project delivery process. Pocock utilizes DOI to measure the level of communication between stakeholders, such as the architect and the construction contractor, in

⁶⁰ Elvin, Integrated Practice in Architecture: Mastering Design-Build, Fast-Track, and Building Information Modeling.

⁶¹ The American Institute of Architects, Integrated Project Delivery: A Guide.

collaborative process and compares it against project outcomes. He finds a direct correlation between DOI, as an indicator of project integration, and increased project performance.⁶²

Dr. Michael Kim proposes a Teleology Hierarchy in building design where the comprehensive designer, through full understanding of project integration, utilizes seven principles for maximization of value-creation: Utility Value, Aesthetic Value, Constructability, Financial Value, Long-term Serviceability, Meronic Value (a term used by Dr. Kim to describe contextual value as part of a whole) and Global Sustainability. The means-end relationship is focused into five levels, from conceptual to completed design, where project goals and design objectives inform design characteristics. Because in large complex projects the primary domain for decision authority for each of the seven principles is shared among multiple-parties, this Teleological Hierarchy provides an interesting framework for assessing value-creation of the IPD process. ⁶³

The Associated General Contractors of America (AGC) in 2009 established an on-going, webinar based Integrated Project Delivery forum. AGC thus far has focused on the contractors' and owners' perspectives of IPD. Although AGC embraces the multi-party IPD concept, it has chosen not to adopt the AIA's standard form IPD contracts and has developed its own competing standard f contract form: ConsensusDOCS 300: Tri-Party Collaborative Agreement.⁶⁴

⁶² James Bryant Pocock, "The Relationship Between Alternate Project Approaches, Integration, and Performance" (Ph. D. diss., Department of Civil and Environmental Engineering, University of Illinois at Urbana-Champaign, 1996), 131.

 ⁶³ Michael Kyong-il Kim, 2011. File: Manuscript in Progress. Integrative Design of Buildings:
Principles and Strategies. University of Illinois at Urbana-Champaign, Unpublished, Chapter 3.
⁶⁴ Bernstein and Hague, "Integrated Project Delivery (IPD): Why Owners Choose Multi-Party."

Bernstein and Hague, "Integrated Project Delivery (IPD): why Owners Choose Multi-Party."

3.2.1. Research in Areas Relevant to Integrated Project Delivery

There are many areas relevant to Integrate Project Delivery, and due to the limited published literature directly relating to Integrated Project Delivery, it is necessary to understand the history, environment, technology and other factors which have shaped the project delivery process. A broad base provides perspective by which to view the concepts of Integrated Project Delivery and provides a framework for deductive reasoning. One must understand and gauge previous and current project delivery methods to establish benchmarks for comparative analysis.

Understanding the origins of the master builder and the inherently integrated nature od project delivery at that time is important. The foundation for the logic of Integrated Project Delivery lies with understanding these early influential authors: Vitruvius, Alberti, Filerete, Palladio, and others.^{65,66,67,68} Their writings influenced design and construction for many centuries. Ross King's book "Brunelleschi's Dome" gives a rich account of the master-builders' work and all the stakeholders involved in what was at the time the most technically complex construction project to date.⁶⁹ From such literature many parallels to today's challenges to an Integrated Project Delivery process are observed.

Epistemology of Design. Authors such as Guach and Kuhn discuss concepts of paradigms in context of revolutionary change in science; these concepts also apply to revolutionary change in project delivery. The evolution of the project delivery process such as DBB, DB, and IPD can be viewed as paradigm shifts in the design and construction industry.

⁶⁵ Vitruvius, The Ten Books on Architecture.

⁶⁶ Alberti, On the Art of Building in Ten Books.

⁶⁷ Andrea Palladio, *The Four Books on Architecture*, 1997, translated by Robert Tavernor and Richard Schofield (Cambridge, MA: The MIT Press, c1570), 436.

⁶⁸ Filarete (aka Antonio di Piero Averlino), *Filarete's Treatise on Architecture*, c1460, translated by John R. Spencer (New Haven: Yale University Press, 1965).

⁶⁹ Ross King, *Brunelleschi's Dome: How a Renaissance Genius Reinvented Architecture* (New York: Penguin Books, 2000).

(Gauch Ch3) Both Kim and Brown discuss integration principles within the context of comprehensive design.^{70,71}

3.3. Conclusions from Literature

What is known?

- Much of the evidence that IPD is a superior method of project delivery is anecdotal; there has been little formal research.
- Leading design and construction professional organizations have both recognized a trend for rapidly increasing demand for IPD services.
- Leading design and construction professional organizations do not currently agree on how IPD multi-party contracting should be structured.
- Emerging technology and tools are minimizing the barriers to IPD.
- There is a premium cost to providing for IPD.

What is not known?

- Universally accepted definition of IPD.
- Value metrics for many aspects of the IPD process

3.3.1. Contribution to the Literature

The research provides a better understanding of how the IPD process creates value relative to other project delivery processes. The contribution to the fields of design and construction is the demonstration and defense of Integrated Project Delivery as a superior method of project delivery for large complex projects, a better understanding of value-creation, and a foundation for further study and advancement of the project delivery process. Potential of improved project value creation is very important to providing a better built environment for the

⁷⁰ Kim, "Countermodeling as a Strategy for Decision Making: Epistemological Problems in Design."

⁷¹ Dale Ray Brown, "A Designing Image: Integrating Design, Planning, and Decision Theories with Cognitive Processes" (Ph. D. diss., Architecture, University of California, Berkley, 1992).

world in an economic climate that demands better, more efficient ways of doing business. The value-creation approach to studying the project delivery process will shed light on many aspects of the project delivery process which may be studied in the future, in more depth to further refine the project delivery process at micro-levels but with macro-level impacts. This study's research of the relationship between value creation and the Integrated Project Delivery process will argue for a normative model for optimal project delivery of large complex projects, and will serve as a platform for future literary contributions on the topic of IPD.

CHAPTER 4.

THEORETICAL FRAMEWORK

4.1. Introduction

Chapter four discusses the key concepts which provide have guided this research project from the very beginning. The reason for a better project delivery process is a desire to maximize value-creation. The argument is not that the traditional project delivery method or any other is bad, rather the argument is that there is always a better way. It is our nature to be innovators always striving to improve the status quo. The following provide the conceptual basis for the research approach and method:

- The relationships between goals and value-creation
- Stakeholder perceptions of relative importance and value-creation
- Macro versus micro viewpoints of measuring project success
- Logically superior effectiveness of Integrated Project Delivery
- Implementability

4.2. Relationship between Goals and Value-Creation

Relationship between goals and value creation, or more specifically the gap between goals established at earliest design conception and the ultimate value of the completed building in traditional project delivery methods, provides the framework by which to evaluate the effectiveness of Integrated Project Delivery to maximize value creation.

Goals change as the design progresses:

"In particular, Rittel with Webber (1973) points out that every design problem may be understood as a symptom of another problem, and therefore, the goal or goals of a design problem depend on the level of perception of the 'whole' at which the 'whole' is conceptualized. Furthermore, he says, whether or not a problem is worth solving depends on how it is going to be solved, as the side and after effects of certain solutions could be utterly undesirable. Yet the 'whole,' including the side and after effects, cannot be known until the designer works on the problem. Therefore, the goal changes as the designer moves along."⁷²

A value of an IPD process is limiting fluctuation of goals. It is critical to establish the integrated process at the time the "whole is conceptualized". All IPD stakeholders must contribute to the definition and selection of goals. If the IPD process does in fact limit the designer's predictive variables as discussed above, then there should be a corresponding calming to goal fluctuation and increased optimization of value-creation.

4.2.1. Perspective of Stakeholder Goals

Each participant in an Integrated Project Delivery process brings his/her own unique background of expertise and experience together in a collaborative process. Likewise each stakeholder brings their own agenda and extra-project goals. The professional qualifications for each stakeholder which make them best suited for the collaborative team also establish unique foundations from which each gains individual perspective of the risk, rewards, and requirements of the project delivery process. Each stakeholder in a collaborative, multi-party process will have different contractual obligations, scopes of work, deliverables, and associated goals. The rationale behind an integrated project delivery process is to optimize the process so that it maximizes benefit to all parties to meet their goals and a common goal. The result is to create a Win-Win scenario where individual strengths of each party mitigate other's weaknesses and thereby increases optimization of the overall project delivery process. The relationship between

⁷² Kim, "Countermodeling as a Strategy for Decision Making: Epistemological Problems in Design", 17.

goals of each party of the Integrated Project Delivery team falls into two categories: (1) independent goals and (2) dependent goals.

This study seeks to identify which goal relationships between owner, designer, builder and user are most important to individual and overall project success. Project stakeholders are surveyed to identify which critical success factors they believe are most important. Because, project delivery is a dynamic and difficult to model process, there are many individual, discrete activities along the project continuum which lend themselves to optimization through modeling and study of game theory. Examples of some include bid strategy, resource balancing, and scheduling. Typically, game theory simplifies complex problems in order to facilitate a workable model or mathematical formula. ⁷³ In the real-world, the project delivery process is highly complex, and the already dynamic process is further complicated by external forces such as volatile market conditions for construction materials, weather/natural disasters, and unforeseen site conditions. There are also social and political factors which can limit productivity or in a worst-case scenario shut down the project.

All of these factors require the project stakeholders to consider multiple conflicting objectives when making decisions individually and as a project team. Where there is a conflict between the stakeholders' objectives, a problem of value tradeoffs exists where the achievement of one objective is traded off against another objective. ⁷⁴ For this reason it is important for the stakeholders to have an understanding not only of the objectives and goals of the project team as a whole, but also of each of the other stakeholders to allow for a more informed collaborative

⁷³ Elliott Mendelson, *Introducing Game Theory and Its Applications* (Boca Raton, FL: Chapman & Hall/CRC, 2004), 169.

⁷⁴ Ralph L. Keeney, Raiffa, *Decisions with Multiple Objectives: Preferences and Value Tradeoffs* (New York: John Wiley & Sons, 1976), 66.

environment, to better understand the framework of goals within the context of the project delivery team and the different types of goal relationships that exist.

4.2.2. Independent Goals

It must be recognized that each party in a collaborative process will most certainly have internal goals, or personal agendas independent of all other stakeholders. Such goals may be as basic as corporate survival and desire to maximize profit, or they might nest into a larger set of long-term organizational goals across a wide spectrum of projects and project types outside of a specific collaborative process. Independent goals do not impede the collaborative process or jeopardize or impede goals of other stakeholders, they do not have a negative effect on the project delivery process. And, they need not be fair or equitable when they are brought into the collaborative discussion.

4.2.3. Dependent Goals

In simplest terms dependent goals are akin to the old saying "You scratch my back, and I'll scratch your back". Not back-scratching in a quid pro quo sense, but in a way where supporting your collaborator's goal is necessarily beneficial to achieving your own party's goal. The network of dependant goals in practice is often much more complex than simple reciprocation between two parties. Often second, third, or greater order dependencies must play out to confer the desired benefit to the intended parties. These goals are very important to identify because they represent the key to a win-win collaborative interaction between parties brought together by the Integrated Project Delivery process. The establishment of a symbiotic relationship is not out of necessity when each party is in itself is capable of efficient and productive contribution to the success of project execution. Rather it is a means of achieving a higher level of optimization where the whole is greater than the sum of the parts. In the spirit of

55

collaboration, dependent goals ideally should be fair and equitable to the parties involved. The dependency relationship between project stakeholders may be formal (contractual) or informal (partnering). For the purposes of this study, dependent goals are divided into the sub-categories of obligatory (intra) and discretionary (extra). These subcategories are relative to the collaborative framework of each party's formal contractual obligations.

4.2.4. Obligatory Dependant Goals

Obligatory goals are those dependent relationships which are clearly articulated in the general conditions and/or supplemental conditions of the contracts between the owner and the Integrated Project Delivery members. Unlike traditional DBB, the collaborative integrated process is dependent on non-linear interaction between the Designer and Builder. Design activities that would be solely the Designer's responsibility in a traditional DBB process are shared responsibilities in an IPD process. The Designer is dependent on the contractor's input and agreement on incremental design packages. Likewise the contractor is dependent on a synchronized and continuous flow of design packages to allow scheduled progress of construction. In game theory this concept is known as goal interdependence, a concept where no one group member can achieve his/her goal unless each other member achieves their own.⁷⁵

The building owner provides the institutional mission and vision which establishes the stated end goal of what purpose a project is to fulfill. The mission and vision are directed goals (obligatory) which all participants must adopt to meet the needs of the owner and satisfy the terms of their individual contracts. Directed goals must be shared by all parties, in accordance with contract, to successfully execute the project delivery process. Beyond the owner mandated mission and vision, Dr. Kim identifies four levels of teleological hierarchy in building projects

⁷⁵ Morton Deutsch, *The Handbook of Conflict Resolution: Theory and Practice*, edited by Morton Deutsch and Peter T. Coleman (San Francisco, CA: Jossey-Bas Publishers, 2000), 42.

which identify goals of integrated design: Project Goals, Design Objectives, Required/Desired Characteristics, and Resulting Design itself.⁷⁶ Likewise for the integrated construction element, as in the IPD project delivery process, the Construction Objectives such as quality, time, cost, safety, as contractual requirements must be added to the inclusive list for both the designer and builder without affecting others outside of the project.⁷⁷

4.3. Goals Define Value-Creation

Successful completion of goals is a requirement of value-creation. But, what are the values which result from design and construction of a building project? Classical definition of value stems from Vitruvius' Ten Books on Architecture where he listed firmitas, utilitas, and venustas as required qualities for buildings.⁷⁸ To Vitruvius structural soundness, operational utility, and aesthetic beauty are the values conceived by design and created by construction. Vitruvius provides what he believed to be the three categories of value created by a building. Firmitas, utilitas, and venustas are certainly important goals, yielding great value, but they are not all encompassing of modern requirements and realities. As Dr. Kim suggests there are additional aspects of value created by design and construction of buildings which Vitruvius did not consider: Economic Value, Constructability, Long-term Serviceability, Sustainability, and Contextual (Meronic) Value.⁷⁹

 ⁷⁶ Kim, Michael Kyong-il. 2011. File: Manuscript in Progress. Integrative Design of Buildings: Principles and Strategies. University of Illinois at Urbana-Champaign, Unpublished, Chapter 3.
⁷⁷ Michael Kyong-il Kim, ARCH 525 Class Notes, Spring Semester 2003, University of Illinois Urbana-Champaign

⁷⁸ Vitruvius, *The Ten Books on Architecture*, 331.

⁷⁹ Michael Kyong-il Kim, "What Would You Say Now, Mr. Vitruvius? Building Design Teleology, Then and Now," paper presented at the ConnectEd 2010 - 2nd International Conference on Design Education, 28 June - 1 July University of New South Wales, Sydney, Australia, 2010, 4.

4.3.1. Relationship of Goals

The formal and informal relationships that exist between each party to the Integrated Project Delivery process creates a framework through which both independent and dependent goals are formed and acted upon. It is this complex web of interrelated goals which creates



Figure 4.1. Goal relationships

desired value from the perspectives of each IPD team member. Figure 4.1 depicts each project team party (owner, designer, & builder) and the goal relationships for traditional Design-Bid-Build, Design-Build, and Integrated Project Delivery models. Areas depicted with the letter "I" represent each parties independent goals."D1" represents those goals which are formally defined

by terms of the prime contract(s) between the owner, designer, and/or builder. Note that in the Design-Build project delivery process there is no privity⁸⁰ between the government Owner/User and the internal design element of the Design-Build contractor. In some rare cases the designer may be the prime contractor with an internal construction group or partner, but it is unlikely for a designer to be the dominate partner of a military Design-Build contract. The IDBB process has a similar contractual relationship as traditional DBB where the Owner holds separate prime



Figure 4.2. Users as part of the project delivery team

contracts with both the designer and builder. However, the terms of each separate contract stipulate integrated practice and shared project goals -- D1 obligatory contractual goals. The relationship is more complex but also affords more opportunity for D2 Discretionary/Partnering

⁸⁰ Sweet and Schneier, *Legal Aspects of Architecture, Engineering, and the Construction Process*, 54.

goals to be achieved through partnering for increased value-creation amongst compatible independent goals. It is this 'extracurricular' mutual goal support which may offer greater overall value-creation from the perspective of all parties involved. This research takes and adds the users group to the list of primary stakeholder groups as depicted in Figure 4.2.

4.3.2. Measuring Value of Goal Accomplishment

Over the course of the project delivery process there are many goals. Of the numerous goals, this study will categorize goals which will facilitate measurement of value-creation. Preliminary study surveying project stakeholders will be utilized to narrow the focus to the goal(s) in practice which the integrated project teams identify as most important to overall project value-creation. Later discussions in the research methods chapter will provide further details of how the research will be conducted.

Once a project goal is determined to be critical, it will be further analyzed in the context of value-creation. For example, constructability may be a critical project goal. In theory an integrated collaborative process would completely eliminate any errors in design communication between the designer and builder thus enhancing constructability. Elimination of design errors/omissions and contractor misinterpretation of design intent prevents costly change orders and delays to construction. The cause of change requirements is most often breakdowns in communication. Communication is a critical tool to breaking down collaboration barriers. Minimizing communication errors by increasing the degree of interaction among the parties increases collective value creation while maintaining balance of individual benefits.⁸¹

⁸¹ Pocock, "The Relationship Between Alternate Project Approaches, Integration, and Performance", 131.
4.4. Macro Viewpoint

Perceptions of project success vary by stakeholder. The Owner and/or other stakeholders may suffer losses, and from their perspective the project may have failed. However, from the User perspective the project may be an outstanding success. Or conversely in the short term the project may be an absolute success in terms of cost, schedule, and quality, but fail to produce the utility demanded by the users and customers. It is easy in a micro-context to define success or failure to meet project goals or objectives deemed critical, and overlook the overall sum of all other factors when assessing overall project value. It is natural to obsess over financial aspects of the construction phase of the project delivery process. In isolation there is nothing more important than the big three: cost, schedule, and quality. No doubt the reality of capital cost cannot be discounted. However, there are significant considerations often overlooked when assessing success at only a micro-level and ignoring the macro-level.

A problem with traditional project delivery methods and traditional assessments of project success is the failure to consider long-term impacts outside the micro-views of the design or construction phases of a project. In fact, the two most critical phases for assessment of valuecreation and project success from a macro-level are completely outside the design and construction phases of project delivery. It is the conceptual phase and operational phase which define project success, and determine a measure of value-creation.

Ultimately, there are only two criteria for assessing project success. The first criterion is project completion. Either the project is completed or it is not. Relative success may then be assessed based on conditions of completion, or how well it was completed. The second criterion,

after project completion, is satisfaction. A measure of how well the completed project meets the original concept in practice.⁸²

Any project delivery method may successfully complete a project, and meet the minimal conditions for micro-viewpoint success and possibly satisfy original goals. When considering the macro viewpoint it is important to consider when and how the conceptual up-front decisions are made. Are the decisions informed by preconceptions of inherent limitations in the project delivery process? Are the decisions made with incomplete information? What was the accuracy of the assumptions and estimates?

The macro-viewpoint is defined by the two phases which bookend the project delivery process timeline. The conceptual phase and the operational phases are the alpha and the omega in determining project success. The points in between, the myriad of design and construction activities, form the basis for the micro-viewpoint of project success. The majority of the body of research into the project delivery process has focused on the micro-viewpoint in relation to the big three: cost, schedule, and quality. ⁸³

A problem with looking exclusively at the micro-view, especially within the context of the traditional project delivery method, is two-fold. First, the gaps between the conceptual phase, where goals which define the ultimate project success are established, subsequent design and construction activities are problematic due to (a) the lack of continuity between stakeholders and (b) often significant time lapse between conceptual decisions and project execution. The conceptual development team is at risk of inaccurate estimates/assumptions.

 ⁸² C.S. Lim and M. Zain Mohamed, "Criteria of Project Success: An Exploratory Re-Examination," *International Journal of Project Management* 17, no. 4 (August 1999): 245.
⁸³ Lim and Mohamed, "Criteria of Project Success: An Exploratory Re-Examination," 243.

Measuring "project success" purely from the micro-viewpoint may not tell the whole story. Often projects are grossly underestimated during conceptual phases, and decision makers bite off more than they can chew. Usually by making over-optimistic estimates of projected efficiencies, cost savings, or schedule durations. Think of the old saying... "If you want it bad, you'll get it bad". Such mistakes are an open invitation to opportunistic bidding, especially in a highly competitive bid environment. Opportunistic bidding is where a contractor low-bids knowing he/she can make up the difference in beyond contract rewards through change-orders and/or claims. ⁸⁴ Opportunistic bidding may enrich one stakeholder, but diminishes the overall project value-creation.

In cases where over-optimistic, under-estimated conceptual decisions move forward to project execution, the design and construction phases have an up-hill battle that may or may not be won at the micro-level. Is it fair to gauge micro project success against flawed concepts? It is not fair when considering the conceptual mistakes that were made outside the control of subsequent stakeholders. A better way would be to measure success from a rectified benchmark of what the conceptual baseline should have been.⁸⁵ A better solution is to utilize an Integrated Project Delivery method to provide continuity of stakeholders throughout the entire process.

Often project success is viewed only from the micro-viewpoint. The micro-viewpoint relies on project completion criteria (cost, schedule, & quality related activities) to assess success. Lim and Mohamed argue that satisfaction criteria are what set apart the macro and

⁸⁴ W. Lo, C. L. Lin, and M. R. Yan, "Contractor's Opportunistic Bidding Behavior and Equilibrium Price Level in the Construction Market," *Journal of Construction Engineering and Management* 133, no. 6 (1 June 2007): 409-16.

⁸⁵ Terry Williams and Knut Samset, "Issues in Front-End Decision Making on Projects," *Project Management Journal* 41, no. 2 (April 2010): 38-49.

micro viewpoints.⁸⁶ A macro-viewpoint takes into account how well the operationalized product provides outcomes which satisfy the conceptual goals which launched the project delivery process.

This research investigates the macro-view of the project delivery process. Adding the User stakeholder group to the collection of key project stakeholders is a necessary step in adopting the macro-viewpoint. As stated above, the macro-view of project success and value-creation emphasizes two phases in which the user stakeholder group is very importance: the conceptual phase and the operational phase. Additionally, the research will include factors preceding and transcending the design and construction phases of project delivery. This is the motivation behind including the *Process* and *Impact* categories of critical success factors along with the *Design* and *Construction* categories which are used to categorize the critical success factors in the round 1 survey to be discussed later.

4.5. Research Concept:

A detailed discussion of the research methods appears in Chapter 6, the following provides the conceptual basis for the research. The research investigates the effectiveness of Integrated Project Delivery to create additional value over traditional project delivery. Significant project stakeholders across the spectrum of the project delivery process were surveyed to get their thoughts on the project delivery process. The primary observation from the surveys was that each stakeholder group appeared to evaluate differently cost, benefit, and/or value from each of unique perspectives of owner, designer, builder, and user groups. To further investigate how the differing perceptions relate to value-creation and efficacy of project delivery a pilot survey was initiated and further developed into the research survey discussed in this

⁸⁶ Lim and Mohamed, "Criteria of Project Success: An Exploratory Re-Examination," 243.

dissertation. Value is assessed relative to important project goals which should be shared by all stakeholders in an integrated process.

The research takes a comprehensive look at the entire project delivery process from planning through construction through operations. Because the Integrated Project Delivery process is non-linear, a comprehensive approach is necessary to understand the dynamic interaction between design and construction agents. The broader focus enables a means to find hidden opportunities for additional optimization of the project delivery process, more so than if a narrower focus is taken. This study focuses almost exclusively on large scale healthcare projects, but also draws on lessons learned from one additional non-medical project of similar scope and complexity to provide a larger sample population. The complexity and significant design and construction challenges associated with planning, designing, and building healthcare facilities presents significantly more opportunities for process optimization than simpler or more routine facility types.

Although this study is not centered on *game theory*, a broad range of game theory principles was reviewed to provide a basis for understanding the impact which interrelation between project delivery stakeholder groups has towards goal attainment and ultimately valuecreation. The game theory review raised the concept of evaluating each primary research question from the perspectives involved in collaborative games: the balance of attaining goals with a beneficial outcome for each partner and the group as a whole.

Goals are abstract general intentions which the parties, either independently or collectively, wish to achieve.⁸⁷ Typically the primary goals of project delivery center on cost, quality, and schedule. These goals in their broad context are not measureable. For example, each

⁸⁷ Kim, 2011. File: Manuscript in Progress. Integrative Design of Buildings: Principles and Strategies. University of Illinois at Urbana-Champaign, Unpublished, Chapter 3.

party likely has a goal to be more profitable, and a strong desire to successfully accomplish this goal. Profit is easily quantifiable by measuring the positive gain after subtracting all expenses, yet measuring the success of accomplishing the goal to be more profitable is not easily quantifiable. The yardstick necessary to measures success is dependent upon subjective, qualitative assessments unique to each project stakeholder's perspective. There are many interrelated aspects of the project delivery process which impact the financial outcomes of each project stakeholder and its resulting measure of profitability. It is not possible to apply a specific measurement to open ended, abstract concepts such as a "goal to be more profitable". How much more profit equals success? Under what circumstances does minimizing loss of profit equal success? In some cases, although profit is a desired goal, there may be other goals which outweigh the pursuit of profit at all costs. It is possible one of the project stakeholders, although desiring short-term profitability, may have a more important long-term strategy where a loss leader⁸⁸ is more beneficial in the broad context of profitability outside a single project.

The example discussed above demonstrates how a clearly quantitative aspect of project delivery, the calculation of profit, is transformed into a more complex qualitative problem when attempting to define and measure project success from a macro-viewpoint. Qualitatively important aspects of project delivery provide even greater challenges to measuring success. Quality is an abstract concept dependent on subjective definition and interpretation. Unlike profit, there is no clear quantitative equation from which to define or derive quality. In the case of quality goals, the aspect of the quality to be measured and the yardstick by which it measures successful attainment are both qualitative and subjectively defined. Akin to the old adage "beauty is in the eye of the beholder", quality success as measured by attainment of even the

⁸⁸ Loss leader is a marketing concept where products or services are sold at a substantial discount, or loss, in order to generate future business leading to a larger net profit.

strictest adherence to plans and specifications may please the owners, designers, and builders, yet greatly disappoint users and customers if the plans and specifications were inadequate to begin with.

Schedule goals require the assessment of time related activities which rely on both quantitative units of measurement (hours, days, months or years) and qualitative aspects of timescale (speed). The schedule estimates and evaluates the durations of all project activities based primarily on the subjective analysis of project requirements and resource allocation.

Goals themselves cannot be measured, but goals do provide a framework of components which may be measured. These components are called objectives. Objectives, unlike goals, are by definition specific, measurable, achievable, relevant, and time-based. Project objectives must be identified and weighted for relative importance, and then analyzed to determine how they may be effectively achieved. Measurement of objective accomplishment requires an unbiased specific yardstick, or metric by which actual performance may be compared. Different parties to the Integrated Project Delivery process may share the same objective, but assess success by different measures.

Success is relative to the perspective from which and by whom it is defined. There may be customary, or industry standard definitions of what success ought to be, but because success is a subjective concept there is no absolute measure of success. Any relevant and measurable definition of "what success is" will suffice. With parameters of success defined, the ultimate success or failure to attain each objective is determined by many project factors. Logically, the more complex a project is, the more it's potential for successful outcomes is impacted by a multitude of factors. The impact each factor has on outcomes varies greatly due to the complex interdependent relationships inherent in planning and completing design and construction

activities. More important than the definition of success is the identification of the Critical Success Factors (CSFs) which have the greatest impact on objective outcomes. It is likely not feasible, and certainly not desirable to exhaustively measure all success factors. Identification and prioritization of CSFs allows weighted analysis of only those factors most important to determining the level of objective attainment in support of project goals.

It is difficult to measure the relative importance each CSF has in relationship to each objective. Other researchers have successfully demonstrated that an Analytical Hierarchy Process (AHP) may be used to meaningfully compare project factors across the range of stakeholders and associated CSFs.^{89,90,91,92,93,94}

The first step towards measuring how well-off the project stakeholders are in terms of creatingvalue is to identify the critical success factors (CSFs) discussed above.

CSFs are identified by a panel of experts through the use of a Delphi method survey.

Three rounds of Delphi surveys were conducted to gather data sufficient to support analysis of factors perceived as important to achieving of project success. The same Delphi panel of experts also provides an assessment of different project delivery methods to successfully attain the most important critical success factors.

 ⁸⁹ Chua, Kog, and Loh, "Critical Success Factors for Different Project Objectives, 142-50.
⁹⁰ Yeung, Chan, and Chan, "Developing a Performance Index for Relationship-Based

Construction Projects in Australia: Delphi Study", 59-68.

⁹¹ Cindy L. Menches and Awad S. Hanna, "Quantitative Measurement of Successful Performance from the Project Manager's Perspective," *Journal of Construction Engineering & Management* 132, no. 12 (December 2006): 1284-93.

 ⁹² Maria Kliniotou, "Identifying, Measuring, and Monitoring Value During Project Development," *European Journal of Engineering Education* 29, no. 3 (September 2004): 367-76.
⁹³ J.K Pinto and J.G. Covin, "Critical Factors in Project Implementation, a Comparison of Construction and R&D Projects," *Technovation* 9, no. 1 (1989): 49-62.

⁹⁴ Florence T.T. Phua and Steve Rowlinson, "How Important is Cooperation to Construction Project Success? A Grounded Empirical Quantification," *Engineering, Construction and Architectural Management* 11, no. 1 (2004): 45-54.

4.6. IPD: Logical Normative Model

4.6.1. Formation of the project delivery team

As noted earlier in Chapter 2, game theory research tells us that the formation of the project team to maximize group utility is extremely important.⁹⁵ Assuming incomplete information, the owner must consider the probability of each stakeholder's contribution to a successful outcome. There must not be a weak link in the chain. Group dynamics is of the utmost importance and weak-links in interpersonal skills or intergroup dynamics must be accounted for in the general condition for administration of the integrated team. The group must have the ability to rapidly address personality issues which become roadblocks to collaboration and cooperation. Stakeholder leaderships must be committed to maximizing group utility. For project delivery to successfully reach its potential for maximizing value-creation, the sum of each stakeholder's utility must be superadditive.⁹⁶ A primary assumption of the Integrated Project Delivery process is the creation of a 'superadditve' project team relative to value-creation where the whole is more important than the sum of its parts. Why shouldn't the benefits be shared with the other project stakeholders? To entice the best performers to engage in the quest for greater value-creation there must be some benefit to do so. Therefore the process must be designed as beneficial to all participants to be implementable.

A purpose of seeking a better process for project delivery is to maximize value-creation. Value-creation for whom? Certainly additional value-creation for the owner, users, customers, public, or anyone else who will utilize the facility during its life-cycle is of great benefit. Especially, for public projects funded by with public dollars. All stakeholders should reap the rewards for any real attainment of the theoretically maximized value-creation. The owner and/or

⁹⁵ Weirich, Collective Rationality: Equilibrium in Cooperative Games, 66.

⁹⁶ Curiel, Cooperative Game Theory and Applications: Cooperative Game Arising from Combinatorial Optimization Problems, 2.

users obviously will benefit from both short-term savings in project costs relative to the original project budget. However, the long-term benefit of lifecycle value-added cannot be ignored. For example, the case study hospital projects are facilities that will provide work environments for thousands of employees, contain complex resource intensive technologies and systems, and in the IDBB pilot-projects when operationalized will interface with more than 10,000 patients/customers per week. The operational costs of such facilities are immense. Any efficiencies or long-term benefit yielded from a more optimal project delivery process is highly desirable.

Which project delivery process best enables collaboration, and has a formal contractual framework to incentivize optimal collaboration where each stakeholder is compensated for foregoing self-interest (and assumed profit) for the overall, best-interest of the project? Logically Integrated Project Delivery, by definition, provides the answer to this question. IPD is a process designed around collaboration across the entire project delivery continuum.

4.6.2. Macro-Focused

Traditional Design-Bid-Build by definition fragments the project delivery process into separate micro-components of design and construction phases. Design-Build provides an improved, and more integrated approach, yet leaves room for additional optimization especially in public sector DB projects which require preliminary design services for adequate documentation to put an RFP out on the street. IPD's collaborative approach assembles the micro-level project phases and stakeholders: owners, design, construction, and users into a process that must adopt a macro-focus to achieve the desired collaboration.

4.6.3. Alignment of Goals and Objectives

Earlier in this chapter was a discussion about the relationship between goals, objectives and value-creation. Establishing that value-creation is the result of successful goal attainment, and likewise results from successfully achieving objectives, the matter of a superior project delivery method rests with establishing what project delivery method is most effective in attaining the most important goals and objectives which both define and determine valuecreation. As discussed in section 4.4 above, it is the conceptual and operational phases of a project the determinants of value-creation. Two questions require consideration:

- (1) What project delivery method is most effective in attaining the most important critical success factors related to value-creation?
- (2) What project delivery method is designed to transcend the micro-phases of project delivery and provides continuity and alignment of goals and objectives from concept to completion and beyond?

Again there is a strong argument that Integrated Project Delivery logically is the most effective at addressing each of these questions.

4.6.4. Implementability

The final logical imperative for Integrated Project Delivery is its implementability. There is no value in IPD if it is only a good idea. Project stakeholders must believe that there is a rational basis for allocating resources for implementing an IPD process. Using a normative approach to evaluating the merits of IPD as compared to other established project delivery methods provides the framework by which project stakeholders may assess relative effectiveness, and infer implementability. More substantial proof that IPD is implementable is the successful completion of IPD projects. It has been done before, and there is desire to do it again. The IDBB pilot-projects are prime examples of Integrated Project Delivery in action.

4.6.5. Linkage to research

This research project is designed to test the logical conclusions that Integrated Project Delivery is the normative model for large complex military construction projects. The first four chapters of this dissertation have provided the background and theoretical framework for the argument that IPD is the normative model for value-creation. Chapter 6 will elaborate on the research method for providing evidence in support the logical conclusion.

CHAPTER 5.

IDBB IN PRACTICE

5.1. Introduction

The IDBB process is not perfect and hasn't been without its problems. It is after all a pilot-program to test Integrated Project Delivery in the context of military construction. Moreover, it is doubtful a more challenging application for the pilot-projects could have been chosen. The enormous scope and cost of the IDBB projects, coupled with the BRAC related fast-track schedule did not allow time for a deliberate entry into the process. As the four projects near completion the Owner stakeholder group already has implemented changes to the MILCON approach to Integrated Project Delivery. The IDBB pilot-projects have spawned the next batch of integrated projects which are now referred to as Early Contractor Involvement (ECI), and the lessons learned from those projects will in turn spawn the next alphabet-soup of Integrated Project Delivery.

It is clear from the survey results discussed in later chapters that key leadership believes an integrated approach is the most effective means of achieving the most important project factors. The leadership challenge remains how to best implement integrated practices into a regulatory environment not optimal for streamlining any process. The progress that has been made in the five years since initiating the IDBB pilot-projects has been impressive. The following is a brief summary of government reports related to the evaluation progress of the IDBB pilot-projects to date.

5.2. Time performance

Three of the four IDBB pilot-projects are on track to be completed on time or ahead of the 15 September 2011 BRAC deadline, including the two largest and most complex projects: The Fort Belvoir Community Hospital, and the National Geospatial Intelligence Agency's (NGA) New Campus East, both at Fort Belvoir, Virginia. The San Antonio Medical Center -North project is making good progress but is behind schedule by at least six months. It is very common for such large and complex projects to go over schedule when considering all the technology and systems which must be commissioned and certified for accredited hospital operations. The real work starts in a hospital after the enclosure is completed.

The Fort Belvoir Community Hospital is on track to start seeing patients in the new facility nearly 60 days prior to the 15 September 2011 BRAC deadline for project completion. Never in the history of the Army Medical Department has hospital as large and complex been designed and built in such a short time. Prior medical MILCON projects of similar size such as the Womack Army Medical Center at Fort Bragg, NC; Brooke Army Medical Center at Fort Sam Houston, TX, and Madigan Army Medical Center at Fort Lewis, WA typically took between two and three years to design and an additional four to five years to construct utilizing the traditional Design-Bid-Build project delivery method. The IDBB process eliminated at least two years off of the project delivery timeline.

NGA reports the project is on time and on budget. It is amazing to consider a \$1.7 billion; 2.4 million square foot campus project to relocate nearly 9,000 workers is on track to be completed in a span of less than five years for both design and construction. To put the NGA project in perspective, an average of nearly a million dollars (\$930,000) of construction had to be placed per day for every day over the five year duration of the project.

5.3. Quality Assessment: Fort Belvoir Community Hospital

In early 2009 an Armed Services Subcommittee was established by Congress to assess the project delivery performance of one of the IDBB pilot-projects and another DB project of similar scope and cost in the same geographic area. The subcommittee released its initial report, "Achieving World Class: An Independent Review of the Design Plans for the Walter Reed National Military Medical Center⁹⁷ and the Fort Belvoir Community Hospital"⁹⁸ in late November 2009. The report has identified performance variances between the IDBB project at Fort Belvoir, VA, and the Design-Build Projects at Bethesda, MD. The IDBB project was found to have produced a higher quality design to meet the congressionally mandated world-class standard for provision of military healthcare, and consideration incorporation of evidence based principles and features into the facility design. A significant challenge for both projects was the "moving target" of the definition of "world class". The IPD concept as discussed in section 2.5.1 (The MacLeamy Curve) allowed the project team to effectively incorporate required changes in design to ensure continuity of evidence based design goals and evolving expectations of "world class". The results of the congressional study support the hypothesis of this research study on at least one aspect of value-creation - perceived quality as benchmarked against project goals. The report was focused on congressional concerns of what the medical operational quality of the completed facilities would be, and if it would meet expectations to provide world class

⁹⁷ The National Naval Medical Center in Bethesda, Maryland will be renamed the Walter Reed National Military Medical Center upon completion of the major addition and alteration projects underway there, and the closure of the Walter Reed Army Medical Center in Washington, D.C. ⁹⁸ Kenneth W. Kizer, Merrily McGowan, and Sheila Bowman, *Achieving World Class: An Independent Review of the Design Plans for the Walter Reed National Military Medical Center and the Fort Belvoir Community Hospital*, National Capital Region Base Realignment and Closure Health Systems Advisory Subcommittee of the Defense Health Board (Office of the Assistant Secretary of Defense for Health Affairs, 2009).

healthcare. Congress has mandated a marco-viewed focus of value-creation to ensure the longterm success operations of the newest military medical facilities.

5.4. NGA Campus East Lessons Learned ⁹⁹

It was noted in the NGA Campus East lessons learned report compiled between 2007 and 2011 that Integrated Project Delivery method was cited as possibly the only method capable of delivering the project on time. However, with no prior experience to benchmark the project against, the stakeholder perspectives varied about the extent to which integration was successful. (Note. the variance of stakeholder perspectives in practice validates the results observed from the survey data). It was noted that the design was already 35% complete when the construction contract was awarded. Stakeholders noted that an earlier start to the collaborative process would have been very beneficial to maximize integration. Stakeholders noted that the integrated project Delivery was selected based primarily on the time performance requirement to meet the BRAC deadline.

Owner and User stakeholders state that they are highly satisfied with the IDBB process and not only would use IPD again if starting over, they would highly recommend Integrated Project Delivery to others.

5.5. Fort Belvoir Community Hospital Lessons Learned¹⁰⁰

The author has been involved with this IDBB project from its inception in 2005 and continues to monitor progress of its process. The Base Realignment and Closure (2005)¹⁰¹

⁹⁹ U.S. Army Corps of Engineers, *NCE Project Lessons Learned 2007-2011* Special Report (2011)

¹⁰⁰ Norfolk District, U.S. Army Corps of Engineers, *Fort Belvoir Community Hospital Lessons Learned Report Phase I: Pre-Design, Design, and Construction* (2010)

legislation imposed statutory project deadlines that could not be achieved by traditional methods. By necessity an Integrated Project Delivery process was selected to allow for hyper-fast-track construction. Construction started as concept designs continued to develop for a 1.3 million square foot, highly complex and technologically advanced military hospital complex. The construction contractor and architectural/engineering firm worked hand in hand to keep project process moving. It was highly chaotic and at more than a few times highly contentious, but the process functioned well and built momentum rapidly. The value added by such an integrated process became apparent at all steps. The ability to rapidly adjust design and project direction as each issue presented itself ensured project momentum was not lost.

5.6. Improving on IDBB

This research has demonstrated that Integrated Project Delivery is the most effective means of attaining the most important critical success factors in pursuit of maximum valuecreation. The panel of experts providing the evaluation in this study came directly from the highest echelons of the project delivery teams responsible for the IDBB pilot-projects. Both the actual performance of the IDBB projects and the assessments from the panel of experts validate the implementability of the IDBB method. IDBB from a macro-level is a resounding success. This study did not focus on the micro-level aspects of the IPD or IDBB, but has ascertained that IDBB process may be improved by further increasing the level of collaboration. As explained in Chapter 2, there are many regulatory impediments to a fully Integrated Project Delivery method. USACE has already attempted to improve on IDBB by placing the focus on earlier contractor involvement in the design process. Recall from Chapter 2 that Federal Acquisition Regulations

¹⁰¹ Department of Defense, Office of the Secretary, Federal Register, *Base Closures and Realignments (BRAC); Notice* 28030 (2005)

requires competitively bid of construction services contracts separate from Title I A/E (Design) service. The IDBB process, and the later Early Contractor Involvement (ECI), method rely on provisions for allowing "Non-Brooks Act, Pre-Construction Services" to enable contractor participation early in the MILCON project delivery process. This workaround approach is highly sub-optimal, and severely limits the benefits of collaboration at the very time it has the most potential for maximizing value-creation.

There are many different perceptions of the success of IDBB within the owner and user stakeholder groups. A separate study is required to isolate and analyze the differences. The assessments seem to break down into two general groups of assessment: (1) the micro-viewpoint or (2) the macro-viewpoint. In the macro-view the assessments are overwhelmingly positive. The micro-viewpoint, however, is highly dependent on stakeholder perception of relative importance. Those who focus on only the construction phase in context of the usual way of doing MILCON business tend to have little good to say about IDBB, mostly because of perceived cost and value issues. These views may be valid in a specific micro-viewpoint context, but really must be seriously considered in the context of lifecycle value-creation. The utility of the completed facilities in the end will determine the level of value-created, and by all accounts there are no early indicators of foreseen problems with utility and functionality of the buildings.

Not every MILCON, or medical MILCON project will be appropriate for implementation of an Integrated Project Delivery method, but there will continue to be future requirements for extremely large and highly complex facilities where the traditional MILCON model is itself an impediment to value-creation. For such projects the acquisition rules must be revised to allow maximum collaboration within the Integrated Project Delivery process.

A sub-category of the MILCON program should be established specifically to address projects which may benefit the most from IPD. The acquisition rules must include exemptions and/or other provisions to enable a fully Integrated Project Delivery process. The only way such a change will be implemented is if the Owners and Users demand change. The Owners are responsible for selecting the project delivery method, and assembling the project delivery team.¹⁰²

The U.S. Army Corps of Engineers is among the best in industry at project delivery, but are an agency purposely focused on the micro-details of the design and construction phases. They do not have a long-term stake in the operational function of the facilities once completed. The operational owners and users, the Tricare Management Activity and the U.S. Army Medical Department in the context of medical MILCON, are the dominate stakeholders in the phases that determine the value-creation.

As discussed in Chapter 4, Section 4, it's the conceptual and operational phases which bookend the project delivery process that defines and measure value-creation. As such the Owners/Users first must integrate their capital investment program fully with the operational planning and execution program as a first step towards optimization of the project delivery process.

¹⁰² Barbara White Bryson and Canan Yetmen, "Why Owners Make The Difference," *ENR: Engineering News-Record* (New York), 02 August 2010, 80.

CHAPTER 6.

RESEARCH METHOD

6.1. Introduction

This study evaluates the value relationship between inputs and outcomes in the Integrated Project Delivery (IPD) process. A modified-Delphi survey method (explained in detail below) is first used to identify and evaluated critical success factors important to overall project valuecreation. The most important critical success factors identified by the investigation then are utilized to evaluate the effectiveness of IPD relative to other project delivery methods: Design-Bid-Build, and Design-Build (DB), to successfully attain the desired outcomes for the various critical success factors. The research explores the differences and commonalities between the individual project stakeholders groups' perception of relative importance for each critical success factor. The relative importance will be explored from two perspectives which are an inextricable dynamic in any collaborative endeavor: "what's in it for me?" (*self-interest*), and "what's best for the project?" (*project-interest*). The dissertation research project employed a three round modified-Delphi survey to identify data for analysis that supports the Integrated Project Delivery approach. This section will describe in detail the concept and implementation of the survey method.

6.2. Survey Problem Statement

This research is based on the belief that the Integrated Project Delivery method may provides a framework for project delivery optimization by means not otherwise achievable by the inherent limitations of other established project delivery methods. By taking a larger,

comprehensive and integrated approach to the delivery process, perhaps each stakeholder's competing interests can be addressed through process optimization to maximize value-creation through a more thoroughly collaborative process. For this research the major project stakeholder groups are defined as the Owners, Designers, Builders, and Users. There certainly are numerous more subgroups within the project delivery process and other important stakeholders external to the project team, but these four are the major stakeholder groups relative to the IDBB pilot-projects. The project delivery method which best allows for process optimization for both the project specific goals, and the self-interest goals of each of these stakeholder groups logically must achieve a higher level of value-creation optimization.

When considering the entire spectrum of value-creation, it is important to identify the different categories of value-creation on which the project delivery process has a significant bearing. *Design Performance* and *Construction Performance* are the two categories which receive the most attention both in practice and in research. Perhaps the quantitative metric analysis of the cost in terms of money, time, materials, labor, and other resources for design and construction contributes to its dominance as the measure of project success. However, design and construction *outputs* are not necessarily easily quantifiable in corresponding units of cost *input*. The *quality* of the building, rather than simplistic output of bricks and mortar (and building systems, equipment, furnishings, etc...) must be considered in the value equation for project delivery process. Market value of the physical bricks and mortar only reflect a portion of the output. The bricks and mortar must be coupled with the operational and functional quality of the building's intended use(s) to assess the output factor in the value equation. Quality output must include more than merely workmanship, materials, and systems. Quality in large part is the legacy which the project delivery team leaves behind after project completion. Unlike cost and

schedule, the qualitative aspects of a building that contribute to long-term value-creation are not easily quantifiable, nor necessarily obvious to any member of the project delivery team, including the owner and user.

This project delivery research goes beyond the design and construction categories typically associated with defining value-creation. Two additional categories play an important part in determining value creation: (1) Process Performance and (2) Project Impact. Process alone has significant affects on cost and quality. Process factors into both the input and output of the value equation. *Impacts* are the important residuals of the project delivery process that endure long after the project delivery team declares victory and goes their separate ways. Some of the residuals affect the lifecycle value of the facility, while others such as professional reputation and long-term business relationships belong solely to the project stakeholders. To not consider these residual values into the overall assessment of value-creation fails to recognize relevant components of value important to each stakeholder group. *Process* and *Impact* have a great deal of influence on value-creation across the spectrums of the project delivery team stakeholders (owner, designer, builder, and users) over the course of the project delivery cycle, and long-term benefit/detriment to all stakeholders associated with the project. The capital investment costs of project delivery, as high as they may be, are fairly insignificant compared to the operational costs for many complex building types such as the IDBB case study projects. In many cases even short-term operational cost of only a few years dwarf the capital costs of the facility housing the operations. When considering the long-term facility costs along with other operational costs such as human resources, adapting for technology and regulatory changes, and maintaining customer satisfaction, the significance of the project delivery team's role in shaping long-term value-creation cannot be ignored.

Complicating the value-creation formula for project delivery is the concept of *quality*. Quality is a relative term, hence the importance of differentiation between quantitative and qualitative analysis in this research. In technical terms the quality of a building is determined by the plans and specifications of the design, and by the successful translation of design intent into a physical object by the builder. Quality is subject to interpretation not only in technical terms, but also in terms of stakeholder perspective. As with the old saying, "Beauty is in the eye of the beholder." The perception of the benefit is in the eye of the project stakeholder.



Figure 6.1 Perception in the value equation

Project delivery output in terms of quality is reliant on *perceived benefit*. From the vantage point of stakeholder perception the concept of value-creation becomes even more abstract. What value to whom, and according to whom? Project delivery is a complex, interrelated web of each project stakeholder's goals. Many goals are distinctly independent and internal relative to the project stakeholder. Other goals are fully shared amongst some or the entire group of project stakeholders. A single stakeholder may perceive benefit where all others perceive none. The hypothesis the Delphi survey corroborates is that Integrated Project Delivery provides a framework for both optimizing shared project goals, but also optimizing independent stakeholder goal attainment for the greater success of the project overall. The survey seeks to assess qualitative aspects and recognize the importance of lifecycle value-creation as a long-term consideration.

As discussed above, there are many aspects of project delivery that are easily quantifiable such as project cost performance, schedule performance, or physical placement of materials and building systems. Because many aspects of project value-creation are qualitative in nature, a method for assessing qualitative data from the project delivery process must be considered when assessing project value-creation. The qualitative assessment must take into account the perceived benefit of all project stakeholders, and the enduring impact beyond completion of the project delivery process. The survey provide a qualitative inquiry into relative importance project success factors based on input from a panel of experts representing each the Owner, Design, Builder, and User stakeholder groups.

The specific problem to be addressed by the modified-Delphi survey, and explained in detail below, is the multi-dimensionality of the project delivery process in regard to the project delivery team. The project delivery team is composed of distinct major stakeholder groups who influence the development, planning and execution of the project delivery process. The formal and informal relationships between the stakeholders are a complex web of intertwining goals and objectives which must be meshed in order to achieve overall project goals and objects. Optimization of the project delivery method. While different project delivery methods may be able to successfully achieve overall project goals in regards to cost, schedule, and/or quality, this limited view of project success only tells part of the story. The missed, or not-identified, opportunities for additional optimization to the project delivery process may tell the rest of the story. One such potential opportunity for further optimization is based on consideration of each stakeholder's perspective(s) and interpretation of what success is. Comprehensively, overall project success

is measured by the sum of each stakeholder group's performance, and is dependent on both project-centered interests and stakeholder self-interests.

An important concept of project integration is optimization through collaboration. Collaboration is best served by a full understanding of each other's position in a collaborative group. In a perfect world each stakeholder would share a common perspective of project goals and objectives, and fully agree on the definition of success. But, this is not a perfect world and it would be reasonable to assume there may be disagreement between a project stakeholder's selfinterest, and the project-interest terms for which each party formally agrees to contractually. Within the four corners of the contract all parties are in agreement. In reality there may be significant "unspoken" disagreement. How could this be? It's a matter of perspective, interpretation, and relative importance. Each stakeholder's individual performance to achieve shared goals is significantly influenced by internal goals of self-interest.

Three rounds of surveys will address the problem discussed above by collecting data from a panel of experts to determine: (1) what critical success factors project leaders believe are most important for project success and by extension value-creation, (2) establish rank-order relative importance of the selected critical success factors, (3) ascertain if there is a difference between self-interest and project-interest for the stakeholder groups, and (4) evaluate the efficacy of different project delivery methods to achieve the selected factors. The survey focuses not only on each stakeholder's individual perspective, but also asks each participant to provide responses considering separately their "self-interest" and "project-interest" perspectives.

6.2.1. Survey Hypothesis

This section is not to be confused with the overall hypothesis of the dissertation; this section refers to the two hypotheses which are addressed in the survey as a supporting element of the overall research. The first hypothesis is that there is a difference between self-interest and project-interest for each project stakeholder. The second hypothesis is that Integrated Project Delivery is more effective than Design-Build, or Design-Bid-Build method for achieving important critical success factors. The relationship to the problem statement discussed at the start of this section centers on recognizing proof that differences, in fact, exist between a stakeholders 'self' and 'project' interests. The use of an appropriate survey model is very important to finding valid information from diverse stakeholders in the project delivery industry.

6.2.2. Survey Specific Literature Review

This section discusses literature particular to the formation of the survey. To determine an appropriate model for conducting this investigation a substantial review of the existing body of related published research was conducted.

Three particular published studies helped to inform and shape the methodology of the surveys: Edward Gibson, Giovanni Migliaccio, and James O'Connor's *Changing Project Delivery Strategy: An implementation Framework* published in Public Works Management and Policy Journal (January 2008) ¹⁰³; D.K.H. Chua, Y.C. Kog, and P.K. Loh's *Critical Success Factors for Different Project Objectives* ¹⁰⁴ and, John Yeung, Albert Chan, and Daniel Chan's

¹⁰³ G. Edward Gibson, Giovanni C. Migliaccio, and James T. O'Connor, "Changing Project Delivery Strategy: An Implementation Framework," *Public Works Management and Policy* 12, no. 3 (January 2008): 483-502.

¹⁰⁴ Chua, Kog, and Loh, "Critical Success Factors for Different Project Objectives", 142-50.

Developing a Performance Index for Relationship-Based Construction Projects in Australia: Delphi Study published in the Journal of Construction Engineering and Management ¹⁰⁵ (May/June 1999, and April 2009 respectively). These three studies demonstrated that the Delphi method is a practical model for investigating qualitative aspects of the project delivery process in vastly different ways, but within a similar context as this study. The commonality of the studies is that each demonstrated how the Delphi process may be used to identify critical success factors, measure relative importance, and possible ways to interpret the results within the complex framework of the project delivery industry. From these examples the seed was planted for ideas on how to expand the Delphi method for more comprehensive approach to the multidimensionality of the project delivery process.

Also key to establishing the methodology for the surveys were Michael K. Kim's *Integrative Design of Buildings: Principles and Strategies*¹⁰⁶, and lecture notes from several of his courses which in large part helped to form the basis for the themes, categories, and several critical success factors used throughout the survey.

6.2.3. Limitations

Nature of Project Delivery: No two projects or designs create the same value. Every project is exposed to unique variables and situations due to the environment in which it is executed. In practice, design and construction means and methods must be adapted, rather than replicated, to account for differing site conditions. Every design must create value specific to its unique situation. A significant variance between each particular project is unavoidable. This variance provides a challenge to providing meaningful comparative analysis between individual

¹⁰⁵ Yeung, Chan, and Chan, "Developing a Performance Index for Relationship-Based Construction Projects in Australia: Delphi Study", 59-68.

¹⁰⁶ Kim, 2011. File: Manuscript in Progress. Integrative Design of Buildings: Principles and Strategies. University of Illinois at Urbana-Champaign, Unpublished, Chapter 3.

projects. The following significantly mitigate the time constraints for successful completion of this study:

- All of the case study projects are part of the same overarching military medical construction program, and in most cases uniform project goals outside of programmatic functional differences.
- The case study projects are clustered around two geographic locations, Washington D.C., and San Antonio, TX. Within each geographic area nearly identical economical environments are assumed, providing a level playing field for comparison of project costs.
- Primary case study projects are all part of the same IDBB Pilot Program

Limited Body of Academic Research: Due to the very limited amount of published academic research directly related to Integrated Project Delivery, this study will have a significant dependence upon that material which is available at this time. Related or complementary research by necessity must be utilized where appropriate to fill gaps. Thoughtful and well reasoned deductive logic is necessary to frame the methodology of this research.

Limited Sample of IDBB Projects: Due to the emergent nature of IPD there is a relatively small body of IPD projects to study, either completed or in execution. Fortunately, IPD has been embraced by the current military construction leadership, and there are several significant projects currently underway which are utilized as a basis for the survey population, and for case study. All of the projects are scheduled to be completed at approximately the same time this study ends in August 2011.

Although the projects are not fully completed at the time of completing this dissertation, they certainly will be substantially complete. Substantial completion affords significant analysis of the measure of success towards completion of the project goals which provide the framework for this study. Additionally, in parallel, an independent government review will be ongoing and will provide an additional source of comparative data.

6.3. Survey Model: Modified-Delphi

The survey method used in this research is conceptually based on the Delphi method. The Delphi method is a process which seeks a consensus opinion of a small panel of subject matter experts through an iterative survey process. The Delphi method is an established and widely used survey process which seeks a consensus opinion of a small panel of subject matter experts through an iterative survey process. The Delphi technique was developed by the Rand Corporation in the mid 1950's as a means of "systematic solicitation and collation of judgments on a particular topic through a set of carefully designed sequential questionnaires interspersed with summarized information and feedback of opinions derived from earlier responses". ¹⁰⁷

A highly beneficial aspect of the Delphi method is the ability to establish a panel of experts over a wide geographic area. The method does not require participants be collocated or to interact directly with each other. For this research the Delphi method allows for the correlation of informed judgments across a wide range of subject-matter expertise to generate a consensus.

This research study modifies the typical Delphi method by establishing four distinct expert groups which then compose a larger, collective panel of experts. The modification is necessary to recognize the four primary stakeholder groups which contribute to the project delivery process. This investigation requires the expanded Delphi format to accommodate and

¹⁰⁷ Delbecq, Van de Ven, and Gustafson, *Group Techniques for Program Planning: A Guide to Nominal Group and Delphi Processes*, 10.

equitably represent the varying range of stakeholder expertise groups engaged within the comprehensive process of project delivery. The primary modification is the formation of a Delphi panel of experts consisting of four distinct stakeholder sub-categories to the project delivery process: Owner, Designer, Builder, and User stakeholder groups

The questions were developed around the concept of a three round set of internet-based surveys. This study was conducted using the internet-based survey services of QuestionPro (www.questionpro.com). The web-based surveys provided a platform for publishing the surveys, communicating with participants, and anonymously collecting participant responses.

6.3.1. Survey Sample

The Delphi process, unlike most survey methods, does not rely on a random sample or distribution based on a large sample size. The Delphi method requires a sharp focus on expertise, because it is *expert opinion*, rather than a large sample size which provides the significance behind the Delphi process.

The typical sample size for a Delphi study ranges between ten to thirty participants. Four or less respondents would be too few to adequately demonstrate significance of the desired consensus. As Delbecq et al. (1975) indicated, "the size of the respondent panel is variable. With a homogenous group of people, ten to fifteen participants might be enough". ¹⁰⁸ A difference with the case of this investigation is the formation of multiple panels of experts. This investigation requires a larger overall sample size than a typical Delphi survey based on the minimum required size for each of the four sub-groups.

¹⁰⁸ Andre L. Delbecq, Andrew H. Van de Ven, and David H. Gustafson, *Group Techniques for Program Planning: A Guide to Nominal Group and Delphi Processes* (Glenview, IL: Scott, Foresman, 1975), 89.

This study approaches each stakeholder group as an independent Delphi group within the larger overall Delphi group consisting of all stakeholder groups. These project stakeholders will include the Owner's, Designers, Builders, and Users. The stakeholder population represents subject matter experts and highly regarded leaders in their respective fields. Access to this population is readily available due to the author's active duty military status and work experience. Significant professional working relationships already have been developed with many of the key individuals and organizations at each of the project sites

As modified, this study is a Delphi within a Delphi. Thus the minimum sample size is increased four-fold. The minimum range for this modified Delphi survey is between 32 and 40 participants. The goal for this survey is to have a minimum of ten respondents from each of the four stakeholder groups for the Round 1 survey, which would allow for an expected and acceptable attrition of respondents over the course of the remaining two surveys. Additionally, the Owner's stakeholder group consisting of a wide range of government agencies would ideally be slightly larger than the other stakeholder groups to adequately represent the most critical layers of authority from each of the many agencies (the Department of Defense, the U.S. Army Corps of Engineers, and the U.S. Army Health Facility Planning Agency to name a few) which comprise the role of the public/government project "owner". The sample sizes achieved for this study were 51, 42, and 40 for the rounds 1, 2, and 3 surveys respectively. Additional details are provided in the individual survey round sections below.

6.3.2. Survey Population

The survey population was limited to only the most qualified and experienced individuals with proven excellence in their respective professions. The focus of this research is to collect data about critical success factors (CSF) for project delivery of large, expensive and complex

projects. The sample for the survey is limited to industry experts with significant experience with the U.S. Army's military construction (MILCON) program, and more specifically related experience with the four Integrated-Design-Bid-Build (IDBB) pilot-projects well under way at the time of this investigation.

Identification and selection of potential survey participants was centered on the executive level of the project delivery teams for the IDBB pilot-projects, and includes senior civilian and military professionals from the Owner, Designer, Builder, and User stakeholder groups. Contacts were gathered from discussions with the program directors of the various projects. Key project executives were then contacted about the proposed research project, and invited to participate in the study. All contacts voiced a strong interest in participating in the survey, but a few could not commit to participation due to business travel plans and workload.

The questions the survey asks are not directly related to assessment of the IDBB pilotprojects, but are meant to ascertain the expert opinion of the highly experienced professionals who have worked in all aspects of project delivery and different project delivery methods over the years with particular emphasis on large complex military construction projects (MILCON). Additionally the sample population has fresh experiences and opinions based on the lessons learned during their involvement with the IDBB projects over the course of several years (between 2005 and 2011).

The Owner's and User's populations include only senior military and civilian project executives directly responsible for the multi-billion dollar MILCON capital investment programs. The designer's and builder's populations include principle level executives of companies and firms which are among the largest, and most successful in industry. Based on the expertise, experience, and participation in the U.S. Army's Integrated Design-Bid-Build pilot-

projects, the sample population of this research project is uniquely, and perhaps exclusively, qualified to evaluate the project delivery process in the context of MILCON mega-projects.

6.4. Procedures

QuestionPro (www.questionpro.com) provided the web-based survey platform through six months of contracted service. All survey development and administration was conducted by the author under the research direction of Professor Michael K. Kim, PhD. The internet-based surveys were conducted in two phases.

Phase one consisted of two rounds of surveys to select and assess relative importance of critical success factors for project delivery. Phase two consisted of a round 3 survey to further evaluate effectiveness of project delivery methods against the CSFs selected in survey rounds 1 and 2. Respondents were identified, as discussed earlier, as having a high degree of expertise and experience with complex military construction projects with an emphasis on medical projects. These participants were then contacted by telephone and/or electronic mail to determine if they would be interested in participating. A validated list of 56 potential participants, providing equal representation for each stakeholder group, was used to populate the email distribution list for the round 1 survey.

Summary of procedures:

- Literature Review
- Pilot study / interviews
- Develop Survey Questionnaire
- Survey of project delivery stakeholders
- Project Case Studies
- Data collection through military sources
- Evaluation of Data

6.5. Survey Overview

Detailed descriptions of the survey questions will be discussed later in each of the individual survey round sections. A brief overview of each survey's questions will be provided here as part of the conceptual overview. Complete copies of the surveys are found in the appendices C, D, and E.

6.5.1. Round 1

Email invitations with links to the web-survey were sent to 56 individuals out of over 60 contacted prior to the start the research. The first round survey was divided into two sections. The first section asked basic questions to validate respondent expertise for participation on the panel of experts: Stakeholder group, Years experience, Level of education, and Professional certifications. The second part of the survey asked the participants to evaluated ten listed critical success factors for each of the 4 categories of the project delivery process: (1) Design, (2) Construction, (3) Process, and (4) Impact. From each list of ten categories, each respondent was required to select the top-5 factors they believed most important for the category. The participants were required to evaluate each category two times, first from a *Self-Interest* perspective and then from a *Project-Interest* perspective. Additionally, survey participants were given the opportunity to select "Other" and include in their 'top-5 selections' one or more CSFs not otherwise listed. Frequency of selection is the measure of relative importance both overall and within each stakeholder.

The categorical breakdown ensures a representative sample of Critical Success Factors (CSFs) across the spectrum of the project delivery process are selected for further analysis in the subsequent Round 2, and Round 3 surveys. Otherwise, it is likely the distribution of selected CSF's in survey Round 1 survey might have centered around individual areas of a particular

stakeholder group's interest, thereby narrowing the focus to stakeholder 'nodes' along the project delivery process timeline. The categorical approach forces the survey participants to take a broad approach in selecting critical success factors outside their normal centers of gravity.

The selection criteria for CSF to be further assessed in the round 2 survey was based on the rank-order percent (frequency based) of CSFs selected in each of the four categories from both the 'self-interest' and 'project-interest' perspectives. To clarify, the respondents picked a top-5 list from eight individual lists of ten critical success factors: 4 Categories evaluated form 2 perspectives equals 8.

Analysis and statistical testing is conducted on all eight groups of ten. The top 25% of rank-ordered factors were selected for further evaluation in survey round 2. Twenty-four critical success factors were selected out of the original forty for further evaluation.

6.5.2. Round 2

The second round used questions developed from responses to the first survey. Prior to the start of the round 2 survey each respondent was provided, via email, the results of the round 1 survey. The published results included listings of the relative importance rankings from each stakeholder group and self/project perspective. The round 2 survey also included an internet link to view the round 1 results before continuing with the round 2 survey.

Round 2 was greatly simplified from round 1. The sample group for round 2 was limited to only those respondents who participated in round 1. Therefore, there was no need to request any further validating demographic information. The survey included only a question to selfidentify with a stakeholder group. The 24 critical success factors selected from each of the four categories (Design, Construction, Process and Impact) in round 1 were consolidated in a single list to be evaluated from both self and project perspectives. In some instances redundancy in

critical success factors between categories was treated by consolidating to a single CSF for the round 2 list.

Round 2 asked the participants to rate each selected Critical Success Factor (CSF) on a 1 to 5 scale (1=least important and 5=most important). Once again the respondents were twice asked each question: once from the 'self-interest' perspective, and a second time from the 'project-interest' perspective.

The mean scores of each critical success factor were used to once again rank-order each CSF for relative importance. Rank-orders were established for each stakeholder group and perspective category. As in survey round 1, a top 25% ranking in any of the stakeholder groups was used as a selection criteria selecting overall important CSFs. The list of critical success factors that started at forty in round 1 was narrowed down to the thirteen most important, as assessed by the panel of experts, to be used in the round 3 survey.

Additionally in round 2 a comparison of inter-rater agreement between survey rounds 1 and 2 was made to determine if consensus had increased among the stakeholder groups and the larger panel of experts. As is reported later in the round 2 section, statistical analysis determined that increased consensus had been achieved. Based on the statistical significance, the iterative Delphi process for evaluating critical success factors was achieved 2 without the requirement for additional any rounds. Had the level of inter-rater agreement not improved between rounds 1 and 2, an additional round, or rounds would have been necessary. Fortunately, the panel of experts performed as expected.

Using the completed data from the rounds 1 and 2, detailed analysis of the differences between stakeholders, and perspectives is provided in the round 2 report.
6.5.3. Round 3

The sample group for the round 3 survey was the same as for round 2. Again, the respondents were provided with results of the previous survey prior to starting round 3. By this time the panel of experts has been provided data that demonstrates a clear difference between stakeholder groups. The focus in round 3 shifts from relative importance of factors, to using the 13 most important factors as a tool for assessing the effectiveness of different project delivery methods. At the start of the survey each respondent was asked to answer questions to validate their particular familiarity with the three project delivery methods (DBB, DB, and IPD). Each responded was asked to select on a scale of 1 to 5 (1. Not at all familiar, 2. Slightly familiar, 3. Somewhat familiar, 4. Moderately family, 5. Very familiar). The sample population was selected based on their broad experience in project delivery and as was expected most respondents answered in the range of high familiarity (3-5). Also as was expected, a couple of respondents in the User group, although experts in their respective areas, selected in the "not familiar at all" to "slightly familiar" range. The User stakeholder group is typically more focused on operational aspects rather than the nuts-and-bolts of project delivery. Two participants that responded with less than high familiarity were excluded from the survey analysis to ensure consistency within the response set.

On a side note, an oversight between rounds 1 and 2 was addressed by a separate questions unrelated from the rest of round 3. One critical success factor (Evidence Based Design) was selected in round 1 but was omitted in round 2. The respondents were asked to rate Evidence Based Design on the same scale as in round 2, and the results were amended to the round 2 data set. For the record, Evidence Based Design did not score high enough to be included in the final list of CSF's selected for round 3.

The main part of Round 3 asked the respondents to assess on a scale of 1 to 5 (1 = not effective at all, 5 = very effective) the effectiveness of three project delivery methods: Design-Bid-Build (DBB), Design-Build (DB), and Integrated-Project-Delivery (IPD) to achieve the selected critical success factors. The participants rated each project delivery method against each of the 13 critical success factors two times, from the self-interest and project-interest perspectives.

Analysis of the data includes rank-ordering of the efficacy each project delivery method for each of the CSFs by stakeholder group and perspective. The data was then aggregated and normalized by weighing for relative importance (determined in round 2). Tests were conducted to determine statistical significance of the resulting relative effectiveness scores.

6.6. Survey Instruments

The survey instruments containing the questions described above are provided in the Appendices. Each survey instrument was developed by the author in discussions with Dr. Michael K. Kim, committee chair and advisor, along with the other committee members.

The instrument for the Round 1 survey was reviewed by the committee and other researchers to establish initial validity. A limited pilot-survey was conducted from 06 December 2010 until 18 December 2010 among a small sample of survey population to provide data for review of the questions, format, and resulting data scale. The rounds 2 and 3 surveys were developed from the responses to the round 1, and were similarly reviewed for initial validity.

The design of the instruments seriously considered the sample population to maximize participation. The participants in this survey are extremely busy professionals at the tops of their professions and involved in several high-stakes mega-projects. Their time is at a premium, and as such the surveys were designed for simplicity and brevity. Each survey was designed to be

completed in approximately 15 minutes, and refinement made during the pilot-survey testing resulted in average completion times meeting or exceeding the target.

Three primary measures are required for population validation, relative importance of critical success factors, and effectiveness of different project delivery methods to achieve critical success factor examples. Each of these measures will utilize the expert opinion of the Delphi panel(s), and are statistically tested using SPSS Statistics 18.

6.6.1. Validity Testing

The instruments (surveys 1, 2, and 3) were reviewed and tested for validity and reliability. Inter-class agreement was measured using Kendall's coefficient of concordance to validate consensus building and level agreement amongst the panel(s) of experts. Intra-class correlation utilizing Cronbach's alpha was used to test for reliability of the questions for the overall group as well as each stakeholder group.

Validation of the sample population, the panel(s) of experts, uses three criteria to evaluate the level of professional expertise in each participant's of expertise: (1) experience, (2) education, and (3) certifications. Additionally, each participant self-identified with a stakeholder group (Owner, Designer, Builder, or User) which indicates membership in a sub-panel within the larger overall panel of experts.

An important aspect when validating the results of a Delphi survey is ensuring the data represents a consensus among the panel of experts. Stated in simple terms, the results are only valid if it can be shown that there is an increasing degree of consensus through the iterative Delphi survey process. Each survey participant is considered a rater. For example think of a panel of judges scoring an event. Consensus is achieved by inter-rater agreement. One statistical method for measuring inter-class, or inter-rater, agreement is Kendall's coefficient of

concordance (W). Kendall's W provides a way to assess inter-rater agreement among a small panel of three or more judges (appropriate for the Delphi method) using the mean rank orders of each rater's assessment of the cases. In the case of this survey rank-order analysis is more reliable due to the small sample size of the Delphi survey. Kendall's W is a coefficient where zero equals no agreement, and one equals perfect agreement among raters. Using non-parametric analysis in SPSS, Kendall's W was calculated for the overall panel of experts, and for each sub-panel of stakeholder group experts (owners, designer, builders, and users) to determine the level of consistency. The Kendall's W results from different rounds of surveys can then be compared to determine if the level of agreement increases, or not, from one survey to the next. In the case of this research project, the inter-rater agreement increases with each survey round. Results are reported in each individual survey section to follow.

Reliability, or intra-class correlation (differing from the "inter-class" correlation in the Kendall's W discussion above) is assessed using Cronbach's Alpha as a scale measure. Cronbach's alpha, a coefficient based on the number of cases, and a ratio of average inter-case covariance to the average case variance, and measures consistency when evaluating multiple raters on ordered category scales. ¹⁰⁹ The measure of Cronbach's alpha is reported as a coefficient between zero and one, where zero is interpreted as follows: zero equals no reliability; less than 0.6 equals not reliable; 0.6 is the minimum level for a reliable scale; 0.7 is reasonably reliable; 0.8 is strongly reliable; 0.9 is very reliable; and levels approaching 1.0 (> 0.98) are potentially over reliable. For this study, minimum reliability levels were generally achieved in all three surveys. The surveys consistently scored high in reliability for the larger overall "panel

¹⁰⁹ J. Marilee Bresciani et al., "Examining Design and Inter-Rater Reliability of a Rubric Measuring Research Quality Across Multiple Disciplines," *Practical Assessment, Research & Evaluation* 14, no. 12 (May 2009).

of experts", there are some instances in the round #1 survey where individual sub-panel (stakeholder groups) failed to individually achieve the minimum reliability (Cronbach's alpha) score of 0.6. In the subsequent rounds two and three all measures of reliability exceed 0.6. Details of reliability are reported in the individual survey sections to follow.

In the Delphi process the typical major statistics for determining significance are the measures of central tendency: mean, median, and mode. In literature, median scores based on a Likert scale are favored.¹¹⁰ When appropriate more rigorous statistical tests are used to provide a more robust argument for the superiority of Integrated Project Delivery. For example a Friedman test with post-hoc tests is used in round 3 to determine significance for the results of the relative effectiveness evaluations.

6.6.2. Confidentiality

Responses to the Delphi questionnaires were treated with complete confidentiality. The identity of participants of the Delphi Survey was not shared among other participants, and the surveys did not contain information that personally identified participants. All responses were completely anonymous. No ID key was used to link individual participants to survey results. Responses were categorized by participant's stakeholder group only. Stakeholder categories include: Owner/Owner's representative, Designer, Builder, and facility User. The participant sample included multiple individuals for each category from each of the case study projects. QuestionPro's (www.questionpro.com) "Respondent Anonymity Assurance" feature was enabled on each of the web-based surveys to allow the ability to track who has responded to the survey and who has not for the purposes of sending out reminder emails, and administration

¹¹⁰ Chia-Chien Hsu and Brian A. Sandford, "The Delphi Technique: Making Sense of Consensus,". *Practical Assessment, Research & Evaluation* 125, no. 10 August 2007, 09 September 2010 http://pareonline.net/pdf/v12n10.pdf>.

of the survey process; and for human subjects protocols ensuring that email identifications are not linked to the response data.

An informed consent agreement was included as the first page in all web-surveys. The informed consent agreement outlined the precautions taken to ensure confidentiality. Respondents could either accept the terms of the agreement or exit the survey. The survey could not be started without respondent agreement.

Confidentiality was an important consideration voiced by participants during the preinvitation discussions. Several participants stated that they would participate only if the survey results were anonymous. Beyond protecting the interests of the participants, the anonymity of the surveys also increased the quality of the data yielded a enabling more critical evaluation among the various stakeholder groups. To maintain participant confidentiality, the dissertation refers to participants only by their stakeholder group and does not directly mention the names of the design firms, or construction companies with they are associated.

6.6.3. Institutional Review Board

This research project was reviewed by the University of Illinois Urbana-Champaign's Institutional Review Board and approved on 02 November 2010 (IRB Protocol Number: 11122). This research project was reviewed by the Institutional Review Board, Office of the Vice Chancellor for Research, University of Illinois at Urbana-Champaign in accordance with policy on human subject research. On 02 November 2010, the IRB determined that this research meets the criteria for exemption because (Category 2) online survey methods are used to assess perceptions of various experts regarding evaluation of various aspects of military construction projects, and (Category 4) because only retrospective analysis of construction project documents will occur. The Category 4 exemption limits analysis of project construction documents to only

those produced before 02 November 2010. The exempt protocols of this research project are approved for a maximum of three years, ending on 01 November 2013. For further information on IRB requirements please visit the IRB website at http://www.irb.illinois.edu. A copy of the IRB approval letter is located in Appendix B.

CHAPTER 7.

ROUND 1 SURVEY REPORT



Figure 7.1. Survey round 2 overall data results

7.1. Introduction

The round 1 report presents data collected from the panel(s) of experts. Please note that analysis in round 1 is kept to a minimal level because the primary purpose is preparation for the second round survey where more detailed analyses will be conducted. Whenever possible charts tables and graphs are used to summarize the data and great effort was taken to graphically represent data and analysis whenever possible. One such example is the data summary on the first page of the report. Note the linearity of the response set in figure 6.1, an early indicator of agreement among the panel of expert when ordering the relative importance of the selected critical success factors. Figure 7.1. above gives a visual pre-view of the data to be discussed below.

Overview:

The round 1 report begins by stating the purpose of the survey, and administrative data before addressing questions and responses. Questions are addressed in the order they were encountered in the survey, and are immediately followed by response data when possible. Where repetitive questions are used to evaluate each of the factor categories separately the questions will be grouped ahead of the data and analysis.

The data and analysis for the round 1 is organized in the following order: sample data, participant validation, critical success factor data and evaluation, and finally a summary of round 1 findings. All sections of data and analysis will be prefaced by the survey question, or a summary of the survey questions, to be addressed in the particular section.

Data and analysis will be organized around the four categories used to represent different aspects of the project delivery process: Design, Impact, Process, and Impact. As discussed in

research methods earlier in chapter 6, the four categories serve the dual purpose of simplifying the survey for the respondents, and providing the means of ensuring and selecting a broad range of factors across the spectrum of the project delivery process for further analysis in the round 2 survey. Additionally, within each of the four categories, the data and analysis is further subdivided around the self-interest and project-interest categories as well.

Each categories' data set includes: frequency tables of the raw data, inter-rater agreement tables for validation purposes, intra-class correlation tables to test survey reliability, correlation matrices, and relative importance tables based on rank-ordering. The next to last section of the report will provide a summary and consolidation of each categories relative importance rank-orders. Any factor scoring in the top-25% (based on a fractional rank-order) of any stakeholder group's rank-ordered list will be selected for further analysis in the round 2 survey. Finally, the report concludes with a summary of round 1 findings.

Purpose:

The round 1 survey served two purposes. The first purpose is to validate the qualification of each individual participant to serve on the panel of experts. The Delphi method is dependent on the experience and expertise of the panel of experts voluntarily participating in this research study. The second, and primary, purpose of the round 1 survey is to select critical success factors for determining relative importance. From the selection of critical success factors and resulting determination of relative importance, the resulting data set is used to identify differences between stakeholder groups and their different perspectives: self-interest and project-interest. Successful completion of round1 provides both proof of concept for the study and a data set to establish the basis for the round 2 survey.

Administrative:

The round 1 survey was published to the website and invitations were sent to a total of 56 participants on 10 January 2011. Weekly email reminders were automatically sent to remaining participants until the survey was closed at 11:59 pm 35 days later on 14 February 2011. The invitations contained internet links to the survey website survey invitations were sent via email to a total of 56 participants. A complete copy of the survey instrument is located in Appendix C.

Participation Statistics:

- Invited: 56
- Started: 53
- Completed: 51
- Participation Rate: 91.1%
- Completion Rate: 96.2%
- Drop outs (after starting): 2
- Average Time to Complete Survey: 16 minutes

7.2. Round 1 Sample

Round 1 - Question 1: Stakeholder Group Identification

Which of the following categories do you best identify with? (Select only one): Design Team; Facility User or User's Representative; Construction Team; Owner or Owner's Representative.

The participation distribution is not perfectly equal between the stakeholder groups, but

poses no problem to the validity of the Delphi survey process. Each stakeholder group is within

the minimum required number of participants (more than 5). The owner stakeholder group may

appear over represented at 33% of the total panel of experts, but that is not the case. Due to the



Figure 7.2. Round 1 survey participation by stakeholder group

Stakeholder Group	Frequency	Percent
(1)	(2)	(5)
Owner / Owner's Representative	17	33.3
Design Team	10	19.6
Construction Team	11	21.6
Facility User / User's Representative	13	25.5
Total	51	100.0

 Table 7.1. Round 1 survey participation frequency table

multiple government agencies which comprise the ownership group the sample set included invitations to more owner's than the other stakeholder groups. All comparisons are made between stakeholder groups as a whole utilizing normalized data. The 92% participation rate, as described in the administrative section above, was better than expected. The survey was designed to perform at a minimum of 70% participation, a consideration necessary given the busy schedules of the population sample.

7.3. Participant Validation

The round 1 survey participants were asked a series of questions to validate expertise and experience. All candidates were screened for a combination of professional certification(s), education level, and years of experience. Although education and professional certifications are important indicators of expertise, the greatest weighting was primarily placed on years of experience, and secondarily on a combination of education and professional certifications. Many of the participants, by virtue of years experience and age, began their careers when education requirements for professional certifications and licensures were much different than today. Likewise the career paths some of the most experience panel members provides a wealth of knowledge beyond any graduate level education. Based on the impressive range of qualifications, no respondent needed to be screened out to ensure integrity of the panel. None the less, the screening process was important for two reasons. First, and most important, the Delphi method is predicated on the qualifications of the participants who comprise the panel of experts. Even though only the most qualified individuals were invited to participate as members of the panel of experts, the validation ensured the credentials of the anonymous respondents matched up with the skill sets of the invited participants.

Round 1 -- Question 2: Professional Certifications

What are your professional certifications? (Select all that apply): Registered Architect; Professional Engineer; Certified Design Professional; Certified Construction/Project Manager; Licensed Healthcare Provider; Not Applicable; Other (list if selected).

The owners' stakeholder group includes a broad range of professional certification which may be unique to sample population. Nearly all are either registered architects, professional engineers, and/or certified project/construction managers. These Owners are likely more

							Profe	ssi	onal C	ert	ifica	tions		
Stakeholo Group	ler	Re Ai	gistered rchitect	Pro H	ofessional Engineer	Co Ma	nstruction /Project anagement	Li He Pi	censed althcare rovider			Арр	Not olicable	
	Ν	n	%	n	%	n	%	n	%	n	%	Description	n	%
(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)	(13)	(14)	(15)
Owner	17	4	24%	5	29%	3	18%	0	0%	6	35%	Fellow, American College of Healthcare Executives (x2); Master's Degree in Healthcare Administration; PhD; Engineer in Training (EIT); Project Management Professional: Certified Design Professional	4	24%
Designer	10	6	60%	3	30%	0	0%	0	0%	1	10%	LEED AP; Evidence Based Design Accreditation and Certification (EDAC)	1	10%
Builder	11	0	0%	4	36%	6	55%	0	0%	2	18%	LEED AP; Project Management Professional	3	27%
User	13	0	0%	0	0%	0	0%	9	69%	3	23%	PhD Architecture; Healthcare Administrator; American Institute of Certified Planners (AICP)	1	8%



Table 7.2. Professional certification frequency table

sophisticated than the average owner in knowledge of project delivery. The Users stakeholder group is comprised exclusively of healthcare professions. However, these healthcare professionals are not ordinary clinicians. These are doctors, nurses, and administrators who have specialized in representing or advocating for the medical operators throughout the planning, programming, and design of numerous large hospital projects. Some are professional nursemethods-analysts who specialize in design of medical operations and translate the requirements into facility requirements and standards. Like the Owners stakeholder group, the User group is also a very sophisticated in project delivery.

The Builders and Designers are principle level executives at Engineering News Report (ENR) top-25 construction companies and design firms.

Round 1 - Question 3: Education

What level of formal education (related to your area of expertise) have you completed? (Select only one): Associates Degree or other training leading to professional certification; Bachelors Degree; Graduate/Professional Degree; Doctorate Degree; Not Applicable.

Educational				Sta	akehold	er Grou	ps			
Educational	Ow	vner	Des	igner	Bui	lder	Us	ser	А	11
Deglee	Freq.	%	Freq.	%	Freq.	%	Freq.	%	Freq.	%
(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)
Associates	1	5.9%	1	10.0%	0	0.0%	0	0.0%	2	3.9%
Bachelors	2	11.8%	4	40.0%	4	36.4%	0	0.0%	10	19.6%
Graduate	11	64.7%	5	50.0%	7	63.6%	9	69.2%	32	62.7%
Doctorate	3	17.6%	0	0.0%	0	0.0%	4	30.8%	7	13.7%
Totals	17	100%	10	100%	11	100%	13	100%	51	100%

 Table 7.3. Participant education frequency table



Figure 7.3. Education histogram

Overall the group is highly educated with over 82% possessing advance degrees above the bachelor level. Education is only one measure of expertise. For example, the two individuals who responded as having associates degrees also have more than 30 years experience each. Many professionals who started careers in the same era as these individuals attained professional certifications prior to the current degree requirements for professional registration as and architect or engineer. Based on cross analysis of experience and professional certification there is no cause to question the expertise of those two particular individuals. The distribution among the stakeholder groups is fairly even. The owner and user groups are especially well educated, an indication that those groups are more sophisticated in the project deliver than most owners and users.



Figure 7.4. Education by stakeholder group

Round 1 - Question 4: Experience

How many years experience do you have in your area of expertise? (Select only one): Less than 10 years; 10-15 years; 15-20 years; 20-25 years; 25-30 years; 30-35 years; More than 35 years.

Vears				Sta	akehol	der Group	S			
Experience	Ov	vner	De	signer	B	uilder	J	Jser		All
Experience	Freq. %		Freq.	%	Freq.	%	Freq.	%	Freq.	%
(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)
< 10	0	0.0%	0	0.0%	0	0.0%	0	0.0%	0	0.0%
10 - 15	1	5.9%	1	10.0%	0	0.0%	1	7.7%	3	5.9%
15 - 20	4	23.5%	1	10.0%	0	0.0%	0	0.0%	5	9.8%
20 - 25	1	5.9%	1	10.0%	1	9.1%	4	30.8%	7	13.7%
25 - 30	6	35.3%	2	20.0%	5	45.5%	2	15.4%	15	29.4%
30 - 35	3	17.6%	2	20.0%	3	27.3%	5	38.5%	13	25.5%
> 35	2	11.8%	3	30.0%	2	18.2%	1	7.7%	8	15.7%
Totals	17	100.0%	10	100.0%	11	100.0%	13	100.0%	51	100.0%

Table 7.4. Participant experience frequency table

The experience level of the panel experts is highly outstanding across the board. Over 70% of the survey respondents have over 25 years of professional experience. Minimally the

participant panel represents more than 1,290 years of combined professional experience! The median range of experience for the panel is between 25 and 30 years, with an average of approximately 27 years.



Figure 7.5. Experience histogram (overall panel of experts)

The experience screening criteria this survey was less than 10 years. Any respondent with less than 10 years of experience regardless of education and/or professional certification would have been excluded from the response set. All respondents, as expected exceeded the screening criteria. The distribution of experience among the stakeholder groups gives no reason to question the collective experience of any group.



Figure 7.6. Participant experience by stakeholder group

7.4. Selection of Important Critical Success Factors

The survey participants were asked in sequence the following set of questions twice: (1) from the stakeholder's "self-interest", and (2) from the "project-interest" perspective. The questions are broken down into four categories: Design, Construction, Process, and Impact as discussed in detail earlier. For brevity questions will only be listed once below, followed by data and analysis of the data set. Special instructions were provided in the survey to prompt the participants to focus their responses accordingly for each of the "self-interest" and "project-interest" sections.

Round 1 - Question 5:

Which of the following <u>Design Category</u> factors do you believe are the most important for project value-creation? (Select exactly 5): Design innovation and creativity; Achieve World-Class"; Aesthetics; Constructability; Community impact and acceptance; Clear and realistic objectives; Owner's vision; Utility and functionality; Sustainability; Evidence Based Design; Other Design Factor (description required if selected).

Round 1 - Question 6:

Which of the following <u>Construction Category</u> factors do you believe are most important for project value-creation? (Select exactly 5): Change-order management; Responsive administration and decision support; Design accuracy; Cost performance; Innovative construction means and methods; Productivity; Time performance; Safety; Constructability of Design; Quality; Other (description required if selected).

Round 1 - Question 7:

Which of the following <u>Process Category</u> factors do you believe are most important for project value-creation? (Select exactly 5): Competency / capability of project delivery team; Trust and respect; Dispute avoidance and resolution; Effective communication; Owner/User participation; Project Planning; Alignment of project objectives; Top management commitment; Risk identification and equitable allocation; Other (description required if selected).

Round 1 - Question 8:

Which of the following <u>Impact Category</u> factors do you believe are most important for project value creation? (Select exactly 5): Job satisfaction; Profit and financial objectives; Avoid contractual penalties; Contract incentives/rewards; Long-term building success / Lifecycle value of facility; Long-term business relationships; Professional reputation or image; Profession/Industry recognition or awards; Litigation avoidance; Owner/User satisfaction; Other (description required if selected).



Figure 7.7. Round 1 example question from web-survey

The responses to these questions are contained in frequency tables in the following sections. The data and analysis will be presented in the following order: (1) Overall data summary, (2) Design category, (3) Construction category, (4) Process category, (5) Impact category, and (6) a summary of the top 25% critical success factors selected for inclusion in the round 2 survey.

7.5. Overall: Data & Analysis

			S	EL	F-II	NTER	EST									
<u> </u>			Fre	eau	enc	v			Rank				Rank	Frac	tiona	1
Cat.	Critical Success Factor	0	D	В	U	ALL	0	D	В	U	ALL	0	D	В	U	ALL
	Utility and functionality	17	8	10	12	47	1.0	2.0	1.0	1.0	1.0	1.00	0.90	1.00	1.00	1.00
	Owner's vision	10	7	9	6	32	4.0	3.0	2.5	6.5	2.0	0.70	0.80	0.85	0.45	0.90
	Clear and realistic objectives	12	4	8	7	31	2.5	7.0	4.0	4.5	3.0	0.85	0.40	0.70	0.65	0.80
Ч	Sustainability	12	6	4	7	29	2.5	4.5	6.0	4.5	4.0	0.85	0.65	0.50	0.65	0.70
<u>.</u>	Evidence based design	8	5	2	11	26	5.5	6.0	9.0	2.0	5.0	0.55	0.50	0.20	0.90	0.60
es	Constructability	8	6	9	1	24	5.5	4.5	2.5	10.0	6.0	0.55	0.65	0.85	0.10	0.50
Ω	Design innovation and creativity	4	9	5	3	21	8.5	1.0	5.0	9.0	7.0	0.25	1.00	0.60	0.20	0.40
	Achieve World Class	7	1	3	4	15	7.0	10.0	7.5	8.0	8.0	0.40	0.10	0.35	0.30	0.30
	Aesthetics	4	2	0	8	14	8.5	8.5	10.0	3.0	9.0	0.25	0.25	0.10	0.80	0.20
	Community impact and acceptance	1	2	3	6	12	10.0	8.5	7.5	6.5	10.0	0.10	0.25	0.35	0.45	0.10
	Quality	15	8	7	12	42	1.0	1.5	2.5	1.0	1.0	1.00	0.95	0.85	1.00	1.00
	Responsive Admin. & Decision Support	10	8	6	10	34	4.5	1.5	4.5	2.0	2.0	0.65	0.95	0.65	0.90	0.90
nc	Time Performance	13	7	3	9	32	2.0	4.0	9.5	3.0	3.0	0.90	0.70	0.15	0.80	0.80
ţi,	Cost Performance	12	4	7	8	31	3.0	7.0	2.5	4.0	4.0	0.80	0.40	0.85	0.70	0.70
nc	Constructability of Design	10	7	9	3	29	4.5	4.0	1.0	9.5	5.0	0.65	0.70	1.00	0.15	0.60
Ħ	Change-Order Management	7	7	4	6	24	6.0	4.0	7.5	5.0	6.0	0.50	0.70	0.35	0.60	0.50
nS	Design Accuracy	6	5	6	5	22	7.0	6.0	4.5	6.5	7.0	0.40	0.50	0.65	0.45	0.40
R)	Productivity	3	3	4	3	13	9.5	8.0	7.5	9.5	8.5	0.15	0.30	0.35	0.15	0.25
	Safety	3	0	5	5	13	9.5	10.0	6.0	6.5	8.5	0.15	0.10	0.50	0.45	0.25
	Innovative Constr. Means & Methods	4	1	3	4	12	8.0	9.0	9.5	8.0	10.0	0.30	0.20	0.15	0.30	0.10
	Collaboration of Project Team	13	8	8	12	41	2.5	1.5	2.5	1.5	1.5	0.85	0.95	0.85	0.95	0.95
	Effective Communication	15	8	6	12	41	1.0	1.5	5.0	1.5	1.5	1.00	0.95	0.60	0.95	0.95
	Competency & Capability of Team	13	6	9	6	34	2.5	4.0	1.0	6.0	3.0	0.85	0.70	1.00	0.50	0.80
S	Trust & Respect	9	3	8	8	28	4.0	8.5	2.5	3.5	4.0	0.70	0.25	0.85	0.75	0.70
Se la	Alignment of Project Objectives	8	5	7	5	2.5	5.0	5.0	4.0	7.0	5.0	0.60	0.60	0.00	0.40	0.60
0 0	Owner / User Participation	5	7	4	8	24	8.0	3.0	7.5	3.5	6.0	0.30	0.80	0.35	0.75	0.50
Pr	Project Planning	5	4	4	7	20	8.0	6.5	7.5	5.0	7.0	0.30	0.45	0.35	0.60	0.40
	Top Management Commitment	5	3	4	4	16	8.0	8.5	7.5	8.0	8.0	0.30	0.25	0.35	0.30	0.30
	Risk ID & Equitable Allocation	6	4	4	1	15	6.0	6.5	7.5	10.0	9.0	0.50	0.45	0.35	0.10	0.20
	Dispute Avoidance & Resolution	4	2	1	2	9	10.0	10.0	10.0	9.0	10.0	0.10	0.10	0.10	0.20	0.10
	Owner / User satisfaction	16	10	10	12	48	15	15	1.0	1.5	10.0	0.95	0.95	1.00	0.95	1.00
	Long-term Success / Lifecycle Value	16	6	7	12	41	1.5	4.5	4.0	1.5	2.0	0.95	0.65	0.70	0.95	0.90
	Professional Reputation & Image	5	8	8	9	30	7.0	3.0	3.0	3.0	3.0	0.40	0.80	0.80	0.80	0.80
t	Long-term Business Relationships	7	10	9	3	29	6.0	1.5	2.0	8.5	4.0	0.50	0.95	0.90	0.25	0.70
)a	Profit & Financial Objectives	11	6	5	3	25	3.0	4.5	6.0	8.5	5.0	0.80	0.65	0.50	0.25	0.60
Π	Litigation Avoidance	8	3	6	4	21	5.0	7.0	5.0	6.0	6.0	0.60	0.40	0.60	0.50	0.50
II	Contract Incentives & Rewards	9	1	4	2	16	4.0	9.0	7.0	10.0	7.0	0.70	0.20	0.40	0.10	0.40
	Job Satisfaction	3	4	2	6	15	9.0	6.0	9.0	4.0	8.0	0.20	0.50	0.20	0.70	0.30
	Avoid Contractual Penalties	3	0	3	4	10	9.0	10.0	8.0	6.0	9.5	0.20	0.10	0.30	0.50	0.15
<u> </u>	Prof./Industry Recognition & Awards	3	2	1	4	10	9.0	8.0	10.0	6.0	9.5	0.20	0.30	0.10	0.50	0.15
	Stakeholder Groups Type	es: ($) = \overline{(}$	0wr	er: I	$D = \overline{D}$	esigne	er: B =	Build	ler: U	= Use	er: ALI	ī. —			

 Table 7.5. Round 1 -- Overall -- Self-Interest: Frequency and rank table

		Р	RO	JE	СТ	-INT	ERE	ST								
Cat	Critical Success Factor		Fr	equ	enc	y]	Rank]	Rank	Frac	tiona	1
Cat.	Critical Success Factor	0	D	В	U	ALL	0	D	В	U	ALL	0	D	В	U	ALL
	Utility and functionality	11	9	8	7	35	2.5	1.0	3.0	3.5	1.0	0.85	1.00	0.80	0.75	1.00
	Owner's vision	9	8	9	4	30	4.0	2.0	1.5	8.5	4.0	0.70	0.90	0.95	0.25	0.70
	Clear and realistic objectives	11	6	6	10	33	2.5	3.5	4.0	1.0	2.5	0.85	0.75	0.70	1.00	0.85
E	Sustainability	13	6	9	5	33	1.0	3.5	1.5	6.5	2.5	1.00	0.75	0.95	0.45	0.85
:18	Evidence based design	5	4	4	6	19	9.5	6.5	7.5	5.0	8.0	0.15	0.45	0.35	0.60	0.30
Ğ	Constructability	6	3	4	9	22	8.0	8.5	7.5	2.0	6.0	0.30	0.25	0.35	0.90	0.50
С	Design innovation and creativity	8	5	4	7	24	5.5	5.0	7.5	3.5	5.0	0.55	0.60	0.35	0.75	0.60
	Achieve World Class	5	3	5	4	17	9.5	8.5	5.0	8.5	9.0	0.15	0.25	0.60	0.25	0.20
	Aesthetics	7	1	2	3	13	7.0	10.0	10.0	10.0	10.0	0.40	0.10	0.10	0.10	0.10
	Community impact and acceptance	8	4	4	5	21	5.5	6.5	7.5	6.5	7.0	0.55	0.45	0.35	0.45	0.40
	Quality	15	9	9	7	40	1.0	1.0	1.0	4.0	1.0	1.00	1.00	1.00	0.70	1.00
Ц	Responsive Admin. & Decision Support	11	7	6	7	31	2.5	4.0	5.5	4.0	3.5	0.85	0.70	0.55	0.70	0.75
ō	Time Performance	8	8	8	7	31	5.5	2.5	3.0	4.0	3.5	0.55	0.85	0.80	0.70	0.75
CF.	Cost Performance	11	8	8	7	34	2.5	2.5	3.0	4.0	2.0	0.85	0.85	0.80	0.70	0.90
Ĕ	Constructability of Design	10	5	8	7	30	4.0	5.0	3.0	4.0	5.0	0.70	0.60	0.80	0.70	0.60
sti	Change-Order Management	5	4	5	9	23	8.0	6.0	7.0	1.0	6.0	0.30	0.50	0.40	1.00	0.50
ũ	Design Accuracy	6	3	4	4	17	7.0	7.5	8.0	9.0	8.0	0.40	0.35	0.30	0.20	0.30
r S	Productivity	4	2	1	2	9	9.5	9.0	9.0	10.0	10.0	0.15	0.20	0.20	0.10	0.10
\cup	Safety	8	3	6	5	22	5.5	7.5	5.5	7.5	7.0	0.55	0.35	0.55	0.35	0.40
	Innovative Constr. Means & Methods	4	1	0	5	10	9.5	10.0	10.0	7.5	9.0	0.15	0.10	0.10	0.35	0.20
	Collaboration of Project Team	12	7	7	7	33	2.0	1.5	4.5	4.0	3.0	0.90	0.95	0.65	0.70	0.80
	Effective Communication	4	6	3	3	16	9.5	3.5	8.5	9.0	10.0	0.15	0.75	0.25	0.20	0.10
- 0	Competency & Capability of Team	13	7	8	10	38	1.0	1.5	2.5	1.5	1.0	1.00	0.95	0.85	0.95	1.00
SS	Trust & Respect	4	4	3	6	17	9.5	7.5	8.5	5.5	9.0	0.15	0.35	0.25	0.55	0.20
ce	Alignment of Project Objectives	11	6	9	10	36	3.0	3.5	1.0	1.5	2.0	0.80	0.75	1.00	0.95	0.90
Q	Owner / User Participation	5	5	8	6	24	8.0	5.5	2.5	5.5	5.0	0.30	0.55	0.85	0.55	0.60
\mathbf{P}	Project Planning	6	3	5	4	18	7.0	9.5	6.0	7.0	7.5	0.40	0.15	0.50	0.40	0.35
	Top Management Commitment	10	5	7	8	30	4.5	5.5	4.5	3.0	4.0	0.65	0.55	0.65	0.80	0.70
	Risk I.D. & Equitable Allocation	10	4	1	3	18	4.5	7.5	10.0	9.0	7.5	0.65	0.35	0.10	0.20	0.35
	Dispute Avoidance & Resolution	9	3	4	3	19	6.0	9.5	7.0	9.0	6.0	0.50	0.15	0.40	0.20	0.50
	Owner / User satisfaction	15	8	10	9	42	1.0	1.5	1.0	1.5	1.0	1.00	0.95	1.00	0.95	1.00
	Long-term Success / Lifecycle Value	12	7	9	9	37	3.0	3.5	2.0	1.5	2.0	0.80	0.75	0.90	0.95	0.90
	Professional Reputation & Image	8	7	8	8	31	4.0	3.5	3.0	3.0	3.0	0.70	0.75	0.80	0.80	0.80
ct	Long-term Business Relationships	7	8	6	5	26	6.0	1.5	4.0	7.0	4.0	0.50	0.95	0.70	0.40	0.70
Da	Profit & Financial Objectives	13	4	3	5	25	2.0	6.5	8.5	7.0	5.0	0.90	0.45	0.25	0.40	0.60
I I	Litigation Avoidance	7	4	5	3	19	6.0	6.5	5.5	10.0	6.0	0.50	0.45	0.55	0.10	0.50
II	Contract Incentives & Rewards	7	0	3	6	16	6.0	10.0	8.5	4.5	9.5	0.50	0.10	0.25	0.65	0.15
	Job Satisfaction	4	4	4	6	18	9.5	6.5	7.0	4.5	7.0	0.15	0.45	0.40	0.65	0.40
	Avoid Contractual Penalties	6	3	2	5	16	8.0	9.0	10.0	7.0	9.5	0.30	0.20	0.10	0.40	0.15
	Prof./Industry Recognition & Awards	4	4	5	4	17	9.5	6.5	5.5	9.0	8.0	0.15	0.45	0.55	0.20	0.30
	Stakeholder Groups Typ	es: () = (Own	er; l	D = D	esigne	er; B =	Build	ler; U	= Use	er; ALI				

 Table 7.6. Round 1 -- Overall -- Project-Interest: Frequency and rank table

The tables 7.5. and 7.6. present the overall frequency data, and rank-orders based on the round 1 response set. The data is presented up front to support the validation and reliability testing which follows. Please note that the data and analysis section will first look at the overall

results of the survey to validate the overall panel of experts, and then will be followed by sections for individual data and analysis of each of the round 1 critical success factor categories. Since the questions were assessed as a set by the survey participants, it is important to validate how the panel of experts performed both overall and by the set, or category.

7.5.1. Overall (All Categories): Inter-rater Agreement

An important measure of the success of the Delphi method is demonstrating a consensus among the panel of experts, or in this case the panels of experts. The data collected in round 1 is dichotomous based on respondent selection of critical success factors from lists of options. The data resulting data collected is yes or no for selection. The relative importance of the critical success factors must be based on frequency counts and resulting rank-ordering. Rank-order analysis is also beneficial due to normalization of data between each of the stakeholder groups. The validation question asks how well the respondents agree with each other as a whole group, and as individual stakeholders. A good measure of inter-rater agreement for this study is Kendall's coefficient of concordance, or Kendall's W. Kendall's W is based on the mean ranking of cases within a response set, perfect for this application. Kendall's W is a coefficient ranging between zero and one, where zero equals no agreement at all between raters, and one equals complete agreement between raters.

Inter-rater agreement is calculated in SPSS Statistics 18 using non-parametric analysis of the overall response set frequencies. The table below shows that there is agreement between all

	Round #1 Inter-rater Agreement: Overall (all categories) Self-Interest Project-Interest												
			Se	lf-Inter	est			Proj	ject-In	terest			
Cat.	Critical Success Factors		М	ean Ra	nk			М	lean R	ank			
		0	D	В	U	All	0	D	В	U	All		
	Aesthetics	15.50	14.50	10.64	23.04	20.10	18.97	12.60	14.14	15.88	19.81		
	Community impact and acceptance	11.97	14.50	16.09	19.96	22.06	20.15	18.60	17.77	18.96	22.56		
	Constructability	20.21	22.50	27.00	12.27	22.84	17.79	16.60	17.77	25.12	24.13		
ц	Design innovation and creativity	15.50	28.50	19.73	15.35	24.02	20.15	20.60	17.77	22.04	22.95		
Sig	Evidence based design	20.21	20.50	14.27	27.65	19.31	16.62	18.60	17.77	20.50	17.46		
De	Clear and realistic objectives	24.91	18.50	25.18	21.50	15.39	23.68	22.60	21.41	26.65	14.72		
, ,	Owner's vision	22.56	24.50	27.00	19.96	15.78	21.32	26.60	26.86	17.42	14.32		
	Sustainability	24.91	22.50	17.91	21.50	27.16	26.03	22.60	26.86	18.96	26.48		
	Utility and functionality	30.79	26.50	28.82	29.19	15.78	23.68	28.60	25.05	22.04	19.42		
	Achieve World Class	19.03	12.50	16.09	16.88	23.24	16.62	16.60	19.59	17.42	22.95		
	Change-Order Management	19.03	24.50	17.91	19.96	16.18	16.62	18.60	19.59	25.12	15.89		
	Constructability of Design	22.56	24.50	27.00	15.35	15.39	22.50	20.60	25.05	22.04	19.03		
n	Cost Performance	24.91	18.50	23.36	23.04	20.10	23.68	26.60	25.05	22.04	19.42		
ctic	Responsive Admin. & Decision Support	22.56	26.50	21.55	26.12	18.92	23.68	24.60	21.41	22.04	20.21		
Design Accuracy 17.85 20.50 21.55 18.42 20.88 17.79 16.65 Image: the product of the										17.42	18.25		
nst	Innovative Constr. Means & Methods	15.50	12.50	16.09	16.88	22.84	15.44	12.60	10.50	18.96	23.74		
C	Productivity	28.44	26.50	22.26	20.10	23.24	13.44	14.00	12.52	14.55	22.30		
	Quality	28.44	20.30	25.50	29.19	22.00	20.30	28.00	20.80	12.04	23.74		
		26.00	24.50	19.73	24.58	29.12	20.15	26.60	21.41	22.04	17.46		
	Time Performance	20.09	24.30	22.26	19.42	10.37	20.15	20.00	25.05	22.04	19.25		
	Alignment of Project Objectives	20.21	20.50	23.30	18.42	18.92	23.08	22.60	20.80	20.05	18.25		
	Collaboration of Project Team	26.09	26.50	25.18	29.19	14.61	24.85	24.60	23.23	22.04	17.07		
	Competency & Capability of Team	26.09	22.50	27.00	19.96	22.06	26.03	24.60	25.05	26.65	20.99		
SS	Effective Communication	28.44	26.50	21.55	29.19	16.96	15.44	22.60	15.95	15.88	17.07		
oce	Dispute Avoidance & Resolution	15.50	14.50	12.45	13.81	16.57	21.32	16.60	17.77	15.88	17.85		
\Pr	Top Management Commitment	16.68	16.50	17.91	16.88	26.76	22.50	20.60	23.23	23.58	25.30		
	Owner / User Participation	16.68	24.50	17.91	23.04	29.51	16.62	20.60	25.05	20.50	27.26		
	Project Planning	16.68	18.50	17.91	21.50	20.49	17.79	16.60	19.59	17.42	20.60		
	Risk I.D. & Equitable Allocation	17.85	18.50	17.91	12.27	14.61	22.50	18.60	12.32	15.88	17.46		
	Trust & Respect	21.38	16.50	25.18	23.04	22.45	15.44	18.60	15.95	20.50	22.95		
	Litigation Avoidance	20.21	16.50	21.55	16.88	20.49	18.97	18.60	19.59	15.88	24.91		
	Avoid Contractual Penalties	14.32	10.50	16.09	16.88	26.76	17.79	16.60	14.14	18.96	23.74		
	Long-term Business Relationships	19.03	30.50	27.00	13.35	24.02	18.97	26.60	21.41	18.96	25.70		
tct	Lab Satisfaction	14.32	12.30	1/.91	10.06	20.70	15.44	10.00	17.95	20.50	18.25		
up;	Long-term Success / Lifecycle Value	29.62	22 50	23.36	29.19	16.96	24.85	24 60	26.86	25.12	22.56		
In	Owner / User satisfaction	29.62	30.50	28.82	29.19	20.10	28.38	26.60	28.68	25.12	20.21		
	Profit & Financial Objectives	23.74	22.50	19.73	15.35	18.53	26.03	18.60	15.95	18.96	17.85		
	Prof/Industry Recognition & Awards	14.32	14.50	12.45	16.88	16.57	15.44	18.60	19.59	17.42	17.85		
	Professional Reputation & Image	16.68	26.50	25.18	24.58	21.67	20.15	24.60	25.05	23.58	17.46		
			Se	elf-Intere	est			Pro	iect-In	terest			
	Statistic	0	D	В	U	All	0	D	В	U	All		
	N	17	10	11	13	51	17	10	11	13	51		
	Kendall's W ^a	.256	.300	.233	.264	.172	.143	.204	.221	.121	.115		
	Chi-square	169.43	117.00	100.02	133.60	342.74	94.92	79.56	95.02	61.10	228.15		
	df	39	39	39	39	39	39	39	39	39	39		
	Asymp. Sig.	.000	.000	.000	.000	.000	.000	.000	.000	.013	.000		
a. Ke	Kendall's Coefficient of Concordance												

 Table 7.7. Round 1 -- Overall: Inter-rater agreement

stakeholders groups and overall ranging from 0.115 to 0.300. Overall the test was significant at the 0.001 level for both self-interest and project interest. Significance levels of each of the individual stakeholder groups may also be found in Table 7.7.

Overall the results indicate that there is more agreement in the "Self-Interest" category than in the "Project-Interest" category. It is reasonable that there is a more significant level of agreement within the internal familiarity of each stakeholder group, while having a lesser amount of agreement when taking considerations outside the stakeholder group. The Designers stakeholder group had the highest level of overall agreement in the self-interest category, and the Builders stakeholder group the highest agreement in the project-centered category. Inter-rater agreement will be discussed along with each category in detail as each factor is covered later in the report.

Testing for inter-rater agreement in this survey sets a baseline for comparison with later rounds in determining if an increase in consensus is achieved. A second overall inter-rater agreement calculation will be shown at the end of this section including only the most important critical success factors selected for further analysis in round 2.

7.5.2. Overall: Reliability Testing

Reliability is a scale measure of internal consistency or agreement of values within cases, and is based on the proportion of the variation in the responses to the survey resulting from the different respondents. The answers to a reliable survey should vary because of the different respondents have different opinions, not because of random chance or confusing questions.. As discussed in the survey methodology section, intra-class correlation was calculated in SPSS to

determine Cronbach's Alpha as a measure of reliability. Cronbach's alpha provides an estimated reliability approximating the following scale: zero equals no reliability; less than 0.6 equals not reliable; 0.6 is the minimum level for a reliable scale; 0.7 is reasonably reliable; 0.8 is strongly reliable; 0.9 is very reliable; and levels approaching 1.0 (> 0.98) are potentially over reliable.

Roun	Round#1 Overall (all categories) Reliability Statistics: Intra-class Correlation												
	Stalzahaldar		Cranhaah'a	95% Confide	ence Interval	FЛ	est Tr	ue Value	0				
Perspective	Group	N	Alpha	Lower Bound	Upper Bound	Value	df1	df2	Sig				
	All	50	0.901	0.852	0.940	10.128	39	1,911	0.000				
Salf	Owner	17	0.818	0.724	0.891	5.492	39	624	0.000				
John John John John John John John John	Designer	10	0.741	0.602	0.846	3.857	39	351	0.000				
Interest	Builder	11	0.671	0.497	0.804	3.040	39	390	0.000				
	User	12	0.742	0.606	0.846	3.874	39	429	0.000				
	All	50	0.846	0.769	0.907	6.493	39	1,911	0.000				
Project-	Owner	17	0.626	0.433	0.775	2.673	39	624	0.000				
Project- Interest	Designer	10	0.566	0.335	0.742	2.307	39	351	0.000				
	Builder	11	0.649	0.462	0.790	2.845	39	390	0.000				
	User	12	0.395	0.076	0.638	1.652	39	429	0.010				

Table 7.8. Round 1 - Overall: reliability testing

A reliability test was conducted use Cronbach's alpha as the test statistic. survey round 1 proves reliable at a 0.001 significance level when considering the full panel of all 50 experts. (Note: Of the 51 samples collected, there was one incomplete response set, for purposes of consistency only the 50 complete response sets were included in validation testing) The survey questions and responses were correlated overall, and by individual stakeholder groups for consistency resulting in a Cronbach's alpha score of 0.901 and 0.846 respectively for Self-Interest and Project-Interest. Overall round 1 survey is strongly to very reliable. The individual reliability scores for the stakeholder groups in general range from reliable to strongly reliable. The two exceptions are the Designer and User groups with score 0.556 and 0.395 respectively in the "Project-Interest" category. The Designer score is only marginally unreliable, but the User

score is significantly less than reliable. Overall reliability scores were higher for Self-Interest than for Project-Interest. A reasonable assumption for the lower scores may be a contrast in understanding of the project delivery process inside and outside each participant's own stakeholder group. The overall assessment of reliability is that the round 1 survey is sufficiently reliable.

Survey Round #1: Relative Importance by Frequencies (n)												
		Sel	f-Int	erest			Proje	ect-I	ntere	st		
Design Catagory CSEs		Fr	eque	ency			Fr	eque	ency			
Design Category CSFS	0	D	B	U	ALL	0	D	В	U	ALL		
	N=17	N=10	N=11	N=13	N=51	N=17	N=10	N=11	N=12	N=50		
(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)		
Utility and functionality	17	8	10	12	47	11	9	8	7	35		
Owner's vision	10	7	9	6	32	9	8	9	4	30		
Clear and realistic objectives	12	4	8	7	31	11	6	6	10	33		
Sustainability	12	6	4	7	29	13	6	9	5	33		
Evidence based design	8	5	2	11	26	5	4	4	6	19		
Constructability	8	6	9	1	24	6	3	4	9	22		
Design innovation and creativity	4	9	5	3	21	8	5	4	7	24		
Achieve World Class	7	1	3	4	15	5	3	5	4	17		
Aesthetics	4	2	0	8	14	7	1	2	3	13		
Community impact and acceptance	1	2	3	6	12	8	4	4	5	21		
Other	3 ^a	0	2 ^b	0	5	2 ^c	1 ^d	0	0	3		
Total (n) =	Total (n) = 85 50 55 65 255 85 50 55 60 250											
Stakeholder Groups Types: O = Owner; D	Stakeholder Groups Types: O = Owner; D = Designer; B = Builder; U = User; ALL = O+D+B+U											
Notes: a. Return on Investment; Budget Control; Cc b. Safety; Completeness of Design Documer	Notes: a. Return on Investment; Budget Control; Cost & Schedule b. Safety; Completeness of Design Documents c. Elevibility: Maintainability											

7.6. Design Category: Data & Analysis

elexibility; Maintainability

d. Design Quality (minimize errors & omissions)

Table 7.9. Round 1 D	esign: Frequency table
----------------------	------------------------



Figure 7.8. Round 1 -- Design: Relative importance by frequency

7.6.1. Inter-Rater Agreement Testing

Inter-rate agreement in the design category indicates significant difference in degree of agreement between the self-interest and project-interest categories for all stakeholder groups. Stakeholders continue to demonstrate a higher degree of inter-rater agreement in the self-interest perspective than in the project-interest perspective. In the self-interest category the Builders stakeholder group demonstrated the highest degree of agreement, and the Designers demonstrated the highest degree of agreement in the project-interest category.



Figure 7.9. Round 1 -- Design: Inter-rater agreement by stakeholder groups

Round #1 Inter	Round #1 Inter-rater Agreement: Design Category												
		Se	lf-Inte	rest			Proj	ject-In	terest				
Critical Success Factors		Μ	lean R	ank			М	lean Ra	ank				
	0	D	В	U	All	0	D	В	U	All			
Aesthetics	4.24	4.00	3.09	6.08	4.41	5.12	3.55	3.91	4.35	4.35			
Community impact and acceptance	3.35	i 4.00 4.45 5.31 4.22 5.41 5.05 4.82 5.12											
Constructability	5.41	6.00	7.18	3.38	5.39	4.82	4.55	4.82	6.65	5.24			
Design innovation and creativity	4.24	7.50	5.36	4.15	5.10	5.41	5.55	4.82	5.88	5.43			
Evidence based design	5.41	5.50	4.00	7.23	5.59	4.53	5.05	4.82	5.50	4.94			
Clear and realistic objectives	6.59	5.00	6.73	5.69	6.08	6.29	6.05	5.73	7.04	6.31			
Owner's vision	6.00	6.50	7.18	5.31	6.18	5.71	7.05	7.09	4.73	6.02			
Sustainability	6.59	6.00	4.91	5.69	5.88	6.88	6.05	7.09	5.12	6.31			
Utility and functionality	8.06	7.00	7.64	7.62	7.65	6.29	7.55	6.64	5.88	6.51			
Achieve World Class	5.12	3.50	4.45	4.54	4.51	4.53	4.55	5.27	4.73	4.75			
Ctatiatia		Se	elf-Inte	rest			Pro	ject-Int	terest				
Statistic	0	D	В	U	All	0	D	В	U	All			
N	17	10	11	13	51	17	10	11	13	51			
Kendall's W ^a	.277	.264	.360	.243	.153	.092	.212	.174	.118	.082			
Chi-square	42.35	23.76	35.64	28.38	70.42	14.06	19.12	17.18	13.80	37.68			
df	9	9	9	9	9	9	9	9	9	9			
Asymp. Sig.	.000	.005	.000	.001	.000	.120	.024	.046	.130	.000			
a. Kendall's Coefficient of Concordance													

 Table 7.10. Round 1 -- Design: Inter-rater agreement

7.6.2. Reliability Testing

	Design Ca	ateg	ory Reliab	ility Statisti	cs: Intra-cla	ass Corre	elatio	n	
	Stakahaldar		Cronbooh's	95% Confid	ence Interval	FЛ	Test Tr	ue Value	0
Perspective	Group	N	Alpha	Lower Bound	Upper Bound	Value	df1	df2	Sig
	All	50	0.886	0.756	0.966	8.794	9	441	0.000
Self	Owner	17	0.837	0.640	0.952	6.124	9	144	0.000
Self- Interest	Designer 10		0.690	0.295	0.909	3.228	9	81	0.002
	Builder	11	0.822	0.598	0.948	5.624	9	90	0.000
	User	12	0.692	0.308	0.909	3.245	9	99	0.002
	All	50	0.777	0.522	0.933	4.478	9	441	0.000
Project	Owner	17	0.383	-0.360	0.817	1.620	9	144	0.115
Interest	Designer	10	0.588	0.063	0.880	2.428	9	81	0.017
Interest	Builder	11	0.524	-0.076	0.860	2.100	9	90	0.037
	User	12	0.379	-0.393	0.818	1.611	9	99	0.122

Table 7.11. Round 1 -- Design: Reliability testing

The design category overall scored between reasonably reliable and strongly reliable. Within all stakeholder groups, there is again a difference between the reliability of the selfinterest and project-interest questions. Overall the reliability of the design category is acceptable.

Correlation Matrix among <u>Design Category</u> Stakeholder Groups (for Round #1)											
Correlation Self-Interest						Project-Interest					
Matrix *	Owner	Designer	Builder	User	Owner	Designer	Builder	User			
Owner	1.000	0.368	0.442	0.424	1.000	0.627 ^a	0.532 ^a	0.193			
Designer		1.000	0.506 ^a	0.023		1.000	0.651 ^a	0.262			
Builder			1.000	-0.094			1.000	0.025			
User				1.000				1.000			
Notes:		·	0 0 7 1 1								

7.6.3. Correlation

^a Correlation is significant at the 0.05 level (2-tailed).

* Correlation Coefficient based on rank ordering within CSF category, and calculated in SPSS using Kendall's tau_b. (N = 10 CSFs)

 Table 7.12. Round 1 -- Design: Stakeholder correlation

In the design category the only significance correlation among stakeholders in the selfinterest category is between the Designer, and the Builder groups. At only 0.506 it is not reasonable to infer there is strong agreement between the designers and builder. It would be unexpected that there would be strong correlation between the self-interests of the stakeholder groups. In project-interest there is a slight, but more significant correlation amongst the owners, designers, and builders.

The critical success factor correlation matrix for self-interest demonstrates some significant negative correlations between CSFs. There are perfect negative correlations between

Owner's Vision and Evidence Based Design, and between Constructability and Aesthetics. The former is puzzling, while the latter is not unexpected. It is understandable that aesthetics and constructability could be at odd with each other. Often such examples are among the hard decisions a project team must make in weighing artistic vision against limited resources. More direct study, outside the scope of this research, might be necessary to infer cause and effect for the perfect negative correlation between Owner's Vision and Evidence Based Design. One possible inference may be a strong assumption among the other project stakeholders that the evidence based design movement is at odds with the owner's vision, or the overarching goals of the evidence based design movement are not embraced by all stakeholder groups.

Correlation Matrix among <u>Design Category</u> Critical Success Factors (for Round #1)												
		Self-Interest										
Correlation Matrix *	Utility & function- ality	Owner's vision	Clear & realistic objectives	Sustain- ability	Evidence based design	Construct- ability	Design innovation & creativity	Achieve World Class	Aesthetics	Community impact & acceptance		
Utility & functionality	1.000	-0.236	0.775	0.000	0.236	0.000	-0.707	0.775	0.000	0.236		
Owner's vision		1.000	-0.183	-0.548	-1.000 ^a	0.913	0.667	0.183	-0.913	0.000		
Clear & realistic objectives			1.000	0.400	0.183	0.000	-0.548	0.800	0.000	-0.183		
Sustainability				1.000	0.548	-0.400	-0.183	0.000	0.400	-0.548		
Evidence based design					1.000	-0.913	-0.667	-0.183	0.913	0.000		
Constructability						1.000	0.548	0.400	-1.000 ^a	-0.183		
Design innovation & creativity							1.000	-0.183	-0.548	-0.333		
Achieve World Class								1.000	-0.400	-0.183		
Aesthetics									1.000	0.183		
Community impact & acceptance										1.000		

^a Correlation is significant at the 0.01 level (2-tailed).

* Correlation Coefficient based on rank ordering within CSF category, and calculated in SPSS using Kendall's tau_b. (N = 4 Stakeholder Groups: Owner; Designer; Builder; User)



No matter what the inference may be for the examples above, a significant issue raised is that factors of significant importance to one or more of the stakeholder groups self-interest which there is diametrically opposed by another stakeholder group. Or stated more simply, there is a significant disagreement in terms of relative importance. Imagine what formal, or informal (unspoken), issues such diametrically opposed viewpoints may cause to the efficiency of collaboration. Each party fights for the interests most important to itself. This very issue is central to the overall thesis of this dissertation. What process may be able to best resolve such conflicts in a way that enhances collaboration and maximizes value-creation? And, do the stakeholders agree on what process is most effective? These are questions to be addressed in the subsequent survey rounds.

Correlation Matrix among <u>Design Category</u> Critical Success Factors (for Round #1)											
	Project-Interest										
Correlation Matrix *	Utility & function- ality	Owner's vision	Clear & realistic objectives	Sustain- ability	Evidence based design	Construct- ability	Design innovation & creativity	Achieve World Class	Aesthetics	Community impact & acceptance	
Utility & functionality	1.000	0.183	-0.183	0.000	0.000	-0.913	-0.400	-0.400	0.258	0.258	
Owner's vision		1.000	-1.000 ^a	0.548	-0.913	-0.333	-0.183	0.548	-0.236	-0.236	
Clear & realistic objectives			1.000	-0.548	0.913	0.333	0.183	-0.548	0.236	0.236	
Sustainability				1.000	-0.800	-0.183	-0.800	0.000	0.516	0.516	
Evidence based design					1.000	0.183	0.400	-0.400	0.000	0.000	
Constructability						1.000	0.548	0.183	-0.236	-0.236	
Design innovation & creativity							1.000	0.400	-0.775	-0.775	
Achieve World Class								1.000	-0.775	-0.775	
Aesthetics									1.000	1.000 ^a	
Community impact & acceptance										1.000	
Notes:											
^a Correlation is significant at the 0.01 le	^a Correlation is significant at the 0.01 level (2-tailed).										
* Correlation Coefficient based on rank Builder; User)	* Correlation Coefficient based on rank ordering within CSF category, and calculated in SPSS using Kendall's tau_b. (N = 4 Stakeholder Groups: Owner; Designer; Builder: User)										

Table 7.14. Round 1 -- Design -- Project-Interest: CSF correlation matrix

Comparing the self-interest and project-interest CSF correlation matrices it is obvious that there is a difference between the two perspectives. The most significant correlation in project-interest exists between Owner's Vision and Clear & Realistic Expectation, and Aesthetics and Community Impact & Acceptance. The negative correlation between expectations and owner's vision may indicate belief that the owner's vision contains objectives which are not realistic to the other project stakeholders.

7.6.4. Relative Importance by Rank-Orders

Based on frequency selection critical success factors are ranked on a scale of 1 to 10 where 1 is the highest rank. Ties are reflected by the difference between the higher and lower rank-order of adjacent CSF. For the purposes of future data comparisons, the rank orders are

		Sel	f-Inte	rest		Project-Interest					
Design Category CSFs		%	Rank	a, b		% Rank ^{a, b}					
Design Category Cors	0	D	В	U	ALL	0	D	В	U	ALL	
	N=17	N=10	N=11	N=13	N=51	N=17	N=10	N=11	N=12	N=50	
(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	
Utility and functionality	1.00	0.90	1.00	1.00	1.0	0.85	1.00	0.80	0.75	1.00	
Owner's vision	0.70	0.80	0.85	0.45	0.9	0.70	0.90	0.95	0.25	0.70	
Clear and realistic objectives	0.85	0.40	0.70	0.65	0.8	0.85	0.75	0.70	1.00	0.85	
Sustainability	0.85	0.65	0.50	0.65	0.7	1.00	0.75	0.95	0.45	0.85	
Evidence based design	0.55	0.50	0.20	0.90	0.6	0.15	0.45	0.35	0.60	0.30	
Constructability	0.55	0.65	0.85	0.10	0.5	0.30	0.25	0.35	0.90	0.50	
Design innovation and creativity	0.25	1.00	0.60	0.20	0.4	0.55	0.60	0.35	0.75	0.60	
Achieve World Class	0.40	0.10	0.35	0.30	0.3	0.15	0.25	0.60	0.25	0.20	
Aesthetics	0.25	0.25	0.10	0.80	0.2	0.40	0.10	0.10	0.10	0.10	
Community impact and acceptance	0.10	0.25	0.35	0.45	0.1	0.55	0.45	0.35	0.45	0.40	
Other ^c	Х	Х	Х	Х	Х	Х	Х	Х	X	Х	
Stakeholder Groups Types: O	= Owr	ner; D	= Des	igner;	B = B	Builder	:;U=	User			

Notes:

a. %Rank: normalized rank order where 1.00 = highest ranked.

b. Selection criteria for highlighted cells (Green/Bold): greater than or equal to 0.75

c. "Other" not included in rank-order series

Table 7.15.	Round 1	Design:	Relative i	importance	%Rank
1 4010 7.10.	Itounu I	Designe	iterative i	mportance	/ UIXamix

converted to a fractional ranking (%Rank) below where the rank-order scale is between 0 and 1 where 0 equals not ranked, and 1 equals the highest rank. The selection criteria for this round and round 2 is a %Rank score of 0.75 or higher in any category.



Figure 7.10. Round 1 -- Design: Relative importance by stakeholders

Survey Round #1: Relative Importance by Frequencies (n)										
		Sel	f-Int	erest	;	Project-Interest				
Construction Category CSEs		Fr	eque	ency		Frequency				
Construction Category CSFS	0	D	В	U	ALL	0	D	В	U	ALL
	N=17	N=10	N=11	N=13	N=51	N=17	N=10	N=11	N=12	N=50
(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)
Quality	15	8	7	12	42	15	9	9	7	40
Responsive Admin. & Decision Support	10	8	6	10	34	11	7	6	7	31
Time Performance	13	7	3	9	32	8	8	8	7	31
Cost Performance	12	4	7	8	31	11	8	8	7	34
Constructability of Design	10	7	9	3	29	10	5	8	7	30
Change-Order Management	7	7	4	6	24	5	4	5	9	23
Design Accuracy	6	5	6	5	22	6	3	4	4	17
Productivity	3	3	4	3	13	4	2	1	2	9
Safety	3	0	5	5	13	8	3	6	5	22
Innovative Constr. Means & Methods	4	1	3	4	12	4	1	0	5	10
Other	0	0	1 ^b	0	3	2 ^c	0	0	0	2
Total (n) =	85	50	55	65	255	85	50	55	60	250
Stakeholder Groups Types: O = Owner; D	D = De	signer	r; B =	Builde	er; U =	User;	ALL	= O+]	D+B+	U
Notes: a. Integrated schedule; Transition to operations b. Collaboration c. Communication; Equipment planning										

7.7. Construction Category: Data & Analysis

 Table 7.16. Round 1 -- Construction: Frequency table



Figure 7.11. Round 1 -- Construction: CSF relative importance by frequency

Round #1 Inter-rater Agreement: Construction Category											
		Se	lf-Inte	rest		Project-Interest					
Critical Success Factors	Mean Rank						Mean Rank				
	0	D	В	U	All	0	D	В	U	All	
Change-Order Management	5.12	6.50	4.86	5.31	5.38	4.56	5.00	5.27	6.65	5.33	
Constructability of Design	6.00	6.50	7.14	4.15	5.87	6.03	5.50	6.64	5.88	6.02	
Cost Performance	6.59	5.00	6.23	6.08	6.07	6.32	7.00	6.64	5.88	6.41	
Responsive Admin. & Decision Support	6.00	7.00	5.77	6.85	6.36	6.32	6.50	5.73	5.88	6.12	
Design Accuracy	4.82	5.50	5.77	4.92	5.19	4.85	4.50	4.82	4.73	4.75	
Innovative Constr. Means & Methods	4.24	3.50	4.41	4.54	4.21	4.26	3.50	3.00	5.12	4.06	
Productivity	3.94	4.50	4.86	4.15	4.30	4.26	4.00	3.45	3.96	3.96	
Quality	7.47	7.00	6.23	7.62	7.15	7.50	7.50	7.09	5.88	7.00	
Safety	3.94	3.00	5.32	4.92	4.30	5.44	4.50	5.73	5.12	5.24	
Time Performance	6.88	6.50	4.41	6.46	6.17	5.44	7.00	6.64	5.88	6.12	
Statistic		Se	elf-Inte	rest		Project-Interest					
Statistic	0	D	В	U	All	0	D	В	U	All	
Ν	17	10	11	13	51	17	10	11	13	51	
Kendall's W ^a	.234	.304	.114	.205	.145	.161	.288	.279	.092	.151	
Chi-square		27.36	11.30	23.95	66.34	24.65	25.92	27.65	10.80	69.29	
df	9	9	9	9	9	9	9	9	9	9	
Asymp. Sig.	.000	.001	.256	.004	.000	.003	.002	.001	.290	.000	
a Kandall's Coafficient of Concordance											

7.7.1. Inter-Rater Agreement Testing

 Table 7.17. Round 1 -- Construction: inter-rater agreement



Figure 7.12. Round 1 -- Construction: Inter-rater agreement by stakeholder group

Inter-rater agreement in the construction category departs slightly from the overall trend in agreement levels between self and project interests. The overall agreement in the construction self-interest category is slightly less than in the project-interest category. The self-interest agreement remains somewhat constant while the project-interest agreement improved by 0.07 over the level in the design category. The designers demonstrated similar levels of agreement between self and project interest, while the builders demonstrated the largest difference in agreement between the self and project perspectives. The difference among builders may be inferred as recognizing differing internal goals among the construction entities represented in the builders stakeholder group (different construction companies), but a more universal agreement of what's important to making a project successful.

7.7.2. Reliability Testing

Construction Category Reliability Statistics: Intra-class Correlation										
	Stalzaholdar		Cranbaah'a	95% Confid	ence Interval	F Test True Value 0				
Perspective	Group	N	Alpha	Lower Bound	Upper Bound	Value	df1	df2	Sig	
	All	50	0.873	0.729	0.962	7.905	9	441	0.000	
Salf	Owner	17	0.873	0.729	0.962	7.905	9	441	0.000	
Sell-	Designer 10		0.746	0.421	0.926	3.931	9	81	0.000	
merest	Builder	Builder 11		-0.753	0.772	1.288	9	90	0.254	
	User	12	0.580	0.056	0.876	2.378	9	99	0.018	
	All	50	0.888	0.760	0.967	8.919	9	441	0.000	
Project	Owner	17	0.675	0.283	0.904	3.073	9	144	0.002	
Interest	Designer	10	0.725	0.375	0.920	3.640	9	81	0.001	
merest	Builder	11	0.742	0.417	0.924	3.876	9	90	0.000	
	User	12	0.182	-0.837	0.760	1.222	9	99	0.290	

 Table 7.18. Round 1 -- Construction: Reliability testing

Overall the data in the construction category is strongly reliable for both self and project perspectives. Builder reliability in self-interest, and user reliability is project interest prove unreliable, likely caused by the very low inter-rater agreement as displayed in the section above.
7.7.3. Correlations

In self-interest the only significant correlations are between owners and designer, and the owners and users. In the project interest there is significantly more correlation between stakeholder groups, again emphasizing the significant differences perspective makes when evaluating what is important to the stakeholder groups in regards to the project delivery process.

Correlation Matrix among	Construction Category	Stakeholder	Groups (for	Round #1)
U	0 7		I (,

Correlation		Self-I	nterest		Project-Interest						
Matrix *	Owner	Designer	Builder	User	Owner	Designer	Builder	User			
Owner	1.000	0.579 ^a	0.329	0.595 ^a	1.000	0.699 ^b	0.732 ^b	0.391			
Designer		1.000	0.171	0.519		1.000	0.840 ^b	0.581 ^a			
Builder			1.000	0.048			1.000	0.508			
User				1.000				1.000			
Mada											

Notes:

^a Correlation is significant at the 0.05 level (2-tailed).

^b Correlation is significant at the 0.01 level (2-tailed).

* Correlation Coefficient based on rank ordering within CSF category, and calculated in SPSS using Kendall's tau_b. (N = 10 CSFs)

 Table 7.19 Round 1-- Construction -- Stakeholder correlation matrix

For self-interest critical success factors in the construction category there are two

significant completely negative correlations: Time Performance and Design Accuracy; and Cost

Performance and Change-Order Management. Logically each of these pairs is directly linked

together. Design problems directly cause construction delays, and change-orders drive up cost.

Correlation	Matrix ar	nong <u>Const</u>	ruction Cat	egory Crit	ical Succes	s Factors	(for Roun	d #1)			
		Self-Interest									
Correlation Matrix *	Quality	Responsive Admin. & Decision Support	Time Perform- ance	Cost Perform- ance	Construct- ability of Design	Change- Order Manage- ment	Design Accuracy	Product- ivity	Safety	Innovative Construct- ion Means & Methods	
Quality	1.000	0.516	0.775	-0.707	-0.707	0.707	-0.775	-0.775	-0.516	-0.577	
Responsive Admin. & Decision Support		1.000	0.000	-0.913	-0.183	0.913	0.000	0.000	-0.400	-0.894	
Time Performance			1.000	-0.183	-0.548	0.183	-1.000 ^a	-0.800	-0.400	0.000	
Cost Performance				1.000	0.333	-1.000 ^a	0.183	0.183	0.548	0.816	
Constructability of Design					1.000	-0.333	0.548	0.913	-0.183	0.408	
Change-Order Management						1.000	-0.183	-0.183	-0.548	-0.816	
Design Accuracy							1.000	0.800	0.400	0.000	
Productivity								1.000	0.000	0.224	
Safety									1.000	0.224	
Innovative Constr. Means & Methods										1.000	
Notes:									-		

^a Correlation is significant at the 0.01 level (2-tailed).

* Correlation Coefficient based on rank ordering within CSF category, and calculated in SPSS using Kendall's tau_b. (N = 4 Stakeholder Groups: Owner; Designer; Builder; User)

Table 7.20. Round 1 Co	onstruction: Self-Inte	erest: CSF correlation
------------------------	------------------------	------------------------

Correlation	Matrix ar	nong <u>Const</u>	ruction Cat	egory Crit	ical Succes	s Factors	(for Roun	d #1)				
		Project-Interest										
Correlation Matrix *	Quality	Responsive Admin. & Decision Support	Time Perform- ance	Cost Perform- ance	Construct- ability of Design	Change- Order Manage- ment	Design Accuracy	Product- ivity	Safety	Innovative Construct- ion Means & Methods		
Quality	1.000	-0.516	-0.333	b	-0.516	-0.707	0.775	1.000 ^a	0.577	-0.775		
Responsive Admin. & Decision Support		1.000	-0.516	b	0.000	0.183	0.000	-0.516	-0.224	0.800		
Time Performance			1.000	b	0.258	0.707	-0.516	-0.333	-0.577	0.000		
Cost Performance				1.000	b	b	b	b	b	b		
Constructability of Design					1.000	0.183	-0.800	-0.516	0.224	0.000		
Change-Order Management						1.000	-0.548	-0.707	-0.816	0.548		
Design Accuracy							1.000	0.775	0.224	-0.200		
Productivity								1.000	0.577	-0.775		
Safety									1.000	-0.671		
Innovative Constr. Means & Methods										1.000		

Notes:

^a Correlation is significant at the 0.01 level (2-tailed).

^b Cost is unanimously ranked at #2.

* Correlation Coefficient based on rank ordering within CSF category, and calculated in SPSS using Kendall's tau_b. (N = 4 Stakeholder Groups: Owner; Designer; Builder; User).

Table 7.21. Round 1 -- Construction -- Project-Interest: CSF correlation matrix

		Sel	f-Inte	erest			Proje	ect-In	teres	t
Construction Category CSFs		%	Rank	a, b			%	Rank	a, b	
Construction Category Cors	0	D	В	U	ALL	0	D	В	U	ALL
	N=17	N=10	N=11	N=13	N=51	N=17	N=10	N=11	N=12	N=50
(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)
Quality	1.00	0.95	0.85	1.00	1.0	1.00	1.00	1.00	0.70	1.00
Responsive Admin. & Decision Support	0.65	0.95	0.65	0.90	0.9	0.85	0.70	0.55	0.70	0.75
Time Performance	0.90	0.70	0.15	0.80	0.8	0.55	0.85	0.80	0.70	0.75
Cost Performance	0.80	0.40	0.85	0.70	0.7	0.85	0.85	0.80	0.70	0.90
Constructability of Design	0.65	0.70	1.00	0.15	0.6	0.70	0.60	0.80	0.70	0.60
Change-Order Management	0.50	0.70	0.35	0.60	0.5	0.30	0.50	0.40	1.00	0.50
Design Accuracy	0.40	0.50	0.65	0.45	0.4	0.40	0.35	0.30	0.20	0.30
Productivity	0.15	0.30	0.35	0.15	0.3	0.15	0.20	0.20	0.10	0.10
Safety	0.15	0.10	0.50	0.45	0.3	0.55	0.35	0.55	0.35	0.40
Innovative Construction Means & Methods	0.30	0.20	0.15	0.30	0.1	0.15	0.10	0.10	0.35	0.20
Other ^c	Х	Х	Х	X	Х	Х	X	Х	X	Х
Stakeholder Groups Types: O = Owner; D = Designer; B = Builder; U = User										
Notes:										

7.7.4. Relative Importance by Rank-Orders

a. %Rank: normalized rank order where 1.00 = highest ranked.

b. Selection criteria for highlighted cells (Green/Bold): greater than or equal to 0.75

c. "Other" not included in rank-order series



 Table 7.22. Round 1 -- Construction: relative importance by %Rank

Figure 7.13. Round 1 -- Construction: Relative importance by stakeholder

Survey Round #1: Relat	Survey Round #1: Relative Importance by Frequencies (n)										
		Sel	f-Int	erest	-]	Proje	ect-I	ntere	st	
		Fr	eque	ency			Frequency				
Process Category CSFs	0	D	В	U	ALL	0	D	В	U	ALL	
	N=17	N=10	N=11	N=13	N=51	N=17	N=10	N=11	N=12	N=50	
(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	
Collaboration of Project Team	13	8	8	12	41	12	7	7	7	33	
Effective Communication	15	8	6	12	41	4	6	3	3	16	
Competency & Capability of Team	13	6	9	6	34	13	7	8	10	38	
Trust & Respect	9	3	8	8	28	4	4	3	6	17	
Alignment of Project Objectives	8	5	7	5	25	11	6	9	10	36	
Owner / User Participation	5	7	4	8	24	5	5	8	6	24	
Project Planning	5	4	4	7	20	6	3	5	4	18	
Top Management Commitment	5	3	4	4	16	10	5	7	8	30	
Risk I.D. & Equitable Allocation	6	4	4	1	15	10	4	1	3	18	
Dispute Avoidance & Resolution	4	2	1	2	9	9	3	4	3	19	
Other	3 ^a	0	0	0	3	1 ^b	0	0	0	1	
Total (n) =	85	50	55	65	255	85	50	55	60	250	
Stakeholder Groups Types: $O = Owner$; $D = Designer$; $B = Builder$; $U = User$; $ALL = O+D+B+U$											
Notes: a. Strong contractual cost controls; coherent organized information transfer; change avoidance											

7.8. Process Category: Data & Analysis

Table 7.23. Round 1 -- Process: Frequency table



Figure 7.14. Round 1 -- Process: Relative importance by frequency

Round #1 Inter	-rater	· Agre	emer	nt: Pr	ocess	Categ	ory			
		Se	lf-Inte	rest			Pro	ject-In	terest	
Critical Success Factors		М	lean R	ank		Mean Rank				
	0	D	В	U	All	0	D	В	U	All
Alignment of Project Objectives	5.41	5.50	6.18	4.92	5.47	6.26	6.00	7.09	7.04	6.59
Collaboration of Project Team	6.88	7.00	6.64	7.62	7.04	6.56	6.50	6.18	5.88	6.29
Competency & Capability of Team	6.88	6.00	7.09	5.31	6.35	6.85	6.50	6.64	7.04	6.78
Effective Communication	7.47	7.00	5.73	7.62	7.04	4.21	6.00	4.36	4.35	4.63
Dispute Avoidance & Resolution	4.24	4.00	3.45	3.77	3.90	5.68	4.50	4.82	4.35	4.92
Top Management Commitment	4.53	4.50	4.82	4.54	4.59	5.97	5.50	6.18	6.27	6.00
Owner / User Participation	4.53	6.50	4.82	6.08	5.37	4.50	5.50	6.64	5.50	5.41
Project Planning	4.53	5.00	4.82	5.69	4.98	4.79	4.50	5.27	4.73	4.82
Risk I.D. & Equitable Allocation	4.82	5.00	4.82	3.38	4.49	5.97	5.00	3.45	4.35	4.82
Trust & Respect	5.71	4.50	6.64	6.08	5.76	4.21	5.00	4.36	5.50	4.73
Statiatia		Se	elf-Inte	rest			Pro	ject-In	terest	
Statistic	0	D	В	U	All	0	D	В	U	All
N	17	10	11	13	51	17	10	11	13	51
Kendall's W ^a	.203	.168	.187	.295	.164	.142	.080	.213	.174	.103
Chi-square	31.09	15.12	18.49	34.48	75.23	21.74	7.20	21.11	20.40	47.48
df	9	9	9	9	9	9	9	9	9	9
Asymp. Sig.	.000	.088	.030	.000	.000	.010	.616	.012	.016	.000
a. Kendall's Coefficient of Concordance										

7.8.1. Inter-Rater Agreement Testing

Table 7.24. Round 1-- Process: Inter-rater agreement





In general the trend of more agreement in the self-interest category continues, except for the builders which had a slightly higher level of agreement in the project-interest category.

	Process C	ateg	gory Reliab	ility Statist	ics: Intra-cl	ass Corr	elatio	on	
	Stakeholder		Cronbach's	95% Confid	ence Interval	FΊ	0		
Perspective	Group	N	Alpha	Lower Bound	Upper Bound	Value	df1	df2	Sig
	All	50	0.892	0.769	0.968	9.258	9	441	0.000
Self	Owner	17	0.755	0.460	0.928	4.080	9	144	0.000
Interest	Designer	10	0.450	-0.252	0.839	1.817	9	81	0.077
merest	Builder	11	0.565	0.017	0.872	2.297	9	90	0.023
	User	12	0.770	0.484	0.932	4.349	9	99	0.000
	All	50	0.827	0.629	0.948	5.780	9	441	0.000
Project	Owner	17	0.623	0.168	0.888	2.649	9	144	0.007
Interest	Designer	10	-0.278	-1.907	0.626	0.783	9	81	0.633
merest	Builder	11	0.631	0.167	0.892	2.710	9	90	0.008
	User	12	0.610	0.124	0.885	2.562	9	99	0.011

7.8.2. Reliability Testing

Table 7.25. Round 1-- Process: Reliability testing

Overall the reliability for the process category critical success factors is between reliable and strongly reliable. In general the stakeholder groups individually remain reliable with the exception of the designer stakeholder group.

7.8.3. Correlations

Correlation Matrix among Process Category Stakeholder Groups (for Round #1)											
Correlation	Self-Interest Project-Interest										
Matrix *	Owner	Designer	Builder	User	Owner	Designer	Builder	User			
Owner	1.000	0.578 ^a	0.859 ^b	0.500	1.000	0.434	0.447	0.434			
Designer		1.000	0.489	0.588 ^a		1.000	0.415	0.525			
Builder			1.000	0.483			1.000	0.732 ^b			
User				1.000				1.000			
Matan											

Notes:

^a Correlation is significant at the 0.05 level (2-tailed).

^b Correlation is significant at the 0.01 level (2-tailed).

* Correlation Coefficient based on rank ordering within CSF category, and calculated in SPSS using Kendall's tau b. (N = 10 CSFs)

 Table 7.26. Round 1 -- Process: Stakeholder correlation matrix

The highest level of process category correlation between stakeholders is between the

owners and builders for self-interest, and builders and users for project interest. There is

significant but lesser correlation between the owners / designers, and designers / users.

Correlati	Correlation Matrix among Process Category Critical Success Factors (for Round #1)											
		Self-Interest										
Correlation Matrix *	Collab- oration of Project Team	Effective Commun- ication	Comp- etency & Capability of Team	Trust & Respect	Alignment of Project Objectives	Owner / User Particip- ation	Project Planning	Top Manage- ment Commit- ment	Risk I.D. & Equitable Allocation	Dispute Avoidance & Resolution		
Collaboration of Project Team	1.000	0.577	-0.816	-0.408	-0.671	0.894	0.671	-0.894	-0.577	0.577		
Effective Communication		1.000	-0.707	-0.707	-0.775	0.258	0.000	-0.775	-0.333	0.333		
Competency & Capability of Team			1.000	0.333	0.913	-0.548	-0.548	0.913	0.707	-0.707		
Trust & Respect				1.000	0.183	-0.183	0.183	0.548	-0.236	0.236		
Alignment of Project Objectives					1.000	-0.400	-0.400	0.800	0.775	-0.775		
Owner / User Participation						1.000	0.800	-0.600	-0.516	0.516		
Project Planning							1.000	-0.400	-0.775	0.775		
Top Management Commitment								1.000	0.516	-0.516		
Risk I.D. & Equitable Allocation									1.000	-1.000 ^a		
Dispute Avoidance & Resolution										1.000		
Notes:												

^a Correlation is significant at the 0.01 level (2-tailed).

* Correlation Coefficient based on rank ordering within CSF category, and calculated in SPSS using Kendall's tau_b. (N = 4 Stakeholder Groups: Owner; Designer; Builder; User).

Table 7.27 Round 1-- Process -- Self-Interest: CSF correlation matrix

Correlati	on Matrix	among <u>Pro</u>	cess Categ	ory Critica	l Success I	Factors (f	or Round #	#1)				
		Project-Interest										
Correlation Matrix *	Collab- oration of Project Team	Effective Commun- ication	Comp- etency & Capability of Team	Trust & Respect	Alignment of Project Objectives	Owner / User Particip- ation	Project Planning	Top Manage- ment Commit-	Risk I.D. & Equitable Allocation	Dispute Avoidance & Resolution		
Collaboration of Project Team	1.000	a	-0.894	0.408	0.671	0.671	0.671	0.000	-0.816	0.000		
Effective Communication		1.000	а	а	а	а	а	а	а	a		
Competency & Capability of Team			1.000	-0.183	-0.800	-0.400	-0.800	-0.183	0.548	0.200		
Trust & Respect				1.000	0.183	0.183	-0.183	-0.667	-0.333	0.548		
Alignment of Project Objectives					1.000	0.000	0.400	0.183	-0.183	-0.400		
Owner / User Participation						1.000	0.400	-0.183	-0.913	0.400		
Project Planning							1.000	0.548	-0.548	-0.400		
Top Management Commitment								1.000	0.000	-0.913		
Risk I.D. & Equitable Allocation									1.000	-0.183		
Dispute Avoidance & Resolution										1.000		

Notes:

^a Effective Communication is unanimously ranked at #1.

^b No correlation for this matrix is significant at the 0.05 level (2-tailed).

* Correlation Coefficient based on rank ordering within CSF category, and calculated in SPSS using Kendall's tau_b. (N = 4 Stakeholder Groups: Owner; Designer; Builder; User)

Table 7.28. . Round 1--Process--Project-Interest: CSF correlation matrix

		Sel	f-Inte	rest			Proje	ect-In	terest	t
Process Category CSFs		%	Rank	a, b			%	Rank	a, b	
	0	D	В	U	ALL	0	D	В	U	ALL
	N=17	N=10	N=11	N=13	N=51	N=17	N=10	N=11	N=12	N=50
(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)
Collaboration of Project Team	0.85	0.95	0.85	0.95	1.0	0.90	0.95	0.65	0.70	0.80
Effective Communication	1.00	0.95	0.60	0.95	1.0	0.15	0.75	0.25	0.20	0.10
Competency & Capability of Team	0.85	0.70	1.00	0.50	0.8	1.00	0.95	0.85	0.95	1.00
Trust & Respect	0.70	0.25	0.85	0.75	0.7	0.15	0.35	0.25	0.55	0.20
Alignment of Project Objectives	0.60	0.60	0.70	0.40	0.6	0.80	0.75	1.00	0.95	0.90
Owner / User Participation	0.30	0.80	0.35	0.75	0.5	0.30	0.55	0.85	0.55	0.60
Project Planning	0.30	0.45	0.35	0.60	0.4	0.40	0.15	0.50	0.40	0.35
Top Management Commitment	0.30	0.25	0.35	0.30	0.3	0.65	0.55	0.65	0.80	0.70
Risk I.D. & Equitable Allocation	0.50	0.45	0.35	0.10	0.2	0.65	0.35	0.10	0.20	0.35
Dispute Avoidance & Resolution	0.10	0.10	0.10	0.20	0.1	0.50	0.15	0.40	0.20	0.50
Other ^c	Х	Х	Х	Х	Х	Х	Х	Х	Х	Х
Stakeholder Groups Types: O	Stakeholder Groups Types: O = Owner; D = Designer; B = Builder; U = User									
Notes: a_{2} %Pank: normalized rank order where 1 00 = highest ranked										

7.8.4. Relative Importance by Rank-Orders

b. Selection criteria for highlighted cells (Green/Bold): greater than or equal to 0.75

c. "Other" not included in rank-order series



 Table 7.29. Round 1 - Process: Relative importance by %Rank

Figure 7.16. Round 1 - Process: %Rank by stakeholder

Survey Round #1: Relative Importance by Frequencies (n)												
		Sel	f-Int	erest]	Proje	ect-I	ntere	st		
Impact Category CSEs		Fr	eque	ency			Fr	eque	ency			
impact Category CSI's	0	D	В	U	ALL	0	D	В	U	ALL		
	N=17	N=10	N=11	N=13	N=51	N=17	N=10	N=11	N=12	N=50		
(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)		
Owner / User satisfaction	16	10	10	12	48	15	8	10	9	42		
Long-term Success / Lifecycle Value	16	6	7	12	41	12	7	9	9	37		
Professional Reputation & Image	5	8	8	9	30	8	7	8	8	31		
Long-term Business Relationships	7	10	9	3	29	7	8	6	5	26		
Profit & Financial Objectives	11	6	5	3	25	13	4	3	5	25		
Litigation Avoidance	8	3	6	4	21	7	4	5	3	19		
Contract Incentives & Rewards	9	1	4	2	16	7	0	3	6	16		
Job Satisfaction	3	4	2	6	15	4	4	4	6	18		
Avoid Contractual Penalties	3	0	3	4	10	6	3	2	5	16		
Prof./Industry Recognition & Awards	3	2	1	4	10	4	4	5	4	17		
Other	4 ^a	0	0	1 ^b	5	2 ^c	1 ^d	0	0	3		
Total (n) = 85 50 55 60 250 85 50 55 60 250												
Stakeholder Groups Types: O = Owner; D = Designer; B = Builder; U = User; ALL = O+D+B+U												
Notes:												

7.9. Impact Category: Data & Analysis

a. Change avoidance, commissioning; achieve goals & objectives; swift project delivery

b. User functionality & satisfaction

c. Smooth project team hand-off to operations; swift transition to equipping, outfitting & transition

d. Public/taxpayer acceptance & satisfaction



Table 7.30. Round 1- Impact: Frequency table

Figure 7.17. Round 1 - Impact: Relative importance by frequency

7.9.1.	Inter-Rater	Agreement	Testing
--------	-------------	-----------	---------

Round #1 Inter-rater Agreement: Impact Category												
		Se	lf-Inte	rest			Proj	ject-In	terest			
Critical Success Factors		М	lean R	ank			М	lean Ra	ank			
	0	D	В	U	All	0	D	В	U	All		
Litigation Avoidance	5.47	4.50	5.73	4.77	5.16	5.12	5.05	5.27	4.35	4.94		
Avoid Contractual Penalties	4.00	3.00	4.36	4.77	4.08	4.82	4.55	3.91	5.12	4.65		
Long-term Business Relationships	5.18	8.00	7.09	4.38	5.94	5.12	7.05	5.73	5.12	5.63		
Contract Incentives & Rewards	5.76	3.50	4.82	4.00	4.67	5.12	3.05	4.36	5.50	4.65		
Job Satisfaction	4.00	5.00	3.91	5.54	4.57	4.24	5.05	4.82	5.50	4.84		
Long-term Success / Lifecycle Value	7.82	6.00	6.18	7.85	7.12	6.59	6.55	7.09	6.65	6.71		
Owner / User satisfaction	7.82	8.00	7.55	7.85	7.80	7.47	7.05	7.55	6.65	7.20		
Profit & Financial Objectives	6.35	6.00	5.27	4.38	5.55	6.88	5.05	4.36	5.12	5.53		
Prof./Industry Recognition & Awards	4.00	4.00	3.45	4.77	4.08	4.24	5.05	5.27	4.73	4.75		
Professional Reputation & Image	4.59	7.00	6.64	6.69	6.04	5.41	6.55	6.64	6.27	6.12		
Statistia		Se	elf-Inte	rest			Pro	ject-In	terest			
Statistic	0	D	В	U	All	0	D	В	U	All		
Ν	17	10	11	13	51	17	10	11	13	51		
Kendall's W ^a	.311	.464	.273	.326	.232	.178	.237	.220	.097	.123		
Chi-square	47.65	41.76	27.00	38.20	106.30	27.26	21.29	21.76	11.40	56.30		
df	9	9	9	9	9	9	9	9	9	9		
Asymp. Sig.	.000	.000	.001	.000	.000	.001	.011	.010	.249	.000		
a Kendall's Coefficient of Concordance												

Table 7.31. Round 1 - Impact: Inter-rater agreement



Figure 7.18. Round 1 - Impact: Inter-rater agreement by stakeholder

	Impact Category Reliability Statistics: Intra-class Correlation												
	Stakabaldar		Cronbach'a	95% Confid	ence Interval	F٦	Test Ti	ue Value	0				
Perspective	Group	N	Alpha	Lower Bound	Upper Bound	Value	df1	df2	Sig				
	All	50	0.934	0.859	0.980	15.155	9	441	0.000				
Salf	Owner	17	0.862	0.696	0.959	7.237	9	144	0.000				
Juterest	Designer	10	0.872	0.708	0.962	7.791	9	81	0.000				
merest	Builder	11	0.733	0.398	0.922	3.750	9	90	0.000				
	User	12	0.834	0.627	0.951	6.019	9	99	0.000				
	All	50	0.857	0.694	0.957	7.007	9	441	0.000				
Project	Owner	17	0.712	0.365	0.915	3.468	9	144	0.001				
Interest	Designer	10	0.641	0.184	0.895	2.789	9	81	0.007				
merest	Builder	11	0.645	0.198	0.896	2.818	9	90	0.006				
	User	12	0.230	-0.730	0.774	1.298	9	99	0.248				

7.9.2. Reliability Test

Table 7.32. Round 1 -- Impact: Reliability testing table

Overall the survey question tested for high reliability at a 0.001 significance level.

Among the stakeholder groups there was also mostly strong reliability at a 0.001 significance level for self-interest and 0.05 level for project interest. The single exception is for the users' stakeholder group in the project-interest category which failed to achieve a 0.01 significance level.

7.9.3. Correlations

Correlatio	Correlation Matrix among Impact Category Stakeholder Groups (for Round #1)												
Correlation		Self-Iı	nterest			Project-	Interest						
Matrix *	Owner	Designer	Builder	User	Owner	Designer	Builder	User					
Owner	1.000	0.357	0.489	0.099	1.000	0.359	0.453	0.444					
Designer		1.000	0.705 ^a	0.338		1.000	.0752 ^a	0.312					
Builder			1.000	0.236			1.000	0.434					
User				1.000				1.000					
Notes:													

^a Correlation is significant at the 0.01 level (2-tailed).

* Correlation Coefficient based on rank ordering within CSF category, and calculated in SPSS using Kendall's tau_b. (N = 10 CSFs)

 Table 7.33. Round 1 -- Impact: Stakeholder correlation matrix

Correlation Matrix among Impact Category Critical Success Factors (for Round #1)												
					Self-In	terest						
Correlation Matrix *	Owner / User satisfactio n	Long-term Success / Lifecycle Value	Profession- al Reputation & Image	Long-term Business Relation- ships	Profit & Financial Objectives	Litigation Avoid- ance	Contract Incentives & Rewards	Job Satisfact- ion	Avoid Contract- ual Penalties	Profession- al Industry Recognitio n & Awards		
Owner / User satisfaction	1.000	а	а	а	а	а	а	а	а	а		
Long-term Success / Lifecycle Value		1.000	-0.577	-0.816	0	0.577	0.000	0.408	0.671	0.671		
Professional Reputation & Image			1.000	0.236	-0.707	-0.333	-0.707	0.236	0.000	0.000		
Long-term Business Relationships				1.000	0.333	-0.707	0.000	-0.333	-0.913	-0.548		
Profit & Financial Objectives					1.000	-0.236	0.667	-0.333	-0.548	-0.183		
Litigation Avoidance						1.000	0.236	-0.236	0.775	0.000		
Contract Incentives & Rewards							1.000	-0.667	-0.183	-0.548		
Job Satisfaction								1.000	0.183	0.913		
Avoid Contractual Penalties									1.000	0.400		
Prof./Industry Recognition & Awards										1.000		
INDIES.												

notes.

^a Owner/User Satisfaction is unanimously ranked at #1.

 $^{\rm b}\,$ No correlation for this matrix is significant at the 0.05 level (2-tailed).

* Correlation Coefficient based on rank ordering within CSF category, and calculated in SPSS using Kendall's tau_b. (N = 4 Stakeholder Groups: Owner; Designer; Builder; User)

Table 7.34. Round 1 -- Impact -- Self-Interest: CSF correlation matrix

Correlat	Correlation Matrix among <u>Impact Category</u> Critical Success Factors (for Round #1)													
		Project-Interest												
Correlation Matrix *	Owner / User satisfactio n	Long-term Success / Lifecycle Value	Profession- al Reputation & Image	Long-term Business Relation- ships	Profit & Financial Objectives	Litigation Avoid- ance	Contract Incentives & Rewards	Job Satisfact- ion	Avoid Contract- ual Penalties	Profession- al Industry Recognitio n &				
Owner / User satisfaction	1.000	а	а	а	а	а	а	а	а	а				
Long-term Success / Lifecycle Value		1.000	0.516	-0.548	-0.548	-0.775	0.548	0.548	0.183	-0.224				
Professional Reputation & Image			1.000	0.236	-0.707	-0.333	-0.236	0.707	-0.236	0.577				
Long-term Business Relationships				1.000	0.000	0.707	-1.000 ^b	0.000	-0.667	0.816				
Profit & Financial Objectives					1.000	0.236	0.000	-0.333	0.333	-0.408				
Litigation Avoidance						1.000	-0.707	-0.707	-0.707	0.577				
Contract Incentives & Rewards							1.000	0.000	0.667	-0.816				
Job Satisfaction								1.000	0.333	0.000				
Avoid Contractual Penalties									1.000	-0.816				
Prof./Industry Recognition & Awards										1.000				

Notes:

^a Owner/User Satisfaction is unanimously ranked at #1.

^b Correlation is significant at the 0.01 level (2-tailed).

* Correlation Coefficient based on rank ordering within CSF category, and calculated in SPSS using Kendall's tau_b. (N = 4 Stakeholder Groups: Owner; Designer; Builder; User)

Table 7.35. Round 1 -- Impact -- Project-Interest: CSF correlation matrix

		Sel	f-Inte	rest			Proje	ect-In	teres	t
Impact Category CSEs		%	Rank	a, b			%	Rank	a, b	
impact Category CSI's	0	D	В	U	ALL	0	D	В	U	ALL
	N=17	N=10	N=11	N=13	N=51	N=17	N=10	N=11	N=12	N=50
(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)
Owner / User satisfaction	0.95	0.95	1.00	0.95	1.0	1.00	0.95	1.00	0.95	1.00
Long-term Success / Lifecycle Value	0.95	0.65	0.70	0.95	0.9	0.80	0.75	0.90	0.95	0.90
Professional Reputation & Image	0.40	0.80	0.80	0.80	0.8	0.70	0.75	0.80	0.80	0.80
Long-term Business Relationships	0.50	0.95	0.90	0.25	0.7	0.50	0.95	0.70	0.40	0.70
Profit & Financial Objectives	0.80	0.65	0.50	0.25	0.6	0.90	0.45	0.25	0.40	0.60
Litigation Avoidance	0.60	0.40	0.60	0.50	0.5	0.50	0.45	0.55	0.10	0.50
Contract Incentives & Rewards	0.70	0.20	0.40	0.10	0.4	0.50	0.10	0.25	0.65	0.15
Job Satisfaction	0.20	0.50	0.20	0.70	0.3	0.15	0.45	0.40	0.65	0.40
Avoid Contractual Penalties	0.20	0.10	0.30	0.50	0.2	0.30	0.20	0.10	0.40	0.15
Prof./Industry Recognition & Awards	0.20	0.30	0.10	0.50	0.2	0.15	0.45	0.55	0.20	0.30
Other ^c	X	Х	Х	Х	Х	Х	Х	Х	X	Х
Stakeholder Groups Types: O = C)wner;	D = I	Design	ner; B	= Buil	der; U	I = Us	er; AL	L	
Notes:										

7.9.4. Relative Importance by Rank-Orders

a. %Rank: normalized rank order where 1.00 = highest ranked.

b. Selection criteria for highlighted cells (Green/Bold): greater than or equal to 0.75

c. "Other" not included in rank-order series



Table 7.36.	Round 1	Impact:	relative im	portance b	v %Rank

Figure 7.19. Round 1 -- Impact: relative importance by %Rank by stakeholder

Round 1 Consolidated List	Round 1 Consolidated List of Top-25 Relative Important in All Categories											
		Sel	f-Inter	est			Proj	ect-Int	erest			
Critical Success Factors		%	Rank	a, b			%	Rank	a, b			
Critical Success Factors	0	D	В	U	ALL	0	D	В	U	ALL		
	N=17	N=10	N=11	N=13	N=51	N=17	N=10	N=11	N=12	N=50		
(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)		
Owner / User satisfaction	0.95	0.95	1.00	0.95	1.0	1.00	0.95	1.00	0.95	1.00		
Utility and functionality	1.00	0.90	1.00	1.00	1.0	0.85	1.00	0.80	0.75	1.00		
Quality	1.00	0.95	0.85	1.00	1.0	1.00	1.00	1.00	0.70	1.00		
Long-term Success / Lifecycle Value	0.95	0.65	0.70	0.95	0.9	0.80	0.75	0.90	0.95	0.90		
Competency & Capability of Team	0.85	0.70	1.00	0.50	0.8	1.00	0.95	0.85	0.95	1.00		
Collaboration of Project Team	0.85	0.95	0.85	0.95	1.0	0.90	0.95	0.65	0.70	0.80		
Responsive Admin. & Decision Support	0.65	0.95	0.65	0.90	0.9	0.85	0.70	0.55	0.70	0.75		
Clear and realistic objectives	0.85	0.40	0.70	0.65	0.8	0.85	0.75	0.70	1.00	0.85		
Owner's vision	0.70	0.80	0.85	0.45	0.9	0.70	0.90	0.95	0.25	0.70		
Professional Reputation & Image	0.40	0.80	0.80	0.80	0.8	0.70	0.75	0.80	0.80	0.80		
Cost Performance	0.80	0.40	0.85	0.70	0.7	0.85	0.85	0.80	0.70	0.90		
Time Performance	0.90	0.70	0.15	0.80	0.8	0.55	0.85	0.80	0.70	0.75		
Sustainability	0.85	0.65	0.50	0.65	0.7	1.00	0.75	0.95	0.45	0.85		
Alignment of Project Objectives	0.60	0.60	0.70	0.40	0.6	0.80	0.75	1.00	0.95	0.90		
Long-term Business Relationships	0.50	0.95	0.90	0.25	0.7	0.50	0.95	0.70	0.40	0.70		
Constructability of Design ^d	0.65	0.70	1.00	0.15	0.6	0.70	0.60	0.80	0.70	0.60		
Profit & Financial Objectives	0.80	0.65	0.50	0.25	0.6	0.90	0.45	0.25	0.40	0.60		
Owner / User Participation	0.30	0.80	0.35	0.75	0.5	0.30	0.55	0.85	0.55	0.60		
Effective Communication	1.00	0.95	0.60	0.95	1.0	0.15	0.75	0.25	0.20	0.10		
Change-Order Management	0.50	0.70	0.35	0.60	0.5	0.30	0.50	0.40	1.00	0.50		
Constructability ^e	0.55	0.65	0.85	0.10	0.5	0.30	0.25	0.35	0.90	0.50		
Design innovation and creativity	0.25	1.00	0.60	0.20	0.4	0.55	0.60	0.35	0.75	0.60		
Trust & Respect	0.70	0.25	0.85	0.75	0.7	0.15	0.35	0.25	0.55	0.20		
Evidence based design ^f	0.55	0.50	0.20	0.90	0.6	0.15	0.45	0.35	0.60	0.30		
Dispute Avoidance & Resolution ^g	0.10	0.10	0.10	0.20	0.1	0.50	0.15	0.40	0.20	0.50		
Aesthetics	0.25	0.25	0.10	0.80	0.2	0.40	0.10	0.10	0.10	0.10		
Stakeholder Groups Types: C) = Owi	ner; D =	= Desig	gner; B	= Buil	der; U	= User	; ALL				

7.10. Top 25% Summary

Notes:

a. %Rank: normalized rank order where 1.00 = highest ranked

b. Selection criteria for highlighted areas: Top 25%: Greater than or equal to 0.75 (Shaded & Bold highlight)

c. "Long-term Business Relationships" will not be submitted for further analysis in future survey rounds because it has little direct bearing on project delivery process under the acquisition rules and regulations for public projects (the focus of this research study). In fact, control measures are in place which purposely limit or forbid relationship based procurement to protect the public's interests.

d. Constructability was evaluated in two different categories: "Design" and "Construction". This CSF was evaluated in the "Construction" category as it relates to implementability of designs and designers intent in the field. Going forward in Survey Round #2 the two "constructabilities" will be consolidated using the Construction category values.

e. See note (c) above. This "Design" category CSF (constructability) was intended to assess considerations critical to design decisions, and will be consolidated into a single CSF using the "Construction" category values in subsequent survey rounds.

f. Evidence based design was originally ommitted from the Round #2 survey do to an error. EBD was submitted for evaluation during the Round #3 survey and the results were later amended to the Round 2 survey data set. This explaination provides background for note (g) below.

g. "Dispute Avoidance & Resolution" was erroneously submitted in place of "Evidence Based Design". for assessment in the Round # 2 survey. See note (f) above for background.

Table 7.37. Round 1 top-25% highest relative importance by %Rank

Round #1 Inter-Rater Agreement: Top 25%											
		Se	elf-Inter	rest			Pro	ject-Int	erest		
Critical Success Factors		Ν	1ean Ra	ınk			Ν	/lean Ra	ınk		
	0	D	В	U	All	0	D	В	U	All	
Owner / User Satisfaction	16.47	16.95	16.73	16.23	16.56	16.29	14.80	16.09	14.38	15.47	
Clear & Realistic Objectives	13.65	9.75	14.55	11.62	12.56	13.47	12.40	11.73	15.31	13.35	
Effective Communication	15.76	14.55	12.36	16.23	14.91	14.88	13.60	13.91	15.31	14.53	
Competency & Capability of Project Team	14.35	12.15	15.64	10.69	13.26	14.18	13.60	12.82	12.54	13.35	
Utility & Functionality	17.18	14.55	16.73	16.23	16.32	13.47	16.00	13.91	12.54	13.82	
Collaboration of Project Delivery Team	14.35	14.55	14.55	16.23	14.91	13.47	12.40	15.00	15.31	14.06	
Trust & Respect	11.53	8.55	14.55	12.54	11.85	8.53	10.00	8.45	11.62	9.59	
Alignment of Project Objectives	10.82	10.95	13.45	9.77	11.15	12.76	11.20	12.82	13.46	12.65	
Owner / User Participation	8.71	13.35	10.18	12.54	10.91	9.24	11.20	13.91	11.62	11.24	
Production of Specified Quality	15.76	14.55	13.45	16.23	15.15	16.29	16.00	15.00	12.54	15.00	
Long-term Bldg. Success & Lifecycle Value	16.47	12.15	13.45	16.23	14.91	14.18	13.60	15.00	14.38	14.29	
Owner's vision	12.24	13.35	15.64	10.69	12.79	12.06	14.80	15.00	9.77	12.65	
Project Time Performance	14.35	13.35	9.09	13.46	12.79	11.35	14.80	13.91	12.54	12.88	
Constructability	12.24	13.35	15.64	7.92	12.09	12.76	11.20	13.91	12.54	12.65	
Project Cost Performance	13.65	9.75	13.45	12.54	12.56	13.47	14.80	13.91	12.54	13.59	
Responsive Admin. & Decision Support	12.24	14.55	12.36	14.38	13.26	13.47	13.60	11.73	12.54	12.88	
Sustainability	13.65	12.15	10.18	11.62	12.09	14.88	12.40	15.00	10.69	13.35	
Evidence Based Design	10.82	10.95	8.00	15.31	11.38	11.35	11.20	9.55	12.54	11.24	
Dispute Avoidance & Resolution	8.00	7.35	6.91	7.00	7.38	8.53	12.40	8.45	8.85	9.35	
Profit & Financial Objectives	12.94	12.15	11.27	7.92	11.15	14.88	10.00	8.45	10.69	11.47	
Professional Reputation & Image	8.71	14.55	14.55	13.46	12.32	11.35	13.60	13.91	13.46	12.88	
Aesthetics	8.00	7.35	5.82	12.54	8.56	10.65	6.40	7.36	8.85	8.65	
Design Innovation & Creativity	8.00	15.75	11.27	7.92	10.21	9.24	10.00	9.55	11.62	10.06	
Change Order Reduction	10.12	13.35	10.18	10.69	10.91	9.24	10.00	10.64	14.38	11.00	
Ctatiatia		Se	elf-Inter	rest			Pro	ject-Int	erest		
Statistic	0	D	В	U	All	0	D	В	U	All	
N	17	10	11	13	51	17	10	11	13	51	
Kendall's W ^a	.236	.186	.251	.257	.141	.154	.149	.193	.104	.095	
Chi-square	92.35	42.83	63.57	76.70	164.83	60.22	34.38	48.81	31.23	111.46	
df	23	23	23	23	23	23	23	23	23	23	
Asymp. Sig.	.000	.007	.000	.000	.000	.000	.060	.001	.117	.000	
a. Kendall's Coefficient of Concordance											

7.10.1. Inter-Rater Agreement Testing of Top-25%

 Table 7.38. Round 1 Top-25% inter-rater agreement

Inter-rater agreement was calculated independently for the 24 critical success factors selected for the round 2 survey to provide a baseline for comparing overall inter-rater agreement

between the survey rounds. For the Delphi method to be successful it must be demonstrated that there is an increase in consensus among the members of the panel of experts.

7.11. Round 1 Findings

- The round 1 Delphi survey is a success based on the high participation level of invited respondents, and the outstanding qualifications of the panel(s) of experts assembled.
- The expertise of the panel of experts as a whole, and for each stakeholder group, is strongly validated in terms of education, experience, and professional certifications. The participants' population is extremely qualified for this study.
- 3. The results of the round 1 survey have successfully provided proof of concept that differences exist between each project stakeholder group's (Owner, Designer, Builder, and User) opinion of what are the most important factors for project success, and that relative importance depends on considerations of self-interest or project-interest. Survey round 2 will further investigate the various differences between interpretations of relative importance.
- 4. Twenty-four critical success factors were successfully selected for relative importance for further analysis in the round 2 survey.

- Inter-rater agreement was validated for all categories of the round 1 survey using Kendall's coefficient of concordance as a test statistic at an overall 0.001 level of significance.
- 6. Reliability of the survey was validated at a 0.001 significance level utilizing intra-class correlation of the critical success factors to determine Cronbach's alpha as a test statistic.

CHAPTER 8.

ROUND 2 SURVEY REPORT



Figure 8.1. Round 2: Overall data results

8.1. Introduction

The round 2 survey report is organized in the following order: Purpose, Administrative Information, Questions Asked, Data Collected, Analysis, Findings, Summary and Conclusion. The format of the round 2 report differs from the round 1 report because the critical success factors categories have been consolidated into one comprehensive list. Now the data will be divided into categories based on stakeholder groups (All Groups, Owners, Designers, Builders, & Users) for the remainder of Delphi study. The breakdown between self-interest and projectinterest remains. Table 8.1. above gives a visual pre-view of the data to be discussed below. Note the increased linearity of the response set compared to the corresponding figure at the start of the round 1 report. The increased linearity visually indicates an increase in inter-rater agreement, which will be statistically substantiated later in this report.

8.1.1. Outline of the Report

Descriptive statistic and related analysis charts and tables will be presented and will reflect choices by the overall panel of experts, and then by each individual stakeholder group. Once again the survey is tested for inter-rater agreement among participants, and for reliability. Relative importance is evaluated by rank-order, and the top-25% factors are selected to be included in the round 3 survey. Additional analysis in the round 2 report includes investigation of the differences between the relative importance from the self-interest and project-interest perspectives to include significance testing. Finally the results of rounds 1 and 2 are compared to test for validation of the Delphi process.

8.1.2. Purpose

The round 2 survey served three purposes:

1. To further evaluate the 24 most important critical success factors selected in round 1. The goal is to further refine the list of most important critical success factors by relative importance for the third and final survey round.

2. To compare inter-rater agreement statistics between round 1 and round 2 to determine if the panel of experts have achieved an increase in consensus between rounds.

3. To used the refined data set to identify and measure differences in relative importance between the self-interest and project-interest for each of the stakeholder groups.

8.1.3. Administrative

The round 2 survey was published to the participants on 15 March 2011. Weekly email reminders were automatically sent to remaining participants until the survey was closed 20 days later at 11:59 pm on 04 April 2011. The survey invitations were sent via email to a total of 51 participants. The round 2 survey invitation list was limited to the respondents who participated in the round 1 survey. All participants were provided with the result of the round 1 survey for their consideration prior to starting the round 2 survey.

Participation Statistics:

- Invited: 51
- Started: 44
- Completed: 42
- Participation Rate: 82.3%
- Completion Rate: 95.5%
- Drop outs (after starting): 2
- Average Time to Complete Survey: 9 minutes

8.2. Round 2 Sample

Round 2 - Question 1: Stakeholder Group Identification

Which of the following categories do you best identify with? (Select only one): Design Team; Facility User or User's Representative; Construction Team; Owner or Owner's Representative.



Figure 8.2. Round 2 stakeholder group participation

There was an attrition of 9 participants from rounds 1 to 2. The owners, designers, and users stakeholder groups lost 4, 1, and 3 participants respectively. Given the busy schedules and responsibilities of the sample population such attrition was planned for, and the total numbers for each stakeholder group remains within the normal range for a Delphi survey. The stakeholder

distribution and overall number of participants remains appropriate for the purposes of this research project.

8.3. Evaluation of Critical Success Factors

The primary purpose of the round 3 survey to further evaluate the twenty-four critical success factors selected in round 1 for relative importance on a five-point Likert-scale: from least important to most important. The survey participants were asked the following question twice. Once each for from the self-interest and project-interest perspectives. For brevity the question will only be listed once, but the data set will be divided by self and project interests.

Round 2 - Question 2: Evaluation of Critical Success Factors

In light of the factors listed below, how would you rate the importance of each factor in general from your (a) Stakeholder Perspective (Self-Interest) / (b) Project Perspective (Project-Interest)?

The following 24 factors were then rated on a one to five scale (1 = least important, and 5 = most important):

1. Aesthetics

- 2. Alignment of Project Objectives
- 3. Effective Communication
- 4. Profit and Financial Objectives
- 5. Owner / User Participation
- 6. Competency & Capability of Project Delivery Team
- 7. Trust & Respect
- 8. Change-Order Reduction
- 9. Project Time Performance
- 10. Dispute Avoidance & Resolution
- 11. Utility & Functionality
- 12. Owner's Vision
- 13. Owner / User Satisfaction
- 14. Sustainability
- 15. Long-term Building Success / Lifecycle Value

- 16. Professional Reputation & Image
- 17. Clear & Realistic Objectives
- 18. Production of Specified Quality
- 19. Responsive Administration & Decision Support
- 20. Collaboration of Project Delivery Team
- 21. Design Innovation & Creativity
- 22. Constructability
- 23. Project Cost Performance
- 24. Evidence Based Design*

* Evidence Base Design was erroneously omitted from the round 2 survey. The factor was included as a separate question on the round 3 survey, and the results have been amended to this round 2 report. All data in this report include the mean score as rated by the panel of experts.)

	Least	F	Important		Most
Aesthetics *		6		•	mportan
Alignment of Project Objectives *	ŏ	ŏ	ŏ	ŏ	ŏ
Effective Communication *	ŏ	ŏ	ŏ	ŏ	ŏ
Profit and Financial Objectives *	ŏ	õ	õ	õ	õ
Owner / User Participation *	ŏ	ŏ	ŏ	ŏ	õ
Competency & Capability of Project Delivery Team *	ŏ	ŏ	ŏ	õ	ŏ
Trust & Respect *	ŏ	õ	õ	õ	õ
Charge-Order Reduction *	ŏ	ō	ē	õ	ō
Project Time Performance *	ŏ	ŏ	õ	ŏ	õ
Dispute Avoidance & Resolution *	ě	ē	ē	ō.	ō
Utility & Functionality *		0	0	0	0
Owner's vision *	0	0	0	0	0
Owner / User Satisfaction *	0	0	0	0	0
Sustainability *	ē.	ō	0	ē	0
Long-term Building Success / Lifecycle Value *	õ	ō	õ	õ	Ö
Professional Reputation & Image *		0	0	0	0
Clear & Realistic Objectives *	0	0	0	0	0
Production of Specified Quality *	0	0	0	Ō	0
Responsive Administration & Decision Support *	0	0	0	0	0
Collaboration of Project Delivery Team *	0	Ō	Ō	Ō	Ō
Design Innovation & Creativity *	0	0	0	0	0
Constructability *	õ	ē	0	0	0
Project Cost Performance *	0	0	0	0	0

Figure 8.3 Round 2 example question from web-survey

Data & Analysis

8.3.1. Overall

Round #2 Descrip	tive	e St	tat	isti	ics: (Over	all (O	wner	, Des	igr	ıer	·, E	Build	ler, &	u Use	r)	
					S	elf-Int	terest						Pre	oject-l	Interes	st	
Critical Success Factor	N	Range	Min	Max	Sum	Mean	Std. Error	Std. Dev.	Var.	Range	Min	Max	Sum	Mean	Std. Error	Std. Dev.	Var.
Aesthetics	42	4	1	5	147	3.500	0.142	0.917	0.841	4	1	5	140	3.333	0.143	0.928	0.862
Alignment of Project Objectives	42	3	2	5	171	4.071	0.125	0.808	0.653	2	3	5	179	4.262	0.113	0.734	0.539
Change Order Reduction	42	4	1	5	136	3.238	0.140	0.906	0.820	3	2	5	152	3.619	0.148	0.962	0.925
Collaboration of Project Delivery Team	42	2	3	5	173	4.119	0.124	0.803	0.644	3	2	5	184	4.381	0.123	0.795	0.632
Effective Communication	42	2	3	5	183	4.357	0.112	0.727	0.528	3	2	5	185	4.405	0.118	0.767	0.588
Competency & Capability of Project Team	42	3	2	5	179	4.262	0.118	0.767	0.588	2	3	5	185	4.405	0.103	0.665	0.442
Constructability	42	4	1	5	159	3.786	0.139	0.898	0.807	4	1	5	164	3.905	0.163	1.055	1.113
Project Cost Performance	42	4	1	5	161	3.833	0.132	0.853	0.728	3	2	5	163	3.881	0.141	0.916	0.839
Dispute Avoidance & Resolution	42	3	2	5	143	3.405	0.137	0.885	0.783	3	2	5	162	3.857	0.139	0.899	0.808
Design Innovation & Creativity	42	3	2	5	146	3.476	0.133	0.862	0.743	4	1	5	141	3.357	0.122	0.791	0.625
Evidence Based Design	42	4	1	5	155	3.690	0.147	0.950	0.902	4	1	5	146	3.476	0.141	0.917	0.841
Long-term Bldg. Success & Lifecycle Value	42	2	3	5	171	4.071	0.125	0.808	0.653	3	2	5	166	3.952	0.136	0.882	0.778
Owner / User Participation	42	2	3	5	181	4.310	0.120	0.780	0.609	3	2	5	166	3.952	0.140	0.909	0.827
Owner / User Satisfaction	42	2	3	5	196	4.667	0.081	0.526	0.276	2	3	5	192	4.571	0.097	0.630	0.397
Profit & Financial Objectives	42	4	1	5	145	3.452	0.164	1.064	1.132	4	1	5	149	3.548	0.171	1.109	1.229
Production of Specified Quality	42	2	3	5	173	4.119	0.109	0.705	0.498	3	2	5	173	4.119	0.124	0.803	0.644
Clear & Realistic Objectives	42	2	3	5	184	4.381	0.108	0.697	0.485	3	2	5	185	4.405	0.113	0.734	0.539
Professional Reputation & Image	42	3	2	5	151	3.595	0.160	1.037	1.076	4	1	5	147	3.500	0.157	1.018	1.037
Responsive Admin. & Decision Support	42	2	3	5	159	3.786	0.111	0.717	0.514	3	2	5	161	3.833	0.140	0.908	0.825
Sustainability	42	3	2	5	157	3.738	0.141	0.912	0.832	3	2	5	148	3.524	0.137	0.890	0.792
Project Time Performance	42	3	2	5	161	3.833	0.118	0.762	0.581	3	2	5	171	4.071	0.129	0.838	0.702
Trust & Respect	42	2	3	5	173	4.119	0.114	0.739	0.546	3	2	5	177	4.214	0.125	0.813	0.660
Utility & Functionality	42	3	2	5	183	4.357	0.131	0.850	0.723	4	1	5	175	4.167	0.132	0.853	0.728
Owner's vision	42	2	3	5	175	4.167	0.122	0.794	0.630	4	1	5	153	3.643	0.163	1.055	1.113

Table 8.1. Round 2 - Overall: descriptive statistics



Figure 8.4. Round 2 -- Overall: relative importance by additive mean scores



Figure 8.5. Round 2 - Overall: relative importance by average mean score

8.3.2. Owners

R	Round #2 Descriptive Statistics: Owners Self-Interest Project-Interest																
					S	elf-In	terest		1				Pr	oject-l	Intere	st	
Critical Success Factor	N	Range	Min	Max	Sum	Mean	Std. Error	Std. Dev.	Var.	Range	Min	Max	Sum	Mean	Std. Error	Std. Dev.	Var.
Aesthetics	13	2	3	5	48	3.692	0.175	0.630	0.397	3	2	5	44	3.385	0.241	0.870	0.756
Alignment of Project Objectives	13	2	3	5	52	4.000	0.226	0.816	0.667	2	3	5	55	4.231	0.201	0.725	0.526
Change Order Reduction	13	3	2	5	43	3.308	0.263	0.947	0.897	3	2	5	43	3.308	0.286	1.032	1.064
Collaboration of Project Delivery Team	13	2	3	5	54	4.154	0.222	0.801	0.641	2	3	5	56	4.308	0.208	0.751	0.564
Effective Communication	13	2	3	5	58	4.462	0.215	0.776	0.603	3	2	5	56	4.308	0.263	0.947	0.897
Competency & Capability of Project Team	13	3	2	5	54	4.154	0.249	0.899	0.808	2	3	5	57	4.385	0.213	0.768	0.590
Constructability	13	3	2	5	49	3.769	0.257	0.927	0.859	3	2	5	53	4.077	0.265	0.954	0.910
Project Cost Performance	13	4	1	5	49	3.769	0.303	1.092	1.192	3	2	5	52	4.000	0.277	1.000	1.000
Dispute Avoidance & Resolution	13	3	2	5	41	3.154	0.249	0.899	0.808	3	2	5	49	3.769	0.343	1.235	1.526
Design Innovation & Creativity	13	3	2	5	44	3.385	0.266	0.961	0.923	3	2	5	44	3.385	0.213	0.768	0.590
Evidence Based Design	13	4	1	5	48	3.692	0.286	1.032	1.064	4	1	5	43	3.308	0.308	1.109	1.231
Long-term Bldg. Success & Lifecycle Value	13	2	3	5	56	4.308	0.208	0.751	0.564	3	2	5	53	4.077	0.265	0.954	0.910
Owner / User Participation	13	2	3	5	55	4.231	0.231	0.832	0.692	3	2	5	51	3.923	0.265	0.954	0.910
Owner / User Satisfaction	13	2	3	5	58	4.462	0.183	0.660	0.436	2	3	5	54	4.154	0.222	0.801	0.641
Profit & Financial Objectives	13	4	1	5	41	3.154	0.296	1.068	1.141	2	3	5	50	3.846	0.249	0.899	0.808
Production of Specified Quality	13	2	3	5	53	4.077	0.239	0.862	0.744	2	3	5	55	4.231	0.201	0.725	0.526
Clear & Realistic Objectives	13	2	3	5	59	4.538	0.183	0.660	0.436	3	2	5	58	4.462	0.243	0.877	0.769
Professional Reputation & Image	13	3	2	5	41	3.154	0.249	0.899	0.808	3	2	5	45	3.462	0.291	1.050	1.103
Responsive Admin. & Decision Support	13	1	3	4	44	3.385	0.140	0.506	0.256	3	2	5	44	3.385	0.266	0.961	0.923
Sustainability	13	3	2	5	50	3.846	0.274	0.987	0.974	3	2	5	45	3.462	0.268	0.967	0.936
Project Time Performance	13	2	3	5	48	3.692	0.208	0.751	0.564	3	2	5	49	3.769	0.257	0.927	0.859
Trust & Respect	13	2	3	5	49	3.769	0.201	0.725	0.526	3	2	5	54	4.154	0.296	1.068	1.141
Utility & Functionality	13	1	4	5	58	4.462	0.144	0.519	0.269	3	2	5	54	4.154	0.249	0.899	0.808
Owner's vision	13	2	3	5	57	4.385	0.213	0.768	0.590	4	1	5	45	3.462	0.402	1.450	2.103

Table 8.2. Round 2 -- Owners: descriptive statistics



Figure 8.6. Round 2 -- Owners: relative importance by mean score

8.3.3. Designers

F	Rou	nd	#2	D	escr	iptive	e Stat	istics	: Des	igı	ıer	`S					
					ŗ	Self-Int	erest						Pı	roject-l	nterest	İ	
Critical Success Factor	N	Range	Min	Мах	Sum	Mean	Std. Error	Std. Dev.	Var.	Range	Min	Max	Sum	Mean	Std. Error	Std. Dev.	Var.
Aesthetics	9	1	3	4	34	3.778	0.147	0.441	0.194	3	1	4	29	3.222	0.324	0.972	0.944
Alignment of Project Objectives	9	2	3	5	35	3.889	0.261	0.782	0.611	2	3	5	39	4.333	0.236	0.707	0.500
Change Order Reduction	9	3	1	4	28	3.111	0.309	0.928	0.861	2	3	5	38	4.222	0.222	0.667	0.444
Collaboration of Project Delivery Team	9	2	3	5	40	4.444	0.294	0.882	0.778	1	4	5	43	4.778	0.147	0.441	0.194
Effective Communication	9	2	3	5	38	4.222	0.278	0.833	0.694	2	3	5	40	4.444	0.242	0.726	0.528
Competency & Capability of Project Team	9	2	3	5	40	4.444	0.242	0.726	0.528	1	4	5	42	4.667	0.167	0.500	0.250
Constructability	9	1	4	5	39	4.333	0.167	0.500	0.250	2	3	5	39	4.333	0.236	0.707	0.500
Project Cost Performance	9	1	3	4	34	3.778	0.147	0.441	0.194	2	3	5	36	4.000	0.236	0.707	0.500
Dispute Avoidance & Resolution	9	1	3	4	31	3.444	0.176	0.527	0.278	2	3	5	34	3.778	0.222	0.667	0.444
Design Innovation & Creativity	9	2	3	5	37	4.111	0.200	0.601	0.361	3	1	4	33	3.667	0.333	1.000	1.000
Evidence Based Design	9	3	2	5	33	3.667	0.289	0.866	0.750	3	2	5	32	3.556	0.294	0.882	0.778
Long-term Bldg. Success & Lifecycle Value	9	2	3	5	38	4.222	0.278	0.833	0.694	3	2	5	33	3.667	0.333	1.000	1.000
Owner / User Participation	9	2	3	5	40	4.444	0.242	0.726	0.528	3	2	5	37	4.111	0.309	0.928	0.861
Owner / User Satisfaction	9	1	4	5	44	4.889	0.111	0.333	0.111	1	4	5	44	4.889	0.111	0.333	0.111
Profit & Financial Objectives	9	2	3	5	36	4.000	0.236	0.707	0.500	2	3	5	33	3.667	0.289	0.866	0.750
Production of Specified Quality	9	2	3	5	38	4.222	0.278	0.833	0.694	3	2	5	35	3.889	0.389	1.167	1.361
Clear & Realistic Objectives	9	1	4	5	42	4.667	0.167	0.500	0.250	1	4	5	41	4.556	0.176	0.527	0.278
Professional Reputation & Image	9	2	3	5	37	4.111	0.309	0.928	0.861	3	2	5	32	3.556	0.338	1.014	1.028
Responsive Admin. & Decision Support	9	2	3	5	36	4.000	0.236	0.707	0.500	2	3	5	38	4.222	0.278	0.833	0.694
Sustainability	9	2	3	5	36	4.000	0.167	0.500	0.250	3	2	5	33	3.667	0.289	0.866	0.750
Project Time Performance	9	3	2	5	35	3.889	0.261	0.782	0.611	1	4	5	39	4.333	0.167	0.500	0.250
Trust & Respect	9	2	3	5	39	4.333	0.236	0.707	0.500	2	3	5	39	4.333	0.236	0.707	0.500
Utility & Functionality	9	2	3	5	41	4.556	0.242	0.726	0.528	4	1	5	36	4.000	0.441	1.323	1.750
Owner's vision	9	2	3	5	40	4.444	0.242	0.726	0.528	3	1	4	32	3.556	0.338	1.014	1.028

Table 8.3. Round 2 -- Designers: Descriptive statistics



Figure 8.7. Round 2 -- Designers: relative importance by mean score

8.3.4. Builders

]	Roi	ınc	l # 2	2 D	esci	riptiv	e Sta	tistic	s: Bu	ild	er	5					
					Ś	Self-Int	erest						Pı	roject-l	Interest	t	
Critical Success Factor	N	Range	Min	Мах	Sum	Mean	Std. Error	Std. Dev.	Var.	Range	Min	Max	Sum	Mean	Std. Error	Std. Dev.	Var.
Aesthetics	10	3	1	4	24	2.400	0.267	0.843	0.711	3	1	4	28	2.800	0.327	1.033	1.067
Alignment of Project Objectives	10	1	4	5	45	4.500	0.167	0.527	0.278	2	3	5	44	4.400	0.267	0.843	0.711
Change Order Reduction	10	3	2	5	33	3.300	0.260	0.823	0.678	3	2	5	33	3.300	0.260	0.823	0.678
Collaboration of Project Delivery Team	10	2	3	5	41	4.100	0.180	0.568	0.322	2	3	5	43	4.300	0.300	0.949	0.900
Effective Communication	10	1	4	5	44	4.400	0.163	0.516	0.267	2	3	5	44	4.400	0.221	0.699	0.489
Competency & Capability of Project Team	10	2	3	5	43	4.300	0.213	0.675	0.456	2	3	5	42	4.200	0.249	0.789	0.622
Constructability	10	2	3	5	37	3.700	0.260	0.823	0.678	4	1	5	30	3.000	0.422	1.333	1.778
Project Cost Performance	10	2	3	5	42	4.200	0.200	0.632	0.400	3	2	5	38	3.800	0.327	1.033	1.067
Dispute Avoidance & Resolution	10	3	2	5	39	3.900	0.314	0.994	0.989	2	3	5	40	4.000	0.258	0.816	0.667
Design Innovation & Creativity	10	2	2	4	28	2.800	0.200	0.632	0.400	2	2	4	29	2.900	0.233	0.738	0.544
Evidence Based Design	10	3	2	5	33	3.300	0.335	1.059	1.122	3	2	5	34	3.400	0.340	1.075	1.156
Long-term Bldg. Success & Lifecycle Value	10	2	3	5	37	3.700	0.260	0.823	0.678	2	3	5	40	4.000	0.258	0.816	0.667
Owner / User Participation	10	2	3	5	42	4.200	0.291	0.919	0.844	3	2	5	38	3.800	0.327	1.033	1.067
Owner / User Satisfaction	10	1	4	5	48	4.800	0.133	0.422	0.178	1	4	5	47	4.700	0.153	0.483	0.233
Profit & Financial Objectives	10	4	1	5	35	3.500	0.373	1.179	1.389	4	1	5	33	3.300	0.396	1.252	1.567
Production of Specified Quality	10	2	3	5	41	4.100	0.180	0.568	0.322	1	4	5	43	4.300	0.153	0.483	0.233
Clear & Realistic Objectives	10	2	3	5	41	4.100	0.233	0.738	0.544	2	3	5	40	4.000	0.258	0.816	0.667
Professional Reputation & Image	10	3	2	5	41	4.100	0.348	1.101	1.211	3	2	5	36	3.600	0.306	0.966	0.933
Responsive Admin. & Decision Support	10	2	3	5	40	4.000	0.258	0.816	0.667	2	3	5	41	4.100	0.277	0.876	0.767
Sustainability	10	3	2	5	30	3.000	0.333	1.054	1.111	2	2	4	30	3.000	0.258	0.816	0.667
Project Time Performance	10	2	3	5	43	4.300	0.213	0.675	0.456	2	3	5	43	4.300	0.213	0.675	0.456
Trust & Respect	10	2	3	5	44	4.400	0.221	0.699	0.489	1	4	5	43	4.300	0.153	0.483	0.233
Utility & Functionality	10	3	2	5	36	3.600	0.371	1.174	1.378	2	3	5	41	4.100	0.180	0.568	0.322
Owner's vision	10	2	3	5	39	3.900	0.277	0.876	0.767	3	2	5	39	3.900	0.277	0.876	0.767

Table 8.4. Round 2 -- Builders: descriptive statistics



Figure 8.8. Round 2 -- Builders: Relative importance by mean score

8.3.5. Users	8.3	.5.	Users
---------------------	-----	-----	-------

	Re	our	ıd	#2	Des	cripti	ive St	atisti	ics: U	sei	rs						
					5	Self-Int	erest						Pı	roject-I	nterest	t	
Critical Success Factor	N	Range	Min	Мах	Sum	Mean	Std. Error	Std. Dev.	Var.	Range	Min	Max	Sum	Mean	Std. Error	Std. Dev.	Var.
Aesthetics	10	2	3	5	41	4.100	0.233	0.738	0.544	2	3	5	- 39	3.900	0.180	0.568	0.322
Alignment of Project Objectives	10	3	2	5	39	3.900	0.314	0.994	0.989	2	3	5	41	4.100	0.233	0.738	0.544
Change Order Reduction	10	4	1	5	32	3.200	0.327	1.033	1.067	3	2	5	38	3.800	0.327	1.033	1.067
Collaboration of Project Delivery Team	10	2	3	5	38	3.800	0.291	0.919	0.844	3	2	5	42	4.200	0.291	0.919	0.844
Effective Communication	10	2	3	5	43	4.300	0.260	0.823	0.678	2	3	5	45	4.500	0.224	0.707	0.500
Competency & Capability of Project Team	10	2	3	5	42	4.200	0.249	0.789	0.622	1	4	5	44	4.400	0.163	0.516	0.267
Constructability	10	4	1	5	34	3.400	0.340	1.075	1.156	2	3	5	42	4.200	0.200	0.632	0.400
Project Cost Performance	10	3	2	5	36	3.600	0.306	0.966	0.933	3	2	5	37	3.700	0.300	0.949	0.900
Dispute Avoidance & Resolution	10	3	2	5	32	3.200	0.291	0.919	0.844	2	3	5	39	3.900	0.233	0.738	0.544
Design Innovation & Creativity	10	2	3	5	37	3.700	0.213	0.675	0.456	1	3	4	35	3.500	0.167	0.527	0.278
Evidence Based Design	10	2	3	5	41	4.100	0.233	0.738	0.544	1	3	4	37	3.700	0.153	0.483	0.233
Long-term Bldg. Success & Lifecycle Value	10	2	3	5	40	4.000	0.258	0.816	0.667	2	3	5	40	4.000	0.258	0.816	0.667
Owner / User Participation	10	2	3	5	44	4.400	0.221	0.699	0.489	3	2	5	40	4.000	0.258	0.816	0.667
Owner / User Satisfaction	10	1	4	5	46	4.600	0.163	0.516	0.267	1	4	5	47	4.700	0.153	0.483	0.233
Profit & Financial Objectives	10	4	1	5	33	3.300	0.367	1.160	1.344	4	1	5	33	3.300	0.448	1.418	2.011
Production of Specified Quality	10	2	3	5	41	4.100	0.180	0.568	0.322	3	2	5	40	4.000	0.258	0.816	0.667
Clear & Realistic Objectives	10	2	3	5	42	4.200	0.249	0.789	0.622	1	4	5	46	4.600	0.163	0.516	0.267
Professional Reputation & Image	10	2	2	4	32	3.200	0.291	0.919	0.844	4	1	5	34	3.400	0.371	1.174	1.378
Responsive Admin. & Decision Support	10	2	3	5	39	3.900	0.233	0.738	0.544	3	2	5	38	3.800	0.249	0.789	0.622
Sustainability	10	2	3	5	41	4.100	0.180	0.568	0.322	2	3	5	40	4.000	0.211	0.667	0.444
Project Time Performance	10	2	2	4	35	3.500	0.224	0.707	0.500	3	2	5	40	4.000	0.333	1.054	1.111
Trust & Respect	10	2	3	5	41	4.100	0.233	0.738	0.544	2	3	5	41	4.100	0.277	0.876	0.767
Utility & Functionality	10	1	4	5	48	4.800	0.133	0.422	0.178	1	4	5	44	4.400	0.163	0.516	0.267
Owner's vision	10	2	3	5	39	3.900	0.233	0.738	0.544	2	3	5	37	3.700	0.213	0.675	0.456



Figure 8.9. Round 2 -- Users: relative importance by mean score

8.4. Round 2 -- Inter-Rater Agreement

Round	1 #2 I	nter-l	Rater	Agre	ement					
		Se	elf-Inte	rest			Pro	ject-Int	erest	
Critical Success Factors		Ν	lean Ra	ınk			N	/lean Ra	ınk	
	0	D	В	U	All	0	D	В	U	All
Owner / User Satisfaction	17.08	19.33	19.60	17.95	18.37	14.73	19.28	18.35	18.75	17.52
Clear & Realistic Objectives	17.38	17.00	14.10	14.50	15.83	16.92	16.67	13.80	17.75	16.32
Effective Communication	16.92	13.50	16.70	15.30	15.75	16.38	15.56	16.55	16.35	16.24
Competency & Capability of Project Team	15.23	15.61	15.80	15.45	15.50	16.46	17.39	15.45	16.30	16.38
Utility & Functionality	17.38	16.06	10.40	19.50	15.94	14.65	13.22	14.70	16.00	14.68
Collaboration of Project Delivery Team	14.81	15.44	14.20	11.80	14.08	16.15	18.00	15.95	14.95	16.21
Trust & Respect	11.46	14.50	16.35	14.05	13.89	14.96	14.83	15.80	13.65	14.82
Alignment of Project Objectives	13.23	10.50	17.40	13.30	13.65	14.92	14.00	16.95	12.95	14.74
Owner / User Participation	14.50	15.33	14.90	16.55	15.26	12.58	13.56	12.05	12.60	12.67
Production of Specified Quality	14.08	13.22	14.50	13.95	13.96	14.96	11.94	15.35	12.50	13.82
Long-term Bldg. Success & Lifecycle Value	15.50	13.22	10.30	13.80	13.37	14.08	9.39	13.15	12.30	12.43
Owner's vision	16.81	15.44	12.30	12.35	14.38	11.04	9.00	13.15	8.80	10.57
Project Time Performance	10.35	10.50	15.60	8.50	11.19	11.00	14.44	15.95	13.15	13.43
Constructability	11.23	13.50	11.05	10.00	11.38	13.81	15.11	6.75	14.45	12.56
Project Cost Performance	12.15	9.06	15.05	9.95	11.65	13.00	11.17	12.00	9.80	11.61
Responsive Admin. & Decision Support	8.77	11.33	13.55	11.65	11.14	8.58	13.39	14.50	10.90	11.57
Sustainability	12.46	11.06	6.30	13.95	11.05	9.12	9.33	6.35	12.45	9.30
Evidence Based Design	12.00	8.72	8.25	13.85	10.85	9.04	7.94	9.40	9.45	8.99
Dispute Avoidance & Resolution	7.12	6.61	12.90	6.70	8.29	11.42	8.72	13.75	11.20	11.35
Profit & Financial Objectives	7.69	11.72	10.65	8.70	9.50	11.81	9.83	9.00	8.00	9.81
Professional Reputation & Image	6.73	12.28	14.45	7.70	9.99	9.00	8.72	11.05	8.35	9.27
Aesthetics	10.73	9.06	3.05	13.45	9.19	8.96	6.00	6.20	11.20	8.20
Design Innovation & Creativity	8.58	12.00	4.60	9.70	8.63	9.19	9.67	5.70	7.40	8.04
Change Order Reduction	7.81	5.00	8.00	7.35	7.14	7.23	12.83	8.10	10.75	9.48
Statistic		Se	elf-Inte	rest			Pro	ject-Int	erest	
Statistic	0	D	В	U	All	0	D	В	U	All
N	13	9	10	10	42	13	9	10	10	42
Kendall's W ^a	.300	.277	.410	.290	.207	.218	.297	.350	.252	.208
Chi-square	89.72	57.39	94.38	66.77	199.95	65.31	61.39	80.48	57.98	201.17
df	23	23	23	23	23	23	23	23	23	23
Asymp. Sig.	.000	.000	.000	.000	.000	.000	.000	.000	.000	.000
a. Kendall's Coefficient of Concordance										

 Table 8.6. Round 2 inter-rater agreement table

A 0.001 significance level is achieve both overall and by all stakeholder groups

individually in both the self-interest, and project-interest categories. Overall the round 2 survey

demonstrates measures of inter-rater agreement overall and for every stakeholder group. Overall self-interest and project-interest scores for Kendall's coefficient of concordance are highly consistent at 0.207 and 0.208 respectively. Builders indicate the greatest level of agreement within both self and project perspectives. Inter-rater agreement is validated for the round 2 survey.



Figure 8.10. Round 2 inter-rate agreement by stakeholder group

Later on in the round 2 report the inter-rater agreement will be compared overall, and by each stakeholder group to the results from round 1 to determine if an increase in consensus has been successfully achieved by the panel(s) of experts.

8.5. Reliability Test

		I	Round #2: Ir	ntraclass Co	orrelation				
				Rel	liability Stat	tistics			
Derspective	Stakeholder		Cranhaah'a	95% Confid	ence Interval	F Te	est Tru	e Valu	e 0
reispective	Group	N	Alpha	Lower Bound	Upper Bound	Value	df1	df2	Sig
	All	42	0.918	0.881	0.948	12.168	48	1968	0.000
Overall	Owner	13	0.857	0.790	0.909	6.983	48	576	0.000
Average	Designer	9	0.776	0.668	0.859	4.466	48	384	0.000
Average	Builder	10	0.820	0.734	0.886	5.548	48	432	0.000
	User	10	0.667	0.508	0.790	2.999	48	432	0.000
	All	42	0.900	0.833	0.949	10.011	23	943	0.000
	Owner	13	0.786	0.634	0.893	4.669	23	276	0.000
Self-Interest	Designer	9	0.671	0.430	0.837	3.041	23	184	0.000
	Builder	10	0.837	0.719	0.919	6.141	23	207	0.000
	User	10	0.720	0.517	0.861	3.572	23	207	0.000
	All	42	0.895	0.824	0.947	9.498	23	943	0.000
	Owner	13	0.648	0.399	0.824	2.839	23	276	0.000
Project-Interest	Designer	9	0.665	0.419	0.834	2.981	23	184	0.000
	Builder	10	0.798	0.651	0.899	4.939	23	207	0.000
	User	10	0.616	0.337	0.809	2.602	23	207	0.000

Table 8.7. Round 2 Reliability test

Round #2	Intraclass Corr	elati	on Summ	ary
	Stakeholder		Cronbach	's Alpha
CSF Category	Group	Ν	Self-	Project-
	Group		Interest	Interest
	All	42	0.900	0.895
	Owner	13	0.786	0.648
ALL	Designer	9	0.671	0.665
	Builder	10	0.837	0.798
	User	10	0.720	0.616

Table 8.8. Round 2 Reliability summary by stakeholder

The round 2 survey tests as very reliable at a 0.001 significance level. Each stakeholder group scored as reliable with a range from minimally reliable to strong reliability for the designer and builder stakeholder groups respectively. Based on the Cronbach's alpha scores the round 2 survey is validated for reliability.

8.6. Round 2 Correlations

	Rour	nd #2: Co	rrelation	among S	takehold	er Groups	5	
Correlation		Self-Ir	nterest			Project	Interest	
Matrix *	Owner	Designer	Builder	User	Owner	Designer	Builder	User
Owner	1.000	0.722 ^b	0.386	0.798 ^b	1.000	0.679 ^b	0.649 ^b	0.716 ^b
Designer		1.000	0.431 ^a	0.642 ^b		1.000	0.648 ^b	0.730 ^b
Builder			1.000	0.168			1.000	0.599 ^b
User				1.000				1.000
User				1.000				1.000

Notes:

^a Correlation is significant at the 0.05 level (2-tailed).

^b Correlation is significant at the 0.01 level (2-tailed).

* Correlation Coefficient based on mean scores, and calculated in SPSS using Pearsons Correlation. (N = 24 CSFs)

 Table 8.9. Round 2 Stakeholder correlation matrix

A significant degree of correlation is indicated between the owners, designer, and users in

the self-interest category. In the project-interest category there is consistent correlation among

all stakeholders.

						5	Correlatic	n Matri x	among	Critical 3	Success Fa	ictors (for	Round	#2)									
												Self-Into	rest										
Correlation Matrix *	soiteftesA	Abjectives Objectives	Cnange Order Reduction Collaboration of	Project Delivery Team	Entective Communication	Competency & Capability of Project Delivery Team	Constructability	Performance Performance	Resolution Besign Innovation &	Creativity Evidence Based	Design Long-term Build. Success / Lifecycle	Value Owner / User Participa- tion	Owner / User Satisfaction	Profit and Financial Objectives	Production of Specified Quality	Clear & Realistic Objectives	Profess-ional Bration & Image	Responsive Administration & Decision Support	Sustainability	Ргојест Тіте Регfогталсе	Trust & Respect	Utility & Functionality	Owners vision
Aesthetics	1.000 -	0.982 -(0.536 -6	.127 -(0.421	-0.174	0.015 -4	0-93 ^b -0	.916 0.	835 0.9	11 0.741	0.728	-0.373	-0.056	0.212	0.511	-0.606	-0.274	0.995 ^b -	-0.941	0.512 0	0 q 666	376
Alignment of Project Objectives	1	1.000 0).627 -6	0.061 0).496	0.023 -	0.203 0	.953 ^a 0.	871 -0.	910 -0.8	24 -0.810	0 -0.772	0.254	-0.098	-0.368	-0.651	0.477	0.241	-0.995 ^b	0.865 (.447 -0)- ^a 979.	.520
Change Order Reduction	1	-	000 -0	.375 (. 979 ^a	-0.714 -	0.556 (.471 0.	167 -0.	883 -0.4	03 -0.29:	5 -0.955	a -0.583	-0.783	-0.885	-0.473	-0.331	-0.555	-0.599	0.252	0.411 -0	.558 -(.316
Collaboration of Project Delivery Team	1	;)- 000.	0.278	0.722	0.979 ^a (241 0.	219 0.	326 -0.5	05 0.395	0.133	0.548	0.733	0.746	0.755	0.633	0.065	-0.035	0.412 (.256 -0	.142 0	.795
Effective Communication	1	1	;	-	. 000.1	-0.742 -	0.462 (.369 0.	025 -0.	780 -0.3	59 -0.09	6 -0.928	-0.668	-0.793	-0.847	-0.291	-0.408	-0.705	-0.479	0.157 -0	0.554 -0	.450 -(.131
Competency & Capability of Project	I	:	:	1	1	1.000	0.797 0	.264 0.	499 0.	392 -0.3	49 -0.09	0 0.478	0.956	0.993	0.918	0.330	0.889	0.730	-0.085	0.487 (.800 -0	.157 0	279
Constructability	1	;	;	;	1	;	000.1	0.102 0.	147 0.	492 -0.3	56 0.434	1 0.331	0.610	0.824	0.861	0.789	0.629	0.169	0.110	0.303 (0 308 0	005 0	.790
Project Cost Performance	1	1	;	1	1	1	-	.000 0.	927 -0.	775 -0.9	49 -0.68	4 -0.688	0.439	0.149	-0.113	-0.415	0.674	0.288	-0.977 ^a 0).971 ^a (.542 -0)- ^b 1948.	.280
Dispute Avoidance & Resolution	:	:	:	:	;	;	:	-	000 -0.	590 -0.8	27 -0.803	2 -0.393	0.695	0.390	0.115	-0.474	0.832	0.626	-0.889 0).957 ^a (.813 -0)- 106.	.400
Design Innovation & Creativity	:	:	:	:	;	;	;	:	-	0.0 0.6	32 0.683	0.924	0.170	0.499	0.714	0.707	-0.072	0.107	0.883 -	-0.601 -0	0.063 0	841 0	560
Evidence Based Design	:	:	:	:	1	:	:	:	-	- 1.0	00 0.423	0.655	-0.44(-0.253	-0.038	0.111	-0.701	-0.141	0.875 -	-0.952 4 -1	0.437 0	922 -(037
Long-term Build. Success / Lifecycle Value	;	;	;	;	;	;	;	;	-	-	. 1.000	0.358	-0.378	0.014	0.277	0.896	-0.426	-0.623	0.767 -	0.626 -1	0.662 0	714 0	868
Owner / User Participation	;	;	;	;	1	1	1	;	1	•	1	1.000	0.347	0.565	0.713	0.405	0.052	0.430	0.767 -	0.509 (0.213 0	750 0	227
Owner / User Satisfaction	1	;	;	;	1	;	1	;	-	-	1	1	1.000	0.919	0.774	0.042	0.949	0.866	-0.298	0.630 (.940 -0	349 0	100.
Profit and Financial Objectives	1	1	;	1	1	1	1	1	1	-		1	1	1.000	0.959 ^b	0.414	0.828	069.0	0.035	0.383 (0- 138 -0	.040 0	349
Production of Specified Quality	1	;	;	1	;	:	:	:	1	•	•	1	1	1	1.000	0.612	0.641	0.533	0.303	0.129 (0.535 0	224 0	.523
Clear & Realistic Objectives	1	1	;	1	1	1	1	1	1	-		1	1	1	ł	1.000	0.019	-0.344	0.576 -	-0.277 -1	0.301 0	488 0	.982 ^a
Professional Reputation & Image	1	:	:	1	:	:	:	:	1	-	•	:	1		:	:	1.000	0.726	-0.532	0.830 0	.891 -0	591 0	.039
Responsive Administration & Decision	;	;	;	;	1	1	1	;	1	•	1	1	1	1	1		;	1.000	-0.237	0.428 (.951 8 -0	.235 -(.433
Sustainability	1	;	:	-	:	-	-	:	-	-	-	:	1	1	:		:	-	1.000 -	-0.903	0.463 0	994 ^b 0	.440
Project Time Performance	;	;	;	;	1	;	1	;	1	•	1	1	1	1	1		;	1	:	1.000 (0- 679)- 856.	.165
Trust & Respect	1	1	;	1	1	1	1	1	1	-		1	1	1	ł		;	1	1	-	000 -0	.480 -(.339
Utility & Functionality	1	;	;	;	1	;	1	;	-	-	1	1	1	1	1		:		:	;	-	0 000	347
Owners vision	1	1	;	1	1	1	1	1	1	-		1	1	1	ł	1	;	1	1	1	:	1	000
Notes:																							
^a Correlation is significant at the 0.05 leve	(2-tailed	1).																					
b Correlation is significant at the 0.01 leve	(2-tailed	.(F																					
* Correlation Coefficient based on mean set	ore by st	akeholde	r groups	, and calc	culated in	SPSS using	Pearsons	Correlatio	n. (N = 4	Stakehold	er Groups:	Owner; De	signer; B	iilder; Use	r)								

 Table 8.10. Round 2 -- Self-Interest: CSF Correlation matrix

 Table 8.11. Round 2 -- Project-Interest: CSF Correlation matrix

			SE	LF-IN	TERE	ST									
Critical Success Easter			Mean					Rank				0	% Ran	k	
Critical Success Factor	0	D	В	U	ALL	0	D	В	U	ALL	0	D	В	U	ALL
Aesthetics	3.692	3.778	2.400	4.100	3.500	17.0	20.5	24.0	9.0	20.0	0.333	0.188	0.042	0.667	0.208
Alignment of Project Objectives	4.000	3.889	4.500	3.900	4.071	11.0	18.5	2.0	14.0	11.5	0.583	0.271	0.958	0.458	0.563
Change Order Reduction	3.308	3.111	3.300	3.200	3.238	21.0	24.0	20.5	23.0	24.0	0.167	0.042	0.188	0.083	0.042
Collaboration of Project Delivery Team	4.154	4.444	4.100	3.800	4.119	8.5	5.5	10.5	16.0	9.0	0.688	0.813	0.604	0.375	0.667
Effective Communication	4.462	4.222	4.400	4.300	4.357	3.0	11.0	3.5	4.0	3.5	0.917	0.583	0.896	0.875	0.896
Competency & Capability of Project Team	4.154	4.444	4.300	4.200	4.262	8.5	5.5	5.5	5.5	6.0	0.688	0.813	0.813	0.813	0.792
Constructability	3.769	4.333	3.700	3.400	3.786	14.0	8.5	16.5	20.0	15.5	0.458	0.688	0.354	0.208	0.396
Project Cost Performance	3.769	3.778	4.200	3.600	3.833	14.0	20.5	7.5	18.0	13.5	0.458	0.188	0.729	0.292	0.479
Dispute Avoidance & Resolution	3.154	3.444	3.900	3.200	3.405	23.0	23.0	14.5	23.0	23.0	0.083	0.083	0.438	0.083	0.083
Design Innovation & Creativity	3.385	4.111	2.800	3.700	3.476	19.5	13.5	23.0	17.0	21.0	0.229	0.479	0.083	0.333	0.167
Evidence Based Design	3.692	3.667	3.300	4.100	3.690	17.0	22.0	20.5	9.0	18.0	0.333	0.125	0.188	0.667	0.292
Long-term Bldg. Success & Lifecycle Valu	4.308	4.222	3.700	4.000	4.071	6.0	11.0	16.5	12.0	11.5	0.792	0.583	0.354	0.542	0.563
Owner / User Participation	4.231	4.444	4.200	4.400	4.310	7.0	5.5	7.5	3.0	5.0	0.750	0.813	0.729	0.917	0.833
Owner / User Satisfaction	4.462	4.889	4.800	4.600	4.667	3.0	1.0	1.0	2.0	1.0	0.917	1.000	1.000	0.958	1.000
Profit & Financial Objectives	3.154	4.000	3.500	3.300	3.452	23.0	16.0	19.0	21.0	22.0	0.083	0.575	0.250	0.16/	0.125
Clear & Production of Specified Quality	4.077	4.222	4.100	4.100	4.119	10.0	11.0	10.5	9.0	9.0	1.000	0.585	0.604	0.00/	0.067
Clear & Realistic Objectives	4.538	4.00/	4.100	4.200	4.381	22.0	2.0	10.5	22.0	2.0	1.000	0.958	0.604	0.813	0.938
Professional Reputation & Image	2 2 9 5	4.111	4.100	3.200	2.393	10.5	15.5	12.0	14.0	19.0	0.085	0.479	0.004	0.065	0.230
Sustainability	3.846	4.000	3 000	1 100	3 738	12.0	16.0	13.0 22.0	9.0	17.0	0.229	0.375	0.300	0.458	0.390
Project Time Performance	3.692	3 880	4 300	3 500	3 8 3 3	12.0	18.5	5.5	10.0	13.5	0.342	0.373	0.123	0.007	0.555
Trust & Respect	3 769	4 333	4.500	4 100	4 1 1 9	17.0	85	3.5	9.0	9.0	0.333	0.271	0.815	0.250	0.479
Utility & Functionality	4 4 6 2	4 556	3 600	4 800	4 3 5 7	3.0	3.0	18.0	1.0	3.5	0.430	0.000	0.090	1 000	0.896
Owner's vision	4 385	4 4 4 4	3 900	3 900	4 167	5.0	5.5	14.5	14.0	7.0	0.833	0.813	0.438	0.458	0 750
Stakeholder	Groups	Types	O = O	wner: [= Des	igner;	B = B	uilder	; U =	User;	ALL				
						0 /			/						
		21	PROJ	ECT-	INTE	RES	Г		/						
Critical Success Factor			PROJ Mean	ECT-	INTE	RES	Г	Rank		1		0	% Ran	k	1
Critical Success Factor	0	D	PROJ Mean B	ECT-	ALL	RES [®]	Г D	Rank B	U	ALL	0	D	% Ran В	k U	ALL
Critical Success Factor Aesthetics	O 3.385	D 3.222	PROJ Mean B 2.800	ECT- U 3.900	INTE ALL 3.333	RES 0 21.0	Г D 24.0	Rank B 24.0	U 15.5	ALL 24.0	O 0.167	D 0.042	% Ran B 0.042	k U 0.396	ALL 0.042
Critical Success Factor Aesthetics Alignment of Project Objectives	O 3.385 4.231	D 3.222 4.333	PROJ Mean B 2.800 4.400	U 3.900 4.100	ALL 3.333 4.262	RES 0 21.0 5.5	Г D 24.0 7.5	Rank B 24.0 2.5	U 15.5 8.5	ALL 24.0 6.0	O 0.167 0.813	D 0.042 0.729	% Ran B 0.042 0.938	k U 0.396 0.688	ALL 0.042 0.792
Critical Success Factor Aesthetics Alignment of Project Objectives Change Order Reduction	O 3.385 4.231 3.308	D 3.222 4.333 4.222	PROJ Mean B 2.800 4.400 3.300	U 3.900 4.100 3.800	ALL 3.333 4.262 3.619	RES 0 21.0 5.5 23.5	Г D 24.0 7.5 10.5	Rank B 24.0 2.5 19.5	U 15.5 8.5 17.5	ALL 24.0 6.0 18.0	O 0.167 0.813 0.063	D 0.042 0.729 0.604	% Ran B 0.042 0.938 0.229	k U 0.396 0.688 0.313	ALL 0.042 0.792 0.292
Critical Success Factor Aesthetics Alignment of Project Objectives Change Order Reduction Collaboration of Project Delivery Team	O 3.385 4.231 3.308 4.308	D 3.222 4.333 4.222 4.778	PROJ Mean B 2.800 4.400 3.300 4.300	U 3.900 4.100 3.800 4.200	ALL 3.333 4.262 3.619 4.381	RES 0 21.0 5.5 23.5 3.5	Г D 24.0 7.5 10.5 2.0	Rank B 24.0 2.5 19.5 5.5	U 15.5 8.5 17.5 6.5	ALL 24.0 6.0 18.0 5.0	O 0.167 0.813 0.063 0.896	D 0.042 0.729 0.604 0.958	% Ran B 0.042 0.938 0.229 0.813	k U 0.396 0.688 0.313 0.771	ALL 0.042 0.792 0.292 0.833
Critical Success Factor Aesthetics Alignment of Project Objectives Change Order Reduction Collaboration of Project Delivery Team Effective Communication	0 3.385 4.231 3.308 4.308 4.308	D 3.222 4.333 4.222 4.778 4.444	PROJ Mean B 2.800 4.400 3.300 4.300 4.400	U 3.900 4.100 3.800 4.200 4.500	ALL 3.333 4.262 3.619 4.381 4.405	RES 0 21.0 5.5 23.5 3.5 3.5	D 24.0 7.5 10.5 2.0 5.0	Rank B 24.0 2.5 19.5 5.5 2.5	U 15.5 8.5 17.5 6.5 3.0	ALL 24.0 6.0 18.0 5.0 3.0	0 0.167 0.813 0.063 0.896	D 0.042 0.729 0.604 0.958 0.833	% Ran B 0.042 0.938 0.229 0.813 0.938	k 0.396 0.688 0.313 0.771 0.917	ALL 0.042 0.792 0.292 0.833 0.917
Critical Success Factor Aesthetics Alignment of Project Objectives Change Order Reduction Collaboration of Project Delivery Team Effective Communication Competency & Capability of Project Team	O 3.385 4.231 3.308 4.308 4.308 4.308 4.385	D 3.222 4.333 4.222 4.778 4.444 4.667	PROJ Mean B 2.800 4.400 3.300 4.300 4.400 4.200	U 3.900 4.100 3.800 4.200 4.500 4.400	ALL 3.333 4.262 3.619 4.381 4.405 4.405	RES 0 21.0 5.5 23.5 3.5 3.5 2.0	D 24.0 7.5 10.5 2.0 5.0 3.0	Rank B 24.0 2.5 19.5 5.5 2.5 8.0	U 15.5 8.5 17.5 6.5 3.0 4.5	ALL 24.0 6.0 18.0 5.0 3.0 3.0	O 0.167 0.813 0.063 0.896 0.896 0.958	D 0.042 0.729 0.604 0.958 0.833 0.917	% Ran B 0.042 0.938 0.229 0.813 0.938 0.708	k 0.396 0.688 0.313 0.771 0.917 0.854	ALL 0.042 0.792 0.292 0.833 0.917 0.917
Critical Success Factor Aesthetics Alignment of Project Objectives Change Order Reduction Collaboration of Project Delivery Team Effective Communication Competency & Capability of Project Team Constructability	O 3.385 4.231 3.308 4.308 4.308 4.308 4.385 4.077	D 3.222 4.333 4.222 4.778 4.444 4.667 4.333	PROJ Mean B 2.800 4.400 3.300 4.300 4.400 3.300 4.200 3.000	U 3.900 4.100 3.800 4.200 4.500 4.400 4.200	ALL 3.333 4.262 3.619 4.381 4.405 3.905	O 21.0 5.5 23.5 3.5 3.5 2.0 10.5	D 24.0 7.5 10.5 2.0 5.0 3.0 7.5	Rank B 24.0 2.5 19.5 5.5 2.5 8.0 21.5	U 15.5 8.5 17.5 6.5 3.0 4.5 6.5	ALL 24.0 6.0 18.0 5.0 3.0 13.0	O 0.167 0.813 0.063 0.896 0.896 0.958 0.604	D 0.042 0.729 0.604 0.958 0.833 0.917 0.729	% Ran B 0.042 0.938 0.229 0.813 0.938 0.708 0.146	k 0.396 0.688 0.313 0.771 0.917 0.854 0.771	ALL 0.042 0.292 0.833 0.917 0.917 0.500
Critical Success Factor Aesthetics Alignment of Project Objectives Change Order Reduction Collaboration of Project Delivery Team Effective Communication Competency & Capability of Project Team Constructability Project Cost Performance	0 3.385 4.231 3.308 4.308 4.308 4.308 4.385 4.077 4.000	D 3.222 4.333 4.222 4.778 4.444 4.667 4.333 4.000	PROJ Mean B 2.800 4.400 3.300 4.300 4.400 3.000 3.000 3.800	U 3.900 4.100 3.800 4.200 4.500 4.400 4.200 3.700	ALL 3.333 4.262 3.619 4.381 4.405 4.405 3.905 3.881	RES 0 21.0 5.5 23.5 3.5 3.5 2.0 10.5 12.0	D 24.0 7.5 10.5 2.0 5.0 3.0 7.5 13.5	Rank B 24.0 2.5 19.5 5.5 2.5 8.0 21.5 15.5	U 15.5 8.5 17.5 6.5 3.0 4.5 6.5 20.0	ALL 24.0 6.0 18.0 5.0 3.0 3.0 13.0 14.0	O 0.167 0.813 0.063 0.896 0.896 0.958 0.604 0.542	D 0.042 0.729 0.604 0.958 0.833 0.917 0.729 0.479	% Ran B 0.042 0.938 0.229 0.813 0.938 0.708 0.146 0.396	k U 0.396 0.688 0.313 0.771 0.917 0.854 0.771 0.208	ALL 0.042 0.292 0.833 0.917 0.917 0.500 0.458
Critical Success Factor Aesthetics Alignment of Project Objectives Change Order Reduction Collaboration of Project Delivery Team Effective Communication Competency & Capability of Project Team Constructability Project Cost Performance Dispute Avoidance & Resolution	0 3.385 4.231 3.308 4.308 4.308 4.308 4.308 4.308 4.307 4.000 3.769	D 3.222 4.333 4.222 4.778 4.444 4.667 4.333 4.000 3.778	PROJ Mean B 2.800 4.400 3.300 4.300 4.400 3.000 3.800 4.000	U 3.900 4.100 3.800 4.200 4.500 4.400 4.200 3.700 3.900	ALL 3.333 4.262 3.619 4.381 4.405 3.905 3.881 3.857	RES 0 21.0 5.5 23.5 3.5 3.5 2.0 10.5 12.0 15.5	Г 24.0 7.5 10.5 2.0 5.0 3.0 7.5 13.5 16.0	Rank B 24.0 2.5 19.5 5.5 2.5 8.0 21.5 15.5 12.0	U 15.5 8.5 17.5 6.5 3.0 4.5 6.5 20.0 15.5	ALL 24.0 6.0 18.0 5.0 3.0 13.0 13.0 14.0 15.0	O 0.167 0.813 0.063 0.896 0.896 0.958 0.604 0.542 0.396	D 0.042 0.729 0.604 0.958 0.833 0.917 0.729 0.479 0.375	% Ran B 0.042 0.938 0.229 0.813 0.938 0.708 0.146 0.396 0.542	k U 0.396 0.688 0.313 0.771 0.917 0.854 0.771 0.208 0.396	ALL 0.042 0.792 0.292 0.833 0.917 0.917 0.917 0.500 0.458 0.417
Critical Success Factor Aesthetics Alignment of Project Objectives Change Order Reduction Collaboration of Project Delivery Team Effective Communication Competency & Capability of Project Team Constructability Project Cost Performance Dispute Avoidance & Resolution Design Innovation & Creativity	O 3.385 4.231 3.308 4.308 4.308 4.308 4.308 4.308 4.308 4.308 4.077 4.000 3.769 3.385	D 3.222 4.333 4.222 4.778 4.444 4.667 4.333 4.000 3.778 3.667	PROJ Mean B 2.800 4.400 3.300 4.300 4.400 3.000 3.000 3.000 3.800 4.000 2.900	U 3.900 4.100 3.800 4.200 4.500 4.400 4.200 3.700 3.900 3.500	ALL 3.333 4.262 3.619 4.381 4.405 3.905 3.881 3.857 3.357	O 21.0 5.5 23.5 3.5 3.5 2.0 10.5 12.0 15.5 21.0	Г 24.0 7.5 10.5 2.0 5.0 3.0 7.5 13.5 16.0 18.5	Rank B 24.0 2.5 19.5 5.5 2.5 8.0 21.5 15.5 12.0 23.0	U 15.5 8.5 17.5 6.5 3.0 4.5 6.5 20.0 15.5 22.0	ALL 24.0 6.0 18.0 5.0 3.0 13.0 14.0 15.0 23.0	O 0.167 0.813 0.063 0.896 0.896 0.958 0.604 0.542 0.396 0.167	D 0.042 0.729 0.604 0.958 0.833 0.917 0.729 0.479 0.375 0.271	% Ran B 0.042 0.938 0.229 0.813 0.938 0.708 0.146 0.396 0.542 0.083	k 0.396 0.688 0.313 0.771 0.917 0.917 0.854 0.771 0.208 0.396 0.396	ALL 0.042 0.792 0.292 0.833 0.917 0.917 0.500 0.458 0.417 0.083
Critical Success Factor Aesthetics Alignment of Project Objectives Change Order Reduction Collaboration of Project Delivery Team Effective Communication Competency & Capability of Project Team Constructability Project Cost Performance Dispute Avoidance & Resolution Design Innovation & Creativity Evidence Based Design	O 3.385 4.231 3.308 4.308 4.308 4.308 4.308 4.308 4.308 4.308 4.308 3.308 3.369 3.385 3.308	D 3.222 4.333 4.222 4.778 4.444 4.667 4.333 4.000 3.778 3.667 3.556	PROJ Mean B 2.800 4.400 3.300 4.300 4.400 3.000 3.800 4.000 2.900 3.400	U 3.900 4.100 3.800 4.200 4.500 4.200 3.700 3.700 3.500 3.500 3.700	ALL 3.333 4.262 3.619 4.381 4.405 4.405 3.905 3.881 3.857 3.357 3.476	O 21.0 5.5 23.5 3.5 3.5 2.0 10.5 12.0 15.5 21.0 15.5 21.0	D 24.0 7.5 10.5 2.0 5.0 3.0 7.5 13.5 16.0 18.5 22.0 18.5	Rank B 24.0 2.5 19.5 5.5 2.5 8.0 21.5 15.5 12.0 23.0 18.0	U 15.5 8.5 17.5 6.5 3.0 4.5 6.5 20.0 15.5 22.0 20.0	ALL 24.0 6.0 18.0 5.0 3.0 13.0 14.0 15.0 23.0 22.0	O 0.167 0.813 0.063 0.896 0.896 0.958 0.604 0.542 0.396 0.167 0.063	D 0.042 0.729 0.604 0.958 0.833 0.917 0.729 0.479 0.375 0.271 0.125	% Ran B 0.042 0.938 0.229 0.813 0.938 0.708 0.146 0.396 0.542 0.083 0.292	k 0.396 0.688 0.313 0.771 0.917 0.854 0.771 0.208 0.396 0.125 0.208	ALL 0.042 0.792 0.292 0.833 0.917 0.917 0.917 0.500 0.458 0.417 0.083 0.417
Critical Success Factor Aesthetics Alignment of Project Objectives Change Order Reduction Collaboration of Project Delivery Team Effective Communication Competency & Capability of Project Team Constructability Project Cost Performance Dispute Avoidance & Resolution Design Innovation & Creativity Evidence Based Design Long-term Bldg. Success & Lifecycle Valu	O 3.385 4.231 3.308 4.308 4.308 4.308 4.308 4.385 4.077 4.000 3.769 3.385 3.308 4.077	D 3.222 4.333 4.222 4.778 4.444 4.667 4.333 4.000 3.778 3.667 3.556 3.667	PROJ Mean B 2.800 4.400 3.300 4.300 4.400 3.000 3.800 4.000 2.900 3.400 4.000	U 3.900 4.100 3.800 4.200 4.500 4.400 4.200 3.700 3.900 3.500 3.700 4.000	ALL 3.333 4.262 3.619 4.381 4.405 4.405 3.905 3.881 3.857 3.357 3.476 3.952	O 21.0 5.5 23.5 3.5 3.5 10.5 12.0 15.5 21.0 23.5 12.0 15.5 21.0 23.5 10.5 12.0 15.5 10.5 10.5 10.5 10.5	Г 24.0 7.5 10.5 2.0 5.0 3.0 7.5 13.5 16.0 18.5 22.0 18.5 22.0	Rank B 24.0 2.5 19.5 5.5 2.5 8.0 21.5 15.5 12.0 23.0 18.0 12.0	U 15.5 8.5 17.5 6.5 20.0 15.5 22.0 20.0 12.0	ALL 24.0 6.0 18.0 5.0 3.0 13.0 14.0 15.0 23.0 22.0 21.15	O 0.167 0.813 0.063 0.896 0.958 0.604 0.542 0.396 0.167 0.063 0.604	D 0.042 0.729 0.604 0.958 0.833 0.917 0.729 0.479 0.479 0.375 0.271 0.125 0.271	% Ran B 0.042 0.938 0.229 0.813 0.938 0.708 0.708 0.708 0.396 0.542 0.083 0.292 0.542	k U 0.396 0.688 0.313 0.771 0.917 0.854 0.771 0.208 0.396 0.125 0.208 0.542	ALL 0.042 0.792 0.833 0.917 0.917 0.917 0.500 0.458 0.417 0.083 0.125 0.563
Critical Success Factor Aesthetics Alignment of Project Objectives Change Order Reduction Collaboration of Project Delivery Team Effective Communication Competency & Capability of Project Team Constructability Project Cost Performance Dispute Avoidance & Resolution Design Innovation & Creativity Evidence Based Design Long-term Bldg. Success & Lifecycle Valu Owner / User Participation	O 3.385 4.231 3.308 4.308 4.308 4.308 4.308 4.308 3.385 3.769 3.385 3.308 4.077 3.908 4.077 3.923	D 3.222 4.333 4.222 4.778 4.444 4.667 4.333 4.000 3.778 3.667 3.556 3.667 4.111	PROJ Mean B 2.800 4.400 3.300 4.300 4.400 3.000 3.000 3.000 3.000 3.800 4.000 3.400 4.700	U 3.900 4.100 3.800 4.200 4.500 4.400 4.200 3.700 3.700 3.500 3.700 4.000 4.000	ALL 3.333 4.262 3.619 4.381 4.405 4.405 3.905 3.881 3.857 3.357 3.476 3.952 3.952	O 21.0 5.5 23.5 3.5 3.5 10.5 12.0 15.5 21.0 23.5 10.5 10.5 11.0 13.0 8.0	D 24.0 7.5 10.5 2.0 5.0 3.0 7.5 13.5 16.0 18.5 22.0 18.5 12.0	Rank B 24.0 2.5 19.5 5.5 2.5 8.0 21.5 12.0 23.0 18.0 12.0 15.5	U 15.5 8.5 17.5 6.5 3.0 4.5 6.5 22.0 20.0 15.5 22.0 20.0 12.0 12.0	ALL 24.0 6.0 18.0 5.0 3.0 3.0 13.0 15.0 23.0 22.0 11.5 11.5 11.5	O 0.167 0.813 0.063 0.896 0.958 0.604 0.542 0.396 0.167 0.063 0.604 0.500	D 0.042 0.729 0.604 0.958 0.833 0.917 0.729 0.479 0.375 0.271 0.125 0.271 0.125 0.271	% Ran B 0.042 0.938 0.229 0.813 0.938 0.708 0.708 0.396 0.542 0.083 0.292 0.542 0.396	k U 0.396 0.688 0.313 0.771 0.917 0.854 0.771 0.208 0.396 0.125 0.208 0.542 0.542	ALL 0.042 0.792 0.292 0.833 0.917 0.917 0.500 0.458 0.417 0.083 0.125 0.563 0.563
Critical Success Factor Aesthetics Alignment of Project Objectives Change Order Reduction Collaboration of Project Delivery Team Effective Communication Competency & Capability of Project Team Constructability Project Cost Performance Dispute Avoidance & Resolution Design Innovation & Creativity Evidence Based Design Long-term Bldg. Success & Lifecycle Valu Owner / User Participation Owner / User Satisfaction Desoft & Simmeria Objectives	O 3.385 4.231 3.308 4.308 4.308 4.308 4.308 4.308 4.308 4.308 4.077 3.385 3.308 4.077 3.923 4.154 2.846	D 3.222 4.333 4.222 4.778 4.444 4.667 4.333 4.000 3.778 3.667 3.556 3.667 4.111 4.889 2.667	PROJ Mean B 2.800 4.400 3.300 4.300 4.400 3.000 3.000 3.000 3.000 3.000 3.000 3.000 3.800 4.000 3.400 4.000 3.800 4.700	U 3.900 4.100 3.800 4.200 4.200 4.200 4.200 4.200 3.700 3.700 3.500 3.700 4.000 4.000 4.000	ALL 3.333 4.262 3.619 4.381 4.405 4.405 3.905 3.881 3.857 3.357 3.476 3.952 3.952 4.571 2.548	O 21.0 5.5 23.5 3.5 2.0 10.5 12.0 15.5 21.0 20.1 10.5 10.5 10.5 10.5 13.0 8.0 8.0	D 24.0 7.5 10.5 2.0 5.0 3.0 7.5 13.5 16.0 18.5 22.0 18.5 12.0 1.0	Rank B 24.0 2.5 19.5 5.5 2.5 8.0 21.5 12.0 23.0 18.0 12.0 15.5 1.0 5.5	U 15.5 8.5 17.5 6.5 3.0 4.5 6.5 22.0 20.0 15.5 22.0 20.0 12.0 12.0 12.0	ALL 24.0 6.0 18.0 5.0 3.0 3.0 13.0 15.0 22.0 11.5 11.5 1.0	O 0.167 0.813 0.063 0.896 0.958 0.604 0.542 0.396 0.167 0.063 0.604 0.500 0.708 0.458	D 0.042 0.729 0.604 0.958 0.833 0.917 0.729 0.479 0.375 0.271 0.125 0.271 0.542 1.000	% Ran B 0.042 0.938 0.229 0.813 0.938 0.708 0.146 0.396 0.542 0.083 0.292 0.542 0.396 1.000	k 0.396 0.688 0.313 0.771 0.917 0.854 0.771 0.208 0.396 0.125 0.208 0.542 0.542 1.000 0.942	ALL 0.042 0.792 0.292 0.833 0.917 0.917 0.500 0.458 0.417 0.083 0.125 0.563 0.563 1.000
Critical Success Factor Aesthetics Alignment of Project Objectives Change Order Reduction Collaboration of Project Delivery Team Effective Communication Competency & Capability of Project Team Constructability Project Cost Performance Dispute Avoidance & Resolution Design Innovation & Creativity Evidence Based Design Long-term Bldg. Success & Lifecycle Valu Owner / User Participation Owner / User Satisfaction Profit & Financial Objectives Design fine of Service Advectives	O 3.385 4.231 3.308 4.308 4.308 4.308 4.308 4.308 3.385 3.769 3.385 3.308 4.077 3.923 4.154 3.846	D 3.222 4.333 4.222 4.778 4.444 4.667 4.333 4.000 3.778 3.667 3.556 3.667 4.111 4.889 3.667	PROJ Mean B 2.800 4.400 3.300 4.300 4.400 3.000 3.000 3.000 3.000 3.000 3.800 4.000 3.800 4.700 3.800	U 3.900 4.100 3.800 4.200 4.200 4.200 4.200 4.200 3.700 3.700 3.500 3.700 4.000 4.000 4.000 4.000	ALL 3.333 4.262 3.619 4.381 4.405 4.405 3.905 3.881 3.857 3.357 3.476 3.952 3.952 4.571 3.548 4.405	O 21.0 5.5 23.5 3.5 2.0 10.5 21.0 10.5 12.0 15.5 21.0 23.5 10.5 10.5 13.0 8.0 14.0	Г 24.0 7.5 10.5 2.0 5.0 3.0 7.5 13.5 16.0 18.5 12.0 18.5 12.0 18.5 12.0	Rank B 24.0 2.5 5.5 2.5 8.0 21.5 15.5 12.0 23.0 18.0 12.0 15.5 1.0 19.5 5.5	U 15.5 8.5 17.5 6.5 3.0 4.5 6.5 22.0 20.0 15.5 22.0 20.0 12.0 12.0 12.0 1.0 .0 12.0	ALL 24.0 6.0 18.0 5.0 3.0 13.0 14.0 15.0 23.0 22.0 11.5 1.0 19.0 9.0	O 0.167 0.813 0.063 0.896 0.958 0.604 0.542 0.396 0.167 0.063 0.604 0.500 0.708 0.450	D 0.042 0.729 0.604 0.958 0.833 0.917 0.729 0.479 0.375 0.271 0.125 0.271 0.542 1.000 0.271	% Ran B 0.042 0.938 0.229 0.813 0.938 0.708 0.146 0.396 0.542 0.083 0.292 0.542 0.396 1.000 0.229	k U 0.396 0.688 0.313 0.771 0.917 0.854 0.771 0.208 0.396 0.125 0.208 0.542 0.542 1.000 0.042	ALL 0.042 0.792 0.292 0.833 0.917 0.917 0.917 0.500 0.458 0.417 0.083 0.125 0.563 0.563 1.000 0.250
Critical Success Factor Aesthetics Alignment of Project Objectives Change Order Reduction Collaboration of Project Delivery Team Effective Communication Competency & Capability of Project Team Constructability Project Cost Performance Dispute Avoidance & Resolution Design Innovation & Creativity Evidence Based Design Long-term Bldg. Success & Lifecycle Valu Owner / User Participation Owner / User Participation Owner / User Satisfaction Profit & Financial Objectives Production of Specified Quality Claar & Bealistic Objectives	O 3.385 4.231 3.308 4.308 4.308 4.308 4.308 4.308 4.308 4.308 4.308 4.308 4.308 4.308 4.308 3.308 4.077 3.923 4.154 3.846 4.231	D 3.222 4.333 4.222 4.778 4.444 4.667 4.333 4.000 3.778 3.667 3.556 3.667 4.111 4.889 3.667 3.889 4.556	PROJ Mean B 2.800 4.400 3.300 4.300 4.400 4.200 3.000 3.800 4.000 3.800 4.000 3.800 4.700 3.800 4.700 3.800 4.300	U 3.900 4.100 3.800 4.200 4.200 4.200 4.200 4.200 3.700 3.700 3.500 3.700 4.000 4.000 4.000 4.000 4.000	ALL 3.333 4.262 3.619 4.381 4.405 4.405 3.905 3.881 3.857 3.357 3.476 3.952 3.952 4.571 3.548 4.119	RES 0 21.0 5.5 23.5 3.5 2.0 10.5 12.0 15.5 21.0 23.5 10.5 13.0 8.0 8.0 14.0 5.5	F 24.0 7.5 10.5 2.0 5.0 3.0 7.5 13.5 16.0 18.5 12.0 1.0 18.5 15.0 4.0 4.0 4.0 4.0 4.0 4.0 4.0 4	Rank B 24.0 2.5 19.5 5.5 2.5 8.0 21.5 15.5 12.0 23.0 18.0 12.0 15.5 1.0 19.5 5.5	U 15.5 8.5 17.5 6.5 3.0 4.5 6.5 22.0 20.0 12.0 12.0 12.0 12.0 12.0 224.0	ALL 24.0 6.0 18.0 5.0 3.0 13.0 14.0 15.0 22.0 11.5 11.5 1.0 9.0 9.0	O 0.167 0.813 0.063 0.896 0.896 0.958 0.604 0.542 0.396 0.167 0.063 0.604 0.500 0.708 0.458 0.813 1.000	D 0.042 0.729 0.604 0.958 0.833 0.917 0.729 0.479 0.375 0.271 0.125 0.271 0.542 1.000 0.271 0.542	% Ran B 0.042 0.938 0.229 0.813 0.938 0.708 0.146 0.396 0.542 0.083 0.292 0.542 0.396 1.000 0.229 0.813 0.292 0.542	k U 0.396 0.688 0.313 0.771 0.917 0.854 0.771 0.208 0.396 0.125 0.208 0.542 0.542 0.542 0.542 0.542	ALL 0.042 0.792 0.292 0.833 0.917 0.917 0.500 0.458 0.417 0.083 0.125 0.563 0.563 1.000 0.250 0.250
Critical Success Factor Aesthetics Alignment of Project Objectives Change Order Reduction Collaboration of Project Delivery Team Effective Communication Competency & Capability of Project Team Constructability Project Cost Performance Dispute Avoidance & Resolution Design Innovation & Creativity Evidence Based Design Long-term Bldg. Success & Lifecycle Valu Owner / User Participation Owner / User Participation Profit & Financial Objectives Production of Specified Quality Clear & Realistic Objectives Professional Peropetation & Image	O 3.385 4.231 3.308 4.308 4.308 4.308 4.308 4.308 4.308 4.308 4.308 4.308 4.308 4.308 4.308 3.308 4.077 3.923 4.154 3.846 4.462 3.462	D 3.222 4.333 4.222 4.778 4.444 4.667 4.333 4.000 3.778 3.667 3.556 3.667 4.111 4.889 3.667 3.889 4.556	PROJ Mean B 2.800 4.400 3.300 4.300 4.400 4.200 3.000 3.800 4.000 3.800 4.000 3.800 4.000 3.800 4.000 3.800 4.300 4.300 4.300 4.300	U 3.900 4.100 3.800 4.200 4.200 4.200 4.200 4.200 3.700 3.700 3.700 3.500 3.700 4.000 4.000 4.000 4.000 4.600 3.400	ALL 3.333 4.262 3.619 4.381 4.405 4.405 3.905 3.881 3.857 3.357 3.476 3.952 3.952 4.571 3.548 4.119 4.405	RES 0 21.0 5.5 23.5 3.5 2.0 10.5 12.0 15.5 21.0 23.5 10.5 11.0 23.5 13.0 8.0 8.0 14.0 5.5	Г 24.0 7.5 10.5 2.0 5.0 3.0 7.5 13.5 16.0 18.5 12.0 1.0 18.5 12.0 1.0 1.0 4.0 22.0 2.0 1.0 5.0 5.0 5.0 5.0 5.0 5.0 5.0 5	Rank B 24.0 2.5 19.5 5.5 2.5 8.0 21.5 12.0 23.0 18.0 12.0 19.5 5.5 12.0 19.5 5.5 12.0 19.5 5.5	U 15.5 8.5 17.5 6.5 20.0 15.5 22.0 20.0 12.0 12.0 12.0 12.0 12.0 12.0	ALL 24.0 6.0 18.0 5.0 3.0 13.0 14.0 15.0 23.0 22.0 11.5 1.0 19.0 9.0 3.0	O 0.167 0.813 0.063 0.896 0.958 0.604 0.542 0.396 0.167 0.063 0.604 0.500 0.708 0.458 0.813 1.000	D 0.042 0.729 0.604 0.958 0.833 0.917 0.729 0.479 0.375 0.271 0.125 0.271 0.542 1.000 0.271 0.542 1.000 0.271 0.417 0.815	% Ran B 0.042 0.938 0.229 0.813 0.938 0.708 0.146 0.396 0.542 0.083 0.292 0.542 0.396 1.000 0.229 0.813 0.542 0.329	k U 0.396 0.688 0.313 0.771 0.917 0.854 0.771 0.208 0.396 0.125 0.208 0.542 0.542 0.542 0.542 0.542 0.9042	ALL 0.042 0.792 0.292 0.833 0.917 0.917 0.500 0.458 0.417 0.083 0.125 0.563 0.563 1.000 0.250 0.667 0.917
Critical Success Factor Aesthetics Alignment of Project Objectives Change Order Reduction Collaboration of Project Delivery Team Effective Communication Competency & Capability of Project Team Constructability Project Cost Performance Dispute Avoidance & Resolution Design Innovation & Creativity Evidence Based Design Long-term Bldg. Success & Lifecycle Valu Owner / User Participation Owner / User Participation Owner / User Satisfaction Profit & Financial Objectives Production of Specified Quality Clear & Realistic Objectives Professional Reputation & Image Responsive Admin & Decision Support	O 3.385 4.231 3.308 4.308 4.308 4.308 4.308 4.308 4.308 4.308 4.308 4.308 4.308 4.308 4.308 3.365 3.308 4.077 3.923 4.154 3.846 4.231 4.462 3.462	D 3.222 4.333 4.222 4.778 4.444 4.667 4.333 4.000 3.778 3.667 3.556 3.667 4.111 4.889 3.667 3.889 4.556 3.556	PROJ Mean B 2.800 4.400 3.300 4.300 4.400 4.200 3.000 3.800 4.000 3.800 4.000 3.800 4.000 3.800 4.000 3.800 4.300 3.800 4.300 3.800 4.300 3.800 4.300 3.800 4.300 3.800 4.300 3.800 4.300 3.800 4.300 3.800 4.300 3.800 4.300 3.800 4.300 3.800 3.800 4.000 3.800 4.000 3.800 4.000 3.800 4.000 3.800 4.000 3.800 4.000 3.800 4.000 3.800 4.000 3.800 4.000 3.800 4.000 3.800 4.300 4.000 3.800 4.000 3.800 4.000 3.800 4.300 4.000 3.800 4.300 4.000 3.800 4.300 3.600 4.300 3.600 4.300 4.300 3.600 3.600 3.600 4.300 3.6000 3.6000 3.6000 3.6000 3.6000 3.6000 3.6000 3.6000 3.6000 3.60000 3.6000 3.60000 3.600000 3.60000000000	U 3.900 4.100 3.800 4.200 4.200 4.200 4.200 4.200 3.700 3.700 3.700 3.500 3.700 4.000 4.000 4.000 4.000 4.600 3.800	ALL 3.333 4.262 3.619 4.381 4.405 3.905 3.881 3.857 3.357 3.476 3.952 3.952 4.571 3.548 4.119 4.405 3.508	RES 0 21.0 5.5 23.5 3.5 2.0 10.5 12.0 15.5 21.0 23.5 10.5 13.0 8.0 8.0 14.0 5.5 1.0 18.0 21.0	Г 24.0 7.5 10.5 2.0 5.0 3.0 7.5 13.5 16.0 18.5 12.0 1.0 18.5 12.0 1.0 1.0 5.0 1.0 1.0 1.0 1.0 1.0 1.0 1.0 1	Rank B 24.0 2.5 19.5 5.5 2.5 8.0 21.5 12.0 23.0 18.0 12.0 19.5 5.5 12.0 19.5 5.5 12.0 19.5 10.0 19.5 5.5 12.0 23.0 17.0 9.5	U 15.5 8.5 17.5 6.5 20.0 15.5 22.0 20.0 12.0 12.0 12.0 12.0 12.0 12.0	ALL 24.0 6.0 18.0 5.0 3.0 13.0 14.0 15.0 23.0 22.0 11.5 1.0 19.0 9.0 3.0 21.0	O 0.167 0.813 0.063 0.896 0.958 0.604 0.542 0.396 0.167 0.063 0.604 0.500 0.708 0.458 0.813 1.000 0.292 0.167	D 0.042 0.729 0.604 0.958 0.833 0.917 0.729 0.479 0.375 0.271 0.125 0.271 0.542 1.000 0.271 0.542 1.000 0.271 0.417 0.875 0.125	% Ran B 0.042 0.938 0.229 0.813 0.938 0.708 0.146 0.396 0.542 0.083 0.292 0.542 0.396 1.000 0.229 0.813 0.229 0.813 0.229 0.813 0.229	k U 0.396 0.688 0.313 0.771 0.854 0.771 0.208 0.396 0.125 0.208 0.54200000000000000000000000000000000000	ALL 0.042 0.792 0.292 0.833 0.917 0.917 0.500 0.458 0.417 0.083 0.125 0.563 0.563 1.000 0.250 0.667 0.917 0.167
Critical Success Factor Aesthetics Alignment of Project Objectives Change Order Reduction Collaboration of Project Delivery Team Effective Communication Competency & Capability of Project Team Constructability Project Cost Performance Dispute Avoidance & Resolution Design Innovation & Creativity Evidence Based Design Long-term Bldg. Success & Lifecycle Valu Owner / User Participation Owner / User Participation Owner / User Satisfaction Profit & Financial Objectives Production of Specified Quality Clear & Realistic Objectives Professional Reputation & Image Responsive Admin. & Decision Support	O 3.385 4.231 3.308 4.308 4.308 4.308 4.308 4.308 4.308 4.308 4.308 4.308 4.308 4.308 4.308 4.308 4.308 4.000 3.769 3.385 3.308 4.077 3.923 4.154 3.846 4.231 4.462 3.462 3.385	D 3.222 4.333 4.222 4.778 4.444 4.667 4.333 4.000 3.778 3.667 3.556 3.667 4.111 4.889 3.667 3.889 4.556 3.556 3.556 3.556	PROJ Mean B 2.800 4.400 3.300 4.400 4.200 3.000 3.000 3.000 3.000 3.000 3.000 3.800 4.000 3.800 4.700 3.800 4.700 3.300 4.300 4.000 3.600 4.100	U 3.900 4.100 3.800 4.200 4.200 4.200 4.200 4.200 3.700 3.700 3.700 4.000 4.000 4.000 4.000 4.600 3.400 3.800 3.800	ALL 3.333 4.262 3.619 4.381 4.405 3.905 3.881 3.857 3.357 3.476 3.952 3.952 4.571 3.548 4.119 4.405 3.508 4.119 4.405 3.508 4.119 4.405 3.508 4.119 4.405 3.508 4.571 3.548 4.119 4.405 3.508 3.524 4.571 3.524 4.571 3.524 4.571 3.524 4.571 3.524 5.5245 5.524 5.5255 5.525 5.525 5.5255 5.5255 5.5255 5.5255 5.5255 5.5255 5.5255 5.5255 5.5255 5.5255 5.52555 5.52555 5.	RES 0 21.0 5.5 23.5 3.5 2.0 10.5 12.0 15.5 21.0 23.5 10.5 13.0 8.0 14.0 5.5 1.0 8.0 18.0 21.0 18.0	Г 24.0 7.5 10.5 2.0 5.0 3.0 7.5 13.5 16.0 18.5 12.0 1.0 18.5 12.0 1.0 18.5 15.0 22.0 1.0 1.0 1.0 1.0 1.0 1.0 1.0 1	Rank B 24.0 2.5 19.5 5.5 2.5 8.0 21.5 12.0 23.0 18.0 12.0 19.5 5.5 12.0 19.5 5.5 12.0 19.5 5.5 12.0 23.0 19.5 5.5 12.0 23.0 17.0 7.0 29.5 5.5 2.5 12.5 12.5 12.5 12.5 12.5 12.5	U 15.5 8.5 17.5 6.5 20.0 15.5 22.0 20.0 12.0 12.0 12.0 12.0 12.0 12.0	ALL 24.0 6.0 18.0 5.0 3.0 13.0 14.0 15.0 23.0 22.0 11.5 1.0 19.0 9.0 3.0 21.0 16.0 20.0	O 0.167 0.813 0.063 0.896 0.896 0.958 0.604 0.542 0.396 0.167 0.063 0.604 0.500 0.708 0.458 0.813 1.000 0.292 0.167	D 0.042 0.729 0.604 0.958 0.833 0.917 0.729 0.479 0.375 0.271 0.125 0.271 0.542 1.000 0.271 0.542 1.000 0.271 0.417 0.875 0.125 0.604 0.275	% Ran B 0.042 0.938 0.229 0.813 0.938 0.708 0.146 0.396 0.542 0.083 0.292 0.542 0.396 1.000 0.229 0.813 0.542 0.333 0.542 0.333 0.542	k U 0.396 0.688 0.313 0.771 0.917 0.854 0.771 0.208 0.396 0.125 0.208 0.54200000000000000000000000000000000000	ALL 0.042 0.792 0.292 0.833 0.917 0.917 0.500 0.458 0.417 0.083 0.125 0.563 0.563 1.000 0.250 0.667 0.917 0.167 0.375
Critical Success Factor Aesthetics Alignment of Project Objectives Change Order Reduction Collaboration of Project Delivery Team Effective Communication Competency & Capability of Project Team Constructability Project Cost Performance Dispute Avoidance & Resolution Design Innovation & Creativity Evidence Based Design Long-term Bldg. Success & Lifecycle Valu Owner / User Participation Owner / User Satisfaction Profit & Financial Objectives Production of Specified Quality Clear & Realistic Objectives Professional Reputation & Image Responsive Admin. & Decision Support Sustainability Project Time Performance	O 3.385 4.231 3.308 4.308 4.308 4.308 4.308 4.308 4.308 4.308 4.308 4.308 4.308 4.308 4.308 4.308 4.308 3.365 3.308 4.077 3.923 4.154 3.846 4.231 4.462 3.462 3.385 3.385 3.462	D 3.222 4.333 4.222 4.778 4.444 4.667 4.333 4.000 3.778 3.667 3.556 3.667 3.556 3.667 4.111 4.889 3.667 3.889 4.556 3.556 4.222 3.667 4.333	PROJ Mean B 2.800 4.400 3.300 4.400 4.400 4.200 3.000 3.000 3.000 3.000 3.000 3.800 4.000 3.800 4.000 3.800 4.000 3.800 4.000 3.800 4.000 3.800 4.000 3.800 4.000 3.600 4.100 3.000	U 3.900 4.100 3.800 4.200 4.200 4.200 4.200 4.200 3.700 3.700 3.700 4.000 4.000 4.000 4.000 4.000 4.000 3.800 4.000	ALL 3.333 4.262 3.619 4.381 4.405 3.905 3.881 3.857 3.357 3.476 3.952 3.952 4.571 3.548 4.119 4.405 3.508 4.119 4.405 3.533 3.524 4.071	RES 0 21.0 5.5 23.5 3.5 2.0 10.5 12.0 15.5 21.0 23.5 13.0 8.0 14.0 5.5 13.0 8.0 14.0 18.0 21.0 18.0 15.5 10.5	Г 24.0 7.5 10.5 2.0 5.0 3.0 7.5 13.5 16.0 18.5 12.0 1.0 18.5 15.0 4.0 10.5 15.0 15.0 15.0 18.5 15.0 10.5 12.0 18.5 12.0 10.5 12.0 18.5 12.0 10.5 12.0 18.5 12.0 10.5 12.0 10.5 12.0 10.5 12.0 10.5 12.0 10.5 12.0 10.5 12.0 10.5 13.5 15.0 10.5 15.0 10.5 15.0 10.5 15.0 15.0 15.0 15.0 15.0 15.0 15.0 15.0 15.0 15.0 15.0 15.0 15.0 15.0 10.5 15.0 10.5 15.0 10.5 15.0 10.5 15.0 10.5 15.0 10.5 15.0 10.5 15.0 10.5 15.0 10.5 15.0 10.5 15.0 10.5 15.0 10.5 15.0 10.5 15.0 10.5 15.0 10.5 15.0 10.5 15.0 10.5 15.0 10.5 15.0 10.5 15.0 15.5 15.0 15.5 15.	Rank B 24.0 2.5 19.5 5.5 2.5 8.0 21.5 12.0 23.0 18.0 12.0 19.5 5.5 12.0 19.5 5.5 12.0 23.0 19.5 5.5 12.0 23.0 19.5 5.5 5.5 5.5 5.5 5.5 5.5 5.5 5.5 5.5	U 15.5 8.5 17.5 6.5 20.0 15.5 22.0 20.0 12.0 12.0 12.0 12.0 24.0 12.0 23.0 23.0 17.5 12.0 12.0 17.5	ALL 24.0 6.0 18.0 5.0 3.0 13.0 14.0 15.0 23.0 22.0 11.5 1.0 19.0 9.0 21.0 16.0 20.0 10.0	O 0.167 0.813 0.063 0.896 0.958 0.604 0.542 0.396 0.167 0.063 0.604 0.500 0.708 0.458 0.813 1.000 0.292 0.167 0.292	D 0.042 0.729 0.604 0.958 0.833 0.917 0.729 0.479 0.375 0.271 0.125 0.271 0.542 1.000 0.271 0.542 1.000 0.271 0.417 0.875 0.125 0.604 0.271 0.272	% Ran B 0.042 0.938 0.229 0.813 0.938 0.708 0.146 0.396 0.542 0.083 0.292 0.542 0.396 1.000 0.229 0.813 0.542 0.333 0.542 0.333 0.542 0.333	k U 0.396 0.688 0.313 0.771 0.854 0.771 0.854 0.771 0.208 0.396 0.125 0.208 0.542 0.542 0.542 0.542 0.958 0.083 0.313 0.542 0.542	ALL 0.042 0.792 0.292 0.833 0.917 0.917 0.500 0.458 0.417 0.083 0.125 0.563 0.563 1.000 0.250 0.667 0.917 0.167 0.375 0.208
Critical Success Factor Aesthetics Alignment of Project Objectives Change Order Reduction Collaboration of Project Delivery Team Effective Communication Competency & Capability of Project Team Constructability Project Cost Performance Dispute Avoidance & Resolution Design Innovation & Creativity Evidence Based Design Long-term Bldg. Success & Lifecycle Valu Owner / User Participation Owner / User Satisfaction Profit & Financial Objectives Production of Specified Quality Clear & Realistic Objectives Professional Reputation & Image Responsive Admin. & Decision Support Sustainability Project Time Performance Trust & Respect	O 3.385 4.231 3.308 4.308 4.308 4.308 4.308 4.308 4.308 4.308 4.308 4.308 4.308 4.308 4.308 4.308 4.308 4.000 3.769 3.462 3.462 3.462 3.462 3.462 3.462 3.462	D 3.222 4.333 4.222 4.778 4.444 4.667 4.333 4.000 3.778 3.667 3.556 3.667 4.111 4.889 3.667 3.889 4.556 3.556 4.222 3.667 4.222 3.667 4.333	PROJ Mean B 2.800 4.400 3.300 4.400 4.400 4.200 3.000 3.000 3.000 3.000 3.000 3.800 4.000 3.800 4.000 3.800 4.000 3.800 4.000 3.800 4.000 3.800 4.000 3.800 4.000 3.600 4.100 3.000 4.300	U 3.900 4.100 3.800 4.200 4.200 4.200 4.200 4.200 3.700 3.700 3.700 4.000 4.000 4.000 4.000 4.000 4.000 4.000 4.000 4.000 4.000	ALL 3.333 4.262 3.619 4.381 4.405 3.905 3.881 3.857 3.357 3.476 3.952 3.952 4.571 3.548 4.119 4.405 3.503 3.524 4.071 3.530 3.833 3.524 4.071 3.548 4.119 4.405 3.500 3.833 3.524 4.071 3.548 4.071 3.548 4.119 4.2144 4.2144 4.2144 4.2144 4.2144 4.2144 4.2144 4.21	RES 0 21.0 5.5 23.5 3.5 2.0 10.5 12.0 15.5 21.0 23.5 10.5 13.0 8.0 14.0 5.5 1.0 0 21.0 5.5 13.0 8.0 18.0 18.0 21.0 5.5 8.0	Г 24.0 7.5 10.5 2.0 5.0 3.0 7.5 13.5 16.0 18.5 12.0 1.0 18.5 15.0 4.0 10.5 15.0 15.0 15.0 15.0 18.5 15.0 10.5 10.0 10.5 10.	Rank B 24.0 2.5 19.5 5.5 2.5 8.0 21.5 12.0 23.0 18.0 12.0 19.5 5.5 12.0 19.5 5.5 12.0 23.0 19.5 5.5 5.5 5.5 5.5	U 15.5 8.5 17.5 6.5 3.0 4.5 6.5 22.0 20.0 12.0 12.0 12.0 12.0 12.0 23.0 23.0 17.5 12.0 23.0 8.5	ALL 24.0 6.0 18.0 5.0 3.0 13.0 14.0 15.0 23.0 22.0 11.5 1.0 19.0 9.0 21.0 16.0 20.0 10.0 7.0	O 0.167 0.813 0.063 0.896 0.896 0.958 0.604 0.542 0.396 0.167 0.063 0.604 0.500 0.708 0.458 0.813 1.000 0.292 0.167 0.292 0.396 0.396	D 0.042 0.729 0.604 0.958 0.833 0.917 0.729 0.479 0.375 0.271 0.125 0.271 0.542 1.000 0.271 0.542 1.000 0.271 0.417 0.875 0.125 0.604 0.271 0.729	% Ran B 0.042 0.938 0.229 0.813 0.938 0.708 0.146 0.396 0.542 0.083 0.292 0.542 0.396 1.000 0.229 0.813 0.542 0.333 0.646 0.146 0.813	k U 0.396 0.688 0.313 0.771 0.854 0.771 0.854 0.771 0.208 0.396 0.125 0.208 0.542 0.542 0.542 0.958 0.083 0.313 0.542 0.542 0.688	ALL 0.042 0.792 0.292 0.833 0.917 0.917 0.917 0.500 0.458 0.417 0.083 0.125 0.563 0.563 1.000 0.250 0.667 0.917 0.167 0.375 0.208 0.625 0.205
Critical Success Factor Aesthetics Alignment of Project Objectives Change Order Reduction Collaboration of Project Delivery Team Effective Communication Competency & Capability of Project Team Constructability Project Cost Performance Dispute Avoidance & Resolution Design Innovation & Creativity Evidence Based Design Long-term Bldg. Success & Lifecycle Valu Owner / User Participation Owner / User Participation Owner / User Satisfaction Profit & Financial Objectives Production of Specified Quality Clear & Realistic Objectives Professional Reputation & Image Responsive Admin. & Decision Support Sustainability Project Time Performance Trust & Respect Utility & Functionality	O 3.385 4.231 3.308 4.308 4.308 4.308 4.308 4.308 4.308 4.308 4.308 4.308 4.308 4.308 4.308 4.308 4.308 4.000 3.769 3.462 3.385 3.462 3.462 3.462 3.769 4.154	D 3.222 4.333 4.222 4.778 4.444 4.667 4.333 4.000 3.778 3.667 3.556 3.667 4.111 4.889 3.667 3.889 4.556 3.556 4.222 3.667 4.333 4.333 4.333 4.000	PROJ Mean B 2.800 4.400 3.300 4.400 3.300 4.400 3.000 3.000 3.000 3.000 3.000 3.000 3.800 4.000 3.800 4.000 3.800 4.000 3.800 4.300 4.300 4.300 4.300 4.300	U 3.900 4.100 3.800 4.200 4.200 4.200 4.200 4.200 3.700 3.700 3.700 4.000 4.000 4.000 4.000 4.000 4.000 4.000 4.000 4.000 4.000 4.000	ALL 3.333 4.262 3.619 4.381 4.405 3.905 3.881 3.857 3.357 3.476 3.952 3.952 4.571 3.548 4.119 4.405 3.500 3.833 3.524 4.071 4.214 4.167	RES 0 21.0 5.5 23.5 3.5 2.0 10.5 12.0 15.5 21.0 23.5 13.0 8.0 14.0 5.5 13.0 8.0 18.0 18.0 21.0 8.0 8.0 8.0 10.5 1	Г 24.0 7.5 10.5 2.0 5.0 3.0 7.5 13.5 16.0 18.5 12.0 1.0 18.5 12.0 1.0 18.5 15.0 22.0 1.0 18.5 12.0 1.0 1.0 5.0 1.0 1.0 5.0 1.0 1.0 5.0 1.0 1.0 7.5 1.0 1.0 7.5 1.0 1.0 7.5 1.0 1.0 7.5 1.0 1.0 7.5 1.0 1.0 7.5 1.0 7.5 1.0 1.0 7.5 1.0 7.5 1.0 7.5 1.0 7.5 1.0 7.5 1.0 7.5 1.0 7.5 1.0 7.5 7.5 1.0 7.5 7.5 1.0 7.5 7.5 1.0 7.5 7.5 1.0 7.5 7.5 1.0 7.5 7.5 1.0 7.5 7.5 1.0 7.5 7.5 1.0 7.5 7.5 1.0 7.5 7.5 1.0 7.5 7.5 1.0 7.5 7.5 7.5 1.0 7.5 7.5 7.5 7.5 7.5 7.5 7.5 7.5	Rank B 24.0 2.5 19.5 5.5 2.5 8.0 21.5 12.0 23.0 18.0 12.0 19.5 5.5 12.0 19.5 5.5 12.0 23.0 19.5 5.5 5.5 5.5 9.5	U 15.5 8.5 17.5 6.5 3.0 4.5 22.0 20.0 15.5 22.0 20.0 12.0 12.0 12.0 12.0 12.0 12.0	ALL 24.0 6.0 18.0 5.0 3.0 13.0 14.0 15.0 23.0 22.0 11.5 1.0 19.0 9.0 21.0 16.0 20.0 10.0 7.0 8.0	O 0.167 0.813 0.063 0.896 0.896 0.958 0.604 0.542 0.396 0.167 0.063 0.604 0.500 0.708 0.458 0.813 1.000 0.292 0.167 0.292 0.396 0.708	D 0.042 0.729 0.604 0.958 0.833 0.917 0.729 0.479 0.375 0.271 0.125 0.271 0.542 1.000 0.271 0.542 1.000 0.271 0.417 0.875 0.125 0.604 0.271 0.729 0.604 0.271 0.729 0.729	% Ran B 0.042 0.938 0.229 0.813 0.938 0.708 0.146 0.396 0.542 0.083 0.292 0.542 0.396 1.000 0.229 0.542 0.396 1.000 0.229 0.813 0.542 0.333 0.646 0.146 0.813 0.6813 0.813 0.646	k U 0.396 0.688 0.313 0.771 0.854 0.771 0.208 0.396 0.125 0.208 0.542 0.542 0.542 0.958 0.042 0.542 0.958 0.313 0.542 0.688 0.854	ALL 0.042 0.792 0.292 0.833 0.917 0.917 0.917 0.500 0.458 0.417 0.083 0.125 0.563 0.563 1.000 0.250 0.667 0.917 0.167 0.375 0.208 0.625 0.7500 0.708
Critical Success Factor Aesthetics Alignment of Project Objectives Change Order Reduction Collaboration of Project Delivery Team Effective Communication Competency & Capability of Project Team Constructability Project Cost Performance Dispute Avoidance & Resolution Design Innovation & Creativity Evidence Based Design Long-term Bldg. Success & Lifecycle Valu Owner / User Participation Owner / User Participation Owner / User Satisfaction Profit & Financial Objectives Production of Specified Quality Clear & Realistic Objectives Professional Reputation & Image Responsive Admin. & Decision Support Sustainability Project Time Performance Trust & Respect Utility & Functionality Owner's vision	O 3.385 4.231 3.308 4.308 4.308 4.308 4.308 4.308 4.308 4.308 4.308 4.308 4.308 4.308 4.308 4.308 4.308 4.000 3.769 3.462 3.462 3.769 4.154 3.462	D 3.222 4.333 4.222 4.778 4.444 4.667 4.333 4.000 3.778 3.667 3.556 3.667 4.111 4.889 3.667 3.889 4.556 3.556 4.222 3.667 4.333 4.333 4.333 4.333	PROJ Mean B 2.800 4.400 3.300 4.400 3.300 4.400 3.000 3.000 3.000 3.000 3.000 3.000 3.800 4.000 3.800 4.000 3.800 4.000 3.800 4.000 3.800 4.000 3.600 4.100 3.000 4.300 4.100 3.900	U 3.900 4.100 3.800 4.200 4.200 4.200 4.200 4.200 3.700 3.700 3.700 4.000 4.000 4.000 4.000 4.000 4.000 4.000 4.000 4.000 4.000 3.800 4.000 4.000 4.000 3.700	ALL 3.333 4.262 3.619 4.381 4.405 3.905 3.881 3.857 3.357 3.476 3.952 3.952 4.571 3.548 4.119 4.405 3.500 3.833 3.524 4.071 4.214 4.167 3.643	RES 0 21.0 5.5 23.5 3.5 2.0 10.5 12.0 15.5 21.0 23.5 13.0 8.0 14.0 5.5 13.0 8.0 18.0	Г 24.0 7.5 10.5 2.0 5.0 3.0 7.5 13.5 16.0 18.5 12.0 1.0 18.5 12.0 1.0 18.5 15.0 22.0 1.0 15.0 18.5 12.0 1.0 1.0 5.0 13.5 12.0 1.0 5.0 13.5 12.0 1.0 5.0 13.5 12.0 1.0 5.0 13.5 12.0 1.0 5.0 13.5 12.0 1.0 5.0 13.5 12.0 1.0 5.0 13.5 12.0 1.0 5.0 13.5 12.0 1.0 5.0 1.0 5.0 1.0 5.0 1.0 5.0 1.0 5.0 1.0 5.0 1.0 5.0 1.0 5.0 1.0 5.0 1.0 5.0 1.0 5.0 1.0 5.0 1.0 5.0 1.0 1.0 5.0 1.0 5.0 1.0 1.0 1.0 5.0 1.0 1.0 5.0 1.0 1.0 1.0 5.0 1.0 1.0 5.0 1.0 1.0 5.0 1.0 5.0 1.0 5.0 1.0 5.0 1.0 5.0 1.0 5.0 1.0 5.0 1.0 5.0 1.0 5.0 1.0 5.0 1.0 5.0 1.0 5.0 1.0 5.0 1.0 5.0 1.0 5.0 7.5 1.0 5.0 1.0 5.0 7.5 1.0 5.0 1.0 5.0 7.5 1.0 5.0 7.5 2.2 0.0 5.0 1.0 5.0 5.0 7.5 2.2 0.0 5.0 5.0 5.0 5.0 5.0 5.0 5.0	Rank B 24.0 2.5 19.5 5.5 2.5 8.0 21.5 12.0 23.0 18.0 12.0 19.5 5.5 12.0 19.5 5.5 12.0 23.0 19.5 5.5 5.5 5.5 5.5 5.5 5.5 14.0	U 15.5 8.5 17.5 6.5 3.0 4.5 6.5 22.0 20.0 15.5 22.0 20.0 12.0 12.0 12.0 12.0 12.0 12.0	ALL 24.0 6.0 18.0 5.0 3.0 13.0 14.0 15.0 23.0 22.0 11.5 1.0 19.0 9.0 21.0 16.0 20.0 10.0 7.0 8.0 17.0	O 0.167 0.813 0.063 0.896 0.958 0.604 0.542 0.396 0.167 0.063 0.604 0.500 0.708 0.458 0.813 1.000 0.292 0.167 0.292 0.396 0.708 0.708 0.708 0.708	D 0.042 0.729 0.604 0.958 0.833 0.917 0.729 0.479 0.375 0.271 0.125 0.271 0.542 1.000 0.271 0.542 1.000 0.271 0.417 0.875 0.125 0.604 0.271 0.729 0.729 0.729 0.729 0.729	% Ran B 0.042 0.938 0.229 0.813 0.938 0.708 0.146 0.396 0.542 0.083 0.292 0.542 0.396 1.000 0.229 0.542 0.396 1.000 0.229 0.813 0.542 0.333 0.646 0.146 0.813 0.646 0.458	k U 0.396 0.688 0.313 0.771 0.854 0.771 0.208 0.396 0.125 0.208 0.542 0.542 0.542 0.542 0.542 0.958 0.313 0.542 0.688 0.688 0.6884 0.6884	ALL 0.042 0.792 0.292 0.833 0.917 0.917 0.500 0.458 0.417 0.083 0.125 0.563 0.563 1.000 0.250 0.667 0.917 0.167 0.375 0.208 0.625 0.7508 0.625 0.708 0.633

8.7. Relative Importance by Rank-Orders

 Table 8.12.
 Round 2 rank-order tables




Figure 8.11. Round 2 -- Overall: relative importance by %Rank



Figure 8.12. Round 2 -- Owners: relative importance by %Rank

Figure 8.13. Round 2 -- Designers: relative importance by %Rank



Figure 8.14. Round 2 -- Builders: relative importance by %Rank



Figure 8.15. Round 2 -- Users: Relative Importance by %Rank

		Sel	f-Inte	rest		Project-Interest						
Critical Success Factors		%	Rank	a, b			%	Rank	a, b			
Clitical Success Factors	0	D	В	U	ALL	0	D	В	U	ALL		
	N=17	N=10	N=11	N=13	N=51	N=17	N=10	N=11	N=12	N=50		
(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)		
Owner / User Satisfaction	0.917	1.000	1.000	0.958	1.000	0.708	1.000	1.000	1.000	1.000		
Clear & Realistic Objectives	1.000	0.958	0.604	0.813	0.958	1.000	0.875	0.542	0.958	0.917		
Effective Communication	0.917	0.583	0.896	0.875	0.896	0.896	0.833	0.938	0.917	0.917		
Competency & Capability of Project Delivery Team	0.688	0.813	0.813	0.813	0.792	0.958	0.917	0.708	0.854	0.917		
Utility & Functionality	0.917	0.917	0.292	1.000	0.896	0.708	0.479	0.646	0.854	0.708		
Collaboration of Project Delivery Team	0.688	0.813	0.604	0.375	0.667	0.896	0.958	0.813	0.771	0.833		
Trust Respect	0.458	0.688	0.896	0.667	0.667	0.708	0.729	0.813	0.688	0.750		
Owner / User Participation	0.750	0.813	0.729	0.917	0.833	0.500	0.542	0.396	0.542	0.563		
Alignment of Project Objectives	0.583	0.271	0.958	0.458	0.563	0.813	0.729	0.938	0.688	0.792		
Production of Specified Quality	0.625	0.583	0.604	0.667	0.667	0.813	0.417	0.813	0.542	0.667		
Long-term Building Success & Lifecycle Value	0.792	0.583	0.354	0.542	0.563	0.604	0.271	0.542	0.542	0.563		
Project Time Performance	0.333	0.271	0.813	0.250	0.479	0.396	0.729	0.813	0.542	0.625		
Owner's vision	0.833	0.813	0.438	0.458	0.750	0.292	0.125	0.458	0.208	0.333		
Project Cost Performance	0.458	0.188	0.729	0.292	0.479	0.542	0.479	0.396	0.208	0.458		
Constructability	0.458	0.688	0.354	0.208	0.396	0.604	0.729	0.146	0.771	0.500		
Responsive Administration Decision Support	0.229	0.375	0.500	0.458	0.396	0.167	0.604	0.646	0.313	0.375		
Sustainability	0.542	0.375	0.125	0.667	0.333	0.292	0.271	0.146	0.542	0.208		
Dispute Avoidance Resolution	0.083	0.083	0.438	0.083	0.083	0.396	0.375	0.542	0.396	0.417		
Professional Reputation Image	0.083	0.479	0.604	0.083	0.250	0.292	0.125	0.333	0.083	0.167		
Evidence Based Design	0.333	0.125	0.188	0.667	0.292	0.063	0.125	0.292	0.208	0.125		
Profit & Financial Objectives	0.083	0.375	0.250	0.167	0.125	0.458	0.271	0.229	0.042	0.250		
Change Order Reduction	0.167	0.042	0.188	0.083	0.042	0.063	0.604	0.229	0.313	0.292		
Aesthetics	0.333	0.188	0.042	0.667	0.208	0.167	0.042	0.042	0.396	0.042		
Design Innovation & Creativity	0.229	0.479	0.083	0.333	0.167	0.167	0.271	0.083	0.125	0.083		
Stakeholder Groups Types: O =	Owner;	D = D	esigner	B = B	uilder; U	U = Use	er; ALL					
Notes:												

Notes:

a. %Rank: normalized rank order where 1.00 = highest ranked.

b. Selection criteria for highlighted cells (Green/Bold): greater than or equal to 0.75



Table 8.13 provides a summary of all fractional rank-order scores, and is sorted in order

of overall relative importance (ALL) where 1.000 equals the highest ranking. Highlighting the

top 25% (%Rank greater than or equal to 0.750) visually demonstrates variation is how each

stakeholder group rated the most relatively important factors.

8.8. Top 25% Summary

			Sel	f-Inter	est			Proj	ect-Int	erest	
	Top 25% CSFs	R	ound#	2 % I	Rank	a, b	R	ound#	2 % I	Rank	ı, b
	100 20 /0 0015	0	D	В	U	ALL	0	D	В	U	ALL
		N=17	N=10	N=11	N=13	N=51	N=17	N=10	N=11	N=12	N=50
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)
1	Owner / User Satisfaction	0.92	1.00	1.00	0.96	1.0	0.71	1.00	1.00	1.00	1.00
2	Clear & Realistic Objectives	1.00	0.96	0.60	0.81	1.0	1.00	0.88	0.54	0.96	0.92
3	Effective Communication	0.92	0.58	0.90	0.88	0.9	0.90	0.83	0.94	0.92	0.92
Х	Competency & Capability of Project Delivery Team	0.69	0.81	0.81	0.81	0.8	0.96	0.92	0.71	0.85	0.92
4	Utility & Functionality	0.92	0.92	0.29	1.00	0.9	0.71	0.48	0.65	0.85	0.71
5	Collaboration of Project Delivery Team	0.69	0.81	0.60	0.38	0.7	0.90	0.96	0.81	0.77	0.83
6	Trust Respect	0.46	0.69	0.90	0.67	0.7	0.71	0.73	0.81	0.69	0.75
7	Owner / User Participation	0.75	0.81	0.73	0.92	0.8	0.50	0.54	0.40	0.54	0.56
8	Alignment of Project Objectives	0.58	0.27	0.96	0.46	0.6	0.81	0.73	0.94	0.69	0.79
9	Production of Specified Quality	0.63	0.58	0.60	0.67	0.7	0.81	0.42	0.81	0.54	0.67
10	Long-term Building Success & Lifecycle Value	0.79	0.58	0.35	0.54	0.6	0.60	0.27	0.54	0.54	0.56
11	Project Time Performance	0.33	0.27	0.81	0.25	0.5	0.40	0.73	0.81	0.54	0.63
12	Owner's vision	0.83	0.81	0.44	0.46	0.8	0.29	0.13	0.46	0.21	0.33
13	Constructability	0.46	0.69	0.35	0.21	0.4	0.60	0.73	0.15	0.77	0.50
	Stakeholder Groups Types: O = Owner; D = Designer; B = Builder; U = User; ALL										

Notes:

a. %Rank: normalized rank order where 1.00 = highest ranked

b. Selection criteria for highlighted areas: Top 25%: Greater than or equal to 0.75 (Green/Bold highlight)

c. "Competency & Capability of Project Delivery Team" will not be submitted for further analysis the Round #3. Analysis of effectiveness of different project delivery methods to achieve the selected CSFs will assume competence/capability of the project delivery team.

Table 8.14. Round 2 -- Thirteen critical success factors selected for the round 3 survey

There are fourteen critical success factors which scored 0.75 or higher in at least one stakeholder group. One factor, "Competency & Capability of Project Delivery Team" is noted as significantly important ranking in as the fourth highest ranked out of the factors evaluated, but will not be selected for evaluation in the round 3 survey. The round 3 survey will evaluate different project delivery methods efficacy in attainment of factor goals. Perhaps there is not greater influence on the outcome of a project regardless of the delivery method than the quality of the individuals and leadership on the project team. Round 3 will assume an equally reasonable, competent and capable project delivery team in evaluation of the three different project delivery methods.



Figure 8.16. Round 2 Overall: Difference in %Ranks between Self & Project interest sorted by relative importance

Logically critical success factors with no difference in relative importance between self and project interest considerations pose less of a hindrance to stakeholder collaboration regardless of the project delivery method. Where there is a difference between self-interest and project-interest, as indicated in Figure 8.16. above, there is opportunity for additional optimization of the project delivery process and resulting value-creation. Resolving the conflict of interests, or implementing practices and procedures which mitigate the differences or make them more palatable to all stakeholder parties is sure to increase overall efficiency of the collaboration. Further study is required to prove this concept, but it is logically a sound cause and effect relationship with great potential for increased value-creation through optimization of the project delivery process.

8.9.1. Difference Analysis: Significance Test

Because the list of critical success factors in round 2 are all important, having been narrowed down by relative importance from round 1's fourty factors to the thirteen most important here in round 2, the differences between many of the mean scores is fairly slight, given that most factors recieved ratings ranging over the top three of the five-point scale. When normalized into rank-orders the differences becomes more emphasized. The emphasized difference is beneficial for highlighting differences, but more rigorous testing is required to determine the significance of these differences. A non-parametric test of related samples was conducted in SPSS Statistics 18 using the Wilcoxon Signed Rank Test for paired samples. Each factor was evaluated as a pair including the self-interest and the project-interest ratings for each critical success factor. The testing was run for the overall group rating, and each individual stakeholder's group ratings. Table 8.15 contains the resulting levels of significance for the hypothesis test run in SPSS Statistics 18. The hypothesis for the test is that there is a difference between the self and project interest ratings for each critical success factor. The null hypothesis is that no difference exists between the two ratings for the factor. Ten factors yielded significant differences in at least one stakeholder group (or overall) at a minimum 0.10 significance level.

Therefore the null hypothesis is rejected for the ten factors the 0.10 level. It is therefore concluded that a significant difference exists for ten of the critical success factors.

Significance Testing of the Difference between Self-Interest & Project-Interest											
Null Hypothesis: median of differences between Self-Interest & Project-Interest equals zero for each CSE											
equals zero for	each CS	<u>SF</u>		~							
	Related-S	Samples W	/ilcoxon	Signed Ra	ank Test*						
Critical Success Factors	Owner	Designer	Builder	All							
Owner / User Satisfaction	0.194	1.000	0.317	0.655	0.334						
Production of Specified Quality	0.527	0.480	0.317	0.705	0.888						
Long-term Building Success & Lifecycle Value	0.380	0.157	0.435	1.000	0.551						
Responsive Administration Decision Support	1.000	0.516	0.679	0.705	0.664						
Effective Communication	0.458	0.527	1.000	0.317	0.761						
Project Cost Performance	0.680	0.414	0.279	0.705	0.914						
Clear & Realistic Objectives	0.705	0.655	0.655	0.102	0.837						
Professional Reputation Image	0.206	0.102	0.301	0.317	0.673						
Trust Respect	0.096 ^c	1.000	0.655	1.000	0.371						
Design Innovation & Creativity	1.000	0.257	0.564	0.480	0.371						
Constructability	0.257	1.000	0.168	0.058 ^c	0.694						
Profit & Financial Objectives	0.070 ^c	0.083 ^c	0.603	0.890	0.688						
Sustainability	0.129	0.180	0.914	0.655	0.095 ^c						
Competency & Capability of Project Delivery Team	0.334	0.317	0.564	0.414	0.204						
Project Time Performance	0.655	0.157	1.000	0.236	0.079 ^c						
Aesthetics	0.157	0.157	0.234	0.480	0.313						
Collaboration of Project Delivery Team	0.720	0.257	0.480	0.206	0.108						
Evidence Based Design	0.096 ^c	0.705	0.705	0.194	0.115						
Utility & Functionality	0.157	0.180	0.238	0.102	0.300						
Alignment of Project Objectives	0.317	0.102	0.655	0.480	0.124						
Change Order Reduction	1.000	0.024 ^b	1.000	0.084 ^c	0.018 ^b						
Owner / User Participation	0.102	0.480	0.102	0.157	0.008 ^a						
Dispute Avoidance Resolution	0.033 ^b	0.257	0.783	0.070 ^c	0.008 ^a						
Owner's vision	0.058 ^c	0.023 ^b	0.705	0.317	0.007 ^a						
Notes: * Paired samples; Non-parametric analysis calculat Paired samples; Non-parametric analysis calculat	ed in SPS	S Statistic	s 8								

a. Reject the null hypothesis, significant at the 0.01 level

b. Reject the null hypothesis, significnat at the 0.05 level

c. Reject the nul hypothesis, significant at the 0.10 level

Table 8.15. Round 2 Significance testing of difference between self and project interests



Figure 8.17 Self-Interest & Project-Interest differences. (Black bars represent statistical significance refer to Table 7.15 for level of significance)

Figure 8.17. above provides additional detail as to which "direction" the difference in relative importance moves left or right of 0.0. Here the difference is measured in difference of %Rank between the factors. The bars extending to the left indicate a greater project-interest bias to relative importance, and the bars extending to the right indicate a greater self-interest bias towards relative importance. The dark bars represent the statistical significance listed in Table 8.17.



Figure 8.18. Detail of %Rank differences by stakeholder groups (and overall)



Figure 8.19. Round 2 -- Overall: difference and relative importance matrix

The matrix in Figure 8.19 plots each factor by relative importance and difference measured in fractional rank order (%Rank). Additionally each plot square is labeled for the project delivery category in which the factor was originally grouped in the round 1 survey. Quadrants C and D contain the highest relative importance, and quadrants B and C contain the

greatest measured differences between self-interest and project-interest. Logically, quadrant C becomes the primary area of focus when considering factors around which additional optimization may be achieved.

The factors in quadrant C are in order of relative importance are: Competence of the Project Delivery Team, Utility & Functionality, Collaboration & Capability of the Project Team, Owner/User Participation, Alignment of Project Objectives, and Project Time Performance. Also of note is the fact that four of the six factors in quadrant C are originally from the process category in the round 1 survey. Additionally, 6 out of the 12 most relatively important factors are also from the process category. Factors from any of categories could potentially be influenced by use of different project delivery methods. The process category factors, as the category nomenclature implies, inextricably associated with the project delivery process. A reasonable assumption then is that the process, in this case the project delivery method, is the primary means of addressing the quadrant C factors. The hypothesis of this dissertation is that Integrated Project Delivery is the superior method of project delivery for large complex medical projects. In round 3 I will put this hypothesis to a partial test by asking the panel of experts to evaluated different project delivery methods for effectiveness related to the 13 factors selected as most important. It seems plausible that an integrated approach may benefit the factors in quadrant C (or other factors not evaluated here which meet the same criteria).

8.10. Round 2 Results Comparison of Rounds 1 & 2

When comparing the round 1 and round 2 data visually in Figure 8.20., it is plain to see that the distribution of critical success factor rankings by stakeholder group is more linear in round 2, and that angle of linearity is very nearly a perfect 45 degrees. The data is normalized as fractional rank, where 1 is the highest rank and the lowest rank approaches zero (zero = not ranked). A perfect linear rank ordered distribution would have a nearly 45 degree line from near zero on the left to 1 on the right. The linear line on the round 1 is easily seen as having a clockwise rotation to an angle closer to 30 degrees. It is obvious from the simple visual



Figure 8.20. Relative importance comparison between rounds 1 & 2

inspection of the data sets that round 2 in fact achieve a higher level of inter-rater agreement and increased consensus among the overall panel of experts. The following tables and figures will directly compare the Kendall's W scores between the first two rounds of surveys to provide statistical proof to the 'thumb-nail' assessment above.

The two rounds used different measures to assess relative importance with round 1 using frequency, and round 2 using a mean score based on how the panel rated each critical success factor on a five-point-scale. In both cases Kendall's coefficient of concordance is calculated based on the rank-order of the factors, and provides a consistent measure between the two rounds. Frequencies and mean scores are not readily comparable, each round was normalized by creating weighted averages by dividing each factor's score (frequency, or mean rating) by the sum of score of all factors. The weighted averages help to draw comparisons relative to the four categories in round 1 and the consolidated list in round 2.

Comparison of Rounds One and Two: Overall Self-Interest							Comparison of Rounds One and Two: Overall Project-Intere					iterest	t
		Round	1 #1		Round	1#2	CSE		Round	1 #1		Round	#2
Critical Success Factor	Freq.	Rank	Weight Avg.	Mean	Rank	Weight Avg.	CSF	Freq.	Rank	Weight Avg.	Mean	Rank	Weight Avg.
(1)	(2)	(3)	(4)	(5)	(6)	(7)	(1)	(2)	(3)	(4)	(5)	(6)	(7)
Owner / User Satisfaction	48	1	0.065	4.667	1	0.049	Owner / User Satisfaction	42	1	0.061	4.571	1	0.048
Clear & Realistic Objectives	31	11	0.042	4.381	2	0.046	Effective Communication	16	23	0.023	4.405	3	0.047
Effective Communication	41	4	0.056	4.357	4	0.046	Competency & Capability of Project Team	38	3	0.055	4.405	3	0.047
Utility & Functionality	47	2	0.064	4.357	4	0.046	Clear & Realistic Objectives	33	8	0.048	4.405	3	0.047
Owner / User Participation	24	20	0.033	4.310	5	0.046	Collaboration of Project Delivery Team	33	8	0.048	4.381	5	0.046
Competency & Capability of Project Team	34	7	0.046	4.262	6	0.045	Alignment of Project Objectives	36	5	0.052	4.262	6	0.045
Owner's vision	32	9	0.043	4.167	7	0.044	Trust & Respect	17	22	0.024	4.214	7	0.045
Collaboration of Project Delivery Team	41	4	0.056	4.119	9	0.044	Utility & Functionality	35	6	0.050	4.167	8	0.044
Production of Specified Quality	42	3	0.057	4.119	9	0.044	Production of Specified Quality	40	2	0.058	4.119	9	0.044
Trust & Respect	28	16	0.038	4.119	9	0.044	Project Time Performance	31	11	0.045	4.071	10	0.043
Alignment of Project Objectives	25	18	0.034	4.071	12	0.043	Long-term Bldg. Success & Lifecycle Value	37	4	0.053	3.952	12	0.042
Long-term Bldg. Success & Lifecycle Value	41	4	0.056	4.071	12	0.043	Owner / User Participation	24	17	0.035	3.952	12	0.042
Project Cost Performance	31	11	0.042	3.833	14	0.041	Constructability	30	14	0.043	3.905	13	0.041
Project Time Performance	32	9	0.043	3.833	14	0.041	Project Cost Performance	34	7	0.049	3.881	14	0.041
Constructability	29	14	0.039	3.786	16	0.040	Dispute Avoidance & Resolution	19	20	0.027	3.857	15	0.041
Responsive Admin. & Decision Support	34	7	0.046	3.786	16	0.040	Responsive Admin. & Decision Support	31	11	0.045	3.833	16	0.041
Sustainability	29	14	0.039	3.738	17	0.040	Owner's vision	30	14	0.043	3.643	17	0.039
Evidence Based Design	26	17	0.035	3.690	18	0.039	Change Order Reduction	23	19	0.033	3.619	18	0.038
Professional Reputation & Image	30	13	0.041	3.595	19	0.038	Profit & Financial Objectives	25	16	0.036	3.548	19	0.038
Aesthetics	14	23	0.019	3.500	20	0.037	Sustainability	33	8	0.048	3.524	20	0.037
Design Innovation & Creativity	21	22	0.028	3.476	21	0.037	Professional Reputation & Image	31	11	0.045	3.500	21	0.037
Profit & Financial Objectives	25	18	0.034	3.452	22	0.037	Evidence Based Design	19	20	0.027	3.476	22	0.037
Dispute Avoidance & Resolution	9	24	0.012	3.405	23	0.036	Design Innovation & Creativity	24	17	0.035	3.357	23	0.036
Change Order Reduction	24	20	0.033	3.238	24	0.034	Aesthetics	13	24	0.019	3.333	24	0.035
Number (n)		51 42			Number (n)	(n) 51		42					
Kendall's coefficient of concordance (W) .141 .20		.207	7	Kendall's coefficient of concordance (W)	W) .095		.208						
Level of Significance	Level of Significance (W)			000)	Lavel of Significance	000			000			

Table 8.16. Comparison of rounds 1 & 2

Comparison of Rounds One and Two: Owner Self-Interest							Comparison of Rounds One an	nd Tw	vo: <u>0</u>	wner Pro	ject-In	terest		
		Round	1 #1		Round	#2			Round	1 #1		Round	#2	
CSF	Eroa	Donk	Weight	Moon	Donk	Weight	CSF	Eroa	Donk	Weight	Moon	Donk	Weight	
	rieq.	Ralik	Avg.	wican	Канк	Avg.		rieq.	Ralik	Avg.	Weatt	Kalik	Avg.	
(1)	(2)	(3)	(4)	(5)	(6)	(7)	(1)		(3)	(4)	(5)	(6)	(7)	
Clear & Realistic Objectives	12	9	0.048	4.538	1	0.049	Clear & Realistic Objectives	11	8	0.048	4.462	1	0.048	
Effective Communication	15	4	0.060	4.462	3	0.048	Competency & Capability of Project Team	13	3	0.057	4.385	2	0.047	
Owner / User Satisfaction	16	2	0.064	4.462	3	0.048	Collaboration of Project Delivery Team	12	6	0.052	4.308	4	0.046	
Utility & Functionality	17	1	0.068	4.462	3	0.048	Effective Communication	4	23	0.017	4.308	4	0.046	
Owner's vision	10	13	0.040	4.385	5	0.047	Alignment of Project Objectives	11	8	0.048	4.231	6	0.045	
Long-term Bldg. Success & Lifecycle Value	16	2	0.064	4.308	6	0.046	Production of Specified Quality	15	1	0.065	4.231	6	0.045	
Owner / User Participation	5	20	0.020	4.231	7	0.045	Owner / User Satisfaction	15	1	0.065	4.154	8	0.045	
Collaboration of Project Delivery Team	13	6	0.052	4.154	9	0.045	Trust & Respect	4	23	0.017	4.154	8	0.045	
Competency & Capability of Project Team	13	6	0.052	4.154	9	0.045	Utility & Functionality	11	8	0.048	4.154	8	0.045	
Production of Specified Quality	15	4	0.060	4.077	10	0.044	Constructability	10	13	0.043	4.077	11	0.044	
Alignment of Project Objectives	8	17	0.032	4.000	11	0.043	Long-term Bldg. Success & Lifecycle Value	12	6	0.052	4.077	11	0.044	
Sustainability	12	9	0.048	3.846	12	0.041	Project Cost Performance	11	8	0.048	4.000	12	0.043	
Constructability	10	13	0.040	3.769	14	0.041	Owner / User Participation	5	20	0.022	3.923	13	0.042	
Project Cost Performance	12	9	0.048	3.769	14	0.041	Profit & Financial Objectives	13	3	0.057	3.846	14	0.041	
Trust & Respect	9	16	0.036	3.769	14	0.041	Dispute Avoidance & Resolution	9	14	0.039	3.769	16	0.041	
Aesthetics	4	22	0.016	3.692	17	0.040	Project Time Performance	8	16	0.035	3.769	16	0.041	
Evidence Based Design	8	17	0.032	3.692	17	0.040	Professional Reputation & Image	8	16	0.035	3.462	18	0.037	
Project Time Performance	13	6	0.052	3.692	17	0.040	Sustainability	13	3	0.057	3.462	18	0.037	
Design Innovation & Creativity	4	22	0.016	3.385	20	0.036	Owner's vision	9	14	0.039	3.462	18	0.037	
Responsive Admin. & Decision Support	10	13	0.040	3.385	20	0.036	Aesthetics	7	19	0.030	3.385	21	0.036	
Change Order Reduction	7	19	0.028	3.308	21	0.036	Design Innovation & Creativity	8	16	0.035	3.385	21	0.036	
Dispute Avoidance & Resolution	4	22	0.016	3.154	23	0.034	Responsive Admin. & Decision Support	11	8	0.048	3.385	21	0.036	
Profit & Financial Objectives	11	12	0.044	3.154	23	0.034	Change Order Reduction	5	20	0.022	3.308	24	0.036	
Professional Reputation & Image	5	20	0.020	3.154	23	0.034	Evidence Based Design	5	20	0.022	3.308	24	0.036	
Number (n) 17 13					Number (n) 17				13					
Kendall's coefficient of concordance (W) .236 .300)	Kendall's coefficient of concordance (W) .154			4	.218					
Level of Significance .000				.000			Level of Significance .000				.000			

Table 8.17. Owner's comparison of rounds 1 & 2

Comparison of Rounds One and Two: Designer Self-Interest					terest		Comparison of Rounds One and Two: Designer Project-Interest						<u>t</u>
		Round	1#1		Round	#2			Round	#1		Round	#2
CSF	Freq.	Rank	Weight Avg.	Mean	Rank	Weight Avg.	CSF	Freq.	Rank	Weight Avg.	Mean	Rank	Weight Avg.
(1)	(2)	(3)	(4)	(5)	(6)	(7)	(1) (1)		(3)	(4)	(5)	(6)	(7)
Owner / User Satisfaction	10	1	0.066	4.889	1	0.049	Owner / User Satisfaction	8	3	0.056	4.889	1	0.050
Clear & Realistic Objectives	4	20	0.026	4.667	2	0.047	Collaboration of Project Delivery Team	7	7	0.049	4.778	2	0.049
Utility & Functionality	8	3	0.053	4.556	3	0.046	Competency & Capability of Project Team	7	7	0.049	4.667	3	0.048
Collaboration of Project Delivery Team	8	3	0.053	4.444	6	0.045	Clear & Realistic Objectives	6	12	0.042	4.556	4	0.047
Competency & Capability of Project Team	6	14	0.040	4.444	6	0.045	Effective Communication	6	12	0.042	4.444	5	0.046
Owner / User Participation	7	9	0.046	4.444	6	0.045	Alignment of Project Objectives	6	12	0.042	4.333	8	0.044
Owner's vision	7	9	0.046	4.444	6	0.045	Constructability	5	16	0.035	4.333	8	0.044
Constructability	7	9	0.046	4.333	9	0.044	Project Time Performance	8	3	0.056	4.333	8	0.044
Trust & Respect	3	22	0.020	4.333	9	0.044	Trust & Respect	4	19	0.028	4.333	8	0.044
Effective Communication	8	3	0.053	4.222	11	0.043	Change Order Reduction	4	19	0.028	4.222	11	0.043
Long-term Bldg. Success & Lifecycle Value	6	14	0.040	4.222	11	0.043	Responsive Admin. & Decision Support	7	7	0.049	4.222	11	0.043
Production of Specified Quality	8	3	0.053	4.222	11	0.043	Owner / User Participation	5	16	0.035	4.111	12	0.042
Design Innovation & Creativity	9	2	0.060	4.111	14	0.042	Project Cost Performance	8	3	0.056	4.000	14	0.041
Professional Reputation & Image	8	3	0.053	4.111	14	0.042	Utility & Functionality	9	1	0.063	4.000	14	0.041
Profit & Financial Objectives	6	14	0.040	4.000	16	0.040	Production of Specified Quality	9	1	0.063	3.889	15	0.040
Responsive Admin. & Decision Support	8	3	0.053	4.000	16	0.040	Dispute Avoidance & Resolution	3	23	0.021	3.778	16	0.039
Sustainability	6	14	0.040	4.000	16	0.040	Design Innovation & Creativity	5	16	0.035	3.667	19	0.038
Alignment of Project Objectives	5	18	0.033	3.889	19	0.039	Long-term Bldg. Success & Lifecycle Value	7	7	0.049	3.667	19	0.038
Project Time Performance	7	9	0.046	3.889	19	0.039	Profit & Financial Objectives	4	19	0.028	3.667	19	0.038
Aesthetics	2	23	0.013	3.778	21	0.038	Sustainability	6	12	0.042	3.667	19	0.038
Project Cost Performance	4	20	0.026	3.778	21	0.038	Evidence Based Design	4	19	0.028	3.556	22	0.036
Evidence Based Design	5	18	0.033	3.667	22	0.037	Professional Reputation & Image	7	7	0.049	3.556	22	0.036
Dispute Avoidance & Resolution	2	23	0.013	3.444	23	0.035	Owner's vision	8	3	0.056	3.556	22	0.036
Change Order Reduction	7	9	0.046	3.111	.111 24 0.031		Aesthetics	1	24	0.007	3.222	24	0.033
Number (n)		10			9		Number (n)	r (n) 10			9		
Kendall's coefficient of concordance (W) .186			.277		Kendall's coefficient of concordance (W)	(W) .149		.297					
Level of Significance .007			.000			Level of Significance		.060)	.000			

Table 8.18. Designer's comparison of round 1s & 2

Comparison of Rounds One and Two: Builder Self-Interest							Comparison of Rounds One and Two: Builder Project-Interes					terest	
		Round	1#1		Round	#2			Round	1 #1		Round	#2
CSF	Freq	Rank	Weight	Mean	Rank	Weight	CSF	Freq	Rank	Weight	Mean	Rank	Weight
	r req.	Runk	Avg.	mean	Runk	Avg.		rieq.	Runk	Avg.	wieum	Runk	Avg.
(1)	(2)	(3)	(4)	(5)	(6)	(7)	(1)		(3)	(4)	(5)	(6)	(7)
Owner / User Satisfaction	10	1	0.007	4.800	1	0.052	Owner / User Satisfaction	10	1	0.063	4.700	1	0.051
Alignment of Project Objectives	7	10	0.068	4.500	2	0.049	Alignment of Project Objectives	9	2	0.057	4.400	3	0.048
Effective Communication	6	14	0.095	4.400	4	0.048	Effective Communication	3	21	0.019	4.400	3	0.048
Trust & Respect	8	6	0.041	4.400	4	0.048	Collaboration of Project Delivery Team	7	14	0.044	4.300	6	0.047
Competency & Capability of Project Team	9	3	0.020	4.300	6	0.046	Production of Specified Quality	- 9	2	0.057	4.300	6	0.047
Project Time Performance	3	21	0.143	4.300	6	0.046	Project Time Performance	8	7	0.051	4.300	6	0.047
Project Cost Performance	7	10	0.068	4.200	8	0.045	Trust & Respect	3	21	0.019	4.300	6	0.047
Owner / User Participation	4	18	0.122	4.200	8	0.045	Competency & Capability of Project Team	8	7	0.051	4.200	8	0.046
Collaboration of Project Delivery Team	8	6	0.041	4.100	11	0.044	Responsive Admin. & Decision Support	6	15	0.038	4.100	10	0.045
Production of Specified Quality	7	10	0.068	4.100	11	0.044	Utility & Functionality	8	7	0.051	4.100	10	0.045
Clear & Realistic Objectives	8	6	0.041	4.100	11	0.044	Dispute Avoidance & Resolution	4	18	0.025	4.000	12	0.044
Professional Reputation & Image	8	6	0.041	4.100	11	0.044	Long-term Bldg. Success & Lifecycle Value	9	2	0.057	4.000	12	0.044
Responsive Admin. & Decision Support	6	14	0.095	4.000	13	0.043	Clear & Realistic Objectives	6	15	0.038	4.000	12	0.044
Dispute Avoidance & Resolution	1	23	0.156	3.900	15	0.042	Owner's vision	9	2	0.057	3.900	14	0.042
Owner's vision	9	3	0.020	3.900	15	0.042	Project Cost Performance	8	7	0.051	3.800	16	0.041
Constructability	9	3	0.020	3.700	17	0.040	Owner / User Participation	8	7	0.051	3.800	16	0.041
Long-term Bldg. Success & Lifecycle Value	7	10	0.068	3.700	17	0.040	Professional Reputation & Image	8	7	0.051	3.600	17	0.039
Utility & Functionality	10	1	0.007	3.600	18	0.039	Evidence Based Design	4	18	0.025	3.400	18	0.037
Profit & Financial Objectives	5	16	0.109	3.500	19	0.038	Change Order Reduction	5	17	0.032	3.300	20	0.036
Change Order Reduction	4	18	0.122	3.300	21	0.036	Profit & Financial Objectives	3	21	0.019	3.300	20	0.036
Evidence Based Design	2	22	0.150	3.300	21	0.036	Constructability	8	7	0.051	3.000	22	0.033
Sustainability	4	18	0.122	3.000	22	0.032	Sustainability	9	2	0.057	3.000	22	0.033
Design Innovation & Creativity	5	16	0.109	2.800	23	0.030	Design Innovation & Creativity	4	18	0.025	2,900	23	0.032
Aesthetics	0	24	0.163	2.400	24	0.026	Aesthetics	2	24	0.013	2.800	24	0.030
Number (n) 11 10				Number (n) 11				10					
Kendall's coefficient of concordance (W) .251 .410)	Kendall's coefficient of concordance (W) .193			.350					
Level of Significance .000 .000)	Level of Significance .001					.000		

Table 8.19. Builder's comparison between rounds 1 & 2

Comparison of Rounds One and Two: User Self-Inter					est		Comparison of Rounds One and Two: User Project-Interest							
		Round	1 #1		Round	#2			Round	#1]	Round	#2	
CSF	Freq.	Rank	Weight Avg.	Mean	Rank	Weight Avg.	CSF	Freq.	Rank	Weight Avg.	Mean	Rank	Weight Avg.	
(1)	(2)	(3)	(4)	(5)	(6)	(7)	(1)		(3)	(4)	(5)	(6)	(7)	
Utility & Functionality	12	1	0.063	4.800	1	0.051	Owner / User Satisfaction	9	4	0.056	4.700	1	0.049	
Owner / User Satisfaction	12	1	0.063	4.600	2	0.049	Clear & Realistic Objectives	10	1	0.062	4.600	2	0.048	
Owner / User Participation	8	11	0.042	4.400	3	0.047	Effective Communication	3	22	0.019	4.500	3	0.047	
Effective Communication	12	1	0.063	4.300	4	0.046	Competency & Capability of Project Team	10	1	0.062	4.400	5	0.046	
Competency & Capability of Project Team	6	17	0.031	4.200	6	0.045	Utility & Functionality	7	8	0.043	4.400	5	0.046	
Clear & Realistic Objectives	7	15	0.037	4.200	6	0.045	Collaboration of Project Delivery Team	7	8	0.043	4.200	7	0.044	
Aesthetics	8	11	0.042	4.100	9	0.044	Constructability	7	8	0.043	4.200	7	0.044	
Evidence Based Design	11	7	0.058	4.100	9	0.044	Alignment of Project Objectives	10	1	0.062	4.100	9	0.043	
Production of Specified Quality	12	1	0.063	4.100	9	0.044	Trust & Respect	6	16	0.037	4.100	9	0.043	
Sustainability	7	15	0.037	4.100	9	0.044	Long-term Bldg. Success & Lifecycle Value	9	4	0.056	4.000	12	0.042	
Trust & Respect	8	11	0.042	4.100	9	0.044	Owner / User Participation	6	16	0.037	4.000	12	0.042	
Long-term Bldg. Success & Lifecycle Value	12	1	0.063	4.000	12	0.043	Production of Specified Quality	7	8	0.043	4.000	12	0.042	
Alignment of Project Objectives	5	20	0.026	3.900	14	0.042	Sustainability	5	19	0.031	4.000	12	0.042	
Responsive Admin. & Decision Support	10	8	0.052	3.900	14	0.042	Project Time Performance	7	8	0.043	4.000	12	0.042	
Owner's vision	6	17	0.031	3.900	14	0.042	Aesthetics	3	22	0.019	3.900	16	0.041	
Collaboration of Project Delivery Team	12	1	0.063	3.800	16	0.041	Dispute Avoidance & Resolution	3	22	0.019	3.900	16	0.041	
Design Innovation & Creativity	3	21	0.016	3.700	17	0.040	Change Order Reduction	9	4	0.056	3.800	18	0.040	
Project Cost Performance	8	11	0.042	3.600	18	0.038	Responsive Admin. & Decision Support	7	8	0.043	3.800	18	0.040	
Project Time Performance	9	9	0.047	3.500	19	0.037	Project Cost Performance	7	8	0.043	3.700	20	0.039	
Constructability	3	21	0.016	3.400	20	0.036	Evidence Based Design	6	16	0.037	3.700	20	0.039	
Profit & Financial Objectives	3	21	0.016	3.300	21	0.035	Owner's vision	4	21	0.025	3.700	20	0.039	
Change Order Reduction	6	17	0.031	3.200	23	0.034	Design Innovation & Creativity	7	8	0.043	3.500	22	0.036	
Dispute Avoidance & Resolution	2	24	0.010	3.200	23	0.034	Professional Reputation & Image	8	7	0.049	3.400	23	0.035	
Professional Reputation & Image	9	9	0.047	3.200	23	0.034	Profit & Financial Objectives	5	19	0.031	3.300	24	0.034	
Number (n)		13		10			Number (n)) 13		10				
Kendall's coefficient of concordance (W)	Kendall's coefficient of concordance (W) .257 .2		.290		Kendall's coefficient of concordance (W)	٨) .104			.252					
Level of Significance .000				.000		Level of Significance .117					.000			

 Table 8.20. User's comparison between rounds 2 & 3

Compari	sson Summary	of Rd#1 &	Rd#1 Inte	er-class A	greement								
Group Test Statistics Self-Interest Project-Interest Round #1 Round #2 Round #1 Round #2													
Group	Test Statistics	Round #1	Round #2	Round #1	Round #2								
	N	51	42	51	42								
	Kendall's W ^a	.141	.207	.095	.208								
Overall	Chi-square	164.828	199.951	111.455	201.173								
	df	23	23	23	23								
	Asymp. Sig.	.000	.000	.000	.000								
	N	17	13	17	13								
	Kendall's W ^a	.236	.300	.154	.218								
Owner	Chi-square	92.351	89.721	60.224	65.315								
	df	23	23	23	23								
	Asymp. Sig.	.000	.000	.000	.000								
	N	10	9	10	9								
	Kendall's W ^a	.186	.277	.149	.297								
Designer	Chi-square	42.828	57.388	34.380	61.389								
	df	23	23	23	23								
	Asymp. Sig.	.007	.000	.060	.000								
	N	11	10	11	10								
	Kendall's W ^a	.251	.410	.193	.350								
Builder	Chi-square	63.573	94.384	48.811	80.479								
	df	23	23	23	23								
	Asymp. Sig.	.000	.000	.001	.000								
	N	13	10	13	10								
	Kendall's W ^a	.257	.290	.104	.252								
User	Chi-square	76.697	66.769	31.231	57.984								
	df	23	23	23	23								
	Asymp. Sig.	.000	.000	.117	.000								
a Kendall's	Coefficient of C	oncordance											

 Table 8.21. Summary of rounds 1 & 2 inter-rater agreement by stakeholder group

Figure 8.21 compares the overall inter-rater agreement scores between rounds 1 and 2. Overall, Kendall's W scores increased 47% from 0.147 to 0.207 for self-interest and nearly 120% from 0.095 to 0.208 for project-interest. The null hypothesis for the validation of inter-rater agreement is that the ratings of the factors are unrelated. The computed W's for the overall panel of experts were statistically significant at greater than the 0.001 level for both self-interest and project-interest. The null hypothesis that the overall panel of expert's ratings is unrelated to each other is rejected. The conclusion is that there is a significant amount of agreement among the respondents for both self-interest and project-interest, and no further Delphi rounds are necessary to evaluate relative importance of the critical success factors.



Figure 8.21. Overall comparison of rounds 1 & 2 inter-rater agreement

Figure 8.21. compares the overall inter-rater agreement scores between rounds 1 and 2. Overall, Kendall's W scores increased 47% from 0.147 to 0.207 for self-interest and nearly 120% from 0.095 to 0.208 for project-interest. The null hypothesis for the validation of inter-rater agreement is that the ratings of the factors are unrelated. The computed W's for the overall panel of experts were statistically significant greater than the 0.001 level for both self-interest and project-interest. The null hypothesis that the overall panel of expert's ratings is unrelated to each other is rejected. The conclusion is that there is a significant amount of agreement among the respondents for both self-interest and project-interest, and no further Delphi rounds are necessary to evaluate relative importance of the critical success factors. The Kendall's W scores are also compared by individual stakeholder groups to validate each sub-panel of experts. The owners and builders stakeholder groups, also have W scores for both self-interest and project-interest which are significant at the 0.001 level. The designer group achieved a 0.10 significance level in self-interest category, and a .001 significance level in the project-interest category. Therefore the null hypothesis may also be rejected for the owner, designer, and builder stakeholder groups, and are validated for a significant amount of agreement.



Figure 8.22. Round 1 & 2 inter-rater agreement comparison by stakeholder groups

The user stakeholder group achieved a .001 level of significance in the self-interest category in both the first and second round, and in the second round for project-interest. The user group did not achieve a 0.10 significance lever for project interest in round 1, but did achieve a 0.001 significance level in round 2 along with a 113% increase in the Kendall W score.

The null hypothesis is rejected for self-interest category, but cannot be rejected with 90% confidence for the project-interest category. With further review it is noted that the users only missed achieving a 0.10 significance in round 1 by 0.015. Taking into account (1) the significance of agreement in the self-interest category, (2) the 113% increase in inter-rater agreement score in round 2 at a 0.001 significance level, and (3) the threshold by which the user group missed achieving a 0.10 significance level for project-interest, it is reasonable to conclude the user stakeholder overall demonstrated significant inter-rater agreement.

8.11. Findings & Conclusions

The following is a summary of the findings and conclusions from the round 2 survey.

Summary of Round 2 Findings:

- The Delphi survey process is again successful with an 83% participation rate and a 95% completion rate. The number of overall, and individual stakeholder group, participants remains adequate for implementing the Delphi method.
- The Round 2 inter-rater agreement measured by Kendalls W indicated significant agreement overall, and within each stakeholder group at a .001 significance level.
- 3. The round 2 survey tested using Cronbach's alpha is reliable at the .001 significance level.

- Thirteen critical success factors were selected, based on relative importance, to be included in the round 3 survey.
- The relative importance of the factors was analyzed to identify the level of difference between the self-interest and project interest perspectives.
 Significance test determined that a significant difference between ten evaluated critical success factors exists at a 0.10 significance level.
- 6. Rounds 1 and 2 were inter-rater agreement tests were compared to determine if the Delphi process had resulted in an increased level of consensus amongst the panel(s) of experts. Overall, a significant increase in inter-rater agreement was measured the .001 level for both the self-interest and project interest categories.

Round 2 Conclusions:

It is concluded from the findings that there is significant agreement among the survey participants, and that the survey is reliable. It is concluded that significant difference do exist between self and project perspectives when rating relative importance of the critical success factors. In total the first two round of surveys support the overall dissertation by proving that there are aspects of project delivery which may be addressed in novel ways by taking into consideration differences that exist between project stakeholders within the project delivery team, and also differences within each individual stakeholder group in relation to self-interest and project-interest The theoretical framework of this dissertation is built around value-creation and how the stakeholder groups within a project delivery team collaborate to successfully achieve goals and objectives. Thus far, the surveys have proven the existence of the significant differences based on highly qualified expert opinion and statistical testing.

The surveys have provided data to study the relative importance each stakeholder group places on many important factors for value-creation throughout the project delivery process. Understanding the relative importance provides a framework from which to find ways to optimize the delivery process to maximize the benefit to all stakeholders, and thereby maximize the value-creation.

CHAPTER 9.

ROUND 3 REPORT



Figure 9.1. Round 3 data summary

9.1. Introduction

The round 3 departs from the goal of the first two rounds of survey. In survey rounds 1 and 2 the panel of experts selected critical success factors and assessed relative importance based each of their stakeholder group's perspectives of self-interest and project-interest. At the end of the second round the top 13 most important critical success factors have been identified. In round 3, these top 13 most important critical success factors will now be used for different purpose. The panel of experts is asked in round 3 to assess the effectiveness of three different project delivery methods: design-bid-build (DBB), design-build (DB), and Integrated Project Delivery (IPD). Figure 9.1 provides a summary of how each stakeholder group rated each project delivery method's effectiveness related to the thirteen factors. Close examination shows Integrated Project Delivery consistently rising to the top of the data distribution for relative effectiveness. This report will present the data and analysis in much greater detail below.

An important concept to consider for the round 3 survey is that evaluation is limited to the context of large, expensive, and complex projects similar to the IDBB-pilot projects the panels of experts are affiliated with. The scale of projects in scope, size, and complexity greatly influences not only relative value perception, but also realities of diminishing return on investments to enhance collaboration and/or increase project integration. Small, simple projects with modest budgets could certainly benefit from a better project delivery process, but the evaluation equation is vastly different when compared to a mega-project. By any definition three of the four IDBB pilot-projects ranging in cost between \$700 million and \$1.7 billion are mega-projects, and the fourth IDBB pilot-project at more than \$92 million is well within the lower bound for consideration as a mega-project. ¹¹¹ The panel of experts was established based on experience and participation is some of the largest and most complex projects in the military construction program.

¹¹¹ Li Zhai, Yanfei Xin, and Chaosheng Cheng, "Understanding the Value of Project Management From a Stakeholder's Perspective: Case Study of Mega-Project Management," *Project Management Journal* 40, no. 1 (March 2009): 99-100. Defines mega-projects as "having greater complexity, more stakeholders, and more extensive influences compared with normal projects." Zhai et al cite a range of reports defining mega-projects from a range of \$22 million to \$500 million, with an emphasis on complexity and stakeholder challenges.

9.1.1. Organization

The round 3 survey report is organized in the following order: Purpose, Administrative Information, Questions Asked, Data & Analysis, Findings, Summary and Conclusion. The format of the round 3 report differs from the round 1 report because the critical success factors categories have been consolidated into one comprehensive list. The presentation of the data will be divided into categories based on stakeholder and the three project delivery methods being assessed. The breakdown between self-interest and project-interest remains.

9.1.2. Purpose

The purposes of the round 3 survey:

- To validate each participant's familiarity with the three project delivery methods. Only data from participants validated for familiarity with the project delivery methods evaluated will be include.
- To evaluate the effectiveness of three project delivery methods (DBB, DB, and IPD) to successfully achieve the most important critical success factors selected by the panel of experts.
- 3. To identify how each stakeholder group differs in their assessments of delivery method efficacy to infer predispositions.
- 4. To infer which project delivery method(s) is most effective for achieving the select important critical success factors.

- 5. To infer implementability of the project delivery methods based on perceptions of relative effectiveness. For a project delivery method to create real value, it must be practical in the 'real-world'. A good idea that is not implementable has no real value in practice.
- 6. An unrelated purpose included in the round 3 survey was evaluation of one critical success factor (Evidence Based Design) which was erroneously omitted from the list of factors to be evaluated in round 2. (The results of the evaluation of Evidence Based Design are amended to the round 2 data set, and have been included in the report and analysis for round 2)

9.1.3. Administrative

The round 3 survey was published to the participant population on 26 April 2011. Weekly email reminders were automatically sent to remaining participants until the survey was closed at 21 days later 11:59 pm on 17 May 2011. The survey invitations were sent via email to a total of 51 participants. The round 3 survey invitation list was again limited to the respondents who participated in the round 1 survey. All participants were provided with the result of the round 2 survey for their consideration prior to starting the round 3 survey.

Participation Statistics:

- Invited: 51
- Started: 44
- Completed: 42
- Participation Rate: 82.3%
- Completion Rate: 95.5%

- Drop outs (after starting): 2
- Not validated for participation: 2
- Total validated participants: 40
- Average Time to Complete Survey: 16 minutes

Round 3 Questions

The following is a summary of the survey questions. A complete copy of the survey

instrument is located in Appendix E.

Round 3 - Question 1: Stakeholder Group

Which of the following categories do you best identify with? (Select only one): Design Team; Facility User or User's Representative; Construction Team; Owner or Owner's Representative.



Figure 9.2. Round 3 participation by stakeholder group

Round 3 - Question 2: Evidence Based Design

* The following question was included in the round 3 survey to correct an error of omission in the round 2 survey:

Rate the importance of Evidence Based Design (twice: from the selfperspective and again from the project-interest perspective) on a five-pointscale, where 1 = least important, and 5 = most important?

The resulting data from this question was amended to the round 2 data and is not reported here. Please note that Evidence Based Design did not score high enough in the amended round 2 data for inclusion in the round 3 survey. Evidence Based Design was included as the last factor evaluated in the round 3 survey, in the event that amended round 2 results may have placed Evidence Based Design in the top 25%. Failing to meet the to criteria for inclusion in the round 3 survey, the data collected for Evidence Based Design in round 3 has been removed from the round 3 data set.

9.2. Familiarity with Project Delivery Methods

Round 3 - Question 3: Effectiveness of Project Delivery Methods How familiar are you with the following project delivery methods: Design-Bid-Build (DBB), Design-Build (DB), and Integrated Project Delivery (IPD)? Rate each on the following five-point-scale: 1. Not at all familiar; 2. Slightly Familiar; 3. Somewhat familiar; 4. Moderately familiar; and 5. Very Familiar.

The original validation of survey participants in round 1 established expertise to evaluated critical success factors. The overall sample population was selected for this research project based on both expert qualification and recent and/or ongoing work associated with the military construction Integrated-Design-Bid-Build (IDBB) pilot-projects. The sample population represents the senior leadership of the largest most successful design firms and construction companies. In addition to having extensive Integrated Project Delivery experience through the IDBB pilot-projects, all the survey participants also have extensive experience with traditional design-bid-build, and design-build projects. To protect the survey participants anonymity the names of the companies are not listed in this document, but they are recognized as among the biggest and best in their respective industries.

All builder stakeholder participants are from companies that place highly on ENR's 2010 Top 400 contractors. All are within the top-25 general building contracting companies (including four in the top-ten), within the top-30 design-build contractors, and within the top-20 healthcare contractors (including 3 of the top-four). Additionally the builders represent the top-20 in construction management (CM) at risk which provides integration services to traditional project delivery methods (including 3 in the top-eleven).¹¹² The designers also place high on ENR's 2010 list of Top 500 design firms. The designers all come from top-25 firms including two of the top-ten healthcare designers. ¹¹³ The owners and users have been involved with many billions of dollars worth of MILCON construction using both traditional design-build and design-build. The U.S. Army Corps of Engineers was one of the early adopters for design-build and has completed hundreds of DB projects.

The validation from round 1 leaves no doubt to the absolute expertise and experience of the panel members across each stakeholder group (Owner, Designer, Builder, and User). To ensure consistency in evaluating the project delivery methods in round 3, participants are asked to rate their familiarity with each method. A minimum screening criteria for individuals and groups was established at 4.00 to ensure high familiarity.

 ¹¹² ENR, "The Top 400 Contractors Sourcebook," special issue of *ENR: Engineering News-Record* 265, no. 7 (13 September 2010) (New York: The McGraw-Hill Companies), 35-36.
 ¹¹³ ENR, "The Top 500 Design Firms Sourcebook," special issue of *ENR: Engineering News-Record* 265, no. 1 (05 July 2010) (New York: The McGraw-Hill Companies), 25-26.

Round# 3 Descriptive Statistics: Familiarity of Project Delivery Methods											
Project Delivery	Familiarity Scale	Ow	vner	Des	igner	Bui	ilder	U	ser	A	A11
Method	(1 - 5)	Freq	%	Freq	%	Freq	%	Freq	%	Freq	%
	1. Not at all	0	0%	0	0%	0	0%	0	0%	0	0%
	2. Slightly	0	0%	0	0%	0	0%	0	0%	0	0%
Design-Bid-Build	3. Somewhat	0	0%	0	0%	0	0%	3	30%	3	7%
(DBB)	4. Moderately	1	7%	0	0%	1	11%	1	10%	3	7%
	5. Very	13	93%	9	100%	8	89%	6	60%	36	86%
	Total	14	100%	9	100%	9	100%	10	100%	42	100%
	1. Not at all	0	0%	0	0%	0	0%	0	0%	0	0%
	2. Slightly	0	0%	0	0%	0	0%	0	0%	0	0%
Design Build (DP)	3. Somewhat	0	0%	1	11%	1	11%	4	40%	6	14%
Design-Bund (DB)	4. Moderately	3	21%	2	22%	1	11%	1	10%	7	17%
	5. Very	11	79%	6	67%	7	78%	5	50%	29	69%
	Total	14	100%	9	100%	9	100%	10	100%	42	100%
	1. Not at all	0	0%	0	0%	0	0%	1	10%	1	2%
	2. Slightly	0	0%	0	0%	0	0%	1	10%	1	2%
Integrated Project	3. Somewhat	0	0%	0	0%	0	0%	3	30%	3	7%
Delivery (IPD)	4. Moderately	3	21%	1	11%	1	11%	2	20%	7	17%
	5. Very	11	79%	8	89%	8	89%	3	30%	30	71%
	Total	14	100%	9	100%	9	100%	10	100%	42	100%

 Table 9.1. Round 3 familiarity frequency table

It was expected that the overwhelming majority of the participant would have a high degree of familiarity and experience with all three methods. However, it was also expected that there may be a limited number of participants who may have limited familiarity with the one or more of the delivery methods despite their involvement in the very same project delivery process, especially in the User stakeholder group. Users have a critical role in the project delivery process, but normally serve in advisory roles at best in the technical/contractual workings of the contractual relationships between other stakeholders (Owner, Design, and Builder).

	Descr	iptiv	e Stati	stics					
Stakeholder Group	Project Delivery Method	N	Range	Min.	Max.	Mean	Std. Error	Std. Deviation	Variance
	Design-Bid-Build (DBB)	42	2	3	5	4.79	.087	.565	.319
Overall	Design-Build (DB)	42	2	3	5	4.55	.114	.739	.546
	Integrated Project Delivery (IPD)	42	4	1	5	4.52	.141	.917	.841
	Design-Bid-Build (DBB)	14	1	4	5	4.93	.071	.267	.071
Owners	Design-Build (DB)	14	1	4	5	4.79	.114	.426	.181
	Integrated Project Delivery (IPD)	14	1	4	5	4.79	.114	.426	.181
	Design-Bid-Build (DBB)	9	0	5	5	5.00	.000	.000	.000
Designers	Design-Build (DB)	9	2	3	5	4.56	.242	.726	.528
	Integrated Project Delivery (IPD)	9	1	4	5	4.89	.111	.333	.111
	Design-Bid-Build (DBB)	9	1	4	5	4.89	.111	.333	.111
Builders	Design-Build (DB)	9	2	3	5	4.67	.236	.707	.500
	Integrated Project Delivery (IPD)	9	1	4	5	4.89	.111	.333	.111
	Design-Bid-Build (DBB)	10	2	3	5	4.30	.300	.949	.900
Users	Design-Build (DB)	10	2	3	5	4.10	.314	.994	.989
	Integrated Project Delivery (IPD)	10	4	1	5	3.50	.428	1.354	1.833

Table 9.2. Round 3 familiarity descriptive statistics

	Descriptiv	e Sta	tistics	(Revi	sed)				
Stakeholder Group	Project Delivery Method	N	Range	Min.	Max.	Mean	Std. Error	Std. Deviation	Variance
	Design-Bid-Build (DBB)	40	2	3	5	4.83	.079	.501	.251
Overall	Design-Build (DB)	40	2	3	5	4.63	.106	.667	.446
	Integrated Project Delivery (IPD)	40	2	3	5	4.68	.097	.616	.379
	Design-Bid-Build (DBB)	14	1	4	5	4.93	.071	.267	.071
Owners	Design-Build (DB)	14	1	4	5	4.79	.114	.426	.181
	Integrated Project Delivery (IPD)	14	1	4	5	4.79	.114	.426	.181
	Design-Bid-Build (DBB)	9	0	5	5	5.00	.000	.000	.000
Designers	Design-Build (DB)	9	2	3	5	4.56	.242	.726	.528
	Integrated Project Delivery (IPD)	9	1	4	5	4.89	.111	.333	.111
	Design-Bid-Build (DBB)	9	1	4	5	4.89	.111	.333	.111
Builders	Design-Build (DB)	9	2	3	5	4.67	.236	.707	.500
	Integrated Project Delivery (IPD)	9	1	4	5	4.89	.111	.333	.111
Users	Design-Bid-Build (DBB)	8	2	3	5	4.38	.324	.916	.839
	Design-Build (DB)	8	2	3	5	4.38	.324	.916	.839
	Integrated Project Delivery (IPD)	8	2	3	5	4.00	.327	.926	.857

 Table 9.3. Round 3 revised familiarity descriptive statistics (after screening process)

Two participants in the user stakeholder group responded with 'not familiar at all' and 'slightly familiar', causing the mean score for the user stakeholder group's familiarity of IPD to fall below 4.00 (moderately familiar). These two cases were screened out of the data set of round 3, and the descriptive statistics for the revised sample were calculated to ensure the minimum 4.0

mean score was achieved across the board. As a result of the screening process the overall panel of expert's means score increased as follows: DBB from 4.79 to 4.83, DB from 4.55 to 4.63; and IPD from 4.52 to 4.63.

Overall there was slightly more familiarity with the traditional DBB method, and equal familiarity with DB and IPD methods. The high degree of familiarity among all stakeholder groups for each of the three project delivery methods (not counting the two participants eliminated for not meeting the screening criteria) validates the appropriateness of the panel of experts to participate in round 3.



Figure 9.3. Round 3 revised participation (post-screening)

9.3. Efficacy of Project Delivery Methods

Round 3 - Question 4: Rate the Effectiveness of Project Delivery Methods The respondents were asked the question twice (again from each the selfinterest and project-interest perspectives) for each of the following critical success factors:

- 1. Owner & User Satisfaction
- 2. Clear & Realistic Objectives
- 3. Effective Communication
- 4. Collaboration of Project Delivery Team
- 5. Utility & Functionality
- 6. Trust & Respect
- 7. Alignment of Project Objectives
- 8. Owner & User Participation
- 9. Production of Specified Quality
- 10. Long-term Building Success (Lifecycle Value)
- 11. Project Time Performance
- 12. Owner's Vision
- 13. Constructability

Rate the effectiveness of each project delivery method to successfully achieve/enable/ensure/maintain/promote/maximize the listed critical success factors on the following one-to-five scale: 1. Least effective, 2. Less effective, 3. Effective, 4. More effective, and 5. Most effective.

Project Time Performance					
Maximize <u>PROJECT TIME PER</u> possible schedule duration	RFORMANC	<u>E</u> to acl	nieve tł	ne shor	test
Rate the effectiveness of each project de	elivery method	from your	STAKEHO	LDER	
	Least Effective	Less Effective	Effective	More Effective	Most Effective
Design-Bid-Build (DBB) *	0	0	0	۲	0
Design-Build (BD) *	0	0	0	0	0
Integrated Project Delivery (IPD) *	0	0	0	O	0
Rate the effectiveness of each project do	elivery method	from a <u>PR</u>	OJECT-CE	NTERED	
PERSPECTIVE	Least Effective	Less Effective	Effective	More Effective	Most Effective
Design-Bid-Build (DBB) *	Ø	O	0	O	0
Design-Build (BD) *	Ø	O	0	O	O
a the state is described	600	-	120	100	

Figure 9.4. Example of CSF question from web-survey

Group Critical Success Factor N Secure Mark Mode Mark Solution Values Ver. Range Min. Mark Mean Soluting Ver. Range Min. Mark Mean Soluting Ver. Range Min. Mark Mark	Design-Bid-Build (DBB) Relative Effectiveness																	
United Subcess Factor Nume Manness Marron SubDess Var. Range Man Max	Group Critical Success Factor			Self-Centered							Project-Centered							
UP Owner/Ler Sustriction 40 4 1 5 388 132 1132 1328 4 1 5 3.68 199 1239 1.55 Clear & Realistic Obscients 40 4 1 5 3.02 187 1181 1395 4 5 3.06 163 163 163 163 163 163 163 163 163 163 163 163 163 163 174 15 3.36 163<	Group	Cilical Success Factor	IN	Range	Min	Max	Mean	Std.Error	Std.Dev	Var.	Range	Min.	Max	Mean	Std.Error	Std.Dev	Var.	
Clear & Realistic Objectives 40 4 1 5 3.68 107 1.248 1.58 4 1 5 3.28 1.99 1.280 1.281 1.99 4 1 5 3.20 1.66 1.043 1. 1.11 1.99 4 1 5 3.20 1.66 1.043 1. 1.11		Owner/User Satisfaction	40	4	1	5	3.58	.182	1.152	1.328	4	1	5	3.40	.175	1.105	1.221	
Principle Communication 4 1 5 240 132 143 123 4 1 5 260 Collaboration of Project Team 40 4 1 5 323 44 1 5 333 132 933 44 1 5 333 144 125 133 144 125 145 1331 145 1331 145 1331 145 1331 145 1331 145 1331 145 1331 145 1331 145 1331 145 1331 145 1331 145 1331 145 1331 144 15 1332 154 1331 144 15 2331 156 1331 144 15 2331 156 1331 144 15 2331 156 1331 144 15 2331 156 1331 144 15 2331 156 1331 144 15 2331 1331 144		Clear & Realistic Objectives	40	4	1	5	3.68	.197	1.248	1.558	4	1	5	3.58	.199	1.259	1.584	
United number leam 40 4 1 5 245 152 658 121 4 1 5 235 146 9217 Timus ment of Objectives 40 4 1 5 235 147 11 5 235 147 9217 118 4 1 5 333 141 1 5 333 141 1 5 333 146 1081 1 733 116 118 1 5 333 161 1081 1 116 151 937 931 4 1 5 325 981 1 1 5 333 156 937 153 1 1 5 333 156 937 153 1 1 5 333 156 937 150 150 150 150 150 150 150 150 150 150 150 150 150 150 150 150 150		Effective Communication	40	4	1	5	2.80	.187	1.181	033	4	1	5	2.63	.165	1.043	1.087	
Tinst & Remeet: 40 4 1 5 273 148 933 371 4 1 5 213 1092 Alisment of Objectives 40 4 1 5 333 177 1118 123 1092 1 5 333 161 10971 Prodextion of Securite Quality 40 3 2 5 378 156 9871 4 1 5 333 166 9871 Time & Fortimance 40 4 1 5 335 154 975 951 4 1 5 236 1987 Constructuality 40 4 1 5 328 163 1064 4 1 5 363 144 1 5 343 144 1 5 343 144 1 5 343 144 1 5 343 144 1 5 343 144 1 5 343 144<		Utility & Functionality	40	4	1	5	3.45	.152	.959	.921	4	1	5	3.35	.146	.921	.849	
Vert Alignment of Obscietives 40 4 1 5 308 177 1118 1251 4 1 5 330 177 1118 1251 4 1 5 330 177 1118 1251 4 1 5 330 150 9977 4 1 5 330 155 9981 1 1 5 321 155 981 1 1 5 323 155 981 1 1 5 321 1 1 5 321 1 1 5 321 1 1 3 1 <th1< th=""> 1 1 <</th1<>	—	Trust & Respect	40	4	1	5	2.73	.148	.933	.871	4	1	5	2.75	.147	.927	.859	
Owner/User Participation 40 4 1 5 3 33 177 118 1251 4 1 5 3 30 161 1018 11 Production of Specific Quality 40 4 1 5 325 155 981 1 5 987 1 5 373 150 997 100 997 100 9981 1 1 5 20 981 1 1 5 100 947 100 100 100 100 100 100 100 100 100 100 100 100 100 100 11 100 100 110 110 110 100 110	A	Alignment of Objectives	40	4	1	5	3.08	.173	1.095	1.199	4	1	5	3.03	.174	1.097	1.204	
Production of Specified Quality 40 3 2 2 3 7.8 150 947 4 1 5 3 75 15 4 1 5 325 155 981 Time Performance 40 4 1 5 335 154 975 951 4 1 5 326 155 981 Constructability 40 4 1 5 210 151 955 911 4 5 3 26 122 150 947 Constructability 40 4 1 5 275 163 1032 1064 4 5 3 26 122 1152 142 1152 142 1150 1 Constructability 40 4 1 5 371 339 1227 1148 4 5 3 26 141 227 1150 1 Constructability 40 4 1 5 371 339 1227 1148 4 5 3 24 40 378 141 227 1 Constructability 40 4 1 5 371 339 1227 1604 4 1 5 343 374 1399 1 Clar & Realistic Objectives 14 4 1 5 371 339 1227 1604 4 1 5 324 266 1008 1 Clar & Realistic Objectives 14 4 1 5 307 245 917 841 4 1 5 3248 000 378 1414 2 Effective Communication 14 4 1 5 371 266 229 1109 4 1 5 324 268 1009 1 Clar & Respect 14 4 1 5 366 229 1009 1209 4 1 5 371 286 1009 1 Constructor of Specified Coulity 14 3 2 5 400 277 991 209 4 1 5 371 286 1009 1 Constructor Specified Coulity 14 3 2 5 400 279 9109 1209 4 1 5 370 334 1251 1 Time & Respect 1 4 4 1 5 366 294 1009 1209 4 1 5 370 343 1091 158 1 Time & Respect 1 4 3 2 5 371 2266 994 989 4 1 5 370 343 1091 158 1 Time & Respect 1 4 3 1 4 207 221 829 487 3 1 4 221 239 893 1 Constructability 14 3 2 5 400 238 860 23 991 4 1 5 370 343 209 1158 1 Time Performance 14 3 1 4 207 221 889 4 1 5 370 23 889 261 778 2 Clar & Realistic Objectives 9 2 3 5 471 204 994 1091 209 4 1 5 400 314 1177 1 Constructability 14 4 1 5 274 244 1099 1209 4 1 5 400 314 1177 1 Constructability 14 4 1 5 272 24 697 23 8 1 4 221 220 687 3 Clar & Realistic Objectives 9 2 3 5 473 328 866 750 3 1 4 221 220 687 7 Time Performance 14 3 1 4 207 3 5 474 370 100 1000 3 2 5 374 222 667 7 Time Realistic Objectives 9 2 2 4 4 302 126 44 371 140 1 2 3 5 422 778 833 10000 1 Constructability 9 2 3 5 441 222 2467 13 4 227 266 670 1 Long-term Linexce Value 9 2 3 5 441 222 278 886 12 3 5 378 222 667 1 Long-term Linexce Value 9 2 3 5 433 228 866 750 3 1 4 226 227 88 866 1 Constructability 9 2 3 5 4 43 229 278 833 1000 1 100 3 2 5 374 229 867 3 3 1 4 261 278 866 1 Constructability 1 9 3	,	Owner/User Participation	40	4	1	5	3.33	.177	1.118	1.251	4	1	5	3.30	.161	1.018	1.036	
Dimension Dimension <thdimension< th=""> <thdimension< th=""> <thd< td=""><td></td><td>Production of Specified Quality</td><td>40</td><td>3</td><td>2</td><td>5</td><td>3.78</td><td>.150</td><td>.947</td><td>.897</td><td>4</td><td>1</td><td>5</td><td>3.73</td><td>.156</td><td>.987</td><td>.974</td></thd<></thdimension<></thdimension<>		Production of Specified Quality	40	3	2	5	3.78	.150	.947	.897	4	1	5	3.73	.156	.987	.974	
Dimension 40 4 1 5 2.68 184 1.161 1331 4 1 5 2.60 182 1.75 Constructability Less Assistaction 14 3 2.75 3.33 2.86 1.032 1.064 4 1 5 3.64 3.41 1.277 1. Clark & Realistic Objectives 14 3 2.5 4.07 2.267 .997 .992 4 1 5 3.44 1.399 1. Collaboration of Protect Team 14 4 1 5 3.07 2.245 .917 8.41 4 1 5 3.42 1.089 1. 1.098 1. 1.098 1. 1.098 1. 1.098 1.1 1.099 1 1.5 3.31 2.99 1.41 1.5 3.44 1.5 3.43 3.09 1.158 1.168 1.1 1.099 1.41 1.52 2.79 2.81 1.051 1.1 1.59		Time Performance	40	4	1	5	2.10	.154	.975	.951	4	1	5	2.23	.155	.981	.902	
Constructability 40 4 1 5 2 75 163 1032 1064 4 1 5 2 83 156 984 1 2771 1 Constructability Ber Satisfaction 14 3 2 5 392 286 1072 1.148 4 1 5 343 374 1399 1 Clear & Realistic Objectives 14 3 2 5 307 245 997 995 4 1 5 340 378 1441 2 Effective Communication 14 4 1 5 307 245 997 995 4 1 5 343 374 1399 1 Cubiobaration of Project Team 14 4 1 5 307 245 997 995 4 1 5 343 294 1085 1 1085 1 Timus & Respect 14 4 1 5 364 246 9108 1016 4 1 5 332 286 1069 1 Timus & Respect 14 4 1 5 364 249 1082 1170 4 1 5 343 291 1085 1 1 Long-term / Lifecvele Value 14 3 2 5 307 245 991 209 4 1 5 343 291 1085 1 1 Long-term / Lifecvele Value 14 3 2 5 304 249 1082 1170 4 1 5 343 291 1085 1 1 Long-term / Lifecvele Value 14 3 2 5 307 246 94 889 4 1 5 343 409 1128 4 1 25 379 344 1251 1 Long-term / Lifecvele Value 14 3 2 5 371 266 991 209 4 1 5 343 309 1158 1 Time Performance 14 4 1 5 3 271 246 991 209 4 1 5 343 309 1158 1 Time Performance 14 4 1 5 3 271 246 991 209 4 1 5 343 201 1089 1 1 Constructability 14 3 2 5 371 266 991 209 4 1 5 343 201 7782 Constructability 14 4 1 5 277 244 914 835 4 1 5 279 261 975 Constructability 14 4 1 5 277 244 914 835 4 1 5 279 261 975 Constructability 9 2 3 5 441 300 1000 3 2 5 344 294 882 2 Constructability 9 2 3 5 441 300 288 60 750 3 1 4 221 222 667 1 Constructability 9 2 3 5 441 224 726 528 2 3 5 422 222 667 1 Constructability 9 2 3 5 441 261 782 711 23 5 400 236 707 Time & Realistic Objectives 9 2 2 4 300 228 707 500 2 3 5 422 222 667 1 Constructability 9 2 3 5 440 223 778 833 604 3 2 5 3 378 222 667 1 Constructability 9 2 3 5 440 223 778 833 604 3 2 5 3 378 222 667 1 Constructability 9 3 2 5 322 278 833 604 2 3 5 422 222 667 1 Constructability 9 3 2 4 3 328 288 666 750 3 1 4 2 47 72 833 33 31000 1 Constructability 9 3 2 4 3 322 278 833 604 2 3 5 422 222 667 1 Constructability 9 3 2 4 3 328 261 776 3 3 1 4 2 47 72 1130 1 Constructability 9 3 4 2 5 322 278 833 604 3 2 5 3 328 398 1167 1 130 1 3 1 4 300 333 1000 1 Constructability 9 3 1 4 5 263 371 108 1 4 3		Owner's Vision	40	4	1	5	3.68	.184	1.163	1.353	4	1	5	3.60	.182	1.150	1.323	
Owner/User Satisfaction 14 3 2 5 3/3 286 1/072 1/18 4 1 5 3/3 3/3 1/14 2 Effective Communication 14 4 1 5 3/1 3/3 1/267 1/604 4 1 5 3/3 3/3 1/3/3 <		Constructability	40	4	1	5	2.75	.163	1.032	1.064	4	1	5	2.83	.156	.984	.969	
Clear & Realistic Oblectives 14 3 2 5 407 267 997 997 994 4 1 5 3 400 378 1.414 2 Effective Communication 14 4 1 5 3.07 245 1.604 4 1 5 3.43 374 1.399 1 Collaboration of Protect Team 14 4 1 5 3.07 245 9.07 8.81 4 1 5 3 2.64 269 1.008 1 Utility & Eunstainability 14 3 2 5 3.64 269 1.008 1.016 4 1 5 3 2.79 2.86 1.006 1 Tust & Respect 14 4 1 5 3.64 269 1.008 1.016 4 1 5 3.279 2.86 1.006 1 Attemment of Objectives 14 4 1 5 3.86 294 1.098 1.120 4 1 5 3.371 2.86 1.008 1 Downer/User Participation of a section Online 14 3 2 5 3.86 294 1.098 1.120 4 1 5 3.771 2.86 1.008 1 Downer/User Participation of Section Online 14 3 2 5 4.00 257 9.061 9.23 4 1 5 3.771 2.86 1.009 1 Incode the Colline 14 3 2 5 4.00 257 9.061 9.23 4 1 5 3.771 2.86 1.009 1 Downer/User Participation of 14 4 1 5 2.71 2.246 9.944 9.89 4 1 5 4.00 3.14 1.177 1 Constructability Objectives 9 2 3 5 4.11 3.09 9.28 8.01 2 3 5 3.80 2.61 .975 2 Convert/User Statisfaction 9 2 3 5 4.41 3.291 .098 1.209 1 2.09 4 1 5 4.00 3.14 1.177 1 Constructability Objectives 9 2 3 5 4.56 2.42 .726 5.28 2 3 5 4.22 2.22 6.67 . Effective Communication 9 3 2 5 3.67 .333 1.000 1.000 3 2 5 5 3.44 294 8.82 2 Collaboration Of Protect Team 9 4 1 5 2.78 4.434 1.302 1.604 4 1 5 2.67 4.41 1.332 1 Utility & Functionality 9 2 3 5 4.410 2.21 2.83 6.866 .750 3 1 4 2.67 2.89 .866 .770 2 5 3.378 2.227 6.677 . Time Reminite Objectives 9 2 2 4 3.202 2.278 .833 .604 2 3 5 3.78 2.227 6.677 . Time Reminite Objectives 9 2 2 4 3.10 2.22 2.78 .833 .604 2 3 5 3.78 2.227 6.677 . Time Reminite Objectives 9 2 2 4 3.22 2.278 .833 .604 3 2 5 3.378 2.228 .667 .700 2 3 5 4.22 2.278 .833 .1000 1 . Owner/User Statisfaction 9 2 3 5 4.402 2.2278 .833 .604 2 1 3 2.427 .278 .538 .1007 1. Time Reminite Objective 9 3 4 1 5 3.11 .389 1.167 1.361 3 1 4 4 2.67 .288 .410 .771 . Time Remonite Objective 0 1 2 3 5 4.300 .287 .333 1.000 1. Owner/User Statisfaction 9 2 2 4 3.227 .278 .833 .604 2 3 5 3.78 2.227 .667 . Time Remonite Objective 9 3 1 4 2.56 .377 .276 .571 3 1 4 4 2.67 .333 1.000 1 . Downer/User St		Owner/User Satisfaction	14	3	2	5	3.93	.286	1.072	1.148	4	1	5	3.64	.341	1.277	1.632	
Filective Communication 14 4 4 1 5 3.71 339 1.26/11604 4 1 5 3.43 374 1.399 1.088 1. Utility & Functionality 14 3 2 5 3.64 2.69 1.008 1.016 4 1 5 3.29 2.86 1.069 1. Tots: K Respect 14 4 1 5 2.26 2.94 1.099 1.094 4 1 5 3.29 2.86 1.069 1. Communication 14 4 1 5 3.64 2.89 1.089 1.170 4 1 5 3.43 2.91 1.088 1. Utility & Functionality 14 3 2 5 4.00 2.57 961 1.923 4 1 5 3.79 2.81 1.069 1. Commercing Lifeccycle Value 14 1 5 3.64 2.89 1.082 1.170 4 1 5 3.71 2.86 1.069 1. Evolution of Specified Onality 14 3 2 5 4.00 2.57 961 923 4 1 5 3.79 2.81 1.069 1. Tots: K 4.83 3.09 1.158 1. Time Performance 14 4 1 5 4.14 2.94 2.92 2.21 8.89 4.1 5 3.79 2.34 1.251 1.158 1. Time Performance 14 4 1 5 4.14 2.94 2.99 4.99 4.1 5 4.03 3.09 1.158 1. Time Performance 14 4 1 5 4.14 2.94 1.099 1.209 4 1 5 4.03 3.04 1.1571 1. Constructability 14 4 1 5 4.14 2.94 1.099 1.209 4 1 5 4.03 3.04 1.1571 1. Constructability 14 4 1 5 4.41 2.94 2.92 8.861 2 3 5 4.22 2.22 6.667 1. Fflective Communication 9 2 3 5 4.71 2.84 0.94 8.861 2 3 5 4.12 2.22 2.667 1. Tots: K Realistic Objectives 9 1 2 3 5 4.71 3.00 2.88 8.61 2 3 5 4.10 2.23 6.778 2. Collaboration of Proize Team 9 4 1 5 2.78 4.34 1.1302 1.694 4 1 5 2.67 4.41 1.323 1. Utility & Functionality 9 2 3 5 4.73 2.278 8.433 6.94 3 2 5 3.67 2.36 7.777 5.00 2.3 5 4.22 2.22 6.667 1. Tots: K Respect 9 2 2 2 4 3.22 2.278 8.83 6.94 3 2 5 3.67 2.36 7.707 1. Tots: K Respect 9 2 2 2 4 3.22 2.278 8.83 6.94 3 2 5 3.67 2.36 7.707 1. Time Performance 9 4 1 5 3.21 2.38 1.44 1.302 1.694 4 1 5 2.44 3.77 1.130 1. Owner's Vision 9 2 3 5 4.42 2.22 7.78 8.33 6.94 3 2 5 3.67 2.36 7.707 1. Time Performance 9 4 1 5 3.21 2.38 1.44 1.021 1.694 4 1 5 2.44 3.77 1.130 1. Owner's Vision 9 2 2 3 5 4.43 2.22 7.78 8.33 6.94 3 2 5 3.26 7.28 8.33 1.000 1.00 2.3 5 5.78 2.22 7.78 8.33 1.000 1.00 2.3 5 5.73 8.222 7.78 8.33 1.000 1.00 2.3 5 5.73 8.222 7.78 8.33 1.000 1.00 2.1 3 4 2.21 2.278 8.33 1.000 1.00 2.1 5 5.00 2.1 3 2.22 7.78 8.33 1.000 1.00 2.1 5 5.00 2.1 3 2.22 7.78 8.33 1.000 1.00 2.1 5 5.00 2.1 3 2.22 7.78 8.33 1.000 1.00 3.1 4 2.60		Clear & Realistic Objectives	14	3	2	5	4.07	.267	.997	.995	4	1	5	4.00	.378	1.414	2.000	
Constantiation of Profeed ream 147 4 1 5 3 200 243 9 210 241 4 1 5 3 200 200 1006 1 1069 1 1006 1 1069 1 1076 4 1 5 3 20 286 1069 1 1076 4 1 5 3 20 286 1069 1 1076 4 1 5 3 20 286 1069 1 1076 4 1 5 3 20 286 1069 1 1076 4 1 5 3 20 286 1069 1 1076 4 1 5 3 20 286 1069 1 1076 4 1 5 3 20 286 1069 1 1076 4 1 5 3 20 286 1069 1 1076 4 1 5 3 20 286 1069 1 1076 4 1 5 3 20 286 1069 1 1089 1 1 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0		Effective Communication	14	4	1	5	3.71	.339	1.267	241	4	1	5	3.43	.374	1.399	1.956	
Dep Trust & Respect 14 4 1 5 2.86 2.94 1.099 1.209 4 1 5 2.79 2.81 1.081 1. Owner/User Particitation 14 4 1 5 3.64 2.89 1.099 1.209 4 1 5 3.71 2.86 1.094 1 5 3.71 2.86 9.94 1.98 1.1 1.069 1. Forduction of Specific Quality 14 3 2 5 3.00 2.77 9.61 9.92 4 1 5 3.43 3.09 1.158 1. Time Performance 14 4 1 5 4.10 1.994 1 5 4.20 3.83 3.89 2.61 7.82 Constructability 14 4 1 5 2.78 2.44 9.14 8.35 4 1 5 2.79 2.61 7.82 2 3 5 4.21 2.22 2.66 7.00 <td>ŝ</td> <td>Utility & Functionality</td> <td>14</td> <td>3</td> <td>2</td> <td>5</td> <td>3.64</td> <td>269</td> <td>1 008</td> <td>1 016</td> <td>4</td> <td>1</td> <td>5</td> <td>3 29</td> <td>286</td> <td>1.008</td> <td>1 143</td>	ŝ	Utility & Functionality	14	3	2	5	3.64	269	1 008	1 016	4	1	5	3 29	286	1.008	1 143	
Himment of Objectives 14 4 1 5 3.46 229 1.082 1.170 4 1 5 3.43 291 1.089 1. Owner/User Particination 14 4 1 5 3.86 2.94 1.092 4 1 5 3.71 2.86 1.089 1. 5 3.71 2.86 1.085 3.71 2.86 1.087 3 1.4 2.21 2.91 8.83 1.099 1.209 4 1 5 3.71 2.83 6.87 3 1.4 2.21 2.94 1.099 1.209 4 1 5 2.07 2.61 9.75 Constructability 14 4 1 5 2.71 2.44 9.44 3.5 4.11 2.01 3.5 4.835 4 1.5 2.70 2.61 9.75 Clear & Realistic Obicetives 2 3 5 3.67 3.33 1.0001 1.000 3 2<	er	Trust & Respect	14	4	1	5	2.86	.294	1.099	1.209	4	1	5	2.79	.281	1.051	1.104	
Owner/User Participation 14 4 1 5 3.86 2.94 1.099 1.209 4 1 5 3.71 2.86 1.069 1. Imme Ferformance 14 3 2 5 3.71 2.66 9.94 9.89 4 1 5 3.70 3.31 1.231 1.158 1. Timme Ferformance 14 4 1 5 4.11 2.90 8.87 3 1 4 2.21 2.29 8.93 . Constructability 14 4 1 5 2.17 2.24 1.099 1.2 3 5 4.20 2.3 5 4.20 2.3 5 4.20 2.21 6.67 . 2.85 2.67 4.41 1.5 2.78 4.34 1.302 1.694 4 1.5 2.67 4.41 1.323 1. 1.11 1.33 1.4000 1.3 3.4 2.24 4.300 2.89	vn	Alignment of Objectives	14	4	1	5	3.64	.289	1.082	1.170	4	1	5	3.43	.291	1.089	1.187	
Preduction of Specified Quality 14 3 2 5 4.00 257 961 923 4 1 5 3.79 3.34 1.251 1.158 1.1 Ime Performance 14 3 1 4 2.07 221 829 687 3 1 4 2.21 239 893 Owner/User Stainsfaction 9 2 3 5 4.10 9.928 861 2 3 5 4.20 2.23 5 4.20 2.23 5 4.20 2.22 2.61 .7782 . Colarboration of Proiest Team 9 2 3 5 4.41 1.323 1.1 1.000 1.004 4 1 5 2.67 .411 3.20 2.3 5 4.31 3.20 1.333 1.000 1.000 3 2 5 3.33 3.33 1.000 1.000	2	Owner/User Participation	14	4	1	5	3.86	.294	1.099	1.209	4	1	5	3.71	.286	1.069	1.143	
Long-term / Lifecvel Value 14 3 2 5 3.71 2.00 394 345 4 1 5 3.43 .309 1.138 1. Time Performance 14 4 1 5 4.14 2.94 1.099 1.209 4 1 5 4.00 314 1.177 1. Constructability 14 4 1 5 2.71 2.44 9.94 8.35 4 1 5 2.79 2.61 975 1. Constructability 14 4 1 5 2.71 2.44 9.94 8.35 4 1 5 2.79 2.61 975 1. Clear & Realistic Objectives 9 2 3 5 4.56 2.42 7.26 528 2 3 5 3.48 2.29 6.67 . Effective Communication 9 3 2 5 3.67 3.33 1.000 1.000 3 2 5 3.44 2.94 8.82 . Collaboration of Project Team 9 4 1 5 2.71 2.41 0.94 4 1 5 2.67 4.41 1.323 1. Utility & Functionality 9 2 3 5 4.11 2.01 7.82 6.11 2 3 5 4.00 2.36 7.07 7. Trust & Respect 9 2 2 2 4 3.00 2.89 8.66 7.50 3 1 4 2.67 2.44 1.1233 1.000 1. Utility & Functionality 9 2 3 5 4.00 2.36 7.07 5.00 2 3 5 3.78 2.22 6.67 . Long-term / Lifecvel Value 9 2 3 5 4.40 2.36 7.07 5.00 2 3 5 3.78 2.22 6.67 . Long-term / Lifecvel Value 9 2 3 5 4.40 2.36 7.07 5.00 2 3 5 3.78 2.22 6.67 . Long-term / Lifecvel Value 9 2 3 5 4.40 2.36 7.07 5.00 2 3 5 3.78 2.22 6.67 . Long-term / Lifecvel Value 9 2 3 5 4.40 2.36 7.07 5.00 2 3 5 3.78 2.22 6.67 . Long-term / Lifecvel Value 9 2 3 5 4.40 2.36 7.07 5.00 2 3 5 3.78 2.22 6.67 . Long-term / Lifecvel Value 9 2 3 5 4.44 1.302 1.694 4 1 5 2.44 3.77 1.130 1. Constructability 9 3 2 5 3.22 2.78 8.33 6.94 3 2 5 3.22 2.78 8.33 . Collaboration of Project Team 9 4 1 5 2.311 3.89 1.167 1.361 4 1 5 2.44 3.77 1.130 1. Constructability 9 3 1 4 2.89 2.61 7.82 6.11 3 1 4 2.78 3.64 1.093 1. Collaboration of Project Team 9 1 2 3 3 2.78 1.47 4.41 1.94 2 1 3 2.56 2.42 7.726 . Hittik & Realistic Objectives 9 3 1 4 2.67 2.89 8.66 7.50 2 1 3 2.44 3.22 2.78 8.33 1.000 1. Production of Specified Oullity 9 2 1 3 2.22 2.78 8.63 6.70 2.51 3 4 2.24 3.77 1.130 1. Production of Specified Oullity 9 2 1 3 2.22 2.66 7.76 1.3 1 4 2.67 3.33 1.000 1. Collaboration of Project Team 9 1 2 4 3.20 2.22 6.67 7.44 2 2 4 3.22 2.278 8.33 1.000 1. Production of Specified Oullity 9 2 1 3 2.00 2.89 8.66 7.50 2 1 3 2.22 2.78 8.33 1.000 1. Production of Specified Oullity 9 2 2 4 3.21 2.22 2.	\cup	Production of Specified Quality	14	3	2	5	4.00	.257	.961	.923	4	1	5	3.79	.334	1.251	1.566	
Inter Ferrinance Image 1 Image 1 <thimage 1<="" th=""> Image 1 <thimage 1<="" th=""></thimage></thimage>		Long-term / Lifecycle Value	14	3	2	5 1	2.07	.266	.994	.989	4	1	3	2 21	.309	1.158	707	
Constructability 14 4 1 5 2.71 2.44 9.14 835 4 1 5 2.79 2.61 9.782 Clear & Realistic Objectives 9 2 3 5 4.51 2.309 9.28 8.61 2 3 5 4.22 2.22 6.671 Clear & Realistic Objectives 9 2 3 5 4.52 2.42 7.26 5.28 2 3 5 4.22 2.22 6.671 1.323.11 1.1323.11		Owner's Vision	14	4	1	5	4.14	.294	1.099	1.209	4	1	5	4.00	.314	1.177	1.385	
Owner/User Satisfaction 9 2 3 5 4.11 309 928 861 2 3 5 3.89 2.61 782 Clear & Realistic Obiectives 9 2 3 5 4.56 2.42 7.26 5.18 2 3 5 4.22 2.66 7.67 Collaboration of Project Team 9 4 1 5 2.78 4.34 1.302 1.694 4 1 5 2.67 4.41 1.323 1.000 1.00 2.6 3 5 4.11 2.21 2.78 4.34 1.302 1.694 4 1 5 2.67 4.41 1.323 1.00 1.23 3 3.40 2.67 2.88 8.66 7.50 3 1 4 2.67 2.88 8.66 7.50 2 3 5 3.78 2.22 6.67 7.07 7.07 7.07 7.07 7.07 7.07 7.07 7.07 7.07		Constructability	14	4	1	5	2.71	.244	.914	.835	4	1	5	2.79	.261	.975	.951	
Clear & Realistic Objectives 9 2 3 5 4.22 2.22 6.67 Effective Communication 9 3 2 5 3.67 3.33 1.000 1.000 3 2 5 3.44 2.44 1.322 1.1 Collaboration of Project Team 9 4 1 5 2.78 4.34 1.302 1.694 4 1 5 2.67 4.411 1.323 1.1 Utility & Functionality 9 2 2 4 3.00 2.89 8.66 .750 3 3.1 4.267 2.89 8.866 .750 2 3 3.33 1.000 1.00 1.00 1.00 1.00 2.78 8.33 .694 3 2 5 3.37 2.22 .667 . .707 .500 2 3 5 4.22 .728 .833 .694 3 2 5 3.37 1.222 .667 .707 .500		Owner/User Satisfaction	9	2	3	5	4.11	.309	.928	.861	2	3	5	3.89	.261	.782	.611	
Effective Communication 9 3 2 5 3.67 .333 1.0001 3 2 5 3.44 .294 .882 . Collaboration of Protect Team 9 4 1 5 2.78 .434 1.302 1.1694 4 1 5 2.67 .411 1.323 1.123 5 4.00 .236 .707 .503 1 4 2.67 .249 .866 .703 1 4 2.67 .249 .866 .703 1 4 2.67 .249 .866 .703 1 4 2.67 .241 .333 .331 .0001 .33 .33 .0001 .33 .33 .0001 .33 .33 .000 .23 .5 .400 .236 .707 .500 .23 .5 .378 .228 .866 .7500 .23 .5 .422 .227 .667 .707 .700 .23 .5 .244 .371		Clear & Realistic Objectives	9	2	3	5	4.56	.242	.726	.528	2	3	5	4.22	.222	.667	.444	
Statistic Collaboration of Project Team 9 4 1 5 2.78 4.34 1.302 1.1 1 3 2.06 1.44 1 3 2.06 1.44 1 3 2.06 1.44 1.323 1 4 2.67 2.29 8.66 7.50 3 1 4 2.67 2.29 8.66 7.50 3 1 4 2.67 2.29 8.66 7.50 3 1 4 2.67 2.29 8.66 7.50 3 1 4 2.67 2.29 8.66 7.50 2 3 5 3.33 3.33 1.000 1.000 2 3 5 4.32 2.78 8.33 6.94 2 3 5 4.22 2.28 8.67 2.23 5 4.22 2.28 3.5 4.22 2.28 3.33 1.000 1.000 1.000 1.000 1.000 1.000 1.000 1.000 1.000 1.000		Effective Communication	9	3	2	5	3.67	.333	1.000	1.000	3	2	5	3.44	.294	.882	.778	
D Dimity & Functionality D Z <thz< th=""> Z Z Z</thz<>	IS	Litility & Europionality	9	2	1	5	2.78	.434	1.302	611	4	3	5	2.67	.441	1.323	1./50	
Sty Alignment of Objectives 9 2 2 4 3.22 278 8.33 6.94 3 2 5 3.33 3.33 1.000 1. Owner/User Participation 9 2 3 5 4.00 2.36 7.07 5.00 2 3 5 4.22 2.667 2.3 5 4.33 2.29 .	ne	Trust & Respect	9	2	2	4	3.00	.289	.866	.750	3	1	4	2.67	.289	.866	.750	
Sign Owner/User Participation 9 2 3 5 4.00 2.36 7.07 5.00 2 3 5 3.78 2.22 667 . Production of Specified Quality 9 2 3 5 4.33 .289 .866 .750 2 3 5 4.22 .278 .833 	<u>1</u> <u></u> .	Alignment of Objectives	9	2	2	4	3.22	.278	.833	.694	3	2	5	3.33	.333	1.000	1.000	
Production of Specified Quality 9 2 3 5 4.33 2.289 8.866 7.50 2 3 5 4.22 2.78 8.833 Time Performance 9 4 1 5 2.22 4.44 1.302 1.694 4 1 5 2.44 3.77 1.1301 Owner's Vision 9 2 3 5 4.44 2.42 726 5.28 2 3 5 4.22 2.22 667 Constructability 9 3 2 5 3.22 2.78 8.83 694 3 2 5 3.22 2.78 633 60 3 4 3.11 63 1 4 2.78 61 1.361 4 1 5 2.89 66 7 61 3 2.44 61 62 62 62 726 61 3 1.4 2.44 726	es	Owner/User Participation	9	2	3	5	4.00	.236	.707	.500	2	3	5	3.78	.222	.667	.444	
Long-term / Lifecvcle Value 9 2 3 5 3.78 .2.78 .8.33 .6.94 2 3 5 3.6.0 .2.36 .7.01 Time Performance 9 4 1 5 2.24 3.11 1.302 1.694 4 1 5 2.44 3.77 1.130 1. Owner's Vision 9 2 3 5 4.44 2.42 .726 .528 2 3 5 4.22 .222 .667 . Constructability 9 3 2 5 3.22 .78 .167 1.361 3 1 4 2.78 .833 .694 3 2.44 .304 1.67 1.361 4 1 5 3.89 1.167 1.361 4 1 5 2.89 .864 .707 .133 4 3.11 .111 .333 .266 .700 2 1 3 2.44 .377	Ō	Production of Specified Quality	9	2	3	5	4.33	.289	.866	.750	2	3	5	4.22	.278	.833	.694	
Inter Performance 9 4 1 3 2.22 4.342 7.302 1.394 4 1 3 2.44 3.77 1.1301 Owner's Vision 9 3 2 5 3.22 2.78 8.33 6.94 3 2 5 3.22 2.278 8.33 6.94 3 2 5 3.22 2.278 8.33 Owner/User Satisfaction 9 4 1 5 3.11 3.89 1.167 1.361 3 1 4 2.78 .364 1.093 1. Clear & Realistic Objectives 9 4 1 5 3.11 .389 1.167 1.361 4 1 5 2.89 .366 1.71 1.111 .333 . . 1.14 2.30 .266 .727 .260 1 3 2.44 .377 1.130 1.278 3 1 4 2.44 .377 1.130 1.278 3		Long-term / Lifecycle Value	9	2	3	5	3.78	.278	.833	.694	2	3	5	3.6/	.236	./0/	.500	
Structubility 9 3 2 5 3.11 1.12 <th1.1< td=""><td></td><td>Owner's Vision</td><td>9</td><td>2</td><td>3</td><td>5</td><td>4 4 4 4</td><td>.434</td><td>726</td><td>528</td><td>2</td><td>3</td><td>5</td><td>4 22</td><td>222</td><td>667</td><td>444</td></th1.1<>		Owner's Vision	9	2	3	5	4 4 4 4	.434	726	528	2	3	5	4 22	222	667	444	
Owner/User Satisfaction 9 4 1 5 3.11 .389 1.167 1.361 3 1 4 2.78 .364 1.093 1. Clear & Realistic Objectives 9 4 1 5 3.11 .389 1.167 1.361 4 1 5 2.89 .389 1.167 1.361 4 1 5 2.89 .389 1.167 1.361 4 1 1 1 3 3 4 3.11 .111 .333 . Collaboration of Project Team 9 1 2 3 2.78 .147 .441 .194 2 1 3 2.56 2.42 .726 . Utility & Functionality 9 3 1 4 2.67 .289 .866 .750 2 1 3 2.24 .726 Owner/User Participation 9 3 1 4 2.67 .289 .866 .750 3 <td></td> <td>Constructability</td> <td>9</td> <td>3</td> <td>2</td> <td>5</td> <td>3.22</td> <td>.278</td> <td>.833</td> <td>.694</td> <td>3</td> <td>2</td> <td>5</td> <td>3.22</td> <td>.278</td> <td>.833</td> <td>.694</td>		Constructability	9	3	2	5	3.22	.278	.833	.694	3	2	5	3.22	.278	.833	.694	
Clear & Realistic Objectives 9 4 1 5 3.11 .389 1.167 1.361 4 1 5 2.89 .389 1.167 1. Effective Communication 9 2 2 4 3.00 .167 .500 .250 1 3 4 3.11 .111 .333 . Collaboration of Project Team 9 1 2 3 2.78 .147 .441 .194 2 1 3 2.56 .242 .726 . Trust & Respect 9 2 1 3 2.33 .236 .707 .500 2 1 3 2.44 .242 .726 . Alianment of Objectives 9 3 1 4 2.67 .289 .866 .750 3 1 4 2.07 .333 1.000 1. Owner/User Participation 9 3 1 4 2.07 .833 .694		Owner/User Satisfaction	9	4	1	5	3.11	.389	1.167	1.361	3	1	4	2.78	.364	1.093	1.194	
Effective Communication 9 2 2 4 3.00 .167 .500 .250 1 3 4 3.11 .111 .333 . Collaboration of Project Team 9 1 2 3 2.78 .147 .441 .194 2 1 3 2.56 .242 .726 . Utility & Functionality 9 3 1 4 2.89 .261 .782 .611 3 1 4 2.44 .242 .726 . Alignment of Objectives 9 3 1 4 2.67 .289 .866 .750 3 1 4 2.44 .242 .776 . Owner/User Participation 9 3 1 4 2.67 .289 .866 .750 3 1 4 2.67 .333 1.000 1.00 1.00 1 4 2.67 .333 1.000 1.00 1 4 2.67 .333 1.000 1.00 1 4 2.67 .333 1.000 1.00 <td></td> <td>Clear & Realistic Objectives</td> <td>9</td> <td>4</td> <td>1</td> <td>5</td> <td>3.11</td> <td>.389</td> <td>1.167</td> <td>1.361</td> <td>4</td> <td>1</td> <td>5</td> <td>2.89</td> <td>.389</td> <td>1.167</td> <td>1.361</td>		Clear & Realistic Objectives	9	4	1	5	3.11	.389	1.167	1.361	4	1	5	2.89	.389	1.167	1.361	
Sign Collaboration of Protect leam 9 1 2 3 2.78 1.44 1.441 1.44 2.50 1.289 .866 1.301 1.44 2.301 2.33 2.36 7.707 5.001 2 1 4 2.44 .377 1.130 1. Alignment of Objectives 9 3 1 4 2.67 .289 .866 .750 3 1 4 2.44 .377 1.130 1. 4 2.44 .377 1.130 1. Owner/User Participation 9 2 2 4 3.22 .222 .667 .444 2 2 4 3.00 .333 1.000 1.33 1.000		Effective Communication	9	2	2	4	3.00	.167	.500	.250	1	3	4	3.11	.111	.333	.111	
Trust & Respect 9 2 1 3 2.33 2.36 .707 .501 2 1 3 2.44 .242 .726 . Alignment of Objectives 9 3 1 4 2.56 .377 1.130 1.278 3 1 4 2.44 .377 1.130 1. Owner/User Participation 9 3 1 4 2.67 .289 .866 .750 3 1 4 2.67 .333 1.000 1. Production of Specified Quality 9 2 2 4 3.12 .222 .667 .444 2 2 4 3.22 .222 .667 .444 2 2 4 3.00 .333 1.000 1. .00 .333 1.000 1.000 3 1 4 3.00 .333 1.000 1.001 .000	S	Litility & Europionality	9	3	2	3 4	2.78	.147	.441	.194	2	1	3 4	2.56	.242	./26	.528	
Migment of Objectives 9 3 1 4 2.56 .377 1.130 1.278 3 1 4 2.44 .377 1.130 1. Owner/User Participation 9 3 1 4 2.67 .289 .866 .750 3 1 4 2.67 .333 1.000 1. Production of Specified Quality 9 2 2 4 3.22 .222 .667 .444 2 2 4 3.00 .333 1.000 1. Ime Performance 9 2 1 3 2.22 .278 .833 .694 2 1 3 2.22 .278 .833 .000 .333 1.000 1.000 3 1 4 3.00 .333 1.000 1.000 3 1 4 3.00 .333 1.000 1.000 3 1 4 3.00 .333 1.000 1.000 3 1 4 2.88 .398 1.126 1.268 3 2 5 3.13 .295 8.3	er	Trust & Respect	9	2	1	3	2.33	.236	.707	.500	2	1	3	2.44	.242	.726	.528	
Owner/User Participation 9 3 1 4 2.67 2.89 .866 .750 3 1 4 2.67 .333 1.000 1. Production of Specified Quality 9 2 2 4 3.22 .222 .667 .444 2 2 4 3.22 .222 .667 .444 2 2 4 3.22 .222 .667 .444 2 2 4 3.00 3.33 1.000 1. Time Performance 9 2 1 3 2.22 .278 .833 .694 2 1 3 2.22 .278 .833 .694 2 1 3 2.22 .278 .833 .000 .333 1.000 1.000 3 1 4 3.00 .333 1.000 1.000 3 1 4 3.00 .333 1.000 1.000 3 1 4 2.88 .339 1.26 1.26	ild	Alignment of Objectives	9	3	1	4	2.56	.377	1.130	1.278	3	1	4	2.44	.377	1.130	1.278	
Production of Specified Quality 9 2 2 4 3.22 .222 .667 .444 2 2 4 3.22 .222 .667 .444 2 2 4 3.22 .222 .667 .444 2 2 4 3.22 .222 .667 .444 2 2 4 3.00 .333 1.000 1 Time Performance 9 2 1 3 2.22 .278 .833 .694 2 1 3 2.22 .278 .833 .000 .333 1.000 1 4 3.00 .333 1.000 1 4 3.00 .333 1.000 1 4 3.00 .333 1.000 1 .00 .333 1.000 1 .00 .333 1.000 1 .00 .333 1.000 1 .00 .333 1.000 1 .00 .00 .033 .000 .01 .00 .01 .0	n	Owner/User Participation	9	3	1	4	2.67	.289	.866	.750	3	1	4	2.67	.333	1.000	1.000	
Long-term / Lifecycle Value 9 2 2 4 3.11 .261 .782 .611 3 1 4 3.00 .333 1.000 1. Time Performance 9 2 1 3 2.22 .278 .833 .694 2 1 3 2.22 .278 .833 .694 2 1 3 2.22 .278 .833 .	щ	Production of Specified Quality	9	2	2	4	3.22	.222	.667	.444	2	2	4	3.22	.222	.667	.444	
Interperformance 9 2 1 3 2.22 .278 .833 .094 2 1 3 2.22 .278 .833 094 2 1 3 2.22 .278 833 000 000 3 1 4 3.000 .333 1.000 3.000 000 3 1 4 3.000 333 1.000 1.000 3 1 4 3.000 333 1.000 1.000 3 1 4 3.00 333 1.000 1.000 3 1 4 3.00 333 1.000 1.000 3 1 4 3.00 333 1.000 1.000 3 1 4 3.00 333 1.000 1.000 3 1 4 3.00 22 278 833 1 3 2.02		Long-term / Lifecycle Value	9	2	2	4	3.11	.261	.782	.611	3	1	4	3.00	.333	1.000	1.000	
Constructability 9 2 1 4 2.00 2.29 8.66 7.50 2 1 3 2.22 2.78 8.33 . Constructability 9 2 1 3 2.00 2.29 8.66 7.50 2 1 3 2.22 2.78 8.33 . Owner/User Satisfaction 8 4 1 5 2.63 .460 1.302 1.696 3 1 4 2.88 .350 .991 . Effective Communication 8 2 1 3 2.00 .327 .926 .857 3 1 4 2.63 .420 1.188 1. Collaboration of Project Team 8 3 1 4 2.38 .375 1.061 1.125 3 1 4 2.63 .420 1.188 1. Utility & Functionality 8 2 2 4 3.00 .267 .756 .571		Time Performance	9	2	1	3 4	3.00	.278	.833	.694	2	1	3 4	3.00	.278	.833	.694	
Owner/User Satisfaction 8 4 1 5 2.88 .398 1.126 1.268 3 2 5 3.13 .295 .835 . Clear & Realistic Objectives 8 4 1 5 2.63 .460 1.302 1.696 3 1 4 2.88 .350 .991 . Effective Communication 8 2 1 3 2.00 .327 .926 .857 3 1 4 2.63 .420 1.188 1. Collaboration of Project Team 8 3 1 4 2.38 .375 1.061 1.125 3 1 4 2.63 .420 1.188 1. Utility & Functionality 8 2 2 4 3.00 .267 .756 .571 2 2 4 3.13 .227 .641 . Trust & Respect 8 3 1 4 2.60 .327 .926 <		Constructability	9	2	1	3	2.00	.289	.866	.750	2	1	3	2.22	.278	.833	.694	
Clear & Realistic Objectives 8 4 1 5 2.63 .460 1.302 1.696 3 1 4 2.88 .350 .991 . Effective Communication 8 2 1 3 2.00 .327 .926 .857 3 1 4 2.63 .324 .916 . Collaboration of Project Team 8 3 1 4 2.38 .375 1.061 1.125 3 1 4 2.63 .420 1.188 1. Utility & Functionality 8 2 2 4 3.00 .267 .756 .571 2 2 4 3.13 .227 .641 . Trust & Respect 8 3 1 4 2.63 .324 .916 .839 3 2 5 3.13 .350 .991 . Alignment of Objectives 8 3 1 4 2.50 .327 .926 .		Owner/User Satisfaction	8	4	1	5	2.88	.398	1.126	1.268	3	2	5	3.13	.295	.835	.696	
Effective Communication 8 2 1 3 2.00 .327 .926 .857 3 1 4 2.63 .324 .916 . Collaboration of Project Team 8 3 1 4 2.38 .375 1.061 1.125 3 1 4 2.63 .420 1.188 1. Utility & Functionality 8 2 2 4 3.00 .267 .756 .571 2 2 4 3.13 .227 .641 . Trust & Respect 8 3 1 4 2.63 .324 .916 .839 3 2 5 .3.13 .350 .991 . Alignment of Objectives 8 3 1 4 2.50 .327 .926 .857 3 1 4 2.63 .324 .916 .327 Owner/User Participation 8 2 1 3 2.38 .263 .744 .55	1	Clear & Realistic Objectives	8	4	1	5	2.63	.460	1.302	1.696	3	1	4	2.88	.350	.991	.982	
Collaboration of Project leam 8 3 1 4 2.38 .375 1.06111.125 3 1 4 2.63 .420 1.188 1. Utility & Functionality 8 2 2 4 3.00 .267 .756 .571 2 2 4 3.13 .227 .641 . Trust & Respect 8 3 1 4 2.63 .324 .916 .839 3 2 5 3.13 .350 .991 . Alignment of Objectives 8 3 1 4 2.50 .327 .926 .857 3 1 4 2.63 .744 .554 2 2 4 2.75 .250 .707 . Production of Specified Quality 8 3 2 5 3.38 .324 .916 .839 2 3 5 3.63 .263 .744 . Production of Specified Quality 8 3 2 5 .338 .324 .916 .839 2 3 5	1	Effective Communication	8	2	1	3	2.00	.327	.926	.857	3	1	4	2.63	.324	.916	.839	
Computer nuctionality 6 2 2 4 5.00 .207 .136 .571 2 2 4 5.041 . Trust & Respect 8 3 1 4 2.63 .324 .916 .839 3 2 5 3.13 .350 .991 . Alignment of Objectives 8 3 1 4 2.63 .327 .926 .857 3 1 4 2.63 .324 .916 . <t< td=""><td>1</td><td>Collaboration of Project Team</td><td>8</td><td>3</td><td>1</td><td>4</td><td>2.38</td><td>.375</td><td>1.061</td><td>571</td><td>3</td><td>1</td><td>4</td><td>2.63</td><td>.420</td><td>1.188</td><td>1.411</td></t<>	1	Collaboration of Project Team	8	3	1	4	2.38	.375	1.061	571	3	1	4	2.63	.420	1.188	1.411	
O Alignment of Objectives 8 3 1 4 2.50 327 926 857 3 1 4 2.63 324 916 . Owner/User Participation 8 2 1 3 2.38 .263 .744 .554 2 2 4 2.75 .250 .707 . Production of Specified Quality 8 3 2 5 3.38 .324 .916 .839 2 3 5 3.63 .263 .744 . Long-term / Lifecycle Value 8 2 1 3 2.50 .267 .756 .571 2 2 4 2.75 .250 .707 . Time Performance 8 3 1 4 1.88 .350 .991 .982 3 1 4 2.00 .378 1.069 1	\mathbf{rs}	Trust & Respect	8	3	1	4	2.63	.207	.730	.371	3	2	5	3.13	.227	.041	.982	
Owner/User Participation 8 2 1 3 2.38 .263 .744 .554 2 2 4 2.75 .250 .707 . Production of Specified Quality 8 3 2 5 3.38 .324 .916 .839 2 3 5 3.63 .263 .744 . Long-term / Lifecycle Value 8 2 1 3 2.50 .267 .756 .571 2 2 4 2.75 .250 .707 . Time Performance 8 3 1 4 1.88 .350 .991 .982 3 1 4 2.00 .378 1.069 1.	Se	Alignment of Objectives	8	3	1	4	2.50	.327	.926	.857	3	1	4	2.63	.324	.916	.839	
Production of Specified Quality 8 3 2 5 3.38 .324 .916 .839 2 3 5 3.63 .263 .744 . Long-term / Lifecycle Value 8 2 1 3 2.50 .267 .756 .571 2 2 4 2.75 .250 .707 . Time Performance 8 3 1 4 1.88 .350 .991 .982 3 1 4 2.06 .378 1.069 1.	Ď	Owner/User Participation	8	2	1	3	2.38	.263	.744	.554	2	2	4	2.75	.250	.707	.500	
Long-term / Lifecycle Value 8 2 1 3 2.50 .267 .756 .571 2 2 4 2.75 .250 .707 . Time Performance 8 3 1 4 1.88 .350 .991 .982 3 1 4 2.06 .378 1.069 1.	ľ	Production of Specified Quality	8	3	2	5	3.38	.324	.916	.839	2	3	5	3.63	.263	.744	.554	
Imme reformance 8 3 1 4 1.88 .350 .991 .982 3 1 4 2.00 .378 1.069 1.		Long-term / Lifecycle Value	8	2	1	3	2.50	.267	.756	.571	2	2	4	2.75	.250	.707	.500	
Owner's Vision 8 3 1 4 275 313 886 786 4 1 5 289 209 11261	1	Time Performance	8	3	1	4	1.88	.550	.991	.982	5 1	1	4	2.00	.5/8	1.069	1.143	
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	1	Constructability	8	4	1	5	3,13	.441	1.246	1.554	3	1	4	3.13	.398	1.120	1.268	

9.4. Design-Bid-Build: Data & Analysis

Table 9.4. Round 3 -- DBB Relative effectiveness: descriptive statistics



Figure 9.5. Round 3 -- DBB: relative effectiveness by average mean score

Viewing the distribution of stakeholder group mean scores in Figure 9.5 it is clear the designers rate Design-Bid-Build (DBB) highest above average among the stakeholder groups, followed by the owners in both self-interest and project interest. The builders and users display and opposing trend rating below the average in both self and project categories. The distribution is much tighter around the linear average form project-interest indicating more agreement among the stakeholders overall.

9.4.1. Inter-Rater Agreement Testing

The panel of experts demonstrated the highest levels of agreement within each stakeholder group when evaluating the Design-Bid-Build method. The owner and designer

group had an exceptionally high agreement. Keep the high level of agreement in mind when reviewing the comparisons of how each group rated the methods later in this report. DBB is generally the lowest rated of the three project delivery methods.

Round #3 Inter-Rater Agreement: Design Bid Build (DBB)												
		Se	elf-Inte	rest		Project-Interest						
Critical Success Factors		Ν	lean Ra	ınk			Ν	lean Ra	ınk			
	0	D	В	U	All	0	D	В	U	All		
Owner / User Satisfaction	8.75	8.67	8.11	7.81	8.40	8.68	8.17	7.50	8.00	8.16		
Clear & Realistic Objectives	9.21	10.39	8.22	7.06	8.83	9.71	9.67	7.39	6.50	8.54		
Effective Communication	7.21	6.94	8.22	3.88	6.71	7.18	6.33	8.61	5.94	7.06		
Collaboration of Project Delivery Team	5.14	4.11	7.11	6.06	5.54	4.18	4.50	6.61	5.75	5.11		
Utility & Functionality	7.43	8.44	7.72	9.19	8.08	6.82	8.61	8.33	8.63	7.93		
Trust & Respect	4.64	4.17	4.83	7.69	5.19	5.04	3.56	5.17	8.38	5.40		
Alignment of Project Objectives	7.64	4.83	5.94	6.88	6.48	7.64	6.33	5.61	5.94	6.55		
Production of Specified Quality		8.22	6.94	5.88	7.41	7.82	8.17	6.89	6.63	7.45		
Long-term Bldg. Success & Lifecycle		9.67	8.72	9.88	9.26	9.00	9.83	9.00	10.38	9.46		
Project Time Performance	7.82	7.56	8.72	6.25	7.65	7.50	7.83	8.44	6.56	7.60		
Owner / User Participation	2.61	3.11	4.67	4.00	3.46	3.36	3.22	4.67	3.38	3.63		
Owner's vision	9.46	10.06	8.17	7.63	8.94	9.14	9.78	8.50	6.81	8.68		
Constructability	4.00	4.83	3.61	8.81	5.06	4.93	5.00	4.28	8.13	5.44		
Statistic		Se	elf-Inte	rest		Project-Interest						
Statistic		D	В	U	All	0	D	В	U	All		
Ν		9	9	8	40	14	9	9	8	40		
Kendall's W ^a	.425	.481	.309	.313	.285	.387	.463	.287	.273	.267		
Chi-square	71.41	51.98	33.40	30.01	136.62	65.03	50.05	31.01	26.20	128.35		
df	12	12	12	12	12	12	12	12	12	12		
Asymp. Sig.	.000	.000	.001	.003	.000	.000	.000	.002	.010	.000		
a. Kendall's Coefficient of Concordance												

 Table 9.5. Round 3 -- DBB: inter-rater agreement testing



Figure 9.6. Round 3 -- DBB: Inter rate agreement comparison

9.4.2. Correlations

Round #3: Correlation among Stakeholder Groups for DBB Category												
Correlation Matrix *		Self-I	nterest		Project-Interest							
	Owner	Owner Designer		User	Owner	Designer	Builder	User				
Owner	1.000	0.761 ^b	0.586 ^b	0.149	1.000	0.800 ^b	0.421 ^a	0.224				
Designer		1.000	0.575 ^b	0.368	-	1.000	0.521 ^a	0.395				
Builder	-		1.000	0.209			1.000	0.258				
User				1.000				1.000				

Notes:

^a Correlation is significant at the 0.05 level (2-tailed).

^b Correlation is significant at the 0.01 level (2-tailed).
* Correlation Coefficient based on mean scores, and calculated in SPSS using Pearsons Correlation.

Table 9.6. Round 3 -- DBB: stakeholder correlation matrix

	Correlations: Self-Interest Design Bid Build (DBB)													
										Long-				
		Clear &		Collab-			Align-		Product-	term				
	Owner/	Real-	Effect-	oration	Utility		ment of	Owner/	ion of	Success				
	User	istic	ive	of Proj.	&		Project	User	Specif-	& Life-	Time			
	Satisfac-	Object-	Comm-	Delivery	Functio	Trust &	Objec-	Partic-	ied	cycle	Perform	Owner's	Construct-	
	tion	ives	unication	Team	n-ality	Respect	tives	ipation	Quality	Value	ance	Vision	ability	
Owner/User Satisfaciton	1.000	0.615**	0.516**	0.451**	0.502**	0.532**	0.473**	0.588^{**}	0.451**	0.432**	0.063	0.621**	0.318*	
Clear & Realistic Objectives		1.000	.584**	0.370^{*}	0.468**	0.516**	0.656**	0.665**	0.631**	0.559**	0.243	0.667**	0.393*	
Effective Communication			1.000	0.485**	0.530**	0.330*	0.543**	0.765**	0.500^{**}	0.494**	0.073	0.664**	0.147	
Collaboration of Proj. Delivery Tean				1.000	0.487**	0.705**	0.596**	0.536**	0.482**	0.457**	0.050	0.465**	0.283	
Utility & Functionality					1.000	0.399*	0.504**	0.673**	0.679**	0.704**	0.117	0.732**	0.427**	
Trust & Respect						1.000	0.673**	0.505**	0.479**	0.475**	0.348*	0.459**	0.593**	
Alignment of Project Objectives							1.000	0.733**	0.585**	0.551**	0.287	0.664**	0.448^{**}	
Owner/User Participation								1.000	0.628**	0.692**	0.161	0.872^{**}	0.361*	
Production of Specified Quality									1.000	0.698**	0.139	0.700^{**}	0.545**	
Long-term Success & Lifecycle Valu										1.000	.347*	0.781**	0.217	
Time Performance											1.000	0.238	0.182	
Owner's Vision												1.000	0.294	
Constructability													1.000	
**. Correlation is significant at the 0.01 level (2-tailed).														
*. Correlation is significant at the 0.05 level (2-tailed).														
Pearson Correlation														
Correlations: Project-Interest Design Bid Build (DBB)														
Long-														
		Clear &		Collab-			Align-		Product-	term				
	Owner/	Real-	Effect-	oration	Utility		ment of	Owner/	ion of	Success				
	User	istic	ive	of Proj.	&		Project	User	Specif-	& Life-	Time			
	Satisfac-	Object-	Comm-	Delivery	Functio	Trust &	Objec-	Partic-	ied	cycle	Perform-	Owner's	Construct-	
	tion	ives	unication	Team	n-ality	Respect	tives	ipation	Quality	Value	ance	Vision	ability	
Owner/User Satisfaciton	1.000	.734	.574	.315	.640	.451	.689**	.506**	.621	.521	.353	.613	.585	
Clear & Realistic Objectives		1.000	.614	.428**	.508	.412**	.713**	.603	.688	.649**	.383	.535**	.559**	
Effective Communication			1.000	.430**	.646**	.318	.556**	.498	.628	.577**	0.239	.475**	.360	
Collaboration of Proj. Delivery Tean				1.000	0.250	.571**	.462**	0.281	.350*	0.222	0.194	0.087	.439**	
Utility & Functionality					1.000	0.195	.549**	.459**	.673**	.695**	0.172	.426**	.635**	
Trust & Respect						1.000	.637**	.353*	.484**	.437**	.416**	.337*	.625**	
Alignment of Project Objectives							1.000	.682**	.622**	.614**	.389*	.496**	.669**	
Owner/User Participation								1.000	.493**	.565**	.354*	.806**	.489**	
Production of Specified Quality									1.000	.762**	0.287	.556**	.662**	
Long-term Success & Lifecycle Value										1.000	.435**	.568**	.578**	
Time Performance											1.000	.344*	0.291	
Owner's Vision	L											1.000	.412**	
0														
Constructability													1.000	
Constructability **. Correlation is significant at the 0	 .01 level	 (2-tailed)											1.000	

*. Correlation is significant at the 0.05 level (2-tailed).

Table 9.7. Round 3 -- DBB: CSF correlation matrixes
9.5.	Design-Build	(DB):
------	---------------------	----------------

			D	esig	1-Bui	ld (D	B) Relati	ve Effec	tivene	ss						
Group	Critical Success Factor	N				Self-C	Centered				0	Pr	oject-	Centered		
Group	Ciffical Success Factor	- 1	Range	Min	Max	Mean	Std.Error	Std.Dev	Var.	Range	Min.	Max	Mean	Std.Error	Std.Dev	Var.
	Owner/User Satisfaction	40	4	1	5	2.98	.162	1.025	1.051	4	1	5	3.33	.149	.944	.892
	Effective Communication	40	4	1	5	3.10	.147	.928	.862	4	1	5	3.35	.122	.//0	.592
	Collaboration of Project Team	40	4	1	5	3.03	150	947	.809	4	1	5	3.13	135	853	728
	Utility & Functionality	40	4	1	5	2.93	.145	.917	.840	4	1	5	3.15	.127	.802	.644
	Trust & Respect	40	4	1	5	2.78	.154	.974	.948	4	1	5	2.85	.146	.921	.849
A	Alignment of Objectives	40	4	1	5	2.63	.132	.838	.702	4	1	5	2.90	.123	.778	.605
	Owner/User Participation	40	4	1	5	2.55	.168	1.061	1.126	4	1	5	2.73	.164	1.037	1.076
	Production of Specified Quality	40	4	1	5	3.03	.131	.832	.692	4	1	5	3.03	.131	.832	.692
	Long-term / Lifecycle Value	40	4	2	5	2.85	.137	.864	./40	3	2	5	2.95	.113	./14	.510
	Owner's Vision	40	4	1	5	2.83	156	984	969	4	1	5	2.85	141	893	.830
	Constructability	40	4	1	5	3.55	.124	.783	.613	4	1	5	3.70	.135	.853	.728
	Owner/User Satisfaction	14	4	1	5	2.93	.305	1.141	1.302	4	1	5	3.29	.266	.994	.989
	Clear & Realistic Objectives	14	4	1	5	2.86	.312	1.167	1.363	4	1	5	3.14	.254	.949	.901
	Effective Communication	14	3	1	4	2.64	.269	1.008	1.016	3	1	4	2.79	.261	.975	.951
~	Collaboration of Project Team	14	3	1	4	2.57	.251	.938	.879	3	1	4	2.64	.225	.842	.709
L.C.	Utility & Functionality	14	2	1	3	2.29	.194	.726	.527	3	1	4	2.64	.199	.745	.555
ne	Alignment of Objectives	14	2	1	4	2.43	.231	.938	.879	2	1	3	2.30	.225	.842	.709
≥ ≥	Owner/User Participation	14	2	1	3	1.93	.221	.829	.407	3	1	4	2.00	.257	.961	.923
\circ	Production of Specified Quality	14	2	1	3	2.43	.173	.646	.418	2	2	4	2.64	.169	.633	.401
	Long-term / Lifecycle Value	14	3	1	4	2.64	.225	.842	.709	2	2	4	2.79	.155	.579	.335
	Time Performance	14	2	3	5	4.00	.234	.877	.769	2	3	5	4.00	.234	.877	.769
	Owner's Vision	14	2	1	3	2.21	.214	.802	.643	2	1	3	2.43	.202	.756	.571
	Constructability	14	4	1	5	3.50	.251	.941	.885	4	1	5	3.71	.286	1.069	1.143
	Clear & Paglistic Objectives	9	2	2	3	2.33	.236	./0/	.500	2	2	4	2.78	.278	.833	.694
	Effective Communication	9	2	2	4	2 78	2201	667	444	4	1	5	3.11	351	1.054	1 1 1 1 1
S	Collaboration of Project Team	9	2	2	4	2.89	.261	.782	.611	2	2	4	3.11	.200	.601	.361
- CL	Utility & Functionality	9	2	2	4	3.00	.167	.500	.250	1	3	4	3.22	.147	.441	.194
Ē.	Trust & Respect	9	2	2	4	2.89	.309	.928	.861	2	2	4	2.89	.309	.928	.861
16	Alignment of Objectives	9	1	2	3	2.56	.176	.527	.278	2	2	4	3.00	.167	.500	.250
eso a	Owner/User Participation	9	1	2	3	2.33	.167	.500	.250	2	1	3	2.56	.242	.726	.528
Ω	Production of Specified Quality	9	2	2	4	3.00	.289	.800	./50	3	2	4	3.00	.333	1.000	261
	Time Performance	9	3	2	5	3 3 3	333	1 000	1 000	3	2	5	3 56	377	1 1 3 0	1 278
	Owner's Vision	9	2	2	4	2.89	.261	.782	.611	1	2	3	2.78	.147	.441	.194
	Constructability	9	2	2	4	3.22	.222	.667	.444	2	3	5	3.67	.236	.707	.500
	Owner/User Satisfaction	9	2	3	5	3.78	.278	.833	.694	3	2	5	3.67	.333	1.000	1.000
	Clear & Realistic Objectives	9	2	3	5	3.67	.236	.707	.500	2	3	5	3.78	.222	.667	.444
	Effective Communication	9	2	3	5	3.78	.222	.667	.444	2	3	5	3.78	.222	.667	.444
Ś	Litility & Europia polity	9	2	2	5	3.89	.261	./82	.011	2	3	5	3.89	.201	./82	.611
er	Trust & Respect	9	3	2	5	3 3 3	289	866	750	3	2	5	3 33	2.89	.007	750
lld	Alignment of Objectives	9	3	2	5	3.22	.278	.833	.694	2	3	5	3.33	.236	.707	.500
. E	Owner/User Participation	9	3	2	5	3.56	.294	.882	.778	3	2	5	3.56	.294	.882	.778
ш	Production of Specified Quality	9	2	3	5	3.67	.236	.707	.500	2	3	5	3.67	.236	.707	.500
	Long-term / Lifecycle Value	9	2	3	5	3.44	.242	.726	.528	2	3	5	3.44	.242	.726	.528
	Time Performance	9	2	3	5	4.00	.289	.866	.750	2	3	5	4.00	.289	.866	.750
	Constructability	9	2	2	5	<u>3.30</u> 4.00	236	.882	.//8	2	3	5	3.07	236	.800	500
	Owner/User Satisfaction	8	2	2	4	2.88	.295	.707	.500	2	3	5	3.63	.263	.744	.554
	Clear & Realistic Objectives	8	2	2	4	2.88	.227	.641	.411	1	3	4	3.38	.183	.518	.268
	Effective Communication	8	2	2	4	2.75	.313	.886	.786	1	3	4	3.38	.183	.518	.268
	Collaboration of Project Team	8	2	2	4	3.00	.267	.756	.571	2	2	4	3.13	.227	.641	.411
S	Utility & Functionality	8	3	1	4	3.00	.378	1.069	1.143	2	2	4	3.25	.313	.886	.786
er	Trust & Respect	8	3	1	4	2.63	.375	1.061	1.125	2	2	4	3.13	.295	.835	.696
Js	Augnment of Objectives	0	2	1	_4 _∕	2.13	.300	1.035	1.0/1	2	2	_4 _∕	3.23	.230	./0/	500
	Production of Specified Ouality	8	1	3	4	3.38	.183	518	268	2	2	4	3.00	.250	.756	.500
	Long-term / Lifecvcle Value	8	3	1	4	2.63	.375	1.061	1.125	2	2	4	2.75	.313	.886	.786
	Time Performance	8	2	3	5	3.75	.313	.886	.786	2	3	5	3.88	.295	.835	.696
	Owner's Vision	8	3	1	4	3.00	.378	1.069	1.143	3	1	4	2.75	.366	1.035	1.071
	Constructability	40	4	1	5	3.55	.124	.783	.613	2	2	4	3.38	263	.744	.554

Table 9.8. Round 3 -- DB Relative effectiveness: descriptive statistics



Figure 9.7. Round 3 -- DB: relative effectiveness by average mean score

In the Design-Build (DB) category, not unexpectedly, the builders rate the effectiveness significantly higher than the average of the group. The Owners overall rate effectiveness of DB the lowest among the stakeholder groups, except for "Time Performance" which they rate higher than average. Designers and Users tend to rate the effectiveness near the overall average. Designers depart from the trend for "Owner/User Satisfaction" which they rate DB significantly lower than average.

Round #3 Inte	r-Rate	er Ag	reeme	ent: D	esign	Build	(DB)			
Self-Interest Proj Critical Success Factors Mean Rank M								ject-Int	erest	
Critical Success Factors		N	lean Ra	ınk			Ν	lean Ra	ınk	
	0	D	В	U	All	0	D	В	U	All
Owner / User Satisfaction	8.36	4.67	7.56	6.50	6.98	9.04	5.67	7.00	8.81	7.78
Clear & Realistic Objectives	8.39	7.89	7.06	6.44	7.59	8.79	7.78	7.61	7.31	8.00
Effective Communication	7.18	6.78	7.50	5.63	6.85	7.14	7.17	7.39	7.94	7.36
Collaboration of Project Delivery Team	6.93	7.50	8.22	7.13	7.39	6.36	7.56	8.17	6.44	7.05
Utility & Functionality	5.54	7.83	7.67	7.38	6.90	6.36	7.56	7.61	7.38	7.11
Trust & Respect	6.14	7.44	5.06	5.00	5.96	5.07	6.56	5.00	6.56	5.69
Alignment of Project Objectives	5.21	5.56	4.89	5.69	5.31	5.14	6.67	4.94	7.25	5.86
Production of Specified Quality	4.14	4.50	6.44	6.19	5.15	3.93	4.61	6.39	6.81	5.21
Long-term Bldg. Success & Lifecycle	5.93	7.28	6.78	9.13	7.06	5.82	7.39	6.67	5.81	6.36
Project Time Performance	7.18	6.72	5.56	5.06	6.29	7.07	6.28	5.44	4.44	6.00
Owner / User Participation	10.79	9.11	8.83	10.13	9.84	10.54	8.56	8.89	9.75	9.56
Owner's vision	5.29	6.78	6.56	7.38	6.33	5.57	5.56	7.11	4.69	5.74
Constructability	9.93	8.94	8.89	9.38	9.36	10.18	9.67	8.78	7.81	9.28
Statistic		Se	elf-Inter	rest			Pro	ject-Int	erest	
Statistic	0	D	В	U	All	0	D	В	U	All
N	14	9	9	8	40	14	9	9	8	40
Kendall's W ^a	.350	.174	.202	.324	.188	.392	.160	.210	.212	.186
Chi-square	58.83	18.81	21.76	31.13	90.47	65.89	17.24	22.70	20.36	89.31
df	12	12	12	12	12	12	12	12	12	12
Asymp. Sig.	.000	.093	.040	.002	.000	.000	.141	.030	.061	.000
a Kendall's Coefficient of Concordance										

9.5.1. Inter-Rater Agreement Testing

a. Kendall's Coefficient of Concordance





The Owner's group had a significantly higher level of agreement when evaluating the Design-Build method that the other stakeholders. Overall the level of agreement between self-interest and project-interest is fairly consistent, and are significant at the 0.001 level.

9.5.2. Correlations

Round #3: Correlation among Stakeholder Groups for DB Category												
Correlation Matrix *		Self-Interest Project-Interest										
	Owner	Designer	Builder	User	Owner	Designer	Builder	User				
Owner	1.000	I.000 0.339 0.512 0.260 1.000 0.547 0.548 0.46										
Designer		1.000	0.311	0.381		1.000	0.470 ^a	0.441 ^a				
Builder			1.000	0.606 ^b			1.000	0.346				
User		1.000 1.000										
Notes:												
Notes:												

^a Correlation is significant at the 0.05 level (2-tailed).

^b Correlation is significant at the 0.01 level (2-tailed).
* Correlation Coefficient based on mean scores, and calculated in SPSS using Pearsons Correlation.

Table 9.10. Round 3 -- DB: stakeholder correlation matrix

			Correlat	ions: Seli	-mterest	Design	bulla (Di	B)					
										Long-			
		Clear &		Collab-			Align-		Product-	term			
	Owner/	Real-	Effect-	oration	Utility		ment of	Owner/	ion of	Success			
	User	istic	ive	of Proj.	&		Project	User	Specif-	& Life-	Time		
	Satisfac-	Object-	Comm-	Delivery	Functio	Trust &	Objec-	Partic-	ied	cycle	Perform-	Owner's	Construct-
	tion	ives	unication	Team	n-ality	Respect	tives	ipation	Quality	Value	ance	Vision	ability
Owner/User Satisfaciton	1.000	0.676	0.723	0.635	0.598	0.534	0.496	0.602	0.482	0.575	0.242	0.529	0.497
Clear & Realistic Objectives		1.000	0.539**	0.581	0.612**	0.423**	0.577**	0.542**	0.561	0.595	-0.036	0.637**	0.311
Effective Communication			1.000	0.786**	0.686**	0.665**	0.698 ^{**}	0.677**	0.432**	0.595**	0.290	0.521**	0.495**
Collaboration of Proj. Delivery Tean				1.000	0.770***	0.674**	0.723**	0.701**	0.488^{**}	0.506**	0.214	0.610**	0.431**
Utility & Functionality					1.000	0.641**	0.797**	0.834**	0.574**	0.665**	0.012	0.866**	0.309
Trust & Respect						1.000	0.617**	0.694**	0.355*	0.477***	0.237	0.600**	0.335*
Alignment of Project Objectives							1.000	0.786***	0.529**	0.664**	0.168	0.727***	0.362*
Owner/User Participation			-	1		-		1.000	0.565**	0.568**	0.223	0.733**	0.306
Production of Specified Quality									1.000	0.612**	0.176	0.507**	0.175
Long-term Success & Lifecycle Valu										1.000	0.221	0.572**	0.353*
Time Performance											1.000	0.017	0.517**
Owner's Vision												1.000	0.261
Constructability													1.000
**. Correlation is significant at the 0	.01 level	(2-tailed)											
*. Correlation is significant at the 0.0	05 level (2	2-tailed).											
			orrelatio	ns: Proje	ct-Intere	st Desig	n Build (DB)				-	
										Long-			
		Clear &		Collab-			Align-		Product-	term			
	Owner/	Real-	Effect-	oration	Utility		ment of	Owner/	ion of	Success			
	User	istic	ive	ot Proj.	ð.		1	laor		0 7 . 0			
	Satisfac-	Object-		D.1	E	T	Project	Dentin	Specif-	& Life-	Time	0	Constant
		i	Comminantian	Delivery	Functio	Trust &	Objec-	Partic-	Specif- ied	& Life- cycle	Time Perform-	Owner's	Construct-
	1.000	ives	unication	Delivery Team	Functio n-ality	Trust & Respect	Objec- tives	Partic- ipation	Specif- ied Quality	& Life- cycle Value	Time Perform- ance	Owner's Vision	Construct- ability
Owner/User Satisfaciton	1.000	ives 0.616 ^{**}	unication 0.459**	Delivery Team 0.426 ^{***}	Functio n-ality 0.442***	Trust & Respect 0.441 ^{**}	Objec- tives 0.325 [*]	Partic- ipation 0.434**	Specif- ied Quality 0.316*	& Life- cycle Value 0.253	Time Perform- ance 0.108	Owner's Vision 0.394 [*]	Construct- ability 0.347*
Owner/User Satisfaciton Clear & Realistic Objectives	1.000	ives 0.616 ^{**} 1.000	0.459** 0.446**	Delivery Team 0.426 ^{**} 0.400 [*]	Functio n-ality 0.442 ^{**} 0.536 ^{**}	Trust & Respect 0.441 ^{**} 0.401 [*]	0.325 [*] 0.445 ^{**}	Partic- ipation 0.434 ^{**} 0.605 ^{**}	Specif- ied Quality 0.316 [*] 0.347 [*]	& Life- cycle Value 0.253 0.406 ^{**}	Time Perform- ance 0.108 0.027	Owner's Vision 0.394 [*] 0.451 ^{**}	Construct- ability 0.347 [*] 0.359 [*]
Owner/User Satisfaciton Clear & Realistic Objectives Effective Communication	1.000 	ives 0.616 ^{**} 1.000 	0.459 ^{**} 0.446 ^{**} 1.000	Delivery Team 0.426 ^{**} 0.400 [*] 0.693 ^{**}	Functio n-ality 0.442** 0.536** 0.624**	Trust & Respect 0.441 ^{**} 0.401 [*] 0.586 ^{**}	0.325 [*] 0.445 ^{**}	Partic- ipation 0.434 ^{**} 0.605 ^{**} 0.439 ^{**}	Specif- ied Quality 0.316 [*] 0.347 [*] 0.331 [*]	& Life- cycle Value 0.253 0.406 ^{**} 0.291	Time Perform- ance 0.108 0.027 0.216	Owner's Vision 0.394 [*] 0.451 ^{**} 0.321 [*]	Construct- ability 0.347 [*] 0.359 [*] 0.343 [*]
Owner/User Satisfaciton Clear & Realistic Objectives Effective Communication Collaboration of Proj. Delivery Team	1.000 	ives 0.616 ^{**} 1.000 	0.459** 0.446** 1.000 	Delivery Team 0.426 ^{**} 0.400 [*] 0.693 ^{**} 1.000	Functio n-ality 0.442 ^{**} 0.536 ^{**} 0.624 ^{**}	Trust & Respect 0.441 ^{**} 0.401 [*] 0.586 ^{**} 0.742 ^{**}	0.325 [*] 0.445 ^{**} 0.644 ^{**}	Partic- ipation 0.434 ^{**} 0.605 ^{**} 0.439 ^{**} 0.561 ^{**}	Specif- ied Quality 0.316 [*] 0.347 [*] 0.331 [*] 0.429 ^{**}	& Life- cycle Value 0.253 0.406 ^{**} 0.291 0.389 [*]	Time Perform- ance 0.108 0.027 0.216 0.186	Owner's Vision 0.394 [*] 0.451 ^{**} 0.321 [*] 0.598 ^{**}	Construct- ability 0.347 [*] 0.359 [*] 0.343 [*] 0.299
Owner/User Satisfaciton Clear & Realistic Objectives Effective Communication Collaboration of Proj. Delivery Tean Utility & Functionality	1.000 	ives 0.616 ^{***} 1.000 	unication 0.459 ^{**} 0.446 ^{**} 1.000 	Delivery Team 0.426 ^{**} 0.693 ^{**} 1.000 	Functio n-ality 0.442 ^{**} 0.536 ^{**} 0.624 ^{**} 0.684 ^{**} 1.000	Trust & Respect 0.441 ^{**} 0.401 [*] 0.586 ^{**} 0.742 ^{**} 0.656 ^{**}	0.325* 0.445** 0.644** 0.715** 0.682**	Partic- ipation 0.434 ^{**} 0.605 ^{**} 0.561 ^{**} 0.605 ^{**}	Specif- ied Quality 0.316 [*] 0.347 [*] 0.331 [*] 0.429 ^{**} 0.417 ^{**}	& Life- cycle Value 0.253 0.406 ^{**} 0.291 0.389 [*] 0.550 ^{**}	Time Perform- ance 0.108 0.027 0.216 0.186 0.061	Owner's Vision 0.394 [*] 0.451 ^{**} 0.321 [*] 0.598 ^{**} 0.676 ^{**}	Construct- ability 0.347 [*] 0.359 [*] 0.343 [*] 0.299 0.367 [*]
Owner/User Satisfaciton Clear & Realistic Objectives Effective Communication Collaboration of Proj. Delivery Tean Utility & Functionality Trust & Respect	1.000 	ives 0.616 ^{**} 1.000 	0.459** 0.446** 1.000 	Delivery Team 0.426 ^{**} 0.693 ^{**} 1.000 	Functio n-ality 0.442 ^{**} 0.536 ^{**} 0.624 ^{**} 0.684 ^{**} 1.000 	Trust & Respect 0.441 ^{**} 0.586 ^{**} 0.742 ^{**} 0.656 ^{**} 1.000	0.325* 0.445** 0.644** 0.642** 0.682** 0.766**	Partic- ipation 0.434 ^{**} 0.605 ^{**} 0.561 ^{**} 0.605 ^{**} 0.626 ^{**}	Specif- ied Quality 0.316 [*] 0.347 [*] 0.331 [*] 0.429 ^{**} 0.417 ^{**}	& Life- cycle Value 0.253 0.406 ^{**} 0.291 0.389 [*] 0.550 ^{**} 0.495 ^{**}	Time Perform- ance 0.108 0.027 0.216 0.186 0.061 0.221	Owner's Vision 0.394 [*] 0.451 ^{**} 0.321 [*] 0.598 ^{**} 0.676 ^{**}	Construct- ability 0.347 [*] 0.359 [*] 0.343 [*] 0.299 0.367 [*] 0.300
Owner/User Satisfaciton Clear & Realistic Objectives Effective Communication Collaboration of Proj. Delivery Tean Utility & Functionality Trust & Respect Alignment of Project Objectives	1.000 	ives 0.616 ^{**} 1.000 	0.459 ^{**} 0.446 ^{**} 1.000 	Delivery Team 0.426 ^{**} 0.693 ^{**} 1.000 	Functio n-ality 0.442 ^{**} 0.536 ^{**} 0.624 ^{**} 1.000 	Trust & Respect 0.441 ^{**} 0.586 ^{**} 0.742 ^{**} 0.656 ^{**} 1.000 	0.325* 0.445** 0.644** 0.715** 0.682** 0.766** 1.000	Partic- ipation 0.434 ^{**} 0.605 ^{**} 0.561 ^{**} 0.605 ^{**} 0.626 ^{**} 0.600 ^{**}	Specif- ied Quality 0.316 [*] 0.347 [*] 0.331 [*] 0.429 ^{**} 0.417 ^{**} 0.540 ^{**} 0.440 ^{**}	& Life- cycle Value 0.253 0.406 ^{**} 0.291 0.389 [*] 0.550 ^{**} 0.495 ^{**} 0.452 ^{**}	Time Perform- ance 0.108 0.027 0.216 0.186 0.061 0.221 0.199	Owner's Vision 0.394 [*] 0.451 ^{**} 0.321 [*] 0.598 ^{**} 0.676 ^{**} 0.564 ^{**}	Construct- ability 0.347* 0.359* 0.343* 0.299 0.367* 0.300 0.301
Owner/User Satisfaciton Clear & Realistic Objectives Effective Communication Collaboration of Proj. Delivery Tean Utility & Functionality Trust & Respect Alignment of Project Objectives Owner/User Participation	1.000 	ives 0.616 ^{**} 1.000 	0.459** 0.446** 1.000 	Delivery Team 0.426 ^{**} 0.693 ^{**} 1.000 	Functio n-ality 0.442 ^{**} 0.536 ^{**} 0.624 ^{**} 1.000 	Trust & Respect 0.441 ^{**} 0.586 ^{**} 0.742 ^{**} 0.656 ^{**} 1.000 	0.325* 0.445** 0.644** 0.715** 0.682** 0.766** 1.000 	Partic- ipation 0.434 ^{**} 0.605 ^{**} 0.561 ^{**} 0.605 ^{**} 0.600 ^{**} 1.000	Specif- ied Quality 0.316 [*] 0.347 [*] 0.429 ^{**} 0.417 ^{**} 0.540 ^{**} 0.540 ^{**} 0.543 ^{**}	& Life- cycle Value 0.253 0.406** 0.291 0.389* 0.550** 0.495** 0.495** 0.452**	Time Perform- ance 0.108 0.027 0.216 0.186 0.061 0.221 0.199 0.153	Owner's Vision 0.394 [*] 0.451 ^{**} 0.321 [*] 0.598 ^{**} 0.676 ^{**} 0.564 ^{**} 0.532 ^{**} 0.674 ^{**}	Construct- ability 0.347* 0.359* 0.343* 0.299 0.367* 0.300 0.301 0.020
Owner/User Satisfaciton Clear & Realistic Objectives Effective Communication Collaboration of Proj. Delivery Tean Utility & Functionality Trust & Respect Alignment of Project Objectives Owner/User Participation Production of Specified Quality	1.000 	ives 0.616** 1.000 	0.459** 0.446** 1.000 	Delivery Team 0.426 ^{**} 0.400 [*] 0.693 ^{**} 1.000 	Functio n-ality 0.442** 0.536** 0.624** 1.000 	Trust & Respect 0.441 ^{**} 0.401 [*] 0.586 ^{**} 0.742 ^{**} 1.000 	Project Objec- tives 0.325 [*] 0.644 ^{**} 0.644 ^{**} 0.682 ^{**} 0.682 ^{**} 1.000 	Osei Partic- ipation 0.434** 0.605*** 0.439** 0.561** 0.605*** 0.605*** 0.605*** 0.605*** 0.600*** 1.000	Specified Quality 0.316 [*] 0.347 [*] 0.331 [*] 0.429 ^{**} 0.417 ^{**} 0.540 ^{**} 0.540 ^{**} 0.513 ^{**} 1.000	& Life- cycle Value 0.253 0.406** 0.291 0.389* 0.550** 0.495** 0.452** 0.500* 0.606**	Time Perform- ance 0.108 0.027 0.216 0.186 0.061 0.221 0.199 0.153 0.173	Owner's Vision 0.394 [*] 0.321 [*] 0.598 ^{**} 0.676 ^{**} 0.564 ^{**} 0.532 ^{**} 0.674 ^{**}	Construct- ability 0.347 [*] 0.359 [*] 0.343 [*] 0.299 0.367 [*] 0.300 0.301 0.020 0.228
Owner/User Satisfaciton Clear & Realistic Objectives Effective Communication Collaboration of Proj. Delivery Tean Utility & Functionality Trust & Respect Alignment of Project Objectives Owner/User Participation Production of Specified Quality Long-term Success & Lifecycle Valu	1.000 	ives 0.616** 1.000 	unication 0.459** 0.446** 1.000 	Delivery Team 0.426** 0.693** 1.000 	Functio n-ality 0.442** 0.636** 0.624** 1.000 	Trust & Respect 0.441 ^{**} 0.401 [*] 0.586 ^{**} 0.656 ^{**} 1.000 	Project Objec- tives 0.325 [*] 0.445 ^{**} 0.644 ^{**} 0.715 ^{**} 0.682 ^{**} 0.682 ^{**} 1.000 	Osei Partic- ipation 0.434** 0.605*** 0.439** 0.561** 0.605*** 0.605*** 0.605*** 0.605*** 0.600** 1.000	Specified Quality 0.316 [*] 0.347 [*] 0.429 ^{**} 0.417 ^{**} 0.540 ^{**} 0.540 ^{**} 0.513 ^{**} 1.000 	& Life- cycle Value 0.253 0.406** 0.291 0.389* 0.550** 0.495** 0.452** 0.452** 0.500** 0.606** 1.000	Time Perform- ance 0.108 0.027 0.216 0.186 0.061 0.221 0.199 0.153 0.173 0.148	Owner's Vision 0.394 [*] 0.451 ^{**} 0.598 ^{**} 0.676 ^{**} 0.564 ^{**} 0.532 ^{**} 0.674 ^{**} 0.489 ^{**} 0.631 ^{**}	Construct- ability 0.347 [*] 0.359 [*] 0.343 [*] 0.299 0.367 [*] 0.300 0.301 0.020 0.228 0.311
Owner/User Satisfaciton Clear & Realistic Objectives Effective Communication Collaboration of Proj. Delivery Tean Utility & Functionality Trust & Respect Alignment of Project Objectives Owner/User Participation Production of Specified Quality Long-term Success & Lifecycle Valu	1100 1.000 	ives 0.616** 1.000 	unication 0.459** 0.446** 1.000 	Delivery Team 0.426** 0.693** 1.000 	Functio n-ality 0.442** 0.634** 0.624** 1.000 	Trust & Respect 0.441 ^{**} 0.586 ^{**} 0.742 ^{**} 0.656 ^{**} 1.000 	Project Objec- tives 0.325 [*] 0.445 ^{**} 0.644 ^{**} 0.715 ^{**} 0.682 ^{**} 0.766 ^{**} 1.000 	0.sei partic- ipation 0.434** 0.605*** 0.439** 0.561*** 0.605*** 0.605*** 0.605*** 0.605*** 0.605*** 0.600*** 0.600*** 1.000	Specified Quality 0.316 [*] 0.347 [*] 0.331 [*] 0.429 ^{**} 0.417 ^{**} 0.540 ^{**} 0.440 ^{**} 0.513 ^{**} 1.000	& Life- cycle Value 0.253 0.406 ^{**} 0.291 0.389 [*] 0.495 ^{**} 0.495 ^{**} 0.452 ^{**} 0.500 ^{**} 0.606 ^{**} 1.000	Time Perform- ance 0.108 0.027 0.216 0.186 0.061 0.221 0.199 0.153 0.173 0.148 1.000	Owner's Vision 0.394 [*] 0.451 ^{**} 0.598 ^{**} 0.676 ^{**} 0.564 ^{**} 0.674 ^{**} 0.674 ^{**} 0.671 ^{**}	Construct- ability 0.347* 0.359* 0.343* 0.299 0.367* 0.300 0.301 0.020 0.228 0.311 0.346*
Owner/User Satisfaciton Clear & Realistic Objectives Effective Communication Collaboration of Proj. Delivery Tean Utility & Functionality Trust & Respect Alignment of Project Objectives Owner/User Participation Production of Specified Quality Long-term Success & Lifecycle Valu Time Performance Owner's Vision	1.000 	ives 0.616** 1.000 	unication 0.459** 0.446** 1.000	Delivery Team 0.426 ^{**} 0.400 [*] 0.693 ^{**} 1.000 	Functio n-ality 0.442*** 0.536** 0.624** 1.000 	Trust & Respect 0.441** 0.586** 0.742** 0.656** 1.000 	Project Objec- tives 0.325* 0.445** 0.644** 0.715** 0.682** 1.000 	0.sel partic- ipation 0.434** 0.605*** 0.439** 0.561*** 0.605*** 0.605*** 0.605*** 0.605*** 0.600*** 0.600*** 0.600*** 0.600*** 1.000	Specified Quality 0.316 [*] 0.347 [*] 0.331 [*] 0.429 ^{**} 0.417 ^{**} 0.540 ^{**} 0.540 ^{**} 0.513 ^{**} 1.000	& Life- cycle Value 0.253 0.406 ^{**} 0.291 0.389 [*] 0.495 ^{**} 0.495 ^{**} 0.452 ^{**} 0.500 ^{**} 0.606 ^{**} 1.000	Time Perform- ance 0.108 0.027 0.216 0.021 0.186 0.061 0.221 0.199 0.153 0.173 0.148 1.000	Owner's Vision 0.394 [*] 0.451 ^{**} 0.321 [*] 0.598 ^{**} 0.676 ^{**} 0.564 ^{**} 0.674 ^{**} 0.674 ^{**} 0.631 ^{**} 0.071 1.000	Construct- ability 0.347* 0.359* 0.343* 0.299 0.367* 0.300 0.301 0.020 0.228 0.311 0.346* 0.209
Owner/User Satisfaciton Clear & Realistic Objectives Effective Communication Collaboration of Proj. Delivery Tean Utility & Functionality Trust & Respect Alignment of Project Objectives Owner/User Participation Production of Specified Quality Long-term Success & Lifecycle Valu Time Performance Owner's Vision Constructability	1000 	ives 0.616*** 1.000 	unication 0.459** 0.446** 1.000	Delivery Team 0.426 ^{**} 0.693 ^{**} 1.000 	Functio n-ality 0.442** 0.536** 0.624** 1.000 	Trust & Respect 0.441*** 0.401* 0.586*** 0.656*** 1.000 	Project Objec- tives 0.325* 0.445** 0.644** 0.715** 0.682** 1.000 	Osei Partic- ipation 0.434*** 0.605*** 0.439*** 0.605*** 0.605*** 0.605*** 0.605*** 0.605*** 0.600*** 1.000	Specif- ied Quality 0.316* 0.347* 0.331* 0.429** 0.417** 0.540** 0.440** 1.000 	& Life- cycle Value 0.253 0.406** 0.291 0.389* 0.495** 0.495** 0.495** 0.606** 1.000 1.000	Time Perform- ance 0.108 0.027 0.216 0.186 0.061 0.221 0.199 0.153 0.173 0.148 1.000 	Owner's Vision 0.394 [*] 0.451 ^{**} 0.321 [*] 0.598 ^{**} 0.676 ^{**} 0.532 ^{**} 0.631 ^{**} 0.631 ^{**} 0.631 ^{**}	Construct- ability 0.347* 0.359* 0.343* 0.299 0.367* 0.300 0.301 0.020 0.228 0.311 0.346* 0.209 1.000
Owner/User Satisfaciton Clear & Realistic Objectives Effective Communication Collaboration of Proj. Delivery Tean Utility & Functionality Trust & Respect Alignment of Project Objectives Owner/User Participation Production of Specified Quality Long-term Success & Lifecycle Valu Time Performance Owner's Vision Constructability **. Correlation is significant at the 0	1.000 	ives 0.616 ^{**} 1.000 	unication 0.459** 0.446** 1.000 <tr t=""> <tr tr=""></tr></tr>	Delivery Team 0.426** 0.693** 1.000 	Functio n-ality 0.442** 0.536** 0.624** 1.000 	Trust & Respect 0.441 ^{**} 0.586 ^{**} 0.742 ^{**} 0.656 ^{**} 1.000 	Project Objec- tives 0.325* 0.445*** 0.715** 0.766** 1.000 	Osei Partic- ipation 0.434*** 0.605*** 0.439*** 0.605*** 0.605*** 0.605*** 0.605*** 0.605*** 0.605*** 0.605*** 0.605*** 0.605*** 0.605*** 0.605*** 0.600*** 1.000	Specif- ied Quality 0.316* 0.331* 0.429** 0.417** 0.540** 0.440** 0.513** 1.000 	& Life- cycle Value 0.253 0.406** 0.291 0.389* 0.405** 0.495** 0.495** 0.606** 1.000 	Time Perform- ance 0.108 0.027 0.216 0.186 0.061 0.221 0.199 0.153 0.173 0.148 1.000 	Owner's Vision 0.394 [*] 0.451 ^{**} 0.598 ^{**} 0.576 ^{**} 0.532 ^{**} 0.674 ^{**} 0.631 ^{**} 0.631 ^{**} 0.071 1.000 	Construct- ability 0.347* 0.359* 0.343* 0.299 0.367* 0.300 0.301 0.020 0.228 0.311 0.346* 0.209 1.000

*. Correlation is significant at the 0.05 level (2-tailed).

Table 9.11. Round 3 -- DB: CSF correlation matrixes

		Int	egrate	d Pro	oject	Deliv	ery (IPD)) Relativ	ve Effe	ctiven	ess				
Group	Critical Success Factor	N				Self-0	Centered	Γ	1			Pr	oject-	Centered	
oroup	Ciffical Success Factor	1,	Range	Min	Max	Mean	Std.Error	Std.Dev	Var.	Range	Min.	Max	Mean	Std.Error	Std.Dev Var.
	Owner/User Satisfaction	40	3	2	5	3.88	.157	.992	.984	3	2	5	4.00	.168	1.062 1.128
	Effective Communication	40	4	1	5	4.05	.160	1.011	1.023	4	1	5	4.23	.166	1.050 1.102
	Collaboration of Project Team	40	4	1	5	4.23	.154	.974	.948	4	1	5	4.28	.152	.960 .922
	Utility & Functionality	40	3	2	5	3.93	.149	.944	.892	4	1	5	3.95	.168	1.061 1.126
Π	Trust & Respect	40	4	1	5	3.85	.174	1.099	1.208	4	1	5	3.93	.177	1.118 1.251
A	Alignment of Objectives	40	4	1	5	3.93	.169	1.0/1	1.148	4	1	5	3.93	.1/3	1.095 1.199
	Production of Specified Quality	40	4	1	5	3.83	168	1.145	1.307	4	1	5	3.85	158	1 001 1 003
	Long-term / Lifecycle Value	40	4	1	5	3.88	.169	1.067	1.138	4	1	5	3.78	.166	1.050 1.102
	Time Performance	40	4	1	5	3.85	.162	1.027	1.054	4	1	5	3.85	.162	1.027 1.054
	Owner's Vision	40	4	1	5	3.88	.169	1.067	1.138	4	1	5	3.90	.167	1.057 1.118
	Constructability	40	4	2	5	3.78	.166	1.050	1.102	4	2	5	3.80	.161	1.018 1.036
	Clear & Realistic Objectives	14	3	2	5	3.00	267	997	995	3	2	5	3.93	305	1 141 1 302
	Effective Communication	14	3	2	5	4.07	.267	.997	.995	2	3	5	4.29	.221	.825 .681
	Collaboration of Project Team	14	3	2	5	4.21	.261	.975	.951	3	2	5	4.21	.239	.893 .797
rs	Utility & Functionality	14	3	2	5	3.86	.231	.864	.747	3	2	5	3.79	.239	.893 .797
ne	Trust & Respect	14	3	2	5	3.93	.245	.917	.841	3	2	5	3.93	.267	.997 .995
\mathbf{A}	Alignment of Objectives	14	3	2	5	3.86	.207	.997	.995	2	2	5	3.93	.280	730 533
\circ	Production of Specified Quality	14	3	2	5	3.86	.206	.770	.593	2	3	5	4.00	.148	.555 .308
	Long-term / Lifecycle Value	14	2	3	5	3.93	.221	.829	.687	2	3	5	3.93	.195	.730 .533
	Time Performance	14	2	3	5	4.07	.195	.730	.533	2	3	5	3.93	.195	.730 .533
	Owner's Vision	14	3	2	5	3.64	.248	.929	.863	3	2	5	3.86	.231	.864 .747
	Constructability Owner/User Satisfaction	14 0	2	3	5	4.21	.18/	.699	.489	2	3	5	4.07	.195	./30 .533
	Clear & Realistic Objectives	9	4	1	5	3.22	.338	1.394	1.944	4	1	5	3.56	.408	1.509 2.278
	Effective Communication	9	4	1	5	4.00	.408	1.225	1.500	4	1	5	4.44	.444	1.333 1.778
\mathbf{S}	Collaboration of Project Team	9	4	1	5	4.00	.408	1.225	1.500	4	1	5	4.22	.434	1.302 1.694
lei	Utility & Functionality	9	3	2	5	3.56	.377	1.130	1.278	4	1	5	3.78	.494	1.481 2.194
50	Trust & Respect	9	4	1	5	3.56	.530	1.590	2.528	4	1	5	3.89	.564	1.691 2.861
S1.	Owner/User Participation	9	4	1	5	3.50	553	1.509	2.278	4	1	5	3.67	553	1.581 2.500
e e	Production of Specified Quality	9	4	1	5	3.44	.556	1.667	2.778	4	1	5	3.44	.503	1.509 2.278
	Long-term / Lifecycle Value	9	4	1	5	3.44	.530	1.590	2.528	4	1	5	3.33	.527	1.581 2.500
	Time Performance	9	4	1	5	3.78	.401	1.202	1.444	4	1	5	4.11	.423	1.269 1.611
	Owner's Vision	9	4	1	5	3.89	.512	1.537	2.361	4	1	5	3.89	.564	1.691 2.861
	Owner/User Satisfaction	9	2	3	5	3.33	.408	1.225	750	2	3	5	4 22	.412	833 694
	Clear & Realistic Objectives	9	3	2	5	3.78	.324	.972	.944	2	3	5	4.00	.236	.707 .500
	Effective Communication	9	2	3	5	4.11	.309	.928	.861	2	3	5	4.22	.324	.972 .944
~	Collaboration of Project Team	9	2	3	5	4.44	.294	.882	.778	2	3	5	4.44	.294	.882 .778
ere	Utility & Functionality	9	2	3	5	4.22	.278	.833	.694	2	3	5	4.22	.278	.833 .694
lde	Alignment of Objectives	9	2	3	5	3.89	.261	./82	.611	2	3	5	3.89	.261	./82 .611
In	Owner/User Participation	9	2	3	5	4.22	.201	.782	.694	2	3	5	4.22	.201	833 694
Ē	Production of Specified Quality	9	2	3	5	3.89	.261	.782	.611	2	3	5	3.89	.261	.782 .611
	Long-term / Lifecycle Value	9	2	3	5	4.00	.289	.866	.750	2	3	5	3.89	.261	.782 .611
	Time Performance	9	2	3	5	3.89	.261	.782	.611	2	3	5	3.89	.261	.782 .611
	Owner's Vision	9	2	3	5	4.00	.289	.866	.750	2	3	5	4.00	.289	.866 .750
	Owner/User Satisfaction	9	2	3	5	4 13	.289	.800	./30	2	2	5	3.78	420	.633 .694
1	Clear & Realistic Objectives	8	2	3	5	4.13	.295	.835	.696	3	2	5	3.88	.398	1.126 1.268
1	Effective Communication	8	2	3	5	4.00	.378	1.069	1.143	3	2	5	3.88	.441	1.246 1.554
	Collaboration of Project Team	8	2	3	5	4.25	.313	.886	.786	2	3	5	4.25	.313	.886 .786
S	Utility & Functionality	8	2	3	5	4.13	.350	.991	.982	3	2	5	4.13	.398	1.126 1.268
er	Irust & Respect	8	3	2	5	4.00	.423	1.195	1.429	3	2	5	4.00	.5/8	1.069 1.143
\mathbf{C}	Owner/User Participation	8	3	2	5	4.25	.330	1,165	1.357	3	2	5	4.13	.398	1.126 1.268
1	Production of Specified Quality	8	2	3	5	4.13	.350	.991	.982	3	2	5	4.00	.423	1.195 1.429
1	Long-term / Lifecycle Value	8	2	3	5	4.13	.350	.991	.982	3	2	5	3.88	.398	1.126 1.268
1	Time Performance	8	4	1	5	3.50	.535	1.512	2.286	4	1	5	3.38	.498	1.408 1.982
1	Owner's Vision	8	2	5	5	4.13	.350	.991	.982	2	5	5	3.88	.295	.835 .696
	CONSTRUCTADITITY	0	4	1	3	5.05	.470	1.408	1.702	4	1	3	5.03	.470	1.400 1.982

9.6. Round 3 Integrated Project Delivery (IPD)

 Table 9.12. Round 3 -- IPD Relative effectiveness: descriptive statistics



Figure 9.9. . Round 3 -- IPD: relative effectiveness by average mean score

Integrated Project Delivery scored consistently high across all thirteen critical success factors. For self-interest the Designers trended slightly below average, while the rest of the stakeholder groups clustered about the linear average. In project interest the distribution is very tight and scores on average slightly higher, with less variance, than self-interest. The average scores for both self and project interest are both the highest among the three project delivery methods, and with the least variance. The distribution suggests a high degree of agreement among all stakeholder groups.

Out of the three delivery methods the panel displayed the lowest level of agreement when assessing Integrated Project Delivery. Although the range of mean scores significantly varied,

the range of mean scores is the highest ranked overall than either design-bid-build, or design-

build. Additionally for the majority of the critical success factors IPD ranked highest of the three

methods. The rankings will be discussed in further detail below.

Round #3 Inter-Rater Agreement: Integrated Project Delivery (IPD)										
	Self-Interest Project-Interest Mean Rank Mean Rank								erest	
Critical Success Factors		Ν	lean Ra	ınk			Ν	lean Ra	ınk	
	0	D	В	U	All	0	D	В	U	All
Owner / User Satisfaction	6.50	6.94	7.28	7.19	6.91	7.68	8.33	7.89	6.00	7.54
Clear & Realistic Objectives	6.64	5.78	6.28	7.19	6.48	6.61	6.61	6.56	7.50	6.78
Effective Communication	7.89	8.06	7.39	6.69	7.58	8.54	9.78	7.89	6.31	8.23
Collaboration of Project Delivery Team	8.43	7.89	9.28	8.00	8.41	8.18	8.33	9.17	8.63	8.53
Utility & Functionality	6.64	6.67	8.00	7.25	7.08	6.04	7.00	7.89	7.94	7.05
Trust & Respect	6.82	6.67	6.06	6.69	6.59	6.64	7.50	5.94	7.19	6.79
Alignment of Project Objectives	7.46	6.72	6.44	7.25	7.03	6.71	6.33	6.28	8.63	6.91
Production of Specified Quality	6.54	7.28	7.89	7.94	7.29	6.79	6.44	7.78	7.75	7.13
Long-term Bldg. Success & Lifecycle	6.21	6.39	6.33	7.25	6.49	6.82	5.00	6.17	7.19	6.34
Value										
Project Time Performance	7.00	6.50	6.83	7.31	6.91	6.71	5.06	6.00	6.63	6.16
Owner / User Participation	7.25	7.61	6.72	5.44	6.85	6.54	7.44	6.56	5.00	6.44
Owner's vision	5.61	8.78	6.72	7.19	6.89	6.46	7.67	6.56	6.19	6.70
Constructability	8.00	5.72	5.78	5.63	6.51	7.29	5.50	6.33	6.06	6.43
Statistic		Se	elf-Inter	rest			Pro	ject-Int	erest	
Statistic	0	D	В	U	All	0	D	В	U	All
N	14	9	9	8	40	14	9	9	8	40
Kendall's W ^a	.068	.072	.114	.088	.058	.057	.208	.123	.143	.058
Chi-square	11.34	7.81	12.34	8.47	27.82	9.50	22.50	13.27	13.73	27.82
df	12	12	12	12	12	12	12	12	12	12
Asymp. Sig.	.500	.800	.418	.748	.006	.660	.032	.350	.318	.006
a Kandall's Coefficient of Concordance										

9.6.1. Inter-Rater Agreement

a. Kendall's Coefficient of Concordance

 Table 9.13. Round 3 -- IPD: Inter-rater agreement



Figure 9.10. Round 3 -- IPD: Inter-rater agreement by stakeholder group

9.6.2. Correlations

Round	#3: Corr	elation a	mong Stal	keholder	Groups	for IPD Ca	ategory		
Correlation Matrix *		Self-I	nterest			Project	t-Interest		
	Owner	Designer	Builder	User	Owner	Designer	Builder	User	
Owner	1.000	0.088	-0.215	-0.404	1.000	0.162	0.000	-0.247	
Designer		1.000	0.484 ^a	0.027		1.000	0.439 ^a	-0.024	
Builder			1.000	0.537 ^a			1.000	0.275	
User 1.000 1.000									
Madaa									

Notes:

^a Correlation is significant at the 0.05 level (2-tailed).

* Correlation Coefficient based on mean scores, and calculated in SPSS using Pearsons Correlation.

Table 9.14. Round 3 -- IPD: stakeholder correlation matrix

		Correl	ations: Se	lf-Interes	st Integra	ated Proj	ect Deliv	ery (IPD/)				
										Long-			
		Clear &		Collab-			Align-		Product-	term			
n=40	Owner/	Real-	Effect-	oration	Utility		ment of	Owner/	ion of	Success			
11-40	User	istic	ive	of Proj.	&		Project	User	Specif-	& Life-	Time		
	Satisfac-	Object-	Comm-	Delivery	Functio	Trust &	Objec-	Partic-	ied	cycle	Perform-	Owner's	Construct-
	tion	ives	unication	Team	n-ality	Respect	tives	ipation	Quality	Value	ance	Vision	ability
Owner/User Satisfaciton	1.000	0.695	0.569	0.720**	0.729**	0.594	0.666	0.630	0.589**	0.615	0.460	0.639	0.440**
Clear & Realistic Objectives		1.000	0.648**	0.614**	0.615**	0.579**	0.787**	0.664**	0.753**	0.646**	0.387	0.646**	0.454**
Effective Communication			1.000	.769**	0.621**	0.607**	0.690**	0.733**	0.654**	0.553**	0.378*	0.671**	0.446**
Collaboration of Proj. Delivery Tean				1.000	0.716**	0.703**	0.754**	0.673**	0.611**	0.645**	0.394*	0.596**	0.477**
Utility & Functionality					1.000	0.804**	0.729**	0.687^{**}	0.730**	0.729**	0.491**	0.729**	0.396*
Trust & Respect						1.000	0.839**	0.691**	0.814**	0.793**	.0525**	0.727***	0.615**
Alignment of Project Objectives							1.000	0.752^{**}	0.824**	0.777**	0.596**	0.732**	0.577**
Owner/User Participation								1.000	0.822	0.670^{**}	0.521**	0.817**	0.529**
Production of Specified Quality									1.000	0.774^{**}	0.517**	0.820^{**}	0.632**
Long-term Success & Lifecycle Valu										1.000	0.638**	0.752**	0.593**
Time Performance											1.000	0.474**	0.682**
Owner's Vision												1.000	0.409**
Constructability													1.000
**. Correlation is significant at the 0	.01 level	(2-tailed)											
*. Correlation is significant at the 0.0)5 level (2	2-tailed).											
		C l	·				D. I	· · · / II					
	r – – –	Correlat	Ions: Pro	ject-inter	est mieg	rated Pr	oject Del	livery (ir	·D)	Long	1		
		Clear &		Collab-			Alion-		Product-	term			
	Owner/	Real-	Effect-	oration	Utility		ment of	Owner/	ion of	Success			
	User	istic	ive	of Proj.	&		Project	User	Specif-	& Life-	Time		
	Satisfac-	Object-	Comm-	Delivery	Functio	Trust &	Objec-	Partic-	ied	cycle	Perform-	Owner's	Construct-
	tion	ives	unication	Team	n-ality	Respect	tives	ipation	Quality	Value	ance	Vision	ability
Owner/User Satisfaciton	1.000	0.775***	0.667**	0.603**	0.751**	0.626**	0.617**	0.629**	0.506**	0.575**	0.564**	0.616**	0.474**
Clear & Realistic Objectives		1.000	.661**	0.634**	0.618**	0.604**	0.700**	0.593**	0.550	0.667**	0.403**	0.614**	0.489**
Effective Communication			1.000	0.853**	0.747^{**}	0.692**	0.684^{**}	0.688^{**}	0.667^{**}	0.629^{**}	0.460^{**}	0.760^{**}	0.547**
Collaboration of Proj. Delivery Tean				1.000	0 794**	0.784**	0.727**	0.653**	0.684	0.597**	0.355*	0.735**	0.451**
Utility & Functionality					1.000	0.000	0.725**	0.764**	0.717**	0.726**	0.370*	0.773**	0.489**
Trust & Despect						1.000	0.701**	0.682**	0.701**	0.750**	0.414**	0.706**	0.527**
must & Respect							0.791	0.082	0./91	0.750	0.414	0.790	0.327
Ali museut of Dupingt Ohiostings							1 000	0.761**	0.000**	0.742**	0 422**	0.760**	0.460
Alignment of Project Objectives							1.000	0.761**	0.808**	0.743**	0.423**	0.768**	0.469
Alignment of Project Objectives Owner/User Participation							1.000	0.761 ^{**} 1.000	0.808**	0.743 ^{***} 0.677 ^{***}	0.423 ^{**} 0.531 ^{**}	0.768 ^{***} 0.833 ^{***}	0.469
Alignment of Project Objectives Owner/User Participation Production of Specified Quality							1.000 	0.761*** 1.000	0.808 ^{***} 0.783 ^{***} 1.000	0.743 ^{**} 0.677 ^{**} 0.845 ^{**}	0.423 ^{**} 0.531 ^{**} 0.427 ^{**}	0.768 ^{***} 0.833 ^{***} 0.785 ^{***}	0.469 0.629 ^{**} 0.599 ^{**}
Alignment of Project Objectives Owner/User Participation Production of Specified Quality Long-term Success & Lifecycle Valu	 	 	 	 	 		1.000 	0.761 ^{**} 1.000 	0.808 ^{***} 0.783 ^{***} 1.000 	0.743 ^{**} 0.677 ^{**} 0.845 ^{**} 1.000	0.423 ^{***} 0.531 ^{***} 0.427 ^{***} 0.372 [*]	0.768 ^{***} 0.833 ^{***} 0.785 ^{***} 0.742 ^{***}	0.469 0.629 ^{**} 0.599 ^{**} 0.581 ^{**}
Alignment of Project Objectives Owner/User Participation Production of Specified Quality Long-term Success & Lifecycle Valu Time Performance	 	 	 	 	 		1.000 	0.761 ^{**} 1.000 	0.808 ^{***} 0.783 ^{***} 1.000 	0.743 ^{***} 0.677 ^{***} 0.845 ^{***} 1.000 	0.423 ^{***} 0.531 ^{**} 0.427 ^{**} 0.372 [*] 1.000	0.768 ^{**} 0.833 ^{**} 0.785 ^{**} 0.742 ^{**} 0.482 ^{**}	0.469 0.629** 0.599** 0.581** 0.609**
Alignment of Project Objectives Owner/User Participation Production of Specified Quality Long-term Success & Lifecycle Valu Time Performance Owner's Vision	 	 	 	 	 		1.000 	0.761 ^{**} 1.000 	0.808 ^{**} 0.783 ^{**} 1.000 	0.743 ^{**} 0.677 ^{**} 0.845 ^{**} 1.000 	0.423 ^{**} 0.531 ^{**} 0.427 ^{**} 0.372 [*] 1.000 	0.768 ^{***} 0.833 ^{***} 0.785 ^{***} 0.742 ^{***} 0.482 ^{***} 1.000	0.469 0.629** 0.599** 0.581** 0.609** 0.529**
Alignment of Project Objectives Owner/User Participation Production of Specified Quality Long-term Success & Lifecycle Valu Time Performance Owner's Vision Constructability	 	 	 	 	 		1.000 	0.761 ^{**} 1.000 	0.808 ^{**} 0.783 ^{**} 1.000 	0.743 ^{**} 0.677 ^{**} 0.845 ^{**} 1.000 	0.423** 0.531** 0.427** 0.372* 1.000 	0.768 ^{***} 0.833 ^{***} 0.785 ^{***} 0.742 ^{***} 0.482 ^{***} 1.000 	0.469 0.629** 0.599** 0.581** 0.609** 0.529** 1.000

Table 9.15. Round 3 -- IPD: CSF correlation matrix

9.7. Round 3 -- Overall Data (DBB, DB, & IPD)

The trends displayed in Figure 9.15. for self and project interest categories display an overall consistently higher rating for IPD with very little variation below the overall average. DBB and DB display a large degree of variation about the overall average. DB generally rates slightly higher than DBB. DBB mostly falls below the overall average rating for effectiveness, with the exception of the Designer ratings. The data will be further analyzed later in this chapter.



Figure 9.11. Round 3 -- Overall: relative effectiveness by average mean score

Rou	ind #.	3 Inter-	-Rater	Agree	ment:	All (D	BB, D	B, &	IPD)		
			Se	elf-Intere	est			Pro	ject-Inte	rest	
Critical Success Fac	ctors		Ν	lean Ran	k			Ν	1ean Ran	k	
		0	D	В	U	All	0	D	В	U	All
Owner / User	DBB	24.96	26.72	16.83	17.06	21.95	23.07	23.06	14.22	17.75	20.01
Satisfaction	DB	16.64	9.67	22.17	16.38	16.26	19.75	13.06	21.06	24.19	19.43
Sutisfuetion	IPD	24.75	22.22	26.22	29.13	25.39	27.21	26.50	28.39	22.44	26.36
Clear & Realistic	DBB	26.82	31.50	16.89	14.50	23.18	26.71	26.83	14.50	16.06	21.86
Objectives	DB	16.04	16.72	21.61	16.19	17.48	18.32	16.50	22.89	20.56	19.39
objectives	IPD	25.46	19.28	23.72	29.13	24.41	25.14	22.89	25.56	25.94	24.89
Effective	DBB	22.25	22.72	14.67	7.88	17.78	20.14	18.17	15.11	13.44	17.23
Communication	DB	13.14	14.61	22.22	14.50	15.79	13.96	16.50	22.06	21.63	17.89
communeation	IPD	27.79	26.33	25.94	27.50	26.99	29.71	30.94	27.33	23.88	28.29
Collaboration of	DBB	15.64	13.28	11.44	11.81	13.40	11.21	11.94	10.56	12.75	11.54
Project Delivery	DB	12.07	16.28	23.39	17.63	16.68	12.36	16.11	23.33	17.19	16.64
Team	IPD	28.93	26.56	30.67	30.88	29.18	29.57	28.22	30.61	30.31	29.65
	DBB	21.75	26.94	14.22	18.44	20.56	18.11	24.78	15.50	17.88	18.98
Utility &	DB	9.25	15.61	22.28	17.50	15.26	11.96	15.67	22.22	19.00	16.51
Functionality	IPD	25.18	21.56	28.39	28.94	25.84	24.04	24.11	28.39	28.19	25.86
	DBB	14.64	15.00	8.17	14.44	13.23	13.25	11.50	8.56	18.69	12.89
Trust & Respect	DB	11.39	16.50	16.67	13.00	14.05	9.96	14.89	16.50	17.69	14.09
	IPD	25.96	22.61	24.39	26.88	25.04	25.57	25.78	24.39	26.81	25.60
	DBB	22.57	17.61	12.28	13.31	17.29	19.89	18.56	11.11	13.44	16.33
Alignment of Project	DB	8.68	11.67	16.50	14.06	12.19	9.86	14.28	17.00	19.50	14.39
Objectives	IPD	27.18	23 22	24 56	28.94	26.05	25.43	23 50	24.11	30.31	25.68
	DBB	6.96	9.17	19.94	15.44	12.08	24.14	24.89	12.78	10.25	18.98
Production of	DB	24.86	23.28	28.50	30.13	26.38	7.82	9.89	19.83	19.25	13.28
Specified Quality	IPD	22.57	23.22	12.94	16.00	19.24	26.36	23.28	28.50	27.75	26.43
Long-term Bldg.	DBB	26.29	29.61	16.61	21.69	23.94	24.21	27.06	16.56	24.38	23.16
Success & Lifecycle	DB	10.32	15.56	20.78	21.81	16.15	11.36	16.00	20.56	15.13	15.23
Value	IPD	24.39	21.39	24.61	28.94	24.68	26.54	20.44	24.33	26.19	24.60
	DBB	22.93	23.28	16.61	12.19	19.44	20.04	20.83	15.94	14.00	18.09
Project Time	DB	12.79	13.72	18.00	12.81	14.18	13.79	13.33	17.89	12.75	14.40
Performance	IPD	25.82	22.06	25.28	29.25	25.54	25.96	19.72	23.83	24.88	23.86
o (11	DBB	7.50	9.50	8.11	6.94	7.98	8.11	9.28	8.00	6.69	8.06
Owner / User	DB	25.71	20.56	24.94	26.13	24.46	25.46	20.94	25.11	27.00	24.68
Participation	IPD	26.64	25.06	24.94	22.00	24.98	25.75	26.83	24.78	19.13	24.45
	DBB	28.14	30.28	16.22	16.13	23.54	25.86	26.83	16.28	15.88	21.93
Owner's vision	DB	8.75	14.61	20.61	17.50	14.49	10.04	11.89	21.83	13.44	13.79
	IPD	23.25	26.11	24.94	29.31	25.49	25.89	26.06	24.67	25.00	25.48
G	DBB	12.54	17.22	6.67	19.56	13.68	12.86	15.78	7.39	18.44	13.40
Constructability	DB	21.32	18.61	25.11	23.00	21.90	23.29	21.56	24.94	20.31	22.68
	IPD	28.11	20.17	21.94	23.13	23.94	27.29	21.01	23.39	21.94	24.00
			Se	lf-Intere	st			Pro	iect-Inte	rest	
Statistic		0	D	B	U	All	0	D	B	U	All
Ν		14	9	9	8	40	14	9	- 9	- 8	40
Kendall's W ^a		.437	.293	.361	.463	.266	.418	.286	.388	.306	.260
Chi-square		232.44	100.06	123.51	140.68	404.81	222.22	97.90	132.74	93.03	395.76
df		38	38	38	38	38	38	38	38	38	38
Asymp. Sig.		.000	.000	.000	.000	.000	.000	.000	.000	.000	.000
a. Kendall's Coefficien	nt of Co	ncordan	ce								

9.7.1. Round 3 Inter-Rater Agreement

 Table 9.16. Round 3 Inter-rater agreement



Figure 9.12. Round 3 -- Overall: Inter-rater agreement

Overall the panel of experts displayed levels of inter-rater agreement higher than that achieved in the first two rounds of surveys. For the round 3 survey, inter-rater agreement is only to note that the participants continue to demonstrate a measure of consensus. It is not appropriate to directly compare the level of agreement between rounds 1 and 2 with round 3 because the topic of the discussion has changed from relative importance of critical success factors to effectiveness of project delivery methods to achieve the CSFs. The individual stakeholder groups displayed exceptionally high degrees of inter-rater agreement with Kendall's coefficient of concordance scores ranging between 0.286 and 0.463. Inter-rater agreement specific to the panel(s) evaluation of each of the three project delivery method follows.

Intraclass Correlation													
				Rel	iability Sta	tistics							
Perspective	Stakeholder		Cronbach's	95% Confid	ence Interval	F Te	est Tru	ie Valu	e 0				
reispeeuve	Group	Ν	Alpha	Lower	Upper	Value	df1	df2	Sig				
.8				Bound	Bound	v ulter	un	012	NIS.				
	All	40	0.917	0.877	0.949	12.100	41	1599	0.000				
	Owner	14	0.896	0.843	0.937	9.577	41	533	0.000				
Self-Interest	Designer	9	0.710	0.557	0.825	3.447	41	328	0.000				
	Builder	9	0.799	0.694	0.879	4.978	41	328	0.000				
	User	8	0.837	0.750	0.902	6.136	41	287	0.000				
	All	40	0.914	0.873	0.948	11.684	41	1599	0.000				
	Owner	14	0.873	0.808	0.923	7.860	41	533	0.000				
Project-Interest	Designer	9	0.647	0.462	0.788	2.836	41	328	0.000				
	Builder	9	0.813	0.715	0.887	5.343	41	328	0.000				
	User	8	0.715	0.564	0.829	3.513	41	287	0.000				

9.7.2. Round 3 Reliability Testing

Table 9.17. Round 3 Reliability testing

The results of the intraclass correlation calculation indicate a very reliable round 3 survey with Cronbach's Alpha scores of 0.917 and 0.914 respectively for the self-interest and project-interest categories. The stakeholder groups evaluated individually also returned scores ranging from 0.7109 (reasonably reliable) to 0.896 (strongly reliable). Therefore it is validated that the round 3 survey performed reliable in context of both the overall panel of experts, but also when viewed a four separate survey cohorts by stakeholder group.

For the design-bid-build category, there is a large and significant correlation between the owners and the designers, and to a somewhat lesser degree correlation between the designers and builders. Although not statistically significant, the low correlation scores between owners and users may be cause for further investigation. A logical assumption would be that the owners and designers, especially in the cases of the public projects, might have significant positive correlations in their evaluation of the project delivery methods.

Round #3: Correlation among Stakeholder Groups Overall Categories (DBB, DB, & IPD)											
Correlation Matrix *		Self-In	nterest			Project	t-Interest				
	Owner	Designer	Builder	User	Owner	Designer	Builder	User			
Owner	1.000	0.603 ^b	0.208	0.250 ^a	1.000	0.628 ^b	0.340 ^b	0.401 ^b			
Designer		1.000	0.107	0.200		1.000	0.258 ^a	0.280 ^a			
Builder			1.000	0.695 ^b			1.000	0.660 ^b			
User				1.000				1.000			
Notes:											
^a Correlation is sign	ificant at t	he 0.05 lev	vel (2-taile	d).							
^b Correlation is sign	^b Correlation is significant at the 0.01 level (2-tailed).										
* Correlation Coeffi	cient base	d on mean	scores, and	d cal culate	d in SPSS	using Pears	ons Correla	tion. (N=			

 Table 9.18. Round 3 -- Overall: stakeholder correlation matrix

9.8. Significance Testing of Relative Effectiveness

There are two means of evaluating the significance of how each project delivery method is rated for effectiveness in attaining the thirteen critical success factors selected in the round 2 survey. The first means is the direct results of the Delphi process. In the Delphi process the typical major statistics are the measures of central tendency: mean, median, and mode. In literature, median scores based on a Likert scale are favored.¹¹⁴ A more rigorous statistical test is also conducted to provide a more robust argument for the superiority of Integrated Project Delivery. A Friedman test with post-hoc tests is used in this research to test for significance at the 0.05 level.

¹¹⁴ Chia-Chien Hsu and Brian A. Sandford, "The Delphi Technique: Making Sense of Consensus,". *Practical Assessment, Research & Evaluation* 125, no. 10 August 2007, 09 September 2010 http://pareonline.net/pdf/v12n10.pdf>.

9.8.1. Weighting for Relative Importance

The effectiveness ratings are weighted by the relative importance of each of the thirteen critical success factors selected during the round 2 survey. Each sample's thirteen ratings for each factor are multiplied by the weighted average relative importance listed in Table 9.19. Each stakeholder group's weighting is based on their responses as reported in the round 2 survey.

Relative Importance by Weighted Average										
		Self-In	nterest		Project-Interest					
CSFs	O N=17	D N=10	B N=11	U N=13	O N=17	D N=10	B N=11	U N=12		
	Wt Avg	Wt Avg	Wt Avg	Wt Avg	Wt Avg	Wt Avg	Wt Avg	Wt Avg		
Owner / User Satisfaction	0.082	0.086	0.089	0.087	0.078	0.089	0.088	0.086		
Clear & Realistic Objectives	0.084	0.083	0.076	0.079	0.084	0.082	0.075	0.084		
Effective Communication	0.082	0.075	0.082	0.081	0.081	0.080	0.082	0.083		
Utility & Functionality	0.082	0.081	0.067	0.091	0.078	0.072	0.077	0.081		
Collaboration of Project Delivery Team	0.076	0.079	0.076	0.072	0.081	0.087	0.080	0.077		
Trust Respect	0.069	0.077	0.082	0.077	0.078	0.078	0.080	0.075		
Owner / User Participation	0.078	0.079	0.078	0.083	0.074	0.074	0.071	0.073		
Alignment of Project Objectives	0.074	0.069	0.084	0.074	0.079	0.078	0.082	0.075		
Production of Specified Quality	0.075	0.075	0.076	0.077	0.079	0.070	0.080	0.073		
Long-term Building Success & Lifecycle Value	0.079	0.075	0.069	0.075	0.076	0.066	0.075	0.073		
Project Time Performance	0.068	0.069	0.080	0.066	0.071	0.078	0.080	0.073		
Owner's vision	0.081	0.079	0.072	0.074	0.065	0.064	0.073	0.068		
Constructability	0.069	0.077	0.069	0.064	0.076	0.078	0.056	0.077		
Stakeholder Groups Types: $O = Owner: D =$	Designe	r: B = B	Builder: T	J = Use	r: ALL =	= O+D+	B+U			

 Table 9.19. Relative importance weighting from the round 2 survey

Table 9.20. is a summary of the resulting weighted sum aggregation of each participant's total of 78 Likert ratings completed in the round 3 survey. Appendix F contains several tables with all the raw data and weighing calculations resulting in the data in Table 9.20. The weighting of the aggregated data provides two functions. The weighting normalizes the data to ensure the evaluation of relative effectiveness is not skewed due to variation in the sample sizes between stakeholder groups. Additionally, the weighing accounts for the variation in relative importance between the thirteen critical success factors. Logically, an assessment of a project

Relative Effectiveness								
Sample Weighted Sum Aggregated								
•	Se	elf-Intere	est	Project-Interest				
N=40	DBB	DB	IPD	DBB	DB	IPD		
Owner01	3.589	2.678	4.251	3.554	2.681	4.241		
Owner02	2.848	2.523	4.000	2.843	2.537	4.000		
Owner03	3.261	3.166	2.523	3.091	3.162	2.916		
Owner04	3.326	2.693	3.780	3.387	2.531	3.703		
Owner05	4.728	1.501	2.654	4.717	1.518	2.687		
Owner06	3.503	2.450	4.562	3.472	2.294	4.466		
Owner07	4.181	2.586	4.557	3.932	2.903	4.459		
Owner08	4.095	3.630	4.150	3.994	3.638	3.993		
Owner09	3.476	2.411	3.055	2.768	2.840	3.377		
Owner10	1.550	3.085	4.014	1.629	3.000	3.929		
Owner11	3.965	2.854	4.302	3.864	2.760	4.307		
Owner12	3.793	3.288	4.697	3.782	3.382	4.612		
Owner13	3.059	1.414	4.527	1.577	3.006	4.417		
Owner14	3.960	2.839	4.360	3.916	3.325	4.771		
Designer01	3.650	2.517	4.132	3.260	2.819	4.378		
Designer02	4.336	2.695	4.754	3.920	2.777	5.000		
Designer03	3.857	2.711	4.297	3.596	3.085	4.618		
Designer04	4.104	2.605	4.454	3.835	3.296	4.773		
Designer05	4.033	2.923	3.464	4.169	3.062	4.095		
Designer06	3.825	2.623	1.790	3.829	2.638	1.795		
Designer07	3.472	3.303	1.381	3.396	3.294	1.324		
Designer08	3.077	3.454	4.248	2.984	3.712	4.316		
Designer09	2.627	2.831	3.994	2.694	2.907	4.078		
Builder01	3.626	3.217	3.846	3.626	3.204	3.845		
Builder02	3.000	5.000	3.000	3.000	5.000	3.056		
Builder03	1.604	3.441	4.314	1.637	3.441	4.321		
Builder04	2.271	4.091	4.524	1.707	4.079	4.727		
Builder05	2.771	3.848	3.931	2.793	3.865	3.948		
Builder06	2.818	3.775	4.236	3.234	4.009	4.475		
Builder07	2.764	3.546	4.550	2.791	3.546	4.507		
Builder08	3.000	3.000	3.000	3.000	3.000	3.000		
Builder09	3.089	3.058	4.593	2.940	2.970	4.594		
User01	2.858	3.000	2.926	2.853	3.394	3.393		
User02	3.558	3.825	4.768	3.767	3.688	4.697		
User03	1.843	2.358	5.000	1.670	2.783	5.000		
User04	1.977	4.000	4.208	2.653	3.925	3.831		
User05	2.526	3.589	4.926	2.530	3.604	4.932		
User06	2.779	2.683	3.253	3.154	2.772	3.024		
User07	3.609	2.577	4.598	3.688	3.339	4.040		
User08	1.742	1.753	2.740	2.620	2.495	2.378		

 Table 9.20. Round 3 - Relative effectiveness aggregated by weighted sums

delivery method's relative effectiveness must account for the importance of the factors. With all other things being equal, a process which maximizes the most important factors is logically more effective than a process which maximizes only the least important factors. Using individual weighting factors for each stakeholder group accounts for different perceptions between stakeholder groups and provides a macro-view assessment of the project delivery methods which is representative of the entire project delivery team.

9.8.2. Friedman Test with Post-hoc Tests (Wilcoxon Signed-Rank Test)

The Friedman Test is the non-parametric alternative to the F-test used in a one-way ANOVA with repeated measures. The Friedman test indicates whether or not a significant difference between the relative effectiveness of the three project delivery methods exists, but the test does not pin-point which project delivery methods in particular differ from each other. Additional testing is required to determine if IPD is effectively superior to the other methods for attaining the selected critical success factors.

To test the individual differences (IPD-DBB, IPD-DB, & DBB-DB) a post-hoc test is run using Wilcoxon Signed-Rank Tests on the three different combinations of groups. The post-hoc tests should only be run if the Friedman Test was statistically significant.

Because making multiple comparisons increases the likelihood of making a Type I error (declaring a result significant when it is not) a Bonferroni adjustment is necessary to ensure that the overall confidence level is high. The Bonferroni adjustment uses a more stringent confidence level for each interval, and is calculated by dividing the desired significance level (0.05) by the number of tests (3 = one for each comparison).¹¹⁵ For this study the Bonferroni adjusted significance level is 0.008 (one-tailed). One-tailed significance is used because I predict that the relative effectiveness of IPD is greater than the relative effectiveness of DBB or DB. The null

¹¹⁵ Alan Agresti and Barbara Finlay, *Statistical Methods for the Social Sciences* (Upper Saddle River, NJ: Prentice Hall, 2009), 377.

hypothesis is no difference exists between the efficacy of IPD and DBB/DB. The alternate hypothesis is that IPD is more effective than DBB/DB.

9.8.3. Statistical Significance: Self-Interest

The Friedman Test indicates that in the Self-Interest category IPD has the highest median rank overall, as well as in each the upper and lower quartiles¹¹⁶. The Freidman test is statistically significant ($X^2(2)=30.697$, P < 0.001), therefore post-hoc tests may be run to determine if a significant difference exists between each pair of the project delivery methods.

The Wilcoxon Signed-Rank Test, the post-hoc test, is the nonparametric equivalent to the dependent t-test, and is used to compare two sets of scores that come from the same participants. In this case the sets of scores are paired comparisons of each of the project delivery methods evaluated by the panel of experts for efficacy in attaining the 13 critical success factors.

Significance Test Relative Effectiveness: <u>Self-Interest</u>										
]	Descript	tive Statis	tics						
Project Delivery Method		Percentiles				Friedman Test				
Project Denvery Method	Ν	25th	50th	75th	Mean					
			(Median)		Rank	Statistics				
Design-Bid-Build (DBB)	40	2.773	3.294	3.817	1.79	Chi-Square	30.697			
Design-Build (DB)	40	2.591	2.889	3.451	1.53	df	2			
Integrated Project Delivery (IPD)	40	3.306	4.222	4.544	2.69	Exact Significance	.000			

Table 9.21. Round 3 - Self-Interest: Friedman Test descriptive statistics, mean ranks, and test statistics

¹¹⁶ Quartile is described as "percentile" in the SPSS output tables.

Wilcoxon Signed Ranks Test: <u>Self-Interest</u>								
	I	Ranks			Exact Si	gnificance		
Matched- Pairs		Ν	Mean Rank	Sum of Ranks	Z	(1-tailed)		
	Negative Ranks	6 ^a	22.00	132.00				
	Positive Ranks	32 ^b	19.03	609.00	2 450 İ	000 *		
IPD - DBB	Ties	2 ^c			-3.439 ³	.000		
	Total	40						
	Negative Ranks	5 d	18.00	90.00				
	Positive Ranks	34 ^e	20.29	690.00	4 10 C İ	000 *		
IPD - DB	Ties	1 f			-4.180 5	.000		
	Total	40						
	Negative Ranks	24 g	20.00	480.00		.107		
	Positive Ranks	15 ^h	20.00	300.00	1.256 k			
DR - DRR	Ties	1 ⁱ			-1.230			
	Total	40						
Notes:	·	·			Notes:			
a. IPD $< D$	BB				j. Based or	n negative ranks.		
b. IPD $> L$)BB				k. Based of	n positive ranks.		
c. IPD = L)B)RR				* Significa	ant at a 0.08		
$a \cdot I \cdot D < I$ $e \cdot I \cdot P \cdot D > D$)B)B	level (Bonfe	arroni adjusted					
f. IPD = DB					from a .025	significance		
g. DB $<$ D	BB				level (1-taile	ed))		
h. DB $>$ D	BB							
i DB = DI	BB							

 Table 9.22. Round 3 -- Self-Interest: Post-Hoc Wilcoxon Signed Ranks Test

Overall Self-Interest Findings:

There was a statistically significant difference in which project delivery method was perceived most effective for achieving the 13 critical success factors, $X^2(2) = 30.679$, P < 0.001. Post-hoc analysis with Wilcoxon Signed-Rank Tests was conducted with a Bonferroni correction applied, resulting in a significance level set at P < 0.008 (1-tailed). Median perceived efficacy levels for Integrated Project Delivery (IPD), Design-Bid-Build (DBB), and Design-Build (DB) were 4.222, 3.294, and 2.889 respectively. There were significant differences between IPD and DBB (Z = -3.459, P < 0.001), and between IPD and DB (Z = -4.186, P < 0.001). There was not a significant difference between DBB and DB (Z = -1.256, P=0.107). The null hypothesis is rejected in the self-interest category, and it is concluded that IPD is perceived by the overall panel of experts as significantly more effective than either DBB or DB to successfully attain the 13 critical success factors.

Table 9.23. provides the Friedman Test results for each of the individual stakeholder groups in the self-interest category. In the median and upper quartiles all four stakeholder groups rated IPD with highest efficacy. In the lower quartile the Owners, Builders, and Users each rated IPD highest. In the lowest quartile the Designers rated DBB over IPD, but it must be noted that the Designer stakeholder group is the only stakeholder group which does not display statistical significance. The Owners, Builders, and Users each display statistical significance at the 0.05 level. Table 9.23. provides the results of the post-hoc tests.

Significance Test Relative Effectiveness: <u>Self-Interest</u>										
			Descr	iptive Sta	tistics		Friedman Test			
Stakeholder	N	Project Delivery Method	I	Percentile	s					
	IN	r roject Denvery Method	25th 50th 75th		Mean	Statistics				
				(Median)		Rank	Statistics			
		Design-Bid-Build (DBB)	3.211	3.546	3.998	2.14	Chi-Square	17.714		
Owner	14	Design-Build (DB)	2.440	2.686	3.105	1.14	df	2		
		Integrated Project Delivery (IPD)	3.599	4.201	4.535	2.71	Exact Significance	.000		
		Design-Bid-Build (DBB)	3.275	3.825	4.069	2.11	Chi-Square	4.667		
Designer	9	Design-Build (DB)	2.614	2.711	3.113	1.44	df	2		
		Integrated Project Delivery (IPD)	2.627	4.132	4.376	2.44	Exact Significance	.107		
		Design-Bid-Build (DBB)	2.518	2.818	3.045	1.39	Chi-Square	9.484		
Builder	9	Design-Build (DB)	3.138	3.546	3.970	1.89	df	2		
		Integrated Project Delivery (IPD)	3.423	4.236	4.537	2.72	Exact Significance	.006		
		Design-Bid-Build (DBB)	1.877	2.653	3.383	1.25	Chi-Square	10.750		
User	8	Design-Build (DB)	2.413	2.842	3.766	1.88	df	2		
		Integrated Project Delivery (IPD)	3.008	4.403	4.887	2.88	Exact Significance	.002		

 Table 9.23. Round 3 -- Self-Interest: Friedman Test descriptive statistics, mean ranks, and test statistics by stakeholder group

Wilcoxon Signed Ranks Test: <u>Self-Interest</u>									
			Ranks			Exact Sig	gnificance		
Stakeholder	Matched- Pairs		Ν	Mean Rank	Sum of Ranks	Z	1-tailed		
	IPD - DBB	Negative Ranks Positive Ranks Ties	3 a 11 b 0 c	8.67 7.18	26.00 79.00	-1.664 ^j	.052		
Owner	IPD - DB	Negative Ranks Positive Ranks Ties	1 d 1 d 1 3 e 0 f	2.00 7.92	2.00 103.00	-3.170 ^j	.000 *		
	DB - DBB	Negative Ranks Positive Ranks Ties	13 g 1 h 0 i 14	7.23	94.00 11.00	-2.605 ^k	.003 *		
	IPD - DBB	Negative Ranks Positive Ranks Ties Total	$\begin{array}{c} 3 \\ 3 \\ 6 \\ 0 \\ 0 \\ 9 \end{array}$	7.33 3.83	22.00 23.00	059 ^j	.500		
Designer	r IPD - DB	Negative Ranks Positive Ranks Ties Total	2 d 7 e 0 f 9	5.50 4.86	11.00 34.00	-1.362 ^j	.102		
	DB - DBB	Negative Ranks Positive Ranks Ties Total	7 g 2 h 0 i 9	5.71 2.50	40.00 5.00	-2.073 ^k	.020		
	IPD - DBB	Negative Ranks Positive Ranks Ties Total	0 a 7 b 2 c 9	.00 4.00	.00 28.00	-2.366 ^j	.008 *		
Builder	IPD - DB	Negative Ranks Positive Ranks Ties Total	1 d 7 e 1 f 9	8.00 4.00	8.00 28.00	-1.400 ^j	.098		
	DB - DBB	Negative Ranks Positive Ranks Ties Total	2 g 6 h 1 i 9	1.50 5.50	3.00 33.00	-2.100 ^j	.020		
	IPD - DBB	Negative Ranks Positive Ranks Ties Total	0 a 8 b 0 c 8	.00 4.50	.00 36.00	-2.521 ^j	.004 *		
User	IPD - DB	Negative Ranks Positive Ranks Ties Total	1 d 7 e 0 f 8	1.00 5.00	1.00 35.00	-2.380 ^j	.008 *		
	DB - DBB	Negative Ranks Positive Ranks Ties Total	2 g 6 h 0 i 8	4.00 4.67	8.00 28.00	-1.400 ^j	.098		
Notes: a. IPD < DBB b. IPD > DBB c. IPD = DBB d. IPD < DB e. IPD > DB		f. IPD = DB g. DB < DBB h. DB > DBB i. DB = DBB				Notes: j. Based on ne k. Based on po * Significant (Bonferroni adju significance l	gative ranks. sitive ranks. at a .008 level usted from a .025 evel (1-tailed))		

Table 9.24. Round 3 -- Self-Interest: Post-Hoc Wilcoxon Signed Ranks Test by stakeholder group

Stakeholder Self-Interest Findings (See tables 9.23 and 9.24):

Owners: There was a statistically significant difference in which project delivery method was perceived most effective for achieving the 13 critical success factors, $X^2(2) =$ 17.714, P < 0.001. Post-hoc analysis with Wilcoxon Signed-Rank Tests was conducted with a Bonferroni correction applied, resulting in a significance level set at P < 0.008 (1-tailed). Median perceived efficacy levels for Integrated Project Delivery (IPD), Design-Bid-Build (DBB), and Design-Build (DB) were 4.201, 3.546, and 2.686 respectively. There were significant differences between IPD and DBB (Z = - 3.170, P < 0.001), and between DBB and DB (Z = - 2.605, P =0.003). There was not a significant difference between IPD and DBB (Z= -1.664, P=0.052). The null hypothesis is rejected for the comparison of IPD and DB. In the selfinterest category it is concluded that IPD is perceived by the Owners as significantly more effective than DB to successfully attain the 13 critical success factors evaluated. Although the null hypothesis for IPD-DBB cannot be rejected at the 0.008 significance level, the Delphi process indicates Owner consensus, based on the median score, that IPD is perceived as more effective than DBB.

Designers: There was not a statistically significant difference in which project delivery method was perceived most effective for achieving the 13 critical success factors, $X^2(2) = 4.667$, P =0.107. Because statistical significance is not achieved in the Friedman Test, Post-hoc analysis with Wilcoxon Signed-Rank Tests is not relevant. Median perceived efficacy levels for Integrated Project Delivery (IPD), Design-Bid-Build (DBB), and Design-Build (DB) were 4.132, 3.825, and 2.711 respectively. Based on median scores the Delphi process indicates a consensus among the Designer's that IPD is more effective than either DBB or DB. **Builders**: There was a statistically significant difference in which project delivery method was perceived most effective for achieving the 13 critical success factors, $X^2(2) = 9.484$, P =0.006. Post-hoc analysis with Wilcoxon Signed-Rank Tests was conducted with a Bonferroni correction applied, resulting in a significance level set at P < 0.008 (1-tailed). Median perceived efficacy levels for Integrated Project Delivery (IPD), Design-Bid-Build (DBB), and Design-Build (DB) were 4.236, 3.546, and 2.818 respectively. There were significant differences between IPD and DBB (Z = - 2.366, P =0.008). There were not significant differences between IPD and DB (Z=-1.400, P=0.098) or between DBB and DB (Z = - 2.100, P =0.020). The null hypothesis is rejected for the comparison of IPD and DBB. In the self-interest category it is concluded that IPD is perceived by the Builders as significantly more effective than DBB to successfully attain the 13 critical success factors evaluated. Although the null hypothesis for IPD-DB cannot be rejected at the 0.008 significance level, the Delphi process indicates Builder consensus, based on the median score, that IPD is perceived as more effective than DB.

Users: There was a statistically significant difference in which project delivery method was perceived most effective for achieving the 13 critical success factors, $X^2(2) = 10.750$, P = 0.002. Post-hoc analysis with Wilcoxon Signed-Rank Tests was conducted with a Bonferroni correction applied, resulting in a significance level set at P < 0.008 (1-tailed). Median perceived efficacy levels for Integrated Project Delivery (IPD), Design-Bid-Build (DBB), and Design-Build (DB) were 4.403, 2.653, and 2.842 respectively. There were significant differences between IPD and DBB (Z = - 2.380, P = 0.004), and between IPD and DB (Z = - 2.380, P =0.008). There was not a significant difference between DB and DBB (Z=-1.40, P=0.098). The null hypothesis is rejected for the comparison of IPD and DBB/DB. In the self-interest category it is concluded that IPD is perceived by the Users as significantly more effective than either DBB or DB to successfully attain the 13 critical success factors evaluated.

9.8.4. Statistical Significance: Project-Interest

The Friedman Test indicates that in the Project-Interest category IPD has the highest mean rank overall, as well as in each the upper and lower quartiles. The Freidman test is statistically significant ($X^2(2)=22.615$, P < 0.001), therefore post-hoc tests may be run to determine if a significant difference exists between each pair of the project delivery methods.

The Wilcoxon Signed-Rank Test, the post-hoc test, is the nonparametric equivalent to the dependent t-test, and is used to compare two sets of scores that come from the same participants. In this case the sets of scores are paired comparisons of each of the project delivery methods evaluated by the panel of experts for efficacy in attaining the 13 critical success factors.

Significance Test Relative Effectiveness: <u>Project-Interest</u>										
]	Descriptive Statistics Percentiles				Friedman Test				
Project Delivery Method	N	25th	50th (Median)	75th	Mean Rank	Statistics				
Design-Bid-Build (DBB)	40	2.774	3.194	3.778	1.78	Chi-Square 22.615				
Design-Build (DB)	40	2.779	3.074	3.520	1.63	df 2				
Integrated Project Delivery (IPD)	40	3.471	4.168	4.572	2.60	Exact Significance .000				

Table 9.25. Round 3 - Project-Interest: Friedman Test descriptive statistics, mean ranks, and test statistics

Overall Project-Interest Findings:

There was a statistically significant difference in which project delivery method was perceived most effective for achieving the 13 critical success factors, $X^2(2) = 22.615$, P < 0.001.

Post-hoc analysis with Wilcoxon Signed-Rank Tests was conducted with a Bonferroni correction applied, resulting in a significance level set at P < 0.008 (1-tailed). Median perceived efficacy levels for Integrated Project Delivery (IPD), Design-Bid-Build (DBB), and Design-Build (DB) were 4.168, 3.194, and 3.074 respectively. There were significant differences between IPD and DBB (Z = -3.824, P < 0.001), and between IPD and DB (Z = -4.075, P < 0.001). There was not a significant difference between DBB and DB (Z = -0.098, P = 0.464). The null hypothesis is rejected in the project-interest category, and it is concluded that IPD is perceived by the overall panel of experts as significantly more effective than either DBB or DB to successfully attain the 13 critical success factors.

Wilcoxon Signed Ranks Test: Project-Interest							
	I	Ranks		Exact Sig	gnificance		
Matched- Pairs		Ν	Mean Rank	Sum of Ranks	Z	(1-tailed)	
	Negative Ranks	8 a	14.50	116.00		.000 *	
IPD - DBB	Positive Ranks	31 b	21.42	664.00	3 824 İ		
	Ties	1 c			-3.024 5		
	Total	40					
	Negative Ranks	7 d	14.00	98.00			
מה חפ	Positive Ranks	32 e	21.31	682.00	4 075 İ	000 *	
IPD - DB	Ties	1 f			-4.075 *	.000	
	Total	40					
	Negative Ranks	22 g	18.05	397.00			
	Positive Ranks	17 h	22.53	383.00	- 098 k	.464	
00-000	Ties	1 i			070		
	Total	40					
Notes:					Notes:		
a. IPD < D	BB				j. Based on r	egative ranks.	
b. IPD $> D$	DBB				k. Based on j	positive ranks.	
c. IPD = D	DBB						
d. IPD $< D$)B						
e. IPD $> D$)B				* Significant	tata .008	
f. IPD = DB					level (Bonferr	oni adjusted	
g. DB < D	BB				from a .025 si	gnificance	
h. DB $>$ D	BB				level (1-tailed))	
i DB = DI	BB						

 Table 9.26. Round 3 -- Project-Interest: Post-Hoc Wilcoxon Signed Ranks Test

Table 8.27. provides the Friedman Test results for each of the individual stakeholder groups in the project-interest category. IPD is rated as most effective by all the stakeholders in each quartile with the exception of the Designers where DBB is highest rated in the lower quartile. The Owners and Builders each display statistical significance at the 0.05 level. The Designers and Users failed to achieve the 0.050 significance level. Only the Owners and Builders will be taken into consideration in the post-hoc testing. Table 9.26. provides the results of the post-hoc tests.

Significance Test Relative Effectiveness: <u>Project-Interest</u>									
	N	N Project Delivery Method		<u>iptive Sta</u> Percentile	tistics s		Friedman Test		
Stakeholder	N			50th (Median)	75th	Mean Rank	Statistics		
		Design-Bid-Build (DBB)	2.824	3.513	3.920	1.93	Chi-Square	13.000	
Owner	14	Design-Build (DB)	2.536	2.872	3.203	1.36	df	2	
		Integrated Project Delivery (IPD)	3.622	4.121	4.461	2.71	Exact Significance	.001	
		Design-Bid-Build (DBB)	3.122	3.596	3.878	2.11	Chi-Square	4.667	
Designer	9	Design-Build (DB)	2.798	3.062	3.295	1.44	df	2	
		Integrated Project Delivery (IPD)	2.937	4.316	4.696	2.44	Exact Significance	.107	
		Design-Bid-Build (DBB)	2.249	2.940	3.117	1.22	Chi-Square	12.250	
Builder	9	Design-Build (DB)	3.102	3.546	4.044	2.00	df	2	
		Integrated Project Delivery (IPD)	3.451	4.321	4.551	2.78	Exact Significance	.001	
		Design-Bid-Build (DBB)	2.553	2.753	3.555	1.75	Chi-Square	1.750	
User	8	Design-Build (DB)	2.775	3.367	3.667	1.88	df	2	
		Integrated Project Delivery (IPD)	3.116	3.936	4.873	2.38	Exact Significance	.531	

 Table 9.27. Round 3 -- Project-Interest: Friedman Test descriptive statistics, mean ranks, and test statistics by stakeholder group

Wilcoxon Signed Ranks Test: <u>Project-Interest</u>									
	Matched_		Ranks			Exact Significance			
Stakeholder	Pairs		N	Mean	Sum of	7	1 tailed		
	1 411 5		11	Rank	Ranks	L	1-taileu		
		Negative Ranks	3 a	5.00	15.00				
	IPD - DBB	Positive Ranks	11 b	8.18	90.00	-2.354 ^j	.008 *		
		Ties	0 c						
		Total	14	1.00	1.00				
		Negative Ranks	12 0	1.00	104.00		*		
Owner	IPD - DB	Positive Ranks	13 e	8.00	104.00	-3.233 ^J	.000 *		
		Ties Total	14	•					
		Negative Ranks	14 10 g	7 70	77.00				
		Positive Ranks	10 S	7.00	28.00	k see k			
	DB - DBB	Ties	0 i	7.00	20.00	-1.538b *	.068		
		Total	14	•					
		Negative Ranks	3 a	6.00	18.00				
		Positive Ranks	6 b	4.50	27.00	522 İ	226		
	IPD - DBB	Ties	0 c			533 ^J	.326		
		Total	9						
		Negative Ranks	2 d	5.00	10.00				
Designer	פת תפו	Positive Ranks	7 e	5.00	35.00	1 481 1	082		
Designer	IF D - DB	Ties	0 f			-1.401 5	.082		
		Total	9						
		Negative Ranks	7 g	5.29	37.00				
	DB - DBB	Positive Ranks	2 h	4.00	8.00	-1 718 ^k	049		
		Ties	0 i			1.710	.019		
		Total	9						
	IPD - DBB	Negative Ranks	() a	.00	.00		-1-		
		Positive Ranks	80	4.50	36.00	-2.521 ^j	.004 *		
		Ties							
		I otal	9	8.00	8.00				
		Regative Ranks	7.6	<u>8.00</u> 4.00	28.00	;			
Builder	IPD - DB	Tion	/ c	4.00	28.00	-1.400 ^J	.098		
		Total	9	ł					
		Negative Ranks	1 g	2.00	2.00				
		Positive Ranks	7 h	4.86	34.00	a a co i			
	DB - DBB	Ties	1 i			-2.240 J	.012		
		Total	9	1					
		Negative Ranks	2 a	1.50	3.00				
	פסת חקו	Positive Ranks	6 b	5.50	33.00	2 100 İ	020		
	и р - рор	Ties	0 c			-2.100 5	.020		
		Total	8						
		Negative Ranks	3 d	2.00	6.00				
User	IPD - DB	Positive Ranks	5 e	6.00	30.00	-1 680 ^j	055		
USUI		Ties	0 f			1.000	.000		
		Total	8		10.01				
		Negative Ranks	4 g	2.50	10.00				
	DB - DBB	Positive Ranks	4 h	6.50	26.00	-1.120 ^j	.156		
		Ties Total	01						
Notes:		10181	δ			Notos:			
a IPD < DD	в	f IPD - DP				i Based on no.	native ranks		
h IPD > DR	B	$\sigma DR < DRR$				k. Based on po	sitive ranks		
c. IPD = DB	B	h DB > DBR				* Significant at	a 008 level		
d. IPD < DB		i DB = DBB				(Bonferroni adiu	sted from a 025		
e. IPD >DB						significance level	1(1-tailed))		

 Table 9.28. Round 3 -- Project-Interest: Post-Hoc Wilcoxon Signed Ranks Test by stakeholder group

Stakeholder Project-Interest Findings (See tables 9.27 and 9.28):

Owners: There was a statistically significant difference in which project delivery method was perceived most effective for achieving the 13 critical success factors, $X^2(2) =$ 13.000, P = 0.001. Post-hoc analysis with Wilcoxon Signed-Rank Tests was conducted with a Bonferroni correction applied, resulting in a significance level set at P < 0.008 (1-tailed). Median perceived efficacy levels for Integrated Project Delivery (IPD), Design-Bid-Build (DBB), and Design-Build (DB) were 4.121, 3.513, and 2.872 respectively. There were significant differences between IPD and DBB (Z = - 2.354, P = 0.008), and between IPD and DB (Z = - 3.233, P < 0.001). There was not a significant difference between DB and DBB (Z = -1.538, P=0.068). The null hypothesis is rejected for the comparison of IPD and DBB/DB. In the project-interest category it is concluded that IPD is perceived by the Owners as significantly more effective than DBB or DB to successfully attain the 13 critical success factors evaluated.

Designers: There was not a statistically significant difference in which project delivery method was perceived most effective for achieving the 13 critical success factors, $X^2(2) = 4.667$, P =0.107. Because statistical significance is not achieved in the Friedman Test, Post-hoc analysis with Wilcoxon Signed-Rank Tests is not relevant. Median perceived efficacy levels for Integrated Project Delivery (IPD), Design-Bid-Build (DBB), and Design-Build (DB) were 4.316, 3.596, and 2.798 respectively. Although the null hypothesis for IPD and DBB/DB cannot be evaluated for the Designer's stakeholder group in the project-interest category, the Delphi process indicates a consensus among the Designer's that IPD is more effective than either DBB or DB based on the median quartile.

Builders: There was a statistically significant difference in which project delivery method was perceived most effective for achieving the 13 critical success factors, $X^2(2) =$

231

12.250, P =0.001. Post-hoc analysis with Wilcoxon Signed-Rank Tests was conducted with a Bonferroni correction applied, resulting in a significance level set at P < 0.008 (1-tailed). Median perceived efficacy levels for Integrated Project Delivery (IPD), Design-Bid-Build (DBB), and Design-Build (DB) were 4.321, 2.940, and 3.546 respectively. There were significant differences between IPD and DBB (Z = -5.521, P =0.004). There were not significant differences between IPD and DB (Z = -1.400, P=0.098) or between DBB and DB (Z = -2.240, P =0.012). The null hypothesis is rejected for the comparison of IPD and DBB. In the project-interest category it is concluded that IPD is perceived by the Builders as significantly more effective than DBB to successfully attain the 13 critical success factors evaluated. Although the null hypothesis for IPD-DB cannot be rejected at the 0.008 significance level, the Delphi process indicates Builder consensus, based on the median score, that IPD is perceived as more effective than DB.

Users: There was not a statistically significant difference in which project delivery method was perceived most effective for achieving the 13 critical success factors, $X^2(2) = 1.750$, P = 0.531. Because statistical significance is not achieved in the Friedman Test, Post-hoc analysis with Wilcoxon Signed-Rank Tests is not relevant. Median perceived efficacy levels for Integrated Project Delivery (IPD), Design-Bid-Build (DBB), and Design-Build (DB) were 3.936, 2.753, and 3.367 respectively. Although the null hypothesis for IPD and DBB/DB cannot be evaluated for the User's stakeholder group in the project-interest category, the Delphi process indicates a consensus among the User's that IPD is more effective than either DBB or DB based on the median quartile.

232



9.8.5. Comparison of Mean Effectiveness Distributions

Figure 9.13. Round 3 comparison of project delivery methods' average mean score distributions

Figure 9.14 compares the distribution of mean scores for relative effectiveness for each of the three delivery methods. It is clear that IPD is consistently rated the most effective overall and displays the least variance about the overall average trend line as visual proof there was significant agreement among the panel of expert.

9.8.6. Overall Comparisons of Effectiveness Ratings

Figure 9.15 the overall average mean scores and trend lines are plotted for self-interest and project interest. It is extremely clear that IPD is consistently rated more effective than DBB or DB for effectiveness in achieving the 13 CSF's.



Figure 9.14. Round 3 -- Overall: Comparison of relative effectiveness by project delivery method

9.8.7. Comparison of Median Weighted Averages

Figure 9.16 plots the median weighted average for each stakeholder group by project delivery method. It is significant that the stakeholder median scores for IPD are rated highest and have the tightest distribution cluster about the overall average score. Not only does the panel of experts have the highest degree of consensus in evaluating IPD, but they also rate the perceived effectiveness of IPD higher than either DBB or DBB. The Friedman Tests discussed above indicates that the comparison of the data plotted in Figure 9.16. is highly significant (P<0.001).



Figure 9.15. Median Weighted Averages (with standard deviation)

9.8.8. Implementability of IPD

Inferred from analysis of Figures 9.14 through 9.16., and reinforced by the significance testing, is that IPD is perceived as a superiorly effective project delivery method compared to either DBB or DB in this study. Another important inference which may be drawn from the data is a measure of implementability or executionability of the project delivery methods. The effective implementation of the project delivery method can influence project success and further realize value-creation for the project.¹¹⁷ Kim argues that "no plan, no matter how good it appears can be good unless it is implementable".¹¹⁸ Assuming IPD is feasible logically, theoretically, technically, and economically there remains one measure of feasibility critical to implementability: acceptance, or buy-in, by the project stakeholders. The acceptance may be viewed as *political* feasibility.¹¹⁹ For a multi-party party collaboration, such as IPD, to be politically feasible each party must perceive a benefit for themselves. If there is no self-interest benefit, there is no incentive for the individual parties to accept the strategy despite an overall benefit to project-interest. The results of this research indicate a *political* consensus among the project stakeholders that IPD is significantly more effective than the other project delivery methods, but more importantly the inter-group differences are the least. The data indicates that IPD is politically feasible among the various stakeholder groups necessary to successfully accomplish the project. In the context of large, complex projects, the research data shown in Figure 9.16. suggests IPD is more implementable for both self-interest and project-interest than DBB or DB.

¹¹⁷ Li Zhai, Yanfei Xin, and Chaosheng Cheng, "Understanding the Value of Project Management From a Stakeholder's Perspective: Case Study of Mega-Project Management," *Project Management Journal* 40, no. 1 (March 2009): 101-02.

¹¹⁸ Kim, "Countermodeling as a Strategy for Decision Making: Epistemological Problems in Design", 27.

¹¹⁹ Kim, 1980, 30.

While it is important that the method be implementable for project-interest, it is even more important for self-interest. As discussed earlier in the dissertation, collaborative game theory calls for a superadditive relationship where optimization of individual stakeholder utility results in higher group utility. A more collaborative project delivery method, such as IPD, provides the framework where each stakeholder may be internally motivated by what is good for them, while externally benefitting the other stakeholders. In practice all parties attempt to portray a "team first" or "project first" attitude, but business survival in the real-world requires self-interest first. To establish the best project team possible to ensure project success, the method must be perceived in self-interest as implementable. Otherwise there is little incentive for the best qualified design and construction business (and individuals) to commit to the IPD process no matter what benefits it may provide to owner and users.

9.9. Round 3 Findings

 The panel of experts was sufficiently familiar with the three project delivery methods to assess relative effectiveness of DBB, DB, and IPD to achieve the thirteen critical success factors selected in rounds 1 and 2. Two participants from the users group were screened out due to failing to meet the high familiarity criterion for participation. Surveys 1 and 2 also revealed that out of 40 CSF's, 13 CSFs or 32.5% were determined to be most significant in importance to project success.

237

- The round 3 survey tested as highly reliable with a Cronbach's Alpha scores of 0.917 for self-interest, and 0.914 for project-interest at a .001 level of significance.
- The round 3 survey tested positive for inter-rater agreement with minimum overall Kendall's W scores of 0.260 for the overall panel of experts, 0.418 for e owners, 0.285 for designers, 0.361 for builders, and 0.306 for users all at 0.001 significance.
- 4. The Friedman Test indicated overall statistical significance for both the selfinterest category ($X^2(2)=30.697$, P < 0.001) and the project-interest category ($X^2(2) = 22.615$, P < 0.001). Post-hoc Wilcoxon Signed Ranks Tests determined statistical significance greatly exceeding the Bonferroni adjusted 0.008 level. It is with strong significance that the superiority of IPD effectiveness over DBB and DB is perceived by the overall panel of experts.
- 5. Analysis of the aggregated median weighted average effectiveness scores (weighted by relative importance) indicates an overwhelming consensus that IPD is perceived more effective than either DBB or DB. IPD displayed the highest median scores in every stakeholder group for both self-interest and project interest. Additional proof of statistical significance is provided by individual stakeholder ratings of relative effectiveness as follows: Owner self-interest IPD over DB (P<0.001), Builder self-interest IPD over DBB

(P<0.008), User self-interest IPD over DBB/DB (P=0.004/0.008), Owner project-interest IPD over DBB/DB (P=0.008/0.000), and Builder project-interest IPD over DBB (P=0.004).

- 6. Based on the findings of the round 3 survey it is strongly concluded that the Delphi panel(s) perceive Integrated Project Delivery as superiorly effective in achieving the thirteen critical success factors in comparison to the design-bidbuild and design-build project delivery methods.
- IPD is logically perceived as more implementable than DBB or DB for attainment of the thirteen factors evaluated. The implementability inference is confined to the research context for large, complex project types.

CHAPTER 10.

CONCLUSIONS AND FUTURE RESEARCH

10.1. Summary of Work

This dissertation has described Integrated Project Deliver as a superior project delivery method for effective value-creation in the context of large complex military construction projects such as the IDBB pilot-projects. The relationship between project goals and value-creation was established as a basis for correlating attainment of critical success factors with evaluating the effectiveness of project delivery methods to successfully create value. A Delphi survey method was modified by expanding the panel of experts to include significant additional sub-panels representing each of the key stakeholder groups (Owner, Designer, Builder, and User) required to provide a macro-level analysis of the project delivery process rather than a micro-level analysis of any single stakeholder group. The survey successfully identified critical success factors representative of the overall project delivery process, and additionally provided an assessment of relative importance of the CSFs. The survey data identified significant differences in perceptions of relative importance between the different project stakeholder groups, and also identified significant differences internal to each stakeholder groups' perceptions of self-interest and project-interest. Taking into consideration the multiple perceptive viewpoints of all project stakeholders from a macro-viewpoint allowed an evaluation of project effectiveness not focused on any single micro-aspect of the project delivery process. The macro-viewed assessment concluded that the collected data corroborated the hypothesis that Integrated Project Delivery provides the normative model for value-creation in large, complex medical military construction projects.
10.2. Contribution to Knowledge

As a nation at war during very difficult economic times, it is critical to find ways to maximize the value of resources expended on provision of world-class healthcare for those who have already sacrificed so much. A goal of this research is to improve the way the U.S. Army approaches project execution for medical projects. This research demonstrates the importance of taking a comprehensive, macro-level viewpoint of the project delivery process, and validates a normative model for project delivery which has great potential for improving our world through enhanced value-creation. This research contributes to knowledge of project delivery in the following ways:

- Developing a means of evaluating the project delivery process based on a macro-viewpoint taking into consideration all phases of project delivery: Conceptual, Design, Construction, and Operational.
- Identifying and seeking to understand the differing perceptions of self-interest and project-interest each stakeholder group have is critical to developing the formal contractual basis for maximizing benefit for all stakeholders.
 Providing a basis for optimizing a collaborative project delivery process by accounting for differences in stakeholder perceptions.
- Providing a knowledge base surrounding the project delivery process and significant practical contributions to the military medical construction program. This research may lead to implementation of process improvements

241

to the Medical MILCON program to better provide for the best environment of care possible for our wounded warriors, and their families.

4. Providing basis of broader application of IPD in general, and IDBB in particular, to other types of large scale projects beyond MILCON.

10.3. Extensions and Future Research

The experience gained in this research may be utilized in the following areas:

 Further study of "Owner's" Stakeholder group for MILCON/public projects by breaking down the multiple parties within the MILCON "Owner's" stakeholder group by key sub-groups. Sub-groups include: Department of Defense, Department of Army, U.S. Army Corps of Engineers, U.S. Army Health Facility Planning Agency, Installation Departments of Public Works, TMA, etc. Identify which CSFs are most important, and any significant variances within the "Home team". Because of the multiple agencies and bureaucracies involved there may be significant variances which limit the optimization within the "Owner's" stakeholder group. Ideally (theoretically) all should be on the same page of music, but in practice each sub-group has different agendas and priorities similar to that between the primary stakeholder groups (Owner, Designer, Builder, & User).

242

- 2. Game Theory. Explore how an Integrated Project Delivery methodology may be advantageous to an owner for mitigating opportunistic bidding. The framework of a contract for multi-party Integrated Project Delivery in theory should virtually eliminate any beyond-contract reward based on design interpretation and/or errors if the builder is a responsible party to the design team.¹²⁰
- 3. Lifecycle Analysis of IPD or IDBB Projects. Due to the currently emergent and experimental nature of IPD there is not a large body of completed comparable projects to assess. At some point in the future, post-occupancy evaluations and lifecycle assessments may be used to compare long-term results of facilities delivered with IPD and traditional methods. The data collected for critical success factor relative importance during this research may be used in the future to create a model for measuring how successful each of the IDBB pilot-projects, or other future projects performed in value-creation. Other researchers have demonstrated models for converting the relative importance data into a performance index by which to evaluate project performance. The methodology used in this research may be utilized to establish a broader range of important critical success factors to evaluate different aspect of project success.

¹²⁰ S. Ping Ho and Liang Y. Liu. "Analytical Model for Analyzing Construction Claims and Opportunistic Bidding." *Journal of Construction Engineering and Management* 130, no. 1 (1 February 2004): 94-104.

4. Development of contractual optimization models based on CSF analysis of project stakeholder groups which best address CSF variance between project stakeholder groups. A more difficult consideration is how best to compensate and incentivize the short-term project stakeholders (Designer and Builder) in light of foregoing self-interest for the benefit of the overall project. The scope of this research project is focused on demonstrating the effectiveness of the Integrated Project Delivery process to successfully attain project goals, and create value. There is much additional research which may be conducted into how best to establish a contractual framework to optimize collaborative interaction of each stakeholder. Game Theory may be used to model how best to compensate stakeholders for their collaborative contributions.

APPENDIX A: INTEGRATED DESIGN-BID-BUILD PILOT-PROJECTS

• Fort Belvoir Community Hospital, Fort Belvoir, VA

~\$1.2 billion program, 1.3 million square feet, U.S. Army Corps of Engineers, IDBB fast-track, awarded at less than 15% design, Start 2007 - End 2011.

• San Antonio Military Medical Center - North, Fort Sam Houston, San Antonio, TX,

~\$703 million construction program, 760,000 square foot expansion of Brook Army Medical Center, and 288,000 square foot renovation of existing medical center, U.S. Army Corps of Engineers, IDBB fast-track, Start 2007 - End 2011.

• Battlefield Health and Trauma Center, Fort Sam Houston, San Antonio, TX,

~\$92 million program, U.S. Army Corps of Engineers, IDBB fast-track, Start 2007 - End 2011.

<u>National Geo-Spatial Intelligence Agency New Campus East, Fairfax County, VA</u>,
 2.4 million square feet, \$1.7 billion, U.S. Army Corps of Engineers, IDBB fast-track,
 awarded at 35% design, Start 2007 - End 2011.

APPENDIX B: INSTITUTIONAL REVIEW BOARD (IRB)

	UNIVERSITY OF ILLINOIS
	AT URBANA-CHAMPAIGN
Office Institu 528 Ea Suite 2 Cham	of the Vice Chancellor for Research ional Review Board it Green Street J3 aign, IL 61820
Nove	nber 2, 2010
Mich Arch 212 T MC-0	ael Kim tecture BH 21
RE:	Integrated Project Delivery: A Value-Creation Study of Integrated-Design-Bid-Build (IDBB) Military Construction (MILCON) pilot projects with emphasis on complext medical programs IRB Protocol Number: 11122
Dear	Michael:
Than Delin (MIL Instit resea Cate; perce const Cate;	(you for submitting the completed IRB application form for your project entited interval in your <i>ary: A Value-Creation Study of Integrated-Design-Bid-Build (IDBB) Military Construction CON) pilot projects with emphasis on complext medical programs.</i> Your project was assigned itional Review Board (IRB) Protocol Number 11122 and reviewed. It has been determined that the rch activities described in this application meet the criteria for exemption at 45CFR46.101(b). gories 2 and 4 apply. Category 2 applies because the study uses online survey methods to assess ptions of various experts regarding evaluation of value creation for design of health related military ruction projects. No information is collected on the surveys that can lead to personal identification. gory 4 applies because retrospective analysis of construction project documentation collected from through 9/9/2010 (last day prior to submission of application for exemption) will occur. This data triginally collected for the purposes of planning and implementing military construction projects r than research purposes.
2006 was o rathe	
2006 was or rathe This appr need mod Ame	determination of exemption only applies to the research study as submitted. Exempt protocols are oved for a maximum of three years. Please note that additional modifications to your project to be submitted to the IRB for review and exemption determination or approval before the fications are initiated. To submit modifications to your protocol, please complete the IRB Research idment Form (see <u>http://irb.illinois.edu/?q=forms-and-instructions/research-amendments.html</u>).
2006 was or rathe This appr need mod Ame We a any of or th	determination of exemption only applies to the research study as submitted. Exempt protocols are oved for a maximum of three years. Please note that additional modifications to your project to be submitted to the IRB for review and exemption determination or approval before the fications are initiated. To submit modifications to your protocol, please complete the IRB Research adment Form (see <u>http://irb.illinois.edu/?q=forms-and-instructions/research-amendments.html</u>). ppreciate your conscientious adherence to the requirements of human subject research. If you have uestions about the IRB process, or if you need assistance at any time, please feel free to contact me > IRB Office, or visit our website at <u>http://www.irb.illinois.edu</u> .
2006 was of rathee This appr need mod Ame We a any of or th Since	determination of exemption only applies to the research study as submitted. Exempt protocols are oved for a maximum of three years. Please note that additional modifications to your project to be submitted to the IRB for review and exemption determination or approval before the fications are initiated. To submit modifications to your protocol, please complete the IRB Research adment Form (see <u>http://irb.illinois.edu/?q=forms-and-instructions/research-amendments.html</u>). ppreciate your conscientious adherence to the requirements of human subject research. If you have uestions about the IRB process, or if you need assistance at any time, please feel free to contact me > IRB Office, or visit our website at <u>http://www.irb.illinois.edu</u> .

APPENDIX C: ROUND 1 SURVEY INSTRUMENT

	Exit Sur
Integrated Projec	t Delivery: Critical Success Factors for Value-Creation
Informed Consent A	greement
The purpose of this research value-creation. LTC Michael B this research as partial fulfill the University of Illinois Urba	study is to assess Integrated Project Delivery (IPD) impacts of Brennan, under the direction of Dr. Michael Kim, is conducting Iment of the requirements for a Ph.D. degree in Architecture a ana-Champaign.
You are invited to participate subject matter expert in plan one of the military construct	e in this research project because you have been identified as nning, design, construction, maintenance, and/or operations o ion Integrated-Design-Bid-Build pilot projects.
Your participation in this res you decide to participate in t decide not to participate in t will not be penalized.	earch study is voluntary. You may choose not to participate. If his research survey, you may withdraw at any time. If you his study or if you withdraw from participating at any time, yo
The procedure involves filling Your responses will be confic your name, email address or through an integrated projec	g an online survey that will take approximately 10-15 minutes dential and we do not collect identifying information such as IP address. The survey questions will be about creating value t delivery process.
We will do our best to keep y protected electronic format. information that will persona scholarly purposes only and representatives.	your information confidential. All data is stored in a password To help protect your confidentiality, the surveys will not conta ally identify you. The results of this study will be used for may be shared with University of Illinois Urbana-Champaign
If you have any questions ab PhD Candidate, University of michael.d.brennan@us.army, mkkim1@illinois.edu / (217) University of Illinois Urbana- subjects.	out the research study, please contact LTC Michael Brennan, Illinois Urbana-Champaign at mdbrenna@illinois.edu, .mil / (703)473-3587 or Dr. Michael Kim at .244-8012. This research has been reviewed according to -Champaign IRB procedures for research involving human
If you have any questions ab University of Illinois" Institu you identify yourself as a res	oout your rights as a participant in this study, please contact tl tional Review Board at 217-333-2670 (collect calls accepted it search participant) or via email at irb@illinois.edu.
Do you agree to participate i	n this survey?
DISAGREE: If you do not wis browser to exit the survey.	h to participate, please decline participation by closing your
AGREE: Check the box below agree to participate, and are	to indicate you have read the above information, voluntarily at least 18 years of age.
I Agree	
Sa	ve Page and Continue Later Continue

<u>Back</u>	Exit Sur
	Questions marked with a * are required
Inte	egrated Project Delivery: Critical Success Factors for Value-Creation
Profession	al Experience (1 of 4)
limited group o will be used to participants.	based on the Delphi Model and is dependent on the collective Wisdom of a of subject matter experts. The data collected from this section of the survey provide a basis to validate the collective expertise and experience of survey
Which of the fo	ollowing categories do you best identify with? (Select only one) *
 Design Team Eacility User 	n or Licer's Representative (i.e. Healthcare Professional)
Owner or Ov	vner's Representative
	Save Page and Continue Later Continue
	14% <u>Respondent Anonymity As</u>

Back	Exit Survey Questions marked with a * are required
	Integrated Project Delivery: Critical Success Factors for Value-Creation
Prof	essional Experience (2 of 4)
What	are your professional certifications? (Select all that apply) *
Ce	rtified Construction/Project Manager
E Lic	ensed Healthcare Provider
Pr	
	rified Design Professional
Ot	rer (If selected, description in box required)
	Save Page and Continue Later
	20% <u>Respondent Anonymity Assura</u>
	$D_{2} = 2 - 610$

Васк	Exi Questions marked with a * are required	t Surve
Integrated	I Project Delivery: Critical Success Factors for Value-Creation	
Professional Ex	sperience (3 of 4)	
What level of formal e	education (related to your area of expertise) have you complete	d?
 Associates Degree o 	or other training leading to professional certification	
Bachelors Degree		
🔘 Graduate / Professio	onal Degree	
Octorate Degree		
Not applicable		
	26% Respondent Anony	mity Ass

Integrated Project Delivery: Critical Success Factors for Value-Creation Professional Experience (4 of 4) More many years experience do you have in your area (and related areas) of expertise? (sees than 10 years) 10-15 years 20-25 years 20-25 years 30-35 years More than 35 years Save Page and Continue Later 22%		Questions marked with a * are required
Professional Experience (4 of 4) How many years experience do you have in your area (and related areas) of expertise? Less than 10 years 10-15 years 20-25 years 22-30 years 30-35 years More than 35 years More than 35 years 32*6 Respondent Anonymity Assume	Integrated	l Project Delivery: Critical Success Factors for Value-Creation
How many years experience do you have in your area (and related areas) of expertise? (Select only one) * Less than 10 years 10-15 years 20-25 years 22-30 years 25-30 years 30-35 years More than 35 years Save Page and Continue Later Continue 32% Respondent Anonymity Assura	Professional Ex	sperience (4 of 4)
Less than 10 years 10-15 years 15-20 years 20-25 years 25-30 years 30-35 years More than 35 years Save Page and Continue Later Continue 32% Respondent Anonymity Assura	How many years expe	erience do you have in your area (and related areas) of expertise?
 10-15 years 15-20 years 20-25 years 25-30 years 30-35 years More than 35 years Save Page and Continue Later Continue 32% Respondent Anonymity Assuration of the second secon	 Less than 10 years 	
 15-20 years 20-25 years 25-30 years 30-35 years More than 35 years Save Page and Continue Later Continue 32% Respondent Anonymity Assuration of the second secon	10-15 years	
 20-25 years 25-30 years 30-35 years More than 35 years Save Page and Continue Later Continue Save Page and Continue Later Continue 32% Respondent Anonymity Assuration of the second se	15-20 years	
 25-30 years 30-35 years More than 35 years Save Page and Continue Later Continue Save Page and Continue Later Save Page and Continue Later Respondent Anonymity Assuration	20-25 years	
 30-35 years More than 35 years Save Page and Continue Later Continue 32% 	25-30 years	
More than 35 years Save Page and Continue Later Continue 32% Respondent Anonymity Assura	30-35 years	
Save Page and Continue Later Continue C	More than 35 years	
32%		Save Page and Continue Later
32% Respondent Anonymity Assura		
		32% Respondent A ponymity A ssurg

Integrated Project Delivery: Critical Success Factors for Value-Creation Survey Overview & Instructions This next series of questions will ask you to select the project delivery factors which you believe are most important to the project team's ability to successfully create value. The listed factors are loosely broken down into four facets of project delivery: Design, Construction, Process, and Impact. In sequence you are asked to select the factors from two distinct perspectives. First from the "Stakeholder Perspective", then from the "Project Team Perspective". (1) STAKEHOLDER PERSPECTIVE: The question to ask yourself is "What factors are most important to overall success of the project and all project stakeholders": (2) PROJECT TEAM PERSPECTIVE: The question to ask yourself is "What factors are most important to overall success of the project and all project stakeholders": Save Page and Continue Later Continue 38%	Back	<u>Exit Surv</u>
Survey Overview & Instructions This next series of questions will ask you to select the project delivery factors which you believe are most important to the project team's ability to successfully create value. The listed factors are loosely broken down into four facets of project delivery: Design, Construction, Process, and Impact. In sequence you are asked to select the factors from two distinct perspectives. First from the "Stakeholder Perspective", then from the "Project Team Perspective". (1) STAKEHOLDER PERSPECTIVE: The question to ask yourself is "What factors are most important to only ' <u>MY</u> ' stakeholder group's success (Owner, Designer, Builder, or User)"? (2) PROJECT TEAM PERSPECTIVE: The question to ask yourself is "What factors are most important to overall success of the project and all project stakeholders"? (2) PROJECT TEAM PERSPECTIVE: The question to ask yourself is "What factors are most important to overall success of the project and all project stakeholders"? Save Page and Continue Later Continue 38%	Integra	ated Project Delivery: Critical Success Factors for Value-Creation
This next series of questions will ask you to select the project delivery factors which you believe are most important to the project team's ability to successfully create value. The listed factors are loosely broken down into four facets of project delivery: Design, Construciton, Process, and Impact. In sequence you are asked to select the factors from two distinct perspectives. First from the "Stakeholder Perspective", then from the "Project Team Perspective": (1) <u>STAKEHOLDER PERSPECTIVE</u> : The question to ask yourself is "What factors are most important to only ' <u>MY</u> ' stakeholder group's success (Owner, Designer, Builder, or User)"? (2) <u>PROJECT TEAM PERSPECTIVE</u> : The question to ask yourself is "What factors are most important to overall success of the project and all project stakeholders"? Save Page and Continue Later Respondent Anonymity Ass	Survey Over	view & Instructions
The listed factors are loosely broken down into four facets of project delivery: Design, Construciton, Process, and Impact. In sequence you are asked to select the factors from <u>two distinct perspectives</u> . First from the "Stakeholder Perspective", then from the "Project Team Perspective": (1) <u>STAKEHOLDER PERSPECTIVE</u> : The question to ask yourself is "What factors are most important to only ' <u>MY</u> ' stakeholder group's success (Owner, Designer, Builder, or User)"? (2) <u>PROJECT TEAM PERSPECTIVE</u> : The question to ask yourself is "What factors are most important to overall success of the project and all project stakeholders"? Save Page and Continue Later Continue <u>38%</u>	This next series of believe are most i	f questions will ask you to select the project delivery factors which you important to the project team's ability to successfully create value.
In sequence you are asked to select the factors from <u>two distinct perspectives</u> . First from the "Stakeholder Perspective"; (1) <u>STAKEHOLDER PERSPECTIVE</u> : The question to ask yourself is "What factors are most important to only ' <u>MY</u> ' stakeholder group's success (Owner, Designer, Builder, or User)"? (2) <u>PROJECT TEAM PERSPECTIVE</u> : The question to ask yourself is "What factors are most important to overall success of the project and all project stakeholders"? Save Page and Continue Later Continue Save Page and Continue Later Continue <u>Continue</u> <u>Save Page and Continue Later</u> Continue <u>Save Page and Continue Later</u> Continue	The listed factors	are loosely broken down into four facets of project delivery: Design,
the "Stakeholder Perspective", then from the "Project Team Perspective": (1) <u>STAKEHOLDER PERSPECTIVE</u> : The question to ask yourself is "What factors are most mportant to only ' <u>MY</u> ' stakeholder group's success (Owner, Designer, Builder, or User)"? (2) <u>PROJECT TEAM PERSPECTIVE</u> : The question to ask yourself is "What factors are most mportant to overall success of the project and all project stakeholders"? Save Page and Continue Later Continue Respondent Anonymity Ass	in sequence you a	rees, and impact. are asked to select the factors from <u>two distinct perspectives</u> . First from
(1) STAKEHOLDER PERSPECTIVE: The question to ask yourself is "What factors are most important to only ' <u>MY</u> ' stakeholder group's success (Owner, Designer, Builder, or User)"? (2) PROJECT TEAM PERSPECTIVE: The question to ask yourself is "What factors are most important to overall success of the project and all project stakeholders"? Save Page and Continue Later Continue 38%	the "Stakeholder	Perspective", then from the "Project Team Perspective":
(2) <u>PROJECT TEAM PERSPECTIVE</u> : The question to ask yourself is "What factors are most important to overall success of the project and all project stakeholders"? 	(1) <u>STAKEHOLDER</u> important to only	<u>{ PERSPECITVE</u> : The question to ask yourself is "What factors are most <u>'MY</u> ' stakeholder group's success (Owner, Designer, Builder, or User)"?
mportant to overall success of the project and all project stakeholders"? Save Page and Continue Later Continue Respondent Anonymity Ass	2) PROJECT TEAM	<u>M PERSPECTIVE</u> : The question to ask yourself is "What factors are most
Save Page and Continue Later Continue Continue Respondent Anonymity Ass	important to over	all success of the project and all project stakeholders"?
Save Page and Continue Later Continue C		
38% Respondent Anonymity Ass		Save Page and Continue Later Continue
38% Respondent Anonymity Ass		
		38% Respondent Anonymity Ass
		38% Respondent Anonymity Ass
		38% Respondent Anonymity Ass
		38% Respondent Anonymity Ass
		38% Respondent Anonymity Ass
		38% Respondent Anonymity Ass
		38% Respondent Anonymity Ass
		38% Respondent Anonymity Ass
		38% Respondent Anonymity Ass

Back	<u>Exit S</u>
Integ	grated Project Delivery: Critical Success Factors for Value-Creation
	SPECIAL INSTRUCTIONS:
perspect	Answer the next 4 questions from the tive of the "Stakeholder" group you selected earlier
Consid	der what is most important to the success of your stakeholder group. (Owner, Designer, Builder, or User)
	Save Page and Continue Later
	41% <u>Respondent Anonymit</u>

Bad	<u>k</u> <u>Exit Survey »</u> Questions marked with a * are required		
	Tetewated Designt Delivery, Oritigal Susana Factors for Value Creation		
	Integrated Project Delivery: Critical Success Factors for Value-Creation		
Sta	akeholder Perspective (1 of 4)		
Fa	ctors most important to only 'my' stakeholder group's success		
Whi	ch of the following "Design" factors do you believe are most important for project value- ation? (Select exactly 5) *		
	Design innovation and creativity		
	Achieve "World-Class"		
	Aesthetics		
1	Constructability		
	Community impact and acceptance		
	Clear and realistic objectives		
	Owner's vision		
	Utility and functionality		
	Sustainability		
	Evidence based design		
	Other: (Add a top-5 "Design" factor not listed above)		
	Save Page and Continue Later Continue		
	47%		
_	Respondent Anonymity Assurance		

Buok	Questions marked with a * are required
In	tegrated Project Delivery: Critical Success Factors for Value-Creation
Stakeholo Factors m	ler Perspective (2 of 4) lost important to only 'my' stakeholder group's success
Which of the	following "Construction" factors do you believe are most important for project
Change-ord	der management
Responsive	administration / decision support
Design acc	uracy
Cost Perfor	mance
Innovative	construction means and methods
Productivit	/
🔲 Time perfo	rmance
Safety	
Constructa	oility of Design
Quality	
Other: (Ad	d a top-5 "Construction" factor not listed above)
	Save Page and Continue Later Continue
	52%

Back Qu	estions marked with a * are required	<u>Exit Surv</u>
Integrated Project [Delivery: Critical Success Factors for Value-Creation	n
Stakeholder Perspecti Factors most importar	ve (3 of 4) it to only 'my' stakeholder group's s	uccess
Which of the following "Proces: value-creation? (Select exactly	s" factors do you believe are most important for pr 5) *	oject
Competency / Capability of pro	oject delivery team	
Collaboration of Integrated Pro	vject Team	
Trust and Respect		
Dispute avoidance and resoluti	on	
Effective communication		
Owner / User participation		
Project planning		
Alignment of project objectives	5	
Top management commitment	:	
Risk: identification and equitab	allocation	
Other: (Add a top-5 "Process"	factor not listed above)	
Save	Page and Continue Later Continue	
[58% Respondent A	nonymity As

<u>« Back</u>	Questions marked with a * are required	
Integra	ted Project Delivery: Critical Success Factors for Value-Creation	
Stakeholder Perspective (4 of 4) Factors most important to only 'my' stakeholder group's success		
Which of the follow creation? (Select e	ving "Impact" factors do you believe are most important for project value- xactly 5) *	
Profit and financi	jal objectives	
Avoid contractua		
Contract Incentiv	ves / Rewards	
Long-term buildi	ng success /Lifecycle Value of facility	
Long-term busin	ess relationships	
Professional repu	utation or image	
Profession/Indus	try Recognition or Awards	
Litigation avoida	nce	
📃 Owner/ user sati	sfaction	
Other: (Add a to	p-5 "Impact" factor not listed above)	
	Save Page and Continue Later	
	64%	

Integrated Project Delivery: Critical Success Factors for Value-Creation
SPECIAL INSTRCTIONS:
Answer the next 4 questions
from the
"Project Team" perspective.
ider what is in the best interest of the project, and the
collective project team as a whole.
section, keep in mind that the building is only a means, not the end. (Kim)
ation encompasses much more than the economic value of the "bricks and morta I value-creation is measured by the ability of the facility, as designed , construct
ated, to achieve its intended purpose and use.
) considering critical success factors (CSF) please keep in mind the big picture of how
project delivery process effects the long-term facility value across the spectrum of
iolders of both the project delivery team (designer, builder, owner, users) and other erm stakeholders (customers, community and environment).
Save Page and Continue Later
70%
70% Respondent Anonymity As
70% <u>Respondent Anonymity As</u>
70% Respondent Anonymity As
70% Respondent Anonymity As
70% Respondent Anonymity As
70% <u>Respondent Anonymity As</u>
70% <u>Respondent Anonymity As</u>
70% Respondent Anonymity As
70% Respondent Anonymity As Page 12 of 19

	Questions marked with a * are required
	Integrated Project Delivery: Critical Success Factors for Value-Creation
Project Factors project	Team Perspective (1 of 4) most important to overall success of the project for all stakeholders
Which of the creation? (re following "Design" factors do you believe are most important for project value- Select exactly 5) *
Evidence	a based design
	ity impact and acceptance
Constru	
Desian i	nnovation and creativity
Owner's	vision
Achieve	"World-Class"
Sustaina	ibility
Clear ar	d realistic objectives
🔲 Utility a	nd functionality
🔲 Other: (Add a top-5 "Design" factor not listed above)
	Save Page and Continue Later Continue
	76% Respondent Anonymity Assura

<u>к васк</u>	Questions marked with a * are required	t Survey
	Integrated Project Delivery: Critical Success Factors for Value-Creation	
Project Factors project	Team Perspective (2 of 4) most important to overall success of the project for a stakeholders	all
Which of th value-creat	e following "Construction" factors do you believe are most important for p ion? (Select exactly 5) *	project
Change-	order management	
Construc	ability of Design	
Cost Perf	ormance	
Time per	formance	
Design a	Jouracy	
	icy	
Safety		
Other: (A	dd a top-5 "Construction" factor not listed above)	
	Save Page and Continue Later Continue	
	82% Respondent Anony	mity Assurat

	Exit Survey »
	Questions marked with a * are required
	Integrated Project Delivery: Critical Success Factors for Value-Creation
Projec	t Team Perspective (3 of 4)
Factor	s most important to overall success of the project for all
projec	t stakeholders
Which of value-cre	the following "Process" factors do you believe are most important for project ation? (Select exactly 5) *
Effecti	ve communication
📃 Alignm	nent of project objectives
🔲 Collab	oration of Integrated Project Team
Disput	e avoidance and resolution
🔲 Risk: i	dentification and equitable allocation
Owner	/ User participation
Projec	t planning
🔲 Top m	anagement commitment
Compe	etency / Capability of project delivery team
Trust a	and Respect
Other:	: (Add a top-5 "Process" factor not listed above)
	Save Page and Continue Later Continue
	88%
	Respondent Anonymity Assurance

Integrated Project Delivery: Critical Success Factors for Value-Creation Project Team Perspective (4 of 4) Factors most important to overall success of the project for all project stakeholders Which of the following "Impact" factors do you believe are most important for project valu creation? (Select exactly 5) *
Project Team Perspective (4 of 4) Factors most important to overall success of the project for all project stakeholders Which of the following "Impact" factors do you believe are most important for project valu creation? (Select exactly 5) *
Factors most important to overall success of the project for all project stakeholders Which of the following "Impact" factors do you believe are most important for project valu creation? (Select exactly 5) *
Which of the following "Impact" factors do you believe are most important for project valu creation? (Select exactly 5) *
Ausid contractivel acceltion
Avoid contractual penalties
Long-term business relationships
Owner/ user satisfaction
Litigation avoidance
Long-term building success /Lifecycle Value of facility
Contract Incentives / Rewards
Job satisfaction
Profession/Industry Recognition or Awards
Professional reputation or image
Profit and financial objectives
Other: (Add a top-5 "Impact" factor not listed above)
Save Page and Continue Later Continue
94% Respondent Anonymity As

t Delivery: Critical Success Fac with the results of th published later in Janu Submit Survey" button your browser.	ctors for Value-Creation appreciated. is survey when f uary. n below to subm	the he nit your
with the results of th bublished later in Janu Submit Survey" button your browser.	appreciated. is survey when t uary. n below to subm	the it your
with the results of th published later in Janu Submit Survey" button your browser.	is survey when t uary. n below to subm	it your
Submit Survey" butto your browser.	n below to subm	it your
Submit Survey" butto your browser.	n below to subm	it your
your browser.		
Submit Survey		
100%	Respondent A	nonvmity Assurat
	100%	100% <u>Respondent A</u>

Thank you for your participation in this pilot survey! The Round #1 survey will be launched shortly after the closure of this preliminary pilot survey. <u>Thank You for completing this survey</u> <u>DILINE SURVEYS POWERED BY</u> <u>DUESTIONPro</u>		Round 1 Survey
Thank you for your participation in this pilot survey! The Round #1 survey will be launched shortly after the closure of this preliminary pilot survey. <u>Thank You for completing this survey</u> <u>DULINE SURVEYS POWERED BY</u> <u>QUESTIONPro</u>		
dosure of this preliminary pilot survey. <u>Thank You for completing this survey</u> <u>ONLINE SURVEYS POWERED BY</u> <u>QuestionPro</u>	Thank you for your particip.	ation in this pilot survey! The Round #1 survey will be launched shortly after th
ONLINE SURVEYS POWERED BY QuestionPro	closure of this preliminary p	pilot survey. <u>Thank You for completing this survey</u>
		OuestionPro
		Questioniro
Kesponderia Anonymity Associate		Respondent Anonymity Assura

Round 1 Survey Like 1936 likes. Sin QuestionPro 1-800-531-0228 · Live Help Free Acc FEATURES SOLUTIONS RESOURCES TAKE A TOUR CONTACT US PRICING LOGIN -Respondent Anonymity Assurance QuestionPro offers a unique quarantee to survey researchers to protect the privacy and confidentiality of the respondents. The Respondent Anonymity Assurance (RAA) is only applicable and valid for our "Corporate" License holders What is RAA? One of the challenges researchers face is the requirement for two directly conflicting issues:-1. The ability to track who has responded to the survey and who has not -- for sending out reminder emails, giving out prizes or compensation etc. AND 2. For human subjects protocols or other privacy reasons, ensuring that email identifications not be linked to the response data. From a technical standpoint, if we need to track who has taken the survey and who has not, the survey researcher also implicitly has the ability to track the response for each individual. To overcome this issue, the Respondent Anonymity Assurance has been introduced, QuestionPro asserts that once RAA is enabled on a survey, although computer generated identification numbers for individuals will be generated, the survey researcher will not have access to both the respondent's email address as well as the response data at the same time. What assertions does QuestionPro make for RAA-enabled surveys? 1. Only QuestionPro personnel can enable this guarantee for any survey Once the RAA is enabled for a survey, at the request of the survey researcher, it will remain perpetual and cannot be 2. rescinded -- by the researcher or anyone else 3. The QuestionPro Survey Software will never present a respondent's email address linked to the response data in any of the analysis tools, reports and data downloads 4. In the exceptionally unlikely event that a breach occurs for any reason, technical or otherwise, QuestionPro will immediately notify the respondent as well as the survey researcher about the issue. If the respondent requests his response data to be removed from the servers as a result of the breach, QuestionPro will comply with the request even if the survey researcher does not agree. Furthermore, QuestionPro will make all reasonable efforts to make sure that such a breach does not occur Are there any Exceptions? Under certain conditions, including but not limited to law enforcement requests and subpoenas, QuestionPro may have to divulge respondent data as well as email address if required to do so by law. QuestionPro is incorporated in the State of Washington, United States. QuestionPro also has offices in India. The legal jurisdiction for any disputes arising out of this or any other agreement shall be King County, WA Limits of Liability QuestionPro shall not be liable in any event for incidental or consequential damages in connection with, or arising out of, the furnishing, performance, or use of this documentation, or any software provided by QuestionPro. QuestionPro may be held liable only in the case of willful negligence on QuestionPro's part. Furthermore, the damages are limited to actual damages and not perceived, punitive, consequential or incidental damages. How do I know if a survey I am taking is protected by this guarantee? For surveys that have RAA enabled, a link to the details about the guarantee will be put on the "Bottom Right" corner of the survey This link cannot be removed or added by the survey researcher. It will be automatically added on surveys that have the anonymity guarantee enabled. Once this link is enabled it cannot be removed, changed or edited. The text for the link will be the following: QuestionPro: Respondent Anonymity Assurance I am taking a survey, and it is not protected by RAA! What do I do? Not all surveys need this level of protection. This only applies to surveys that have a computer generated respondent identification mechanism enabled. Many resenters chose here so no enable this and send out a simple link to the survey without the identification mechanism. In such cases there is no way of identifying who has taken the survey and who has not. Please consult with the owner of the survey to see if respondent identification is enabled or not. In most cases, researchers will disclose this information at the A questionpro.com sales agent is beginning of the survey. available to answer your questions Furthermore, QuestionPro may not be held liable for privacy claims made by survey researchered Page 19 of 19 Friday, 20 May, 2011 https://www.questionpro.com/security/raa.html

APPENDIX D: ROUND 2 SURVEY INSTRUMENT

En Informed Consent Agreement	<u>kit Survey⇒</u>
Informed Consent Agreement	
The purpose of this research study is to compare relative effectiveness of several pro	
of Dr. Michael Kim, is conducting this research as partial fulfillment of the requirement for a Ph.D. degree in Architecture at the University of Illinois Urbana-Champaign.	ject tion nts
Your participation in this research study is voluntary. You may choose not to particip If you decide to participate in this research survey, you may withdraw at any time. If decide not to participate in this study or if you withdraw from participating at any tim you will not be penalized.	ate. 'you ne,
The procedure involves filling an online survey that will take approximately 10 minut Your responses will be confidential and we do not collect identifying information such your name, email address or IP address. The survey questions will be about creating value through an integrated project delivery process.	es. 1 as
We will do our best to keep your information confidential. All data is stored in a passy protected electronic format. To help protect your confidentiality, the surveys will not contain information that will personally identify you. The results of this study will be used for scholarly purposes only and may be shared with University of Illinois Urbana Champaign representatives.	word a-
If you have any questions about the research study, please contact LTC Michael Bren PhD Candidate, University of Illinois Urbana-Champaign at mdbrenna@illinois.edu, michael.d.brennan@us.army.mil / (703)473-3587 or Dr. Michael Kim at mkkim1@illinois.edu / (217) 244-8012. This research has been reviewed according to University of Illinois Urbana-Champaign IRB procedures for research involving human subjects.	nan, o n
f you have any questions about your rights as a participant in this study, please cont he University of Illinois'' Institutional Review Board at 217-333-2670 (collect calls iccepted if you identify yourself as a research participant) or via email at rb@illinois.edu.	tact
Do you agree to participate in this survey?	
DISAGREE: If you do not wish to participate, please decline participation by closing y browser to exit the survey.	our
AGREE: Check the box below to indicate you have read the Informed Consent Agreem voluntarily agree to participate, and are at least 18 years of age.	ient,
I Agree	
Continue	
18% Respondent Anon	ymity Assuran

	Kouna 2 Survey	a an Williams
<u>Back</u>		<u>Exit Survey</u>
Round #1 Survey R	esults	
Thank you again for your pa	articipation in the Round #1 Survey!	
Survey Round #1 Results m	nay viewed at the following link:	
https://netfiles.uiuc.edu/m 20Results.pdf	ndbrenna/www/Survey Results/Round	<u>%20%231%20Survey%</u>
Based on frequency of selec Factors (CSF's) by nearly 50 one of the four stakeholder composite list of CSF's to be Notes: (1) The "Panel of Experts"	ction, the Panel of Experts has narrowed 0%. Only CSF's receiving a 60% or great groups (Owner, Designer, Builder, and U e assessed in this Round #2 Survey. has been strongly validated based on ed	the list of Critical Success ter selection rate by any Jser) is included in a new ucation, professional
certifications, and/or years	of experience. (See results for more inf	ormation)
(2) Cross-tabulation analys Selection" indicates statistic	is between "Stakeholder Groups" and "F ical significance.	requency of CSF
	Continue	
	Continue	
	Continue 31%	Respondent Anonymity Assur
	Continue	Respondent Anonymity Assure
	Continue	<u>Respondent Anonymity Assura</u>
	Continue31%	<u>Respondent Anonymity Assura</u>
	Continue	<u>Respondent Anonymity Assur</u>
	Continue	<u>Respondent Anonymity Assura</u>
	Continue	<u>Respondent Anonymity Assur</u>

<u>Back</u>	Questions marked with a * are required	Exit Survey
Stakeholder Group		
Which of the following cate	gories do you best identify with? (Select o	only one) *
Eacility User or User's Repr	recentative	
Construction Team	esentative	
 Owner or Owner's Represe 	ntative	
	Continue	
	43%	Respondent Anonymity Assura

D I-	Round 2 Survey	
<u>« Back</u>		Exit Survey
Round #2 Survey Ov	erview & Instructions	
You will be asked to assess t "Least Important" to "Most I	he relative importance of each factoring the second s	ctor on a 5-point scale from
As in Survey Round #1, in se <u>perspectives</u> . First from the " Perspective":	quence you are asked to assess t 'Stakeholder Perspective'', then f	he factors from <u>two distinct</u> rom the "Project Centered
(1) <u>STAKEHOLDER PERSPECT</u> to only ' <u>MY</u> ' stakeholder grou	<u>(IVE</u> : The question to ask yoursel ip's success (Owner, Designer, Bu	f is "How important is the factor iilder, or User)"?
(2) <u>PROJECT-CENTERED PERS</u> factor to the overall success o	<u>SPECTIVE</u> : The question to ask yo of the project and all project stak	ourself is "How important is the eholders"?
	Continue	
	56%	
		<u>Respondent Anonymity Assur</u>

	Round 2 Survey		
<u>« Back</u>		Exit Survey	
	SPECIAL INSTRUCTIONS	S:	
Assess t perspective o	the following critical success f of the " <u>STAKEHOLDER</u> " group	actors from the you selected earlier.	
Consider ho	Consider how important each factor is to the success of your stakeholder group. (Owner, Designer, Builder, or User)		
	Continue		
		2 C	
	62%	Respondent Anonymity Assurar	
	D 5 610		

« Back

Questions marked with a * are required

Exit Survey »

In light of the factors listed below, how would you rate the importance of each factor in general from your <u>STAKEHOLDER PERSPECTIVE</u>?

	Least Important		Important		Most Important
Aesthetics *	0	\bigcirc	0	\bigcirc	0
Alignment of Project Objectives *	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
Effective Communication *	\bigcirc	\bigcirc	\odot	\bigcirc	0
Profit and Financial Objectives *	0	\bigcirc	\odot	\bigcirc	0
Owner / User Participation *	\bigcirc	\bigcirc	\odot	\bigcirc	\bigcirc
Competency & Capability of Project Delivery Team *	\bigcirc	\bigcirc	\odot	\bigcirc	0
Trust & Respect *	\bigcirc	\bigcirc	\odot	\bigcirc	\odot
Change-Order Reduction *	\bigcirc	\bigcirc	\odot	\bigcirc	\bigcirc
Project Time Performance *	\bigcirc	\bigcirc	\bigcirc	\bigcirc	0
Dispute Avoidance & Resolution *	\odot	\bigcirc	\odot	\bigcirc	\odot
Utility & Functionality *	\bigcirc	\bigcirc	\bigcirc	\bigcirc	0
Owner's vision *	0	\bigcirc	\bigcirc	\bigcirc	0
Owner / User Satisfaction *	\bigcirc	\bigcirc	\odot	\bigcirc	\odot
Sustainability *	\bigcirc	\bigcirc	\bigcirc	\bigcirc	0
Long-term Building Success / Lifecycle Value *	\bigcirc	\bigcirc	\bigcirc	\bigcirc	0
Professional Reputation & Image *	\bigcirc	\bigcirc	\odot	\bigcirc	0
Clear & Realistic Objectives *	\bigcirc	\bigcirc	0	\bigcirc	\bigcirc
Production of Specified Quality *	\odot	\bigcirc	0	\bigcirc	\odot
Responsive Administration & Decision Support *	\bigcirc	\bigcirc	\bigcirc	\bigcirc	0
Collaboration of Project Delivery Team *	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
Design Innovation & Creativity *	\bigcirc	\bigcirc	\odot	\bigcirc	\odot
Constructability *	0	\bigcirc	\bigcirc	\bigcirc	\odot
Project Cost Performance *	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\odot

Continue

68%

.

Respondent Anonymity Assurance

Page 6 of 10

Round 2 Survey
<u>« Back</u> Exit Survey >
SPECTAL INSTRUCTIONS.
SPECIAL INSTRCTIONS:
Answer the following question from the
"PROJECT-CENTERED" perspective.
Consider how important each factor is to the overall success of the project, and the collective success of project team as a whole.
For this section, keep in mind that the building is only a means, not the end. (Kim)
Value-creation encompasses much more than the economic value of the "bricks and mortar". Successful value-creation is measured by the ability of the facility, as designed , constructed, and operated, to achieve its intended purpose and use.
When considering critical success factors (CSF) please keep in mind the big picture of how the project delivery process effects the long-term facility value across the spectrum of stakeholders of both the project delivery team (designer, builder, owner, users) and other long-term stakeholders (customers, community and environment).
Continue
81% Respondent Anonymity Assurance
81% <u>Respondent Anonymity Assurant</u>
81% <u>Respondent Anonymity Assuran</u>
81% Respondent Anonymity Assurant
81% Respondent Anonymity Assurand
B1%
81% Respondent Anonymity Assurant

« Back

Questions marked with a * are required

Exit Survey »

In light of the factors listed below, how would you rate the importance of each factor in general from a <u>PROJECT-CENTERED PERSPECTIVE</u>?

	Least Important		Important		Most Important
Owner's vision *	0	\bigcirc	0	\bigcirc	0
Alignment of Project Objectives *	\bigcirc	\bigcirc	0	0	\bigcirc
Trust & Respect *	0	\bigcirc	0	\bigcirc	0
Clear & Realistic Objectives *	0	\bigcirc	\odot	\bigcirc	0
Effective Communication *	\odot	0	0	\bigcirc	0
Owner / User Satisfaction *	0	\bigcirc	0	\bigcirc	0
Responsive Administration & Decision Support *	0	\bigcirc	\odot	\bigcirc	0
Dispute Avoidance & Resolution *	0	0	\odot	0	\odot
Design Innovation & Creativity *	0	\bigcirc	0	\bigcirc	\odot
Aesthetics *	0	\bigcirc	0	\bigcirc	0
Collaboration of Project Delivery Team *	0	\bigcirc	0	\bigcirc	0
Constructability *	0	\bigcirc	\odot	\bigcirc	\bigcirc
Competency & Capability of Project Delivery Team *	0	\bigcirc	0	\bigcirc	0
Profit and Financial Objectives *	0	\bigcirc	0	\bigcirc	0
Utility & Functionality *	0	\bigcirc	\bigcirc	\bigcirc	\odot
Professional Reputation & Image *	\bigcirc	0	\odot	\bigcirc	\odot
Owner / User Participation *	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc
Project Cost Performance *	0	\bigcirc	0	\bigcirc	0
Long-term Building Success / Lifecycle Value *	0	\bigcirc	0	0	\odot
Sustainability *	0	\bigcirc	\odot	\bigcirc	\odot
Project Time Performance *	0	\bigcirc	\odot	0	0
Change-Order Reduction *	0	0	\odot	0	\odot
Production of Specified Quality *	\odot	\bigcirc	\bigcirc	\bigcirc	\odot

Continue

87%

Respondent Anonymity Assurance

Page 8 of 10

THANK YOU! Your par	rticipation is greatly ap	<u>Exit Survey »</u> opreciated.
You will be provided Round #3 (final) surv Round #3 will ask the effectiveness of differ most important critic	with the results of this vey is published. e Panel of Experts to w rent project delivery m al success factors.	s survey when the reigh in on the nethods to acheive the
Please click on the "S responses and close y	ubmit Survey" button your browser.	below to submit your
	Submit Survey	



APPENDIX E: ROUND 3 SURVEY INSTRUMENT

	2	
	<u>Exit Surv</u>	
Inform	ed Consent Agreement	
The purpo delivery n of Dr. Micl for a Ph.D	ese of this research study is to compare relative effectiveness of several project nethods for maximum value-creation. LTC Michael Brennan, under the direction hael Kim, is conducting this research as partial fulfillment of the requirements . degree in Architecture at the University of Illinois Urbana-Champaign.	
Your parti If you dec decide not you will no	icipation in this research study is voluntary. You may choose not to participate. ide to participate in this research survey, you may withdraw at any time. If you t to participate in this study or if you withdraw from participating at any time, ot be penalized.	
The proce Your resp your name value thro	dure involves filling an online survey that will take approximately 10 minutes. onses will be confidential and we do not collect identifying information such as e, email address or IP address. The survey questions will be about creating rugh an integrated project delivery process.	
We will do protected contain in used for s Champaig	o our best to keep your information confidential. All data is stored in a password electronic format. To help protect your confidentiality, the surveys will not formation that will personally identify you. The results of this study will be cholarly purposes only and may be shared with University of Illinois Urbana- n representatives.	
If you hav PhD Cand michael.d mkkim1@ University subjects.	re any questions about the research study, please contact LTC Michael Brennan, idate, University of Illinois Urbana-Champaign at mdbrenna@illinois.edu, .brennan@us.army.mil / (703)473-3587 or Dr. Michael Kim at illinois.edu / (217) 244-8012. This research has been reviewed according to r of Illinois Urbana-Champaign IRB procedures for research involving human	
If you hav the Univer accepted i irb@illinoi	re any questions about your rights as a participant in this study, please contact rsity of Illinois" Institutional Review Board at 217-333-2670 (collect calls if you identify yourself as a research participant) or via email at s.edu.	
Do you ag	ree to participate in this survey?	
DISAGREE browser to	: If you do not wish to participate, please decline participation by closing your o exit the survey.	
AGREE: Cl voluntaril	neck the box below to indicate you have read the Informed Consent Agreement, y agree to participate, and are at least 18 years of age.	
🔲 I Agree	e	
	Continue	
	396	
	Respondent Anonymity Ass	
	D 1 000	
	Page 1 of 22	
	Round 3 Survey	
---	---	-----------------------------------
<u>« Back</u>		Exit Survey »
Round #2 Survey	/ Results	
Thank you again for you	r participation on the "Panel of Expe	erts" for this series of surveys.
Survey Round #2 Result	ts may viewed at the following link:	,
* * * Survey Round # 2	<u>Results</u> * * *	
Based on the results of s further narrowed down	Survey Round #2, the list of Critical to those selected as most important	Success Factors (CSF's) has been
The final survey in this s project delivery method	series (Round #3) will ask you to rat is to best achieve the selected CSF's.	te the effectiveness of different
	Castinus	
	Continue	
	6%	
		Respondent Anonymity Assurance
	Page 2 of 22	

		Round 3 Survey	
<u>«</u>	Back	Questions marked with a * are required	<u>Exit Survey</u> »
	Stakeholder Group		
	Once again, please select th Construction Team	e category you best identify with. (Selec	t only one) *
	 Design Team 		
	Facility User or User's Repr	esentative	
	 Owner or Owner's Represer 	ntative	
		Continue	
		9%	Respondent Anonymity Assurance
		Page 3 of 22	
		1 450 5 01 22	

< Back	Exit Se	urvey ×
Questions marked with a	* are required	
** AMENDMENT TO SURVEY ROUND	#2 **	
Due to an oversight, one factor which met qualifying accidentally omitted from Survey Round #2:	g criteria in Survey Round #1, but was	5
EVIDENCE BASED DESIGN (EBD)		
Please assess the relative importance of <u>"Evidence</u> responses will be amended to Survey Round #2 fine	<u>Based Design"</u> below. (Collected Il results)	
In light of the Round #2 factors listed at the botton	of the page:	
Rate the importance of <u>Evidence Based Design</u> from <u>PERSPECTIVE</u> .	your <u>STAKEHOLDER GROUP'S</u>	
Evidence Based Design *	Least Important Mo Important Impor O O O O O	st rtant
Rate the importance of <u>Evidence Based Design</u> from	a <u>PROJECT-CENTERED PERSPECTIVE</u> . Least Important Mo	st
Evidence Based Design *	Important Impor	rtant
Aesthetics Alignment of Project Objectives Change-Order Reduction Clear & Realistic Objectives Collaboration of Project Delivery Team Competency & Capability of Project Delivery Team Constructability Project Cost Performance Design Innovation & Creativity Dispute Avoidance & Resolution Effective Communication Long-term Building Success / Lifecycle Value Owner / User Satisfaction Owner / User Participation Owner / User Participation Owner / User Participation Professional Reputation & Image Profit and Financial Objectives Production of Specified Quality Responsive Administration & Decision Support Sustainability Project Time Performance Trust & Respect Utility & Functionality		
Continue 15%		Assurance

Lack Questions marked with a * are required Project Delivery Methods Design-Bid-Build (DBB), also known as "Traditional Method: Linear process, where the owner first enters into a contract with a designer to produce a follow design, the 100% design documents are then used by the owner to competitively bid and award a separate construction contract. Simultaneous design and construction is not possible of the owner to competitively bid possible of the owner to competitively bid possible. Design-Build (DB) Maile contractor with in-house design, or a pre-established design partner. Note: Mit construction contracts with a design er or contract with a dost (RFP) development. Typically the completion of a 35% concept design is equired to competitively bid the DB contract. Integrated Project Delivery (IPD) / Integrated Design-Bidle (IDBE): The soluborative multi-party contractual relationship where the designer and builder followent with the owner/jusers throughout all phases of design and construction. DBB is blace for the following project delivery (IPD). DibB is blace for the following project deliver with a dost of implement Integrated Project Delivery (IPD). DibB is blace for the owner dust and construction. DBB is blace for the following regioner delivery (IPD). Method mailer are you with the following project delivery methods: Implement Methods: Design-Bid-Build (DB): Implement Method is a more inclusion in the project delivery (IPD) is in the owner (ECI). Implement Methods: Design-Bid-Build (DB): Implement Methods: Implement Methods: Implement						
<section-header></section-header>	<u>Back</u> Que	estions marked wit	h a * are re	equired	ļ	Exit Survey >
<section-header> Design-Bid-Build (DBB), also known as "Traditional Method: Bide consers, where the owner first enters into a contract with a designer to produce as a barber owner to competitively bid observes are then used by the owner to competitively bid observes. Design-Build (DB) Bide contract for design and construction services. Typically the owner contracts with a figure as separate design partner. Note: full observes owner to competitively bid the DB contract. Design-Build (DB) Bide contract for design and construction services. Typically the completion of a 35% concept design services contract for competitively bid the DB contract. Design-Build projects typically first require a separate design services contract for competitively bid the DB contract. Design-Build project stypically first requires a separate design services. Design-Build project stypically first requires as parate design and construction. DBB second as with the somer/users throughout all phases of design and construction. DBB second set with the somer/users throughout all phases of design and construction. Design-Build (DBB) * Nat At Milling Signtly Somewhar Moderately Very Familiar Fa</section-header>	Project Delivery Metho	ds				
	Design-Bid-Build (DBB) , also kn	own as "Traditio	nal Metho	d:		
Design-Build (DB) More familiar are you with the following project delivery methods Not At All Slightly Somewhat Moderately Very Pamiliar Familiar Familiar Familiar Familiar Familiar Design-Build (DB)* Ontinue Ontinue Ontinue Ontinue	Linear process, where the owner 100% design. The 100% design and award a separate constructi possible using the DBB method.	r first enters into documents are t on contract. Sim	a contrac hen used l ultaneous	t with a desi by the owner design and o	gner to prod to competit construction	uce a ively bid is not
A single contract for design and construction services. Typically the owner contracts with a construction contractor with in-house design, or a pre-established design partner. Note: MILCON Design-Build projects typically first require a separate design services contract for required to competitively bid the DB contract. Integrated Project Delivery (IPD) / Integrated Design-Bid-Build (IDBB): TPD is a collaborative multi-party contractual relationship where the designer and builder construction. IDBB is the MILCON piolet method to implement Integrated Project Delivery (IPD). IDBB is also referred to as Early Contractor Involvement (ECI). Mot At All Slightly Somewhat Moderately Very Familiar Fami	Design-Build (DB):					
Integrated Project Delivery (IPD) / Integrated Design-Bid-Build (IDBB): IPD is a collaborative multi-party contractual relationship where the designer and builder collaborate with the owner/users throughout all phases of design and construction. IDBB is the MILCON pilot-project method to implement Integrated Project Delivery (IPD). IDBB is also referred to as Early Contractor Involvement (ECI). How familiar are you with the following project delivery methods? Not At All Slightly Somewhat Moderately Very Familiar Familiar Familiar Familiar Familiar Familiar Familiar Integrated Project Delivery (IPD) * Design-Bid-Build (DBB) * Design-Build (DB) * Integrated Project Delivery (IPD) * Continue 22%	A single contract of weshin and construction contractor with in-I MILCON Design-Build projects ty request-for-proposal (RFP) deve required to competitively bid the	house design, or pically first reque lopment. Typical DB contract.	a pre-esta ire a sepa ly the com	blished design rate design s pletion of a	gn partner. N services cont 35% concep	Note: Tract for t design is
Image: Design-Bid-Build (DBB) * Image: Design-Build (DB) *	Integrated Project Delivery (IPD) / Integrated D	esign-Bid-	Build (IDBB):	
Not At All Slightly Somewhat Moderately Very Familiar Familiar Familiar Familiar Familiar Design-Bid-Build (DBB) * O O O O O Design-Build (DB) * O O O O O Integrated Project Delivery (IPD) * O O O O O O Continue 22%	the MILCON pilot-project method also referred to as Early Contract	d to implement I tor Involvement	elivery m	Project Deliv	/ery (IPD). I	DBB is
Design-Bid-Build (DBB) * Design-Build (DB) * Desi	now fammar are you with the fo	Not At All Familiar	Slightly Familiar	Somewhat Familiar	Moderately Familiar	Very Familiar
Continue 22% Respondent Anonymity Assuran	Design-Bid-Build (DBB) *			0	0	0
Continue 22% Respondent Anonymity Assuran	Integrated Project Delivery (IPD) *	0	0	0	0	0
22% Respondent Anonymity Assuran	Integrated Project Derivery (IPD)					
22% Respondent Anonymity Assuran		Contin	ue			
Respondent Anonymity Assuran		Contin	ue			
		Contin 22%	ue			
		Contin 22%	ue o		<u>Respondent An</u>	onymity Assuran
		Contin 22%	ue D		<u>Respondent An</u>	onymity Assuran
		Contin 22%	ue D		<u>Respondent An</u>	onymity Assuran
		Contin 22%	ue o		<u>Respondent An</u>	onymity Assuranc
		Contin 22%	ue)		<u>Respondent An</u>	onymity Assuran
		Contin 22%	ue j		<u>Respondent An</u>	onymity Assuranc
		Contin 22%	ue j		<u>Respondent An</u>	onymity Assuranc

Back		<u>Exit Survey</u>
Round #3 Survey Ove	erview & Instructions	
NOTE: <u>The conceptual framew</u> <u>COMPLEX PROJECTS ONLY.</u> (so Belvoir, VA and Fort Sam Hous	vork for the following questions is lim cale and scope similar to the MILCON ston/Joint Base San Antonio, TX)	<u>nited to VERY LARGE &</u> I IDBB pilot-projects at Fort
You will be asked to rate the e successfully achieve the most experts in survey rounds 1 & 2	effectiveness of different project deli important critical success factors as 2.	very methods to selected by the panel of
As in the previous surveys, yo methods from <u>two distinct per</u> the "Project Centered Perspec	u are asked to assess the effectivene <u>rspectives</u> . First from the "Stakehold ctive":	ess of the project delivery er Perspective", then from
(1) <u>STAKEHOLDER PERSPECT</u> delivery method for <u>MY stake</u> l	<u>[VE</u> : The question to ask yourself is " <u>holder group's interests</u> ?"	How effective is the project
(2) <u>PROJECT-CENTERED PERS</u> project delivery method for th	<u>PECTIVE</u> : The question to ask yourse le <u>success of the project</u> ?	If is "How effective is the
	Continue	
	Continue	
	Continue	Respondent Anonymity Assura
	Continue 24%	<u>Respondent Anonymity Assuran</u>
	Continue	<u>Respondent Anonymity Assuran</u>
	Continue	<u>Respondent Anonymity Assura</u>
	Continue	<u>Respondent Anonymity Assura</u>
	Continue	<u>Respondent Anonymity Assuran</u>
	Continue	<u>Respondent Anonymity Assurat</u>
	Continue	<u>Respondent Anonymity Assurar</u>
	Continue	<u>Respondent Anonymity Assura</u>

* Each Questions marked with a * are required Owner & User Satisfaction Saccessfully achieve OWNER & USER SATISFACTION Cate the effectiveneess of each project delivery method from your SIAKHOLDER Design-Bid-Build (DBB) * Image: Design-Bid-Build (DBB) * Design-Build (BD) * Image: Design-Bid-Build (DBB) * Design-Bid-Build (DBB) * Image: Design-Bid-Build (DBB) * Design-Bid-Build (DBB) * Image: Design-Bid-Build (DBB) * Design-Bid-Build (DBB) * Image: Design-Bid-Build (DBB) * Design-Bid-Build (DBB) * Image: Design-Bid-Build (DBB) *						
Survey & User Satisfaction Subsective & Sective & Contract Contr	<u>3ack</u> Que	estions marked with a st a	are required	I	Ē	xit Surve
Successfully achieve OWNER & USER SATISFACTION Are the effectiveness of each project delivery method from your STAKEHOLDER Design-Bid-Build (DBB)* Design-Bid (BD)* Design-Bid-Build (DBB)* Design-Bid-Build (DBB)* Design-Bid-Build (DBB)* Design-Bid-Build (DBB)* Design-Build (BD)* Design-Bid-Build (DBB)* Design-Build (BD)* Design-Build	Owner & User Satisfact	tion				
Atte the effectiveness of each project delivery method from your STAKEHOLDER Design-Bid-Build (DBB) * Design-Bid (BD) * Integrated Project Delivery (IPD) * Carter the effectiveness of each project delivery method from a PROJECT-CENTERED Design-Bid-Build (DBB) * Design-Build (BD) * Integrated Project Delivery (IPD) * Continue 29%	Successfully achieve O	WNER & USER S	ATISFA	<u>CTION</u>		
Least Less Effective More Mo Effective Effective ate the effectiveness of each pr <u>PERSPECTIVE</u>	roject delivery method	from your	STAKEHO	LDER		
Design-Build (DBB) * Design-Build (BD) * Integrated Project Delivery (IPD) * Rate the effectiveness of each project delivery method from a PROJECT-CENTERED PERSPECTIVE Least Less Effective More More Effective Effective Effective Effective Effect Design-Build (DBB) * Design-Build (BD) * Integrated Project Delivery (IPD) * Continue 29% Respondent Anonymity 2		Least Effective	Less Effective	Effective	More Effective	Most Effective
Design-Build (BD) * Integrated Project Delivery (IPD) * Rate the effectiveness of each project delivery method from a PROJECT-CENTERED PERSPECTIVE I Least Less Effective Effective Effective Effective Effective Effective Effective Integrated Project Delivery (IPD) * Continue 29%	Design-Bid-Build (DBB) *	0	0	\bigcirc	0	0
Integrated Project Delivery (IPD) *	esign-Build (BD) *	\odot	\odot	\bigcirc	\bigcirc	0
Ate the effectiveness of each project delivery method from a PROJECT-CENTERED PERSPECTIVE Design-Bid-Build (DBB) * Design-Build (BD) * Integrated Project Delivery (IPD) * Continue 29%	ntegrated Project Delivery (IPD) st	0		\bigcirc	\odot	Ô
PERSPECTIVE IN Continue	late the effectiveness of each pr	roject delivery method	from a <u>PR</u>	OJECT-CE	NTERED	
Effective Effect	<u>'ERSPECTIVE</u>	Least	Less	Effective	More	Most
Design-Build (DBB) * O O O O O O O O O O O O O O O O O O		Effective	Effective		Effective	Effective
Design-Build (BD) * Integrated Project Delivery (IPD) * Continue 29% Respondent Anonymity /	esign-Bid-Build (DBB) *	0	0	0	0	0
Integrated Project Delivery (IPD) * Continue 29% Respondent Anonymity /	Design-Build (BD) *	0	0	0	0	0
Continue 29%	ntegrated Project Delivery (IPD) *	0	0	0	0	0
29% Respondent Anonymity /		Continue				
29% Respondent Anonymity /						
		29%		Re	spondent Ano	nymity Assu

	Round 3 Surv	ey			
<u>« Back</u>	Questions marked with a	* are require	d	Ē	xit Survey »
Clear & Realistic Proj	ect Objectives				
Maintain <u>CLEAR & RE</u> the entire project del	ALISTIC PROJEC	T OBJEC	TIVES t	hrough	out
Rate the effectiveness of each PERSPECTIVE	n project delivery meth	od from you	r <u>STAKEHO</u>	<u>ILDER</u>	
	Leas	Less	Effective	More	Most
Decign Rid Ruild (DRR) *	Effecti	ve Effective		Effective	Effective
Design-Build (RD) *	0			0	
Integrated Project Delivery (IPD)	*				
PERSPECTIVE	Leas	: Less	Effective	More	Most
	Effecti	ve Effective		Effective	Effective
Design-Bid-Build (DBB) *		0	0	0	0
Design-Build (BD) *	*	0	0	0	
	Continue	J 			
	35%]	Re	spondent Ano	nymity Assurance

<u> Back</u>	Questions marked	with a * a	re required		E	<u>xit Surve</u>
Effective Communic	ation					
Promote <u>EFFECTIVE</u> stakeholders throug	<u>COMMUNIC</u>	ATION ire pro	betwee ject del	n all pr livery p	oject rocess	
Rate the effectiveness of ea	ich project deliver	y method	from your	STAKEHO	DLDER	
		Least	Less	Effective	More	Most
Docian-Bid-Build (DBB) *		Effective	Effective		Effective	Effective
Design-Build (BD) *			0		0	0
Integrated Project Delivery (IP	D) *	0	0	0	0	0
Rate the effectiveness of ea	ich project deliver	y method	from a <u>PR</u>	OJECT-CE	NTERED	
		Least Effective	Less Effective	Effective	More Effective	Most Effective
Design-Bid-Build (DBB) *		0	0	\bigcirc	0	0
Design-Build (BD) *		\odot	0	\bigcirc	0	\odot
Integrated Project Delivery (IP	D) *	\odot	\bigcirc	\bigcirc	\bigcirc	\bigcirc
	Co	ntinue				
		0%				
				Re	spondent Ano	nymity Assur

Collaboration of Pro	Questions marked	with a * a	re required			
Collaboration of Pro						
	ject Delivery	Team				
Maximize <u>COLLABO</u>	RATION OF P	ROJEC	DELI	ERY TE	AM	
Rate the effectiveness of ea	ch project deliver	y method	from your	STAKEHO	LDER	
		Least	Less	Effective	More	Most
Desian-Bid-Build (DBB) *			O	0	O	O
Design-Build (BD) *		õ	õ	õ	õ	õ
Integrated Project Delivery (IP	D) *	0	0	0	0	0
Rate the effectiveness of ea	ch project deliver	y method	from a <u>PR</u>	OJECT-CE	NTERED	
		Least	Less	Effective	More	Most
B		Effective	Effective		Effective	Effective
Design-Bid-Build (DBB) *		0	0	0	0	0
Integrated Project Delivery (ID)	* (C	0	0	0	0	0
	Co	ontinue				
		45%				
				Res	pondent Ano	nymity Assura

« Back	Questions man	ked with a * a	re required		E	xit Survey
Utility & Functionali	ty					
Maximize the <u>Utility</u>	& Functio	<u>nality</u> val	ue of tl	ne facili	ity prod	luced
Rate the effectiveness of each perspective and perspective and perspective per	ch project deli	very method	from your	STAKEHO	LDER	
		Least	Less	Effective	More	Most
Design-Bid-Build (DBB) *			O	0	Enective	O
Design-Build (BD) *		Õ	0	Õ	Õ	Õ
Integrated Project Delivery (IPE) *	O	0	\bigcirc	\bigcirc	O
Rate the effectiveness of eac	h project deli	very method	from a <u>PR</u>	OJECT-CE	NTERED	
PERSPECTIVE		Least	Less	Effective	More	Most
		Effective	Effective		Effective	Effective
Design-Bid-Build (DBB) *		0	0	0	0	0
Design-Build (BD) *	*	0	0	0	0	0
	(Continue				
		50%		Res	spondent Ano	nymity Assura

Back	Questions marked w	ith a * a	re required		E	<u>xit Surve</u>
Trust & Respect						
Maximize <u>TRUST & R</u>	ESPECT betw	een a	II proje	ct stake	eholder	s
Rate the effectiveness of each <u>PERSPECTIVE</u>	ı project delivery ı	nethod	from your	STAKEHO	LDER	
	F	Least	Less	Effective	More	Most
Design-Bid-Build (DBB) *	E		Cirective	0		Cirective
Design-Build (BD) *		õ	0	0	0	0
Integrated Project Delivery (IPD)	*	\odot	0	0	0	Ô
Rate the effectiveness of each	ı project delivery ı	nethod	from a <u>PR</u>	OJECT-CE	NTERED	
PERSPECTIVE		Least	Less	Effective	More	Most
	E	ffective	Effective		Effective	Effective
Design-Bid-Build (DBB) *		0	\odot	\odot	0	O
Design-Build (BD) *			0	0	0	0
	Cont	inue				
	55	%		_		
				Res	pondent Ano	nymity Assura

<u>« Васк</u>	Questions marke	ed with a * a	re required		Ē	xit Survey
Alignment of Projec	t Objectives					
Maximize the <u>ALIGN</u> project stakeholder	IMENT OF P s across the	ROJECT entire p	OBJEC project	<u>TIVES</u> b deliver	etweer y proce	n all ss
Rate the effectiveness of ea	ch project delive	ry method	from your	STAKEHO	<u>IDER</u>	
		Least Effective	Less Effective	Effective	More Effective	Most Effective
Design-Bid-Build (DBB) *		0	0	\bigcirc	0	0
Design-Build (BD) *		\bigcirc	0	\bigcirc	0	0
Integrated Project Delivery (IPI	D) *	\odot	\odot	\odot	\odot	\odot
		Continue				
		61%				
		0170		Re	spondent Ano	nymity Assura

Auser & User Participation Anable OWNER & USER PARTICIPATION throughout the entite Continue Integrated Project Delivery (IPD) *	Most fective
Enable OWNER & USER PARTICIPATION throughout the entire project delivery process East the effectiveness of each project delivery method from your STAKEHOLDER PERSPECTIVE I Design-Bid-Build (DBB)* Image: Design-Bid-Build (DBB)* Image: Design-Bid-Build (DBB)* Image: Design-Bid-Build (DBB)* Image: Design-Bid-Build (DBB)* Rate the effectiveness of each project delivery method from a PROJECT-CENTERED PERSPECTIVE I Least Less Effective Effective Iffective Effective Effective Effective Effective Effective Effective Effective Effective Iffective Iffective Effective	Most fective
Bate the effective less of each project delivery method from your STAKEHOLDER Design-Bid-Build (DBB) * Design-Build (BD) * Integrated Project Delivery (IPD) * Bases the effective effective Design-Bid-Build (DBB) * Design-Build (BD) * Design-Build (BD) * Design-Build (BD) * Design-Build (BD) * Design-Build (BD) * Design-Build (BD) * Design-Build (BD) * Design-Build (BD) * Design-Build (BD) * Design-Build (BD) * Design-Build (BD) * Design-Build (BD) * Design-Build (BD) * Design-Build (BD) * Design-Build (BD) * Design-Build (BD) *<	Most fective
Least Less Effective Effective More More More Design-Build (DBD) * Integrated Project Delivery (IPD) * Integrated Project Delivery (IPD) * Integrated Project Delivery (IPD) * Rate the effectiveness of each project delivery method from a PROJECT-CENTERED PERSPECTIVE Effective Integrated Project Delivery (IPD) * Design-Build (DBB) * Integrated Project Delivery (IPD) * Design-Build (DBD) * Integrated Project Delivery (IPD) * Integrated Project Delivery (IPD) * Integrated Project Delivery (IPD) *	Most fective
Design-Build (BD) * Integrated Project Delivery (IPD) * Rate the effectiveness of each project delivery method from a PROJECT-CENTERED PERSPECTIVE Least Less Effective Effective Design-Bid-Build (DBB) * Image: Continue Design-Build (BD) * Image: Continue Integrated Project Delivery (IPD) * 66%	0
Rate the effectiveness of each project delivery method from a PROJECT-CENTERED Design-Bid-Build (DBB) * Design-Build (BD) * Integrated Project Delivery (IPD) * 66%	
Least Less Effective More M Effective Effective	
Design-Bid (DBB) * Design-Build (BD) * Integrated Project Delivery (IPD) * Continue 66% Respondent Anonymit	Most
Design-Build (BD) *	O
Integrated Project Delivery (IPD) *	\bigcirc
Continue 66% Respondent Anonymit	0
66% Respondent Anonymit	
	ity Assura

<u> Back</u>				E	xit Survey
Questions ma	arked with a * a	re required			
Production of Specified Qualit	ty				
Successfully achieve the PRO	DUCTION	OF SPE	CIFIED	QUALI	TY
Rate the effectiveness of each project del <u>PERSPECTIVE</u>	ivery method	from your	STAKEHO	LDER	
	Least	Less Effective	Effective	More	Most
Design-Bid-Build (DBB) *	O	O	0	O	©
Design-Build (BD) *	0	0	0	Õ	Õ
Integrated Project Delivery (IPD) $*$	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
Rate the effectiveness of each project del	ivery method	from a <u>PR</u>	OJECT-CE	NTERED	
PERSPECTIVE	Least	Less	Effective	More	Most
	Effective	Effective		Effective	Effective
Design-Bid-Build (DBB) *	0	\odot	\bigcirc	\odot	\bigcirc
Design-Build (BD) *		\odot	\bigcirc	\bigcirc	\bigcirc
Integrated Project Delivery (IPD) *		\odot	\odot	\odot	0
	Continue				
	71%		Res	pondent Ano	numity A ssura
			Ites	spondent Ano	nymny Assura

the facility produced	t ch project delive	ry method	from your	STAKEHO	LDER	<u> </u>
<u>. INGI LUTITI</u> II		Least	Less	Effective	More	Most
Design-Bid-Build (DBB) *				0	enective	
Design-Build (BD) *		0	0	0	0	0
Integrated Project Delivery (IPE) *	\odot	0	0	\odot	\odot
		onunde				
		76%		Re	pondent Ano	nymity Assur:

Project Time Performance Additional and an analysis of the project delivery method from your STACEHOLDER Effective of the form of the	Project Time Performance Additional and an analysis of each project delivery method from your STAKEHOLDER Enspective Bill Design-Build (DBB) * Design-Build (BD) * Design-Build (DBB) * Design-Build (DBB) * Design-Build (DBB) * Design-Build (BD) * Integrated Project Delivery (IPD) * Design-Build (BDB) * Design-Build (BDB) * Design-Build (BDB) * Design-Build (BDB) * Design-Build (BDB) * Design-Build (BDB) * Design-Build (BDB) * Design-Build (BDB) * Design-Build (BDB) * Design-Build (BDB) * Design-Build (BDB) * Design-Build (BDB) * Design-Build (BDB) * Design-Build (BDB) * Design-Build (BD) * Design-Build (BD) * Design-Build (BD) * Design-Build (BD) * Design-Build (BD) * Design-Build (BD) * Design-Build (BD) * Design-Build (BD) * Design-Build (BD) * Design-Build (BD) * Design-Build (BD) * Design-Build (BD) *	RMANC y method Effective o y method Effective o ntinue 81%	E to acl	STAKEHO Effective	he shor	test Most Effective
Maximize PROJECT TIME PERFORMANCE to achieve the shortest possible schedule duration Design-Bid-Build (DBB)* Least Less Effective Effect	Maximize PROJECT TIME PERFORMANCE to achieve the shortest cosible schedule duration Eterefective effective NC y method Least Effective ontinue B1%	to acl	STAKEHO Effective	NTERED	Most Effective	
Acta that effective effec	Act the effectiveness of each project delivery method from your STAKEHOLDER Design-Bid-Build (DBB)* Design-Build (BD)* Integrated Project Delivery (IPD)* Design-Bid-Build (DBB)* Design-Bid-Build (BD)* Integrated Project Delivery (IPD)* Continue	y method Least Effective ontinue B1%	from your Less Effective © from a PR Less Effective © © ©	Effective	More Effective	Most Effective Most Effective
Least Less Effective Effective More Most Design-Bid-Build (DBB) * Integrated Project Delivery (IPD) * Image: Continue Image: Continue Image: Continue Rate the effectiveneess of each project delivery method from a PROJECT-CENTERED PERSPECTIVE Design-Bid-Build (DBB) * Image: Continue Image: Continue Image: Continue Design-Bid-Build (DBB) * Image: Continue Image: Continue Image: Continue Image: Continue Design-Build (BD) * Image: Continue Image: Continue Image: Continue Image: Continue Respondent Anonymity Assume	Least Less Effective More Most Design-Bid-Build (DBB) * Integrated Project Delivery (IPD) * Image: State of the effective rest of the e	Least Effective o y method Least Effective o ontinue 81%	Less Effective © from a PR Less Effective © ©	Effective	More Effective	Most Effective © © Most Effective © ©
Design-Bid-Build (DBB) * Design-Build (BD) * Integrated Project Delivery (IPD) * Rate the effectiveness of each project delivery method from a PROJECT-CENTERED PERSPECTIVE Least Less Effective Effective Effective Effective Effective Effective Effective of the other of the o	Design-Bid-Build (DBB) * Integrated Project Delivery (IPD) * Integrated Project Delivery (IPD) * Rate the effectiveness of each project delivery method from a PROJECT-CENTERED PERSPECTIVE More Most Effective Effective Effective Effective Effective Effective Effective Effective Effective Integrated Project Delivery (IPD) * Design-Bid-Build (DBB) * Integrated Project Delivery (IPD) * Integrated Project Delivery (IPD) * Integrated Project Delivery (IPD) * Integrated Project Delivery (IPD) * Integrated Project Delivery (IPD) * Respondent Anonymity Assuration 81%	y method Least Effective	from a PR Less Effective	COJECT-CE Effective	More Effective	Most Effective
Design-Build (BD) * Integrated Project Delivery (IPD) * Image: Constraint of the second	Design-Build (BD) * Integrated Project Delivery (IPD) * Rate the effectiveness of each project delivery method from a PROJECT-CENTERED PERSPECTIVE Design-Bid-Build (DBB) * Design-Build (BD) * Design-Build (BD) * Integrated Project Delivery (IPD) * Integrated Project Delivery (IPD) * Respondent Anonymity Assura	y method Least Effective	from a PR Less Effective	© COJECT-CE Effective	More Effective	Most Effective
Integrated Project Delivery (IPD) *	Integrated Project Delivery (IPD) *	y method Least Effective	from a <u>PR</u> Less Effective © ©	© Effective © ©	More Effective	Most Effective
Rate the effectiveness of each project delivery method from a PROJECT-CENTERED PERSPECTIVE Design-Bid-Build (DBB) * Design-Build (BD) * Integrated Project Delivery (IPD) * Continue 81%	Rate the effectiveness of each project delivery method from a PROJECT-CENTERED PERSPECTIVE Least Less Effective Effective Design-Bid-Build (DBB) * Image: Continue Design-Build (BD) * Image: Continue Integrated Project Delivery (IPD) * Image: Continue 81% Respondent Anonymity Assura	y method Least Effective Ontinue	from a PR Less Effective © ©	Effective	More Effective	Most Effective © ©
81% Respondent Anonymity Assura	81% Respondent Anonymity Assura	81%				
Kespondent Anonymusy Assur:	Kespondent Anonymity Assura			D	1	
				<u></u>	spondent Ano	<u>nyiinty Assura</u>

« Back	Questions marked	d with a * a	re required		E	<u>xit Survey</u>
Owner's Vision						
Ensure the <u>OWNER'</u> delivery process	<u>S VISION</u> is	fully rea	alized b	y the p	roject	
Rate the effectiveness of ea	ch project delive	ry method	from your	STAKEHO	LDER	
		Least	Less	Effective	More	Most
Desian-Bid-Build (DBB) *		Enective	enective	\bigcirc	enective ()	Enective
Design-Build (BD) *		0	0	0	Õ	0
Integrated Project Delivery (IPI	D) *	0	0	0	0	0
	Ca	ontinue				
		87%		Res	spondent Ano	nymity Assura

Constructability Action action of the construct of the construction of the constr	hrough y method Least Effective © y method Effective © ontinue	from your	entire STAKEHO Effective	PLDER More Effective	Most Effective © © Most Effective © ©
Maximize CONSTRUCTABILITY throughout the entire project clivery process Design-Bid-Build (DBB)* Design-Bid (BD)* Integrated Project Delivery (IPD)* Effective Effecti	hrough y method Least Effective O y method Least Effective O O ontinue	from your Less Effective () () () () () () () () () ()	entire STAKEHO Effective OJECT-CEI Effective O	DIDER More Effective	Most Effective © © Most Effective © ©
State the effective eff	y method Effective	from your	STAKEHO Effective	NTERED More Effective	Most Effective © © Most Effective © ©
Least Effective Less Effective Effective Effective More Effective Most Effective Design-Build (BD) * Integrated Project Delivery (IPD) * Integrated from a PROJECT-CENTERED PERSPECTIVE Image: Continue Rate the effectiveneess of each project delivery method from a PROJECT-CENTERED PERSPECTIVE More Image: Continue Mo	Least Effective	Less Effective	Effective	More Effective	Most Effective
Design-Bid-Build (DBB) * Integrated Project Delivery (IPD) * Integrated Project Delivery (IPD) * Rate the effectiveness of each project delivery method from a PROJECT-CENTERED PERSPECTIVE More Most Effective Effective Effective Effective Effective Effective Effective Effective Effective Effective Effective Effective Integrated Project Delivery (IPD) * Design-Build (BB) * Integrated Project Delivery (IPD) * Integrated Project Delivery (IPD) * Continue 92%	y method Least Effective	from a PR Less Effective	OJECT-CE Effective	More Effective	Most Effective
Design-Build (BD) * Integrated Project Delivery (IPD) * Image: Constraint of the second	y method Least Effective	from a <u>PR</u> Less Effective	© COJECT-CE Effective	More Effective	Most Effective
Integrated Project Delivery (IPD) *	y method Least Effective	from a <u>PR</u> Less Effective © ©	© Effective © ©	More Effective	Most Effective © ©
Rate the effectiveness of each project delivery method from a PROJECT-CENTERED PERSPECTIVE Design-Bid-Build (DBB) * Design-Build (BD) * Integrated Project Delivery (IPD) * Continue 92%	y method Effective	from a <u>PR</u> Less Effective	Effective	More Effective	Most Effective © ©
92% Respondent Anonymity Assura					
92% Respondent Anonymity Assura					
	92%		Res	spondent Ano	nymity Assura
		9 of 22	9 of 22	9 of 22	9 of 22

Round 3 Survey « Back Exit Survey » Questions marked with a * are required **EVIDENCE BASED DESIGN** Maximize the implementation of EVIDENCE BASED DESIGN principles throughout the project delivery process Rate the effectiveness of each project delivery method from your <u>STAKEHOLDER</u> <u>PERSPECTIVE</u> Least Less Effective More Most Effective Effective Effective Effective Design-Bid-Build (DBB) * 0 0 0 0 0 Design-Build (BD) * \bigcirc \bigcirc \bigcirc \bigcirc 0 Integrated Project Delivery (IPD) * \bigcirc 0 0 0 0 Rate the effectiveness of each project delivery method from a **PROJECT-CENTERED** PERSPECTIVE Least Effective More Most Less Effective Effective Effective Effective Design-Bid-Build (DBB) * 0 0 0 \bigcirc 0 Design-Build (BD) * \bigcirc \bigcirc \bigcirc \bigcirc 0 Integrated Project Delivery (IPD) * 0 0 \bigcirc \bigcirc 0 Continue 97% Respondent Anonymity Assurance Page 20 of 22

Round	3 Survey
<u>« Back</u>	Exit Survey »
Once againTHANK YOU for pa Delphi survey! Your assistance You will be provided with the f surveys within the next 60 day	articipating in all 3 rounds of the is greatly appreciated! inal results and analysis of all three ys.
Please click on the "Submit Su responses and close your brow	rvey" button below to submit your rser.
Sub	bmit Survey
	100% Respondent Anonymity Assurance
Page	21 of 22

Round 3 Survey QuestionPro 1-800-531-0228 · Live Help Like 1936 likes. Free Ace FEATURES SOLUTIONS RESOURCES TAKE A TOUR PRICING CONTACT US LOGIN -Respondent Anonymity Assurance QuestionPro offers a unique guarantee to survey researchers to protect the privacy and confidentiality of the respondents. The Respondent Anonymity Assurance (RAA) is only applicable and valid for our "Corporate" License holders. What is RAA? One of the challenges researchers face is the requirement for two directly conflicting issues:-1. The ability to track who has responded to the survey and who has not -- for sending out reminder emails, giving out prizes or compensation etc AND 2 For human subjects protocols or other privacy reasons, ensuring that email identifications not be linked to the response data. From a technical standpoint, if we need to track who has taken the survey and who has not, the survey researcher also implicitly has the ability to track the response for each individual. To overcome this issue, the Respondent Anonymity Assurance has been introduced. QuestionPro asserts that once RAA is enabled on a survey, although computer generated identification numbers for individuals will be generated, the survey researcher will not have access to both the respondent's email address as well as the response data at the same time. What assertions does QuestionPro make for RAA-enabled surveys? 1. Only QuestionPro personnel can enable this guarantee for any survey 2. Once the RAA is enabled for a survey, at the request of the survey researcher, it will remain perpetual and cannot be rescinded -by the researcher or anyone else 3. The QuestionPro Survey Software will never present a respondent's email address linked to the response data in any of the analysis tools, reports and data downloads 4. In the exceptionally unlikely event that a breach occurs for any reason, technical or otherwise, QuestionPro will immediately notify The respondent as well as the survey researcher about the issue. If the respondent requests his response data to be removed from the servers as a result of the breach, QuestionPro will comply with the request even if the survey researcher does not agree. Furthermore, QuestionPro will make all reasonable efforts to make sure that such a breach does not occur again Are there any Exceptions? Under certain conditions, including but not limited to law enforcement requests and subpoenas, QuestionPro may have to divulge respondent data as well as email address if required to do so by law. QuestionPro is incorporated in the State of Washington, United States. QuestionPro also has offices in India The legal jurisdiction for any disputes arising out of this or any other agreement shall be King County, WA Limits of Liability QuestionPro shall not be liable in any event for incidental or consequential damages in connection with, or arising out of, the furnishing, performance, or use of this documentation, or any software provided by QuestionPro QuestionPro may be held liable only in the case of willful negligence on QuestionPro's part. Furthermore, the damages are limited to actual damages and not perceived, punitive, consequential or incidental damages How do I know if a survey I am taking is protected by this guarantee? For surveys that have RAA enabled, a link to the details about the guarantee will be put on the "Bottom Flight" corner of the survey. This link cannot be removed or added by the survey researcher. It will be automatically added on surveys that have the anonymity guarantee enabled. Once this link is enabled it cannot be removed, changed or edited. The text for the link will be the following-QuestionPro: Respondent Anonymity Assurance Page 22 of 22 Friday, 20 May, 2011 https://www.questionpro.com/security/raa.html

APPENDIX F: ROUND 3 WEIGHTED SUM CALCULATIONS

				Desi	gn-Bid	l-Builo	l (DBE	B) Se	lf-Inte	rest					
		Owner/User Satisfaction	Clear & Realistic Objectives	Effective Communication	Collaboration of Project Team	Utility & Functionality	Trust & Respect	Alignment of Objectives	Owner/User Participation	Production of Specified Quality	Long-term Lifecycle Value	Time Performance	Owner's Vision	Constructability	Weighted Sum Aggregated Relatie Effectiveness
Owner01	Raw Weight	5 0.082	4	5 0.082	3	4	2	4	4	4	3	2	4	2	
Owner02	Weighted Score Raw	0.411	0.334	0.411	0.246	0.306	0.139	0.312	0.295	0.300	0.238	0.136	0.323	0.139	3.589
0 11 10 10 2	Weight	0.082	0.084	0.082	0.082	0.076	0.069	0.078	0.074	0.075	0.079	0.068	0.081	0.069	2 9 4 9
Owner03	Raw	5	4	3	3	3	3	3	3	3	3	2	4	3	2.040
	Weighted Score	0.082	0.084	0.082	0.082	0.076	0.069	0.078	0.074	0.075	0.079	0.068	0.081	0.069	3.261
Owner04	Raw Weight	3	5	3	2	2	3	5	4	4	3	2	4	3	
005	Weighted Score	0.002	0.418	0.002	0.164	0.153	0.208	0.390	0.295	0.300	0.238	0.136	0.323	0.208	3.326
Owner05	Weight	0.082	0.084	0.082	0.082	0.076	0.069	0.078	0.074	0.075	0.079	0.068	0.081	0.069	
Owner06	Weighted Score Raw	0.411	0.418	0.411	0.411	0.382	0.347	0.390	0.368	0.375	0.397	0.068	0.404	0.347	4.728
	Weight Weighted Score	0.082	0.084	0.082	0.082	0.076	0.069	0.078	0.074	0.075	0.079	0.068	0.081	0.069	3 503
Owner07	Raw	4	4	4	4	4	4	5	5	5	5	2	5	3	
-	Weighted Score	0.082	0.084	0.082	0.082	0.076	0.069	0.078	0.0/4	0.075	0.079	0.068	0.081	0.069	4.181
Owner08	Raw Weight	5	5	4	3	4	4	4	4	4	4	4	5 0.081	3	
Outmar 00	Weighted Score	0.411	0.418	0.329	0.246	0.306	0.278	0.312	0.295	0.300	0.317	0.272	0.404	0.208	4.095
Owner09	Weight	0.082	0.084	0.082	0.082	0.076	0.069	4 0.078	0.074	0.075	4 0.079	0.068	0.081	0.069	
Owner10	Weighted Score Raw	0.329	0.167	0.246	0.329	0.306	0.208	0.312	0.295	0.300	0.317	0.136	0.323	0.208	3.476
	Weight	0.082	0.084	0.082	0.082	0.076	0.069	0.078	0.074	0.075	0.079	0.068	0.081	0.069	1 550
Owner11	Raw	3	5	5	3	5	2	3	5	5	5	2	5	3	1.550
	Weight Weighted Score	0.082	0.084	0.082	0.082	0.076	0.069	0.078	0.074	0.075	0.079	0.068	0.081	0.069	3.965
Owner12	Raw	3	5	5	3	3	3	4	4	5	4	3	4	3	
	Weighted Score	0.082	0.418	0.082	0.082	0.070	0.009	0.078	0.074	0.075	0.073	0.008	0.081	0.208	3.793
Owner13	Raw Weight	0.082	0.084	0.082	0.082	0.076	0.069	0.078	5 0.074	0.075	0.079	0.068	5 0.081	0.069	
Owner14	Weighted Score	0.246	0.251	0.411	0.246	0.229	0.069	0.234	0.368	0.225	0.238	0.068	0.404	0.069	3.059
	Weight	0.082	0.084	0.082	0.082	0.076	0.069	0.078	0.074	0.075	0.079	0.068	0.081	0.069	2.0.(0
Designer01	Weighted Score Raw	0.411	0.418	0.246	0.246	0.382	0.208	0.234	0.295	0.375	0.397	0.136	0.404	0.208	3.960
	Weight Weighted Score	0.086	0.083	0.075	0.081	0.079	0.077	0.079	0.069	0.075	0.075	0.069	0.079	0.077	3.650
Designer02	Raw	5	5	3	5	5	4	4	4	5	5	3	5	3	
	Weighted Score	0.086	0.083	0.075	0.081	0.079	0.077	0.079	0.069	0.075	0.075	0.069	0.079	0.077	4.336
Designer03	Raw Weight	3	5	4	4	4	4	3	4	5	4	2	5	3	
Dagigmar04	Weighted Score	0.259	0.413	0.299	0.322	0.314	0.306	0.236	0.275	0.373	0.299	0.138	0.393	0.230	3.857
Designer04	Weight	0.086	0.083	0.075	4 0.081	4 0.079	0.077	4 0.079	0.069	0.075	4 0.075	0.069	4 0.079	0.077	
Designer05	Weighted Score Raw	0.432	0.413	0.373	0.322	0.314	0.306	0.314	0.344	0.373	0.299	0.069	0.314	0.230	4.104
	Weight	0.086	0.083	0.075	0.081	0.079	0.077	0.079	0.069	0.075	0.075	0.069	0.079	0.077	4 033
Designer06	Raw	4	4	3	1	4	3	3	4	5	5	5	5	4	4.033
	Weight Weighted Score	0.086	0.083	0.075	0.081	0.079	0.077	0.079	0.069	0.075	0.075	0.069	0.079	0.077	3.825
Designer07	Raw	4	5	4	2	3	3	4	4	3	3	3	4	3	
	Weighted Score	0.080	0.083	0.075	0.081	0.079	0.077	0.079	0.069	0.075	0.073	0.069	0.079	0.077	3.472
Designer08	Raw Weight	<u>3</u> 0.086	<u>3</u> 0.083	<u>4</u> 0.075	2 0.081	4 0.079	2 0.077	2 0.079	<u>3</u> 0.069	4 0.075	4 0.075	2 0.069	4 0.079	<u>3</u> 0.077	
Designer00	Weighted Score	0.259	0.248	0.299	0.161	0.314	0.153	0.157	0.206	0.299	0.299	0.138	0.314	0.230	3.077
Designer09	Weight	0.086	0.083	0.075	0.081	0.079	0.077	0.079	0.069	0.075	0.075	0.069	0.079	0.077	2 (25
	weighted Score	0.259	0.330	1 0.149	0.161	0.236	0.153	0.157	0.206	0.224	0.224	0.138	0.236	0.153	2.627

				Desig	gn-Bid	-Build	I (DBE	3) Se	lf-Inte	rest					
		Owner/User Satisfaction	Clear & Realistic Objectives	Effective Communication	Collaboration of Project Team	Utility & Functionality	Trust & Respect	Alignment of Objectives	Owner/User Participation	Production of Specified Quality	Long-term Lifecycle Value	Time Performance	Owner's Vision	Constructability	Weighted Sum Aggregated Relatie Effectiveness
Builder01	Raw	4	4	4	3	4	3	4	4	3	4	3	4	3	
	Weight	0.089	0.076	0.082	0.06/	0.076	0.082	0.078	0.084	0.076	0.069	0.080	0.072	0.069	2 6 2 6
Builder02	Raw	3	3	3	0.201	3	3	3	3	0.229	3	3	0.290	0.200	5.020
Dunder02	Weight	0.089	0.076	0.082	0.067	0.076	0.082	0.078	0.084	0.076	0.069	0.080	0.072	0.069	
	Weighted Score	0.268	0.229	0.245	0.201	0.229	0.245	0.234	0.251	0.229	0.206	0.240	0.217	0.206	3.000
Builder03	Raw	1	1	3	2	3	1	1	1	3	2	1	1	1	
	Weight	0.089	0.076	0.082	0.067	0.076	0.082	0.078	0.084	0.076	0.069	0.080	0.072	0.069	
	Weighted Score	0.089	0.076	0.245	0.134	0.229	0.082	0.078	0.084	0.229	0.138	0.080	0.072	0.069	1.604
Builder04	Raw	5	3	3	3	1	3	1	2	2	2	1	2	1	
	Weight	0.089	0.076	0.082	0.067	0.076	0.082	0.078	0.084	0.076	0.069	0.080	0.072	0.069	
D 11 05	Weighted Score	0.446	0.229	0.245	0.201	0.076	0.245	0.078	0.167	0.152	0.138	0.080	0.145	0.069	2.271
Builder05	Raw	3	3	3	3	3	2	2	3	3	3	3	3	2	
	Weight	0.089	0.076	0.082	0.06/	0.076	0.082	0.0/8	0.084	0.076	0.069	0.080	0.0/2	0.069	2 771
Duildar06	Weighted Score	0.268	0.229	0.245	0.201	0.229	0.164	0.156	0.251	0.229	0.206	0.240	0.217	0.138	2.//1
Bullder00	Naw Weight	2	4	$\frac{2}{0.082}$	0.067	0.076	0.082	4	2	0.076	0.060	0.080	0.072	0.060	
	Weighted Score	0.039	0.305	0.062	0.007	0.229	0.062	0.312	0.064	0.305	0.005	0.060	0.072	0.009	2.818
Builder07	Raw	3	2	3	3	3	2	2	3	4	3	2	4	2	2.010
	Weight	0.089	0.076	0.082	0.067	0.076	0.082	0.078	0.084	0.076	0.069	0.080	0.072	0.069	
	Weighted Score	0.268	0.152	0.245	0.201	0.229	0.164	0.156	0.251	0.305	0.206	0.160	0.290	0.138	2.764
Builder08	Raw	3	3	3	3	3	3	3	3	3	3	3	3	3	
	Weight	0.089	0.076	0.082	0.067	0.076	0.082	0.078	0.084	0.076	0.069	0.080	0.072	0.069	
	Weighted Score	0.268	0.229	0.245	0.201	0.229	0.245	0.234	0.251	0.229	0.206	0.240	0.217	0.206	3.000
Builder09	Raw	4	5	3	2	3	2	3	3	4	4	2	3	2	
	Weight	0.089	0.076	0.082	0.067	0.076	0.082	0.078	0.084	0.076	0.069	0.080	0.072	0.069	2 0 0 0
Llaam01	Weighted Score	0.357	0.381	0.245	0.134	0.229	0.164	0.234	0.251	0.305	0.275	0.160	0.217	0.138	3.089
User01	Woight	3	0.070	0.091	0.001	0.072	0.077	0.092	0.074	0.077	2	2	0.074	3	
	Weighted Score	0.087	0.079	0.081	0.091	0.072	0.077	0.085	0.074	0.077	0.075	0.132	0.074	0.004	2 858
User02	Raw	5	5	3	3	3	3	3	3	5	3	2	4	4	2.050
030102	Weight	0.087	0.079	0.081	0.091	0.072	0.077	0.083	0.074	0.077	0.075	0.066	0.074	0.064	
	Weighted Score	0.434	0.396	0.243	0.272	0.215	0.232	0.249	0.221	0.387	0.226	0.132	0.294	0.257	3.558
User03	Raw	3	1	1	1	3	1	1	2	3	3	1	3	1	
	Weight	0.087	0.079	0.081	0.091	0.072	0.077	0.083	0.074	0.077	0.075	0.066	0.074	0.064	
	Weighted Score	0.260	0.079	0.081	0.091	0.215	0.077	0.083	0.147	0.232	0.226	0.066	0.221	0.064	1.843
User04	Raw	2	2	1	2	2	2	2	1	4	1	1	1	5	
	Weight	0.087	0.079	0.081	0.091	0.072	0.077	0.083	0.074	0.077	0.075	0.066	0.074	0.064	1.077
LL	Weighted Score	0.1/4	0.158	0.081	0.181	0.143	0.155	0.166	0.074	0.309	0.075	0.066	0.074	0.321	1.977
User05	Kaw Waiaht	0.097	2	2	2	3	3	2	2	0 077	3	2	3	3	
	Weighted Score	0.087	0.079	0.081	0.091	0.072	0.077	0.085	0.074 0.147	0.077	0.075	0.132	0.074	0.004	2 526
User06	Raw	3	2	2	3	4	3	3	3	3	3	1	3	3	2.520
000100	Weight	0.087	0.079	0.081	0.091	0.072	0.077	0.083	0.074	0.077	0.075	0.066	0.074	0.064	
	Weighted Score	0.260	0.158	0.162	0.272	0.287	0.232	0.249	0.221	0.232	0.226	0.066	0.221	0.192	2.779
User07	Raw	3	4	3	4	4	4	4	3	4	3	4	3	4	
	Weight	0.087	0.079	0.081	0.091	0.072	0.077	0.083	0.074	0.077	0.075	0.066	0.074	0.064	
	Weighted Score	0.260	0.317	0.243	0.362	0.287	0.309	0.332	0.221	0.309	0.226	0.264	0.221	0.257	3.609
User08	Raw	1	2	1	1	2	2	2	2	2	2	2	2	2	
	Weight	0.087	0.079	0.081	0.091	0.072	0.077	0.083	0.074	0.077	0.075	0.066	0.074	0.064	
	Weighted Score	0.087	0 158	0.081	0.091	0 143	0.155	0.166	0 147	0 155	0.151	0.132	0 147	0 128	1 742

]	Design	-Bid-l	Build ((DBB)	Pro	ject-In	terest					
		Owner/User Satisfaction	Clear & Realistic Objectives	Effective Communication	Collaboration of Project Team	Utility & Functionality	Trust & Respect	Alignment of Objectives	Owner/User Participation	Production of Specified Quality	Long-term Lifecycle Value	Time Performance	Owner's Vision	Constructability	Weighted Sum Aggregated Relatie Effectiveness
Owner01	Raw	5	4	5	3	4	2	4	4	4	3	2	4	2	
	Weighted Score	0.078	0.084	0.081	0.078	0.081	0.078	0.074	0.079	0.079	0.076	0.071	0.065	0.076	3.554
Owner02	Raw	3	3	2	3	3	3	3	3	3	3	3	3	2	
	Weight	0.078	0.084	0.081	0.078	0.081	0.078	0.074	0.079	0.079	0.076	0.071	0.065	0.076	2 9 4 2
Owner03	Raw	4	4	3	3	3	3	3	3	3	3	2	3	3	2.043
	Weight	0.078	0.084	0.081	0.078	0.081	0.078	0.074	0.079	0.079	0.076	0.071	0.065	0.076	
Owner04	Weighted Score	0.312	0.335	0.242	0.234	0.242	0.234	0.221	0.238	0.238	0.229	0.141	0.195	0.229	3.091
Owner04	Weight	0.078	0.084	0.081	0.078	0.081	0.078	0.074	0.079	0.079	0.076	0.071	0.065	0.076	
	Weighted Score	0.312	0.418	0.242	0.156	0.162	0.234	0.368	0.317	0.317	0.229	0.141	0.260	0.229	3.387
Owner05	Raw	5	5	5	5	5	5	5	5	5	5	1	5	5	
	Weighted Score	0.390	0.418	0.404	0.390	0.404	0.390	0.368	0.397	0.397	0.382	0.071	0.325	0.382	4.717
Owner06	Raw	5	5	4	3	3	3	3	3	4	3	3	4	2	
	Weighted Score	0.078	0.084	0.081	0.078	0.081	0.078	0.074	0.079	0.079	0.076	0.071	0.065	0.076	3 472
Owner07	Raw	5	5	4	3	4	3	4	4	5	4	2	5	3	0.11/2
	Weight	0.078	0.084	0.081	0.078	0.081	0.078	0.074	0.079	0.079	0.076	0.071	0.065	0.076	2.022
Owner08	Weighted Score Raw	0.390	0.418	0.323	0.234	0.323	0.234	0.294	0.317	0.397	0.306	<u>0.141</u> 4	0.325	0.229	3.932
O whereo	Weight	0.078	0.084	0.081	0.078	0.081	0.078	0.074	0.079	0.079	0.076	0.071	0.065	0.076	
0 00	Weighted Score	0.390	0.418	0.323	0.234	0.323	0.234	0.294	0.317	0.317	0.306	0.283	0.325	0.229	3.994
Owner09	Kaw Weight	0.078	2	0.081	2	0.081	0.078	0 074	0.079	0 079	0.076	2	<u> </u>	0.076	
	Weighted Score	0.234	0.167	0.242	0.156	0.242	0.234	0.221	0.238	0.238	0.229	0.141	0.195	0.229	2.768
Owner10	Raw	2	2	1	1	3	1	2	1	2	2	1	1	2	
	Weighted Score	0.078	0.084	0.081	0.078	0.081	0.078	0.0/4	0.079	0.079	0.076	0.071	0.065	0.076	1 629
Owner11	Raw	3	5	5	2	5	2	3	5	5	5	2	5	3	11027
	Weight	0.078	0.084	0.081	0.078	0.081	0.078	0.074	0.079	0.079	0.076	0.071	0.065	0.076	2.0(4
Owner12	Raw	3	0.418	0.404	3	0.404	3	4	4	0.397	4	3	4	0.229	3.804
0	Weight	0.078	0.084	0.081	0.078	0.081	0.078	0.074	0.079	0.079	0.076	0.071	0.065	0.076	
Oruman12	Weighted Score	0.234	0.418	0.404	0.234	0.242	0.234	0.294	0.317	0.397	0.306	0.212	0.260	0.229	3.782
Owner13	Weight	0.078	0.084	0.081	0.078	0.081	0.078	0.074	0.079	0.079	0.076	0.071	0.065	0.076	
	Weighted Score	0.078	0.084	0.081	0.078	0.081	0.078	0.074	0.397	0.079	0.076	0.071	0.325	0.076	1.577
Owner14	Raw Weight	3	5	3	3	3	4	4	4	5	5	3	5	4	
	Weighted Score	0.078	0.418	0.081	0.078	0.081	0.312	0.294	0.317	0.397	0.382	0.212	0.325	0.306	3.916
Designer01	Raw	5	4	5	1	5	1	2	3	4	3	1	5	3	
	Weighted Score	0.089	0.082	0.080	0.072	0.087	0.078	0.074	0.078	0.070	0.066	0.078	0.064	0.078	3 260
Designer02	Raw	4	5	4	5	4	3	4	4	4	4	3	4	3	5.200
	Weight	0.089	0.082	0.080	0.072	0.087	0.078	0.074	0.078	0.070	0.066	0.078	0.064	0.078	2.020
Designer03	weighted Score Raw	0.554	0.412	0.322	0.362	<u>0.346</u> 4	0.235	<u>0.298</u> 4	<u>0.314</u> 4	0.282	<u>0.266</u> 4	2	<u>0.258</u> 4	0.235	3.920
	Weight	0.089	0.082	0.080	0.072	0.087	0.078	0.074	0.078	0.070	0.066	0.078	0.064	0.078	
Designer04	Weighted Score	0.266	0.330	0.322	0.217	0.346	0.235	0.298	0.314	0.352	0.266	0.157	0.258	0.235	3.596
Designer04	Weight	0.089	4	0.080	0.072	4	0.078	0.074	0.078	0.070	0.066	0.078	0.064	0.078	
	Weighted Score	0.354	0.330	0.322	0.290	0.346	0.314	0.298	0.314	0.352	0.266	0.157	0.258	0.235	3.835
Designer05	Raw	5	5	3	3	5	3	5	5	5	3	2	5	5	
	Weighted Score	0.089	0.082	0.241	0.072	0.087	0.078	0.074	0.392	0.352	0.199	0.157	0.322	0.392	4.169
Designer06	Raw	4	4	3	1	4	3	3	4	5	5	5	5	4	
	Weighted Saara	0.089	0.082	0.080	0.072	0.087	0.078	0.074	0.078	0.070	0.066	0.078	0.064	0.078	3 8 20
Designer07	Raw	4	5	3	3	3	3	3	4	3	3	3	4	3	5.029
	Weight	0.089	0.082	0.080	0.072	0.087	0.078	0.074	0.078	0.070	0.066	0.078	0.064	0.078	
Decimar00	Weighted Score	0.354	0.412	0.241	0.217	0.260	0.235	0.223	0.314	0.211	0.199	0.235	0.258	0.235	3.396
Designet08	Weight	0.089	0.082	0.080	0.072	0.087	0.078	0.074	0.078	0.070	0.066	0.078	0.064	0.078	
D :	Weighted Score	0.266	0.247	0.241	0.145	0.346	0.157	0.149	0.235	0.282	0.266	0.157	0.258	0.235	2.984
Designer09	Kaw Weight	0.080	4	2	0.072	3	2	3	3	3	3	2	3	2	
	Weighted Score	0.266	0.330	0.161	0.145	0.260	0.157	0.223	0.235	0.211	0.199	0.157	0.193	0.157	2.694

]	Design	-Bid-I	Build (DBB)	Pro	ject-In	terest					
		Owner/User Satisfaction	Clear & Realistic Objectives	Effective Communication	Collaboration of Project Team	Utility & Functionality	Trust & Respect	Alignment of Objectives	Owner/User Participation	Production of Specified Quality	Long-term Lifecycle Value	Time Performance	Owner's Vision	Constructability	Weighted Sum Aggregated Relatie Effectiveness
Builder01	Raw	4	4	4	3	4	3	4	4	3	4	3	4	3	
	Weight	0.088	0.075	0.082	0.077	0.080	0.080	0.071	0.082	0.080	0.075	0.080	0.073	0.056	2 (2)
Puilder02	Weighted Score	2	0.299	0.329	0.230	2	2	0.284	0.329	2	0.299	2	0.292	0.168	3.020
Builde102	Weight	0.088	0.075	0.082	0.077	0.080	0.080	0.071	0.082	0.080	0.075	0.080	0.073	0.056	
	Weighted Score	0.000	0.073	0.002	0.230	0.000	0.000	0.213	0.002	0.000	0.073	0.000	0.219	0.168	3,000
Builder03	Raw	1	1	3	2	3	1	1	1	3	2	1	1	1	
	Weight	0.088	0.075	0.082	0.077	0.080	0.080	0.071	0.082	0.080	0.075	0.080	0.073	0.056	
	Weighted Score	0.088	0.075	0.247	0.153	0.241	0.080	0.071	0.082	0.241	0.150	0.080	0.073	0.056	1.637
Builder04	Raw	1	2	3	3	1	3	1	1	2	1	1	2	1	
	Weight	0.088	0.075	0.082	0.077	0.080	0.080	0.071	0.082	0.080	0.075	0.080	0.073	0.056	
	Weighted Score	0.088	0.150	0.247	0.230	0.080	0.241	0.071	0.082	0.161	0.075	0.080	0.146	0.056	1.707
Builder05	Raw	3	3	3	3	3	2	2	3	3	3	3	3	2	
	Weight	0.088	0.075	0.082	0.0//	0.080	0.080	0.0/1	0.082	0.080	0.075	0.080	0.073	0.056	2 702
Duildar06	Weighted Score	0.264	0.224	0.247	0.230	0.241	0.161	0.142	0.247	0.241	0.224	0.241	0.219	0.112	2.793
Builde100	Naw Weight	4	0.075	0.082	0.077	4	0.080	4	0.082	0.080	4	2	0.073	0.056	
	Weighted Score	0.000	0.073	0.082	0.077	0.000	0.080	0.284	0.082	0.000	0.075	0.060	0.073	0.050	3 2 3 4
Builder07	Raw	3	2	3	3	3	2	2	3	4	3	2	4	2	01201
	Weight	0.088	0.075	0.082	0.077	0.080	0.080	0.071	0.082	0.080	0.075	0.080	0.073	0.056	
	Weighted Score	0.264	0.150	0.247	0.230	0.241	0.161	0.142	0.247	0.321	0.224	0.161	0.292	0.112	2.791
Builder08	Raw	3	3	3	3	3	3	3	3	3	3	3	3	3	
	Weight	0.088	0.075	0.082	0.077	0.080	0.080	0.071	0.082	0.080	0.075	0.080	0.073	0.056	
	Weighted Score	0.264	0.224	0.247	0.230	0.241	0.241	0.213	0.247	0.241	0.224	0.241	0.219	0.168	3.000
Builder09	Raw	3	5	3	2	3	2	2	3	4	4	2	3	2	
	Weight	0.088	0.075	0.082	0.077	0.080	0.080	0.071	0.082	0.080	0.075	0.080	0.073	0.056	2.0.40
Llaam01	Weighted Score	0.264	0.374	0.247	0.153	0.241	0.161	0.142	0.247	0.321	0.299	0.161	0.219	0.112	2.940
User01	Woight	3	0.084	0.092	0.091	3	0.075	0.072	0.075	0.072	2	2	0.069	0 077	
	Weighted Score	0.080	0.064	0.085	0.081	0.077	0.075	0.075	0.075	0.073	0.075	0.075	0.008	0.077	2 853
User02	Raw	5	4	3	3	3	5	3	3	5	3	3	5	4	2.035
050102	Weight	0.086	0.084	0.083	0.081	0.077	0.075	0.073	0.075	0.073	0.073	0.073	0.068	0.077	
	Weighted Score	0.431	0.338	0.248	0.242	0.231	0.376	0.220	0.226	0.367	0.220	0.220	0.339	0.308	3.767
User03	Raw	2	1	1	1	2	2	1	2	3	2	1	3	1	
	Weight	0.086	0.084	0.083	0.081	0.077	0.075	0.073	0.075	0.073	0.073	0.073	0.068	0.077	
	Weighted Score	0.172	0.084	0.083	0.081	0.154	0.150	0.073	0.150	0.220	0.147	0.073	0.204	0.077	1.670
User04	Raw	3	4	2	4	3	2	2	2	4	2	1	1	4	
	Weight	0.086	0.084	0.083	0.081	0.077	0.075	0.073	0.075	0.073	0.073	0.073	0.068	0.077	2 (52
LL05	Weighted Score	0.259	0.338	0.165	0.323	0.231	0.150	0.14/	0.150	0.294	0.14/	0.073	0.068	0.308	2.653
Userus	Wajaht	0.086	2	0.082	2	0 077	0.075	0 072	0.075	0.072	0.072	$\frac{2}{0.072}$	0.069	0 077	
	Weighted Score	0.080	0.064	0.085	0.081	0.231	0.075	0.073	0.075	0.073	0.073	0.073	0.008	0.077	2 530
User06	Raw	3	3	3	3	4	3	3	3	4	4	1	3	4	2.550
	Weight	0.086	0.084	0.083	0.081	0.077	0.075	0.073	0.075	0.073	0.073	0.073	0.068	0.077	
	Weighted Score	0.259	0.253	0.248	0.242	0.308	0.226	0.220	0.226	0.294	0.294	0.073	0.204	0.308	3.154
User07	Raw	3	3	4	4	4	4	4	4	4	3	4	3	4	
	Weight	0.086	0.084	0.083	0.081	0.077	0.075	0.073	0.075	0.073	0.073	0.073	0.068	0.077	
	Weighted Score	0.259	0.253	0.330	0.323	0.308	0.301	0.294	0.301	0.294	0.220	0.294	0.204	0.308	3.688
User08	Raw	3	3	3	1	3	3	3	3	3	3	2	2	2	
	Weight	0.086	0.084	0.083	0.081	0.077	0.075	0.073	0.075	0.073	0.073	0.073	0.068	0.077	2 (20
	weighten Score	11/79	10/25	IU 74X		1 11 7 1	11//6	• • • 7 7 ()	111//6	i u 770.	1 11 7 70	1114/	1 1 1 1 1 1	11174	2.620

				D	esign-l	Build	(DB)	- Self-l	Interes	st					
		Owner/User Satisfaction	Clear & Realistic Objectives	Effective Communication	Collaboration of Project Team	Utility & Functionality	Trust & Respect	Alignment of Objectives	Owner/User Participation	Production of Specified Quality	Long-term Lifecycle Value	Time Performance	Owner's Vision	Constructability	Weighted Sum Aggregated Relatie Effectiveness
Owner01	Raw	3	3	3	3	2	1	2	1	2	3	5	2	5	
	Weighted Score	0.082	0.084	0.082	0.082	0.153	0.069	0.156	0.074	0.075	0.079	0.008	0.161	0.009	2.678
Owner02	Raw	3	3	3	2	2	2	2	2	2	2	4	2	4	
	Weighted Score	0.082	0.084	0.082	0.082	0.153	0.009	0.078	0.147	0.075	0.079	0.008	0.161	0.009	2.523
Owner03	Raw	4	4	3	3	3	3	3	3	3	3	3	3	3	
	Weighted Score	0.329	0.334	0.082	0.082	0.229	0.009	0.234	0.221	0.225	0.238	0.008	0.031	0.009	3.166
Owner04	Raw Weight	2	3	2	3	3	2	3	2	3	3	3	3	3	
	Weighted Score	0.164	0.251	0.164	0.082	0.229	0.139	0.234	0.147	0.225	0.238	0.008	0.031	0.009	2.693
Owner05	Raw Weight	1	1	1	1	1	1	1	1	3	2	5	1	1	
	Weighted Score	0.082	0.084	0.082	0.082	0.076	0.069	0.078	0.074	0.225	0.159	0.340	0.081	0.069	1.501
Owner06	Raw Weight	3	3	2	2	2	1	2	1	3	4	4	1	4	
	Weighted Score	0.246	0.251	0.164	0.164	0.153	0.069	0.156	0.074	0.225	0.317	0.272	0.081	0.278	2.450
Owner07	Raw Weight	$\frac{2}{0.082}$	3	$\frac{2}{0.082}$	2	2	2	2	$\frac{2}{0.074}$	2	3	5	3	4	
	Weighted Score	0.164	0.251	0.164	0.164	0.153	0.139	0.156	0.147	0.150	0.238	0.340	0.242	0.278	2.586
Owner08	Raw Weight	5	5	4	3	3	4	3	3	3	4	4	3	3	
	Weighted Score	0.411	0.418	0.329	0.246	0.229	0.278	0.234	0.221	0.225	0.317	0.272	0.242	0.208	3.630
Owner09	Raw Weight	3	1	2	2	2	3	2	2	2	2	5	2	4	
	Weighted Score	0.246	0.084	0.164	0.164	0.153	0.208	0.156	0.147	0.150	0.159	0.340	0.161	0.278	2.411
Owner10	Raw Weight	3	4	3	4	3	3	3	3	3	3	3	2	3	
	Weighted Score	0.246	0.334	0.246	0.329	0.229	0.208	0.234	0.221	0.225	0.238	0.204	0.161	0.208	3.085
Owner11	Raw Weight	4	3	3	4	3	3	2	1 0.074	2	2	3	3	4	
	Weighted Score	0.329	0.251	0.246	0.329	0.229	0.208	0.156	0.074	0.150	0.159	0.204	0.242	0.278	2.854
Owner12	Raw Weight	3	0.084	4	3	3	0.069	3	0.074	<u>3</u> 0.075	0.079	0.068	0.081	4	
	Weighted Score	0.246	0.251	0.329	0.246	0.229	0.208	0.234	0.221	0.225	0.238	0.340	0.242	0.278	3.288
Owner13	Raw Weight	0.082	0.084	1	0.082	1	0.069	1	0.074	0.075	0.079	0.068	0.081	0.069	
	Weighted Score	0.082	0.084	0.082	0.082	0.076	0.208	0.078	0.074	0.075	0.079	0.204	0.081	0.208	1.414
Owner14	Raw Weight	4	0.084	4 0.082	0.082	2 0.076	0.069	2 0.078	<u>2</u> 0.074	2 0.075	2 0.079	4 0.068	2 0.081	4 0.069	
D : 01	Weighted Score	0.329	0.251	0.329	0.246	0.153	0.208	0.156	0.147	0.150	0.159	0.272	0.161	0.278	2.839
Designer01	Kaw Weight	0.086	0.083	0.075	0.081	0.079	0.077	0.079	0.069	0.075	0.075	0.069	2 0.079	0.077	
Desis	Weighted Score	0.086	0.165	0.224	0.242	0.236	0.230	0.236	0.206	0.149	0.149	0.206	0.157	0.230	2.517
Designer02	Kaw Weight	<u>2</u> 0.086	<u> </u>	<u>2</u> <u>0.0</u> 75	<u>2</u> 0.081	4 0.079	<u> </u>	<u> </u>	<u> </u>	<u>2</u> <u>0.0</u> 75	<u> </u>	<u>2</u> 0.069	4 <u>0.0</u> 79	<u> </u>	
Designar	Weighted Score	0.173	0.248	0.149	0.161	0.314	0.153	0.236	0.206	0.149	0.224	0.138	0.314	0.230	2.695
Designer03	Weight	0.086	4 0.083	0.075	2 0.081	0.079	2 0.077	2 0.079	2 0.069	4 0.075	0.075	2 0.069	4 0.079	2 0.077	
Designan04	Weighted Score	0.259	0.330	0.149	0.161	0.236	0.153	0.157	0.138	0.299	0.224	0.138	0.314	0.153	2.711
Designer04	Weight	0.086	0.083	0.075	0.081	0.079	0.077	0.079	0.069	0.075	0.075	0.069	0.079	0.077	
Designan05	Weighted Score	0.173	0.330	0.149	0.161	0.157	0.153	0.157	0.138	0.299	0.149	0.275	0.157	0.306	2.605
Designer05	Weight	0.086	0.083	0.075	0.081	0.079	0.077	0.079	0.069	0.075	0.075	0.069	0.079	0.077	
Designant	Weighted Score	0.173	0.248	0.224	0.322	0.236	0.306	0.157	0.138	0.224	0.224	0.206	0.236	0.230	2.923
Designer06	Weight	0.086	0.083	0.075	0.081	<u> </u>	0.077	<u>2</u> 0.079	2 0.069	0.075	0.075	0.069	<u></u> 0.079	0.077	
Designar07	Weighted Score	0.259	0.248	0.224	0.242	0.236	0.153	0.157	0.138	0.224	0.149	0.206	0.157	0.230	2.623
Designer0/	Weight	0.086	0.083	0.075	0.081	<u> </u>	4 0.077	0.079	0.069	4 0.075	<u>4</u> 0.075	4 0.069	<u> </u>	4 0.077	
Designario	Weighted Score	0.259	0.248	0.224	0.242	0.236	0.306	0.236	0.138	0.299	0.299	0.275	0.236	0.306	3.303
Designer08	Weight	0.086	4 0.083	0.075	4 0.081	<u> </u>	4 0.077	<u> </u>	0.069	0.075	0.075	0.069	<u> </u>	4 0.077	
Designer00	Weighted Score	0.259	0.330	0.224	0.322	0.236	0.306	0.236	0.206	0.224	0.224	0.344	0.236	0.306	3.454
Designet09	Weight	0.086	0.083	0.075	0.081	0.079	0.077	0.079	0.069	0.075	0.075		0.079	0.077	
	Weighted Score	0.173	0.165	0.299	0.242	0.236	0.230	0.236	0.138	0.149	0.224	0.275	0.236	0.230	2.831

				D	esign-l	Build	(DB)	- Self-l	Interes	st					
		Owner/User Satisfaction	Clear & Realistic Objectives	Effective Communication	Collaboration of Project Team	Utility & Functionality	Trust & Respect	Alignment of Objectives	Owner/User Participation	Production of Specified Quality	Long-term Lifecycle Value	Time Performance	Owner's Vision	Constructability	Weighted Sum Aggregated Relatie Effectiveness
Builder01	Raw	3	4	3	3	3	3	3	3	3	3	3	4	4	
	Weight	0.089	0.076	0.082	0.067	0.076	0.082	0.078	0.084	0.076	0.069	0.080	0.072	0.069	2 217
Builder02	Raw	0.208	0.305	0.245	0.201	0.229	0.245	0.234	0.231	0.229	0.200	0.240	0.290	0.275	5.217
Dunder02	Weight	0.089	0.076	0.082	0.067	0.076	0.082	0.078	0.084	0.076	0.069	0.080	0.072	0.069	
	Weighted Score	0.446	0.381	0.409	0.335	0.381	0.409	0.390	0.418	0.381	0.344	0.400	0.362	0.344	5.000
Builder03	Raw	3	4	3	4	4	3	3	3	3	3	4	4	4	
	Weight	0.089	0.076	0.082	0.067	0.076	0.082	0.078	0.084	0.076	0.069	0.080	0.072	0.069	
D 11 04	Weighted Score	0.268	0.305	0.245	0.268	0.305	0.245	0.234	0.251	0.229	0.206	0.320	0.290	0.275	3.441
Builder04	Kaw	5	4	4	4	4	4	3	4	4	3	5	4	5	
	Weighted Score	0.089	0.076	0.082	0.067	0.076	0.082	0.078	0.084	0.076	0.069	0.080	0.072	0.069	4 091
Builder05	Raw	4	4	4	5	4	3	4	4	4	3	4	4	3	4.071
Builderot	Weight	0.089	0.076	0.082	0.067	0.076	0.082	0.078	0.084	0.076	0.069	0.080	0.072	0.069	
	Weighted Score	0.357	0.305	0.327	0.335	0.305	0.245	0.312	0.335	0.305	0.206	0.320	0.290	0.206	3.848
Builder06	Raw	4	3	4	4	4	4	2	4	4	4	5	3	4	
	Weight	0.089	0.076	0.082	0.067	0.076	0.082	0.078	0.084	0.076	0.069	0.080	0.072	0.069	
	Weighted Score	0.357	0.229	0.327	0.268	0.305	0.327	0.156	0.335	0.305	0.275	0.400	0.217	0.275	3.775
Builder07	Raw	4	3	4	4	4	3	3	4	3	3	4	3	4	
	Weight	0.089	0.076	0.082	0.06/	0.076	0.082	0.078	0.084	0.076	0.069	0.080	0.0/2	0.069	2 5 4 6
Duilder08	Weighted Score	0.357	0.229	0.327	0.268	0.305	0.245	0.234	0.335	0.229	0.206	0.320	0.217	0.275	3.540
Dullde108	Weight	0.089	0.076	0.082	0.067	0.076	0.082	0.078	0.084	0.076	0.069	0.080	0.072	0.069	
	Weighted Score	0.009	0.229	0.002	0.007	0.229	0.002	0.234	0.004	0.229	0.005	0.000	0.072	0.005	3.000
Builder09	Raw	3	3	4	3	3	2	3	2	4	4	3	2	4	
	Weight	0.089	0.076	0.082	0.067	0.076	0.082	0.078	0.084	0.076	0.069	0.080	0.072	0.069	
	Weighted Score	0.268	0.229	0.327	0.201	0.229	0.164	0.234	0.167	0.305	0.275	0.240	0.145	0.275	3.058
User01	Raw	3	3	3	3	3	3	3	3	3	3	3	3	3	
	Weight	0.087	0.079	0.081	0.091	0.072	0.077	0.083	0.074	0.077	0.075	0.066	0.074	0.064	2.000
LL	Weighted Score	0.260	0.238	0.243	0.272	0.215	0.232	0.249	0.221	0.232	0.226	0.198	0.221	0.192	3.000
User02	Kaw Waight	3	3	4	4	4	4	4	4	4	3	0.066	4	4	
	Weighted Score	0.087	0.079	0.001	0.091	0.072	0.077	0.003	0.074	0.077	0.075	0.000	0.074	0.004	3 825
User03	Raw	2.	3	2	2	3	2	2	1	3	2	3	3	3	0.020
	Weight	0.087	0.079	0.081	0.091	0.072	0.077	0.083	0.074	0.077	0.075	0.066	0.074	0.064	
	Weighted Score	0.174	0.238	0.162	0.181	0.215	0.155	0.166	0.074	0.232	0.151	0.198	0.221	0.192	2.358
User04	Raw	4	4	4	4	4	4	4	4	4	4	4	4	4	
	Weight	0.087	0.079	0.081	0.091	0.072	0.077	0.083	0.074	0.077	0.075	0.066	0.074	0.064	1.000
XX 05	Weighted Score	0.347	0.317	0.325	0.362	0.287	0.309	0.332	0.294	0.309	0.302	0.264	0.294	0.257	4.000
User05	Kaw	4	3	3	3	4	3	3	4	4	4	4	4	4	
	Weighted Score	0.087	0.079	0.081	0.091	0.072	0.077	0.085	0.074	0.077	0.075	0.000	0.074	0.004	3 589
User06	Raw	3	3	2	3	3	2	2	3	3	2	3	3	3	5.557
000100	Weight	0.087	0.079	0.081	0.091	0.072	0.077	0.083	0.074	0.077	0.075	0.066	0.074	0.064	
	Weighted Score	0.260	0.238	0.162	0.272	0.215	0.155	0.166	0.221	0.232	0.151	0.198	0.221	0.192	2.683
User07	Raw	2	2	2	3	2	2	3	2	3	2	5	2	4	
	Weight	0.087	0.079	0.081	0.091	0.072	0.077	0.083	0.074	0.077	0.075	0.066	0.074	0.064	
XX 06	Weighted Score	0.174	0.158	0.162	0.272	0.143	0.155	0.249	0.147	0.232	0.151	0.330	0.147	0.257	2.577
User08	Raw	2	2	2	2	1	1	1	1	3	1	3	1	3	
	Weighted Score	0.08/	0.079	0.081	0.091	0.072	0.077	0.083	0.074	0.077	0.075	0.066	0.074	0.064	1 753

				Des	sign-B	uild (I) - P	roject	-Inter	est					
		Owner/User Satisfaction	Clear & Realistic Objectives	Effective Communication	Collaboration of Project Team	Utility & Functionality	Trust & Respect	Alignment of Objectives	Owner/User Participation	Production of Specified Quality	Long-term Lifecycle Value	Time Performance	Owner's Vision	Constructability	Weighted Sum Aggregated Relatie Effectiveness
Owner01	Raw	3	3	3	3	2	1	2	1	2	3	5	2	5	
	Weighted Score	0.078	0.084	0.081	0.078	0.162	0.078	0.147	0.079	0.159	0.070	0.354	0.003	0.382	2.681
Owner02	Raw	3	3	3	2	2	2	2	2	2	2	4	2	4	
	Weighted Score	0.078	0.084	0.081	0.078	0.081	0.078	0.074	0.079	0.079	0.076	0.071	0.065	0.076	2.537
Owner03	Raw	4	4	3	3	3	3	3	3	3	3	3	3	3	
	Weight Weighted Score	0.078	0.084	0.081	0.078	0.081	0.078	0.074	0.079	0.079	0.076	0.071	0.065	0.076	3,162
Owner04	Raw	2	3	2	3	3	2	2	2	2	3	3	3	3	
	Weighted Score	0.078	0.084	0.081	0.078	0.081	0.078	0.074	0.079	0.079	0.076	0.071	0.065	0.076	2.531
Owner05	Raw	1	1	1	1	1	1	1	1	3	2	5	1	1	21001
	Weight Weighted Score	0.078	0.084	0.081	0.078	0.081	0.078	0.074	0.079	0.079	0.076	0.071	0.065	0.076	1 518
Owner06	Raw	4	3	1	1	2	1	1	1	3	3	3	3	4	11010
	Weight	0.078	0.084	0.081	0.078	0.081	0.078	0.074	0.079	0.079	0.076	0.071	0.065	0.076	2 294
Owner07	Raw	3	3	3	2	3	2	3	2	2	3	5	3	4	2,2)4
	Weight	0.078	0.084	0.081	0.078	0.081	0.078	0.074	0.079	0.079	0.076	0.071	0.065	0.076	2 003
Owner08	Raw	5	5	4	3	4	3	2	4	3	4	4	3	3	2.905
	Weight	0.078	0.084	0.081	0.078	0.081	0.078	0.074	0.079	0.079	0.076	0.071	0.065	0.076	3 6 3 8
Owner09	Raw	4	3	2	3	2	3	3	2	2	3	4	2	4	5.058
	Weight	0.078	0.084	0.081	0.078	0.081	0.078	0.074	0.079	0.079	0.076	0.071	0.065	0.076	2.940
Owner10	Raw	3	3	3	3	3	3	3	3	3	3	3	3	3	2.840
	Weight	0.078	0.084	0.081	0.078	0.081	0.078	0.074	0.079	0.079	0.076	0.071	0.065	0.076	2 000
Owner11	Weighted Score Raw	0.234	0.251	0.242	0.234	0.242	0.234	0.221	0.238	0.238	0.229	0.212	0.195	0.229	3.000
	Weight	0.078	0.084	0.081	0.078	0.081	0.078	0.074	0.079	0.079	0.076	0.071	0.065	0.076	
Owner12	Weighted Score Raw	0.312	0.167	0.242	0.312	0.242	0.234	0.147	0.079	0.159	0.153	0.212	0.195	0.306	2.760
0 W MOT 12	Weight	0.078	0.084	0.081	0.078	0.081	0.078	0.074	0.079	0.079	0.076	0.071	0.065	0.076	
Owner13	Weighted Score Raw	0.234	0.335	0.323	0.234	0.242	0.234	0.221	0.238	0.238	0.229	0.354	0.195	0.306	3.382
0 110115	Weight	0.078	0.084	0.081	0.078	0.081	0.078	0.074	0.079	0.079	0.076	0.071	0.065	0.076	
Owner1/	Weighted Score	0.234	0.251	0.242	0.234	0.242	0.234	0.221	0.079	0.238	0.229	0.354	0.065	0.382	3.006
Owner14	Weight	0.078	0.084	0.081	0.078	0.081	0.078	0.074	0.079	0.079	0.076	0.071	0.065	0.076	
Designer()1	Weighted Score	0.312	0.335	0.323	0.234	0.242	0.234	0.221	0.159	0.317	0.153	0.283	0.130	0.382	3.325
Designet01	Weight	0.089	0.082	0.080	0.072	0.087	0.078	0.074	0.078	0.070	0.066	0.078	0.064	0.078	
Designer()	Weighted Score	0.266	0.247	0.402	0.217	0.346	0.157	0.223	0.078	0.070	0.133	0.235	0.129	0.314	2.819
Designer02	Weight	0.089	0.082	0.080	0.072	0.087	0.078	0.074	0.078	0.070	0.066	0.078	0.064	0.078	
Designer()?	Weighted Score	0.177	0.330	0.080	0.145	0.346	0.157	0.223	0.235	0.141	0.199	0.235	0.193	0.314	2.777
Designer05	Weight	0.089	0.082	0.080	0.072	0.087	0.078	0.074	0.078	0.070	0.066	0.078	0.064	0.078	
Designan04	Weighted Score	0.354	0.330	0.241	0.217	0.260	0.157	0.223	0.235	0.282	0.199	0.157	0.193	0.235	3.085
Designer04	Weight	0.089	0.082	0.080	0.072	0.087	0.078	0.074	0.078	0.070	0.066	0.078	0.064	0.078	
D	Weighted Score	0.177	0.247	0.241	0.217	0.260	0.235	0.223	0.235	0.282	0.199	0.392	0.193	0.392	3.296
Designer05	Weight	0.089	0.082	0.080	0.072	0.087	4 0.078	0.074	0.078	0.070	0.066	0.078	0.064	0.078	
D i O(Weighted Score	0.177	0.247	0.241	0.290	0.260	0.314	0.223	0.235	0.211	0.199	0.235	0.193	0.235	3.062
Designer06	Kaw Weight	0.089	0.082	0.080	0.072	0.087	0.078	0.074	2 0.078	0.070	2 0.066	0.078	0.064	0.078	
D :	Weighted Score	0.266	0.247	0.241	0.217	0.260	0.157	0.149	0.157	0.211	0.133	0.235	0.129	0.235	2.638
Designer07	Raw Weight	0.089	0.082	0.080	0.072	0.087	4	0.074	0.078	4	4	0.078	0.064	4	
	Weighted Score	0.266	0.247	0.241	0.217	0.260	0.314	0.223	0.235	0.282	0.266	0.235	0.193	0.314	3.294
Designer08	Raw Weight	4	4	4	4	3	4 0.078	4	3	3	3	5	3	4	
	Weighted Score	0.354	0.330	0.322	0.290	0.260	0.314	0.298	0.235	0.211	0.199	0.392	0.193	0.314	3.712
Designer09	Raw Weight	2	2	3	3	3	3	3	2	3	3	5	3	3	
	Weighted Score	0.177	0.165	0.241	0.217	0.260	0.235	0.223	0.157	0.211	0.199	0.392	0.193	0.235	2.907

				Des	sign-B	uild (I)B) - P	roject	-Inter	est					
		Owner/User Satisfaction	Clear & Realistic Objectives	Effective Communication	Collaboration of Project Team	Utility & Functionality	Trust & Respect	Alignment of Objectives	Owner/User Participation	Production of Specified Quality	Long-term Lifecycle Value	Time Performance	Owner's Vision	Constructability	Weighted Sum Aggregated Relatie Effectiveness
Builder01	Raw	3	4	3	3	3	3	3	3	3	3	3	4	4	
	Weight	0.088	0.075	0.082	0.077	0.080	0.080	0.071	0.082	0.080	0.075	0.080	0.073	0.056	2 204
Duildar02	Weighted Score	0.264	0.299	0.247	0.230	0.241	0.241	0.213	0.247	0.241	0.224	0.241	0.292	0.224	5.204
Duide102	Weight	0.088	0.075	0.082	0.077	0.080	0.080	0.071	0.082	0.080	0.075	0.080	0.073	0.056	
	Weighted Score	0.000	0.374	0.002	0.383	0.000	0.000	0.355	0.002	0.000	0.374	0.000	0.364	0.030	5,000
Builder03	Raw	3	4	3	4	4	3	3	3	3	3	4	4	4	
	Weight	0.088	0.075	0.082	0.077	0.080	0.080	0.071	0.082	0.080	0.075	0.080	0.073	0.056	
	Weighted Score	0.264	0.299	0.247	0.307	0.321	0.241	0.213	0.247	0.241	0.224	0.321	0.292	0.224	3.441
Builder04	Raw	5	4	4	4	4	4	3	4	4	3	5	4	5	
	Weight	0.088	0.075	0.082	0.077	0.080	0.080	0.071	0.082	0.080	0.075	0.080	0.073	0.056	
	Weighted Score	0.439	0.299	0.329	0.307	0.321	0.321	0.213	0.329	0.321	0.224	0.402	0.292	0.280	4.079
Builder05	Raw	4	4	4	5	4	3	4	4	4	3	4	4	3	
	Weight	0.088	0.075	0.082	0.077	0.080	0.080	0.071	0.082	0.080	0.075	0.080	0.073	0.056	2.965
Duildan06	Weighted Score	0.351	0.299	0.329	0.383	0.321	0.241	0.284	0.329	0.321	0.224	0.321	0.292	0.168	3.865
Builderuo	Woight	4	4	4	4	4	4	0.071	4	4	4	0.090	4	4	
	Weighted Score	0.000	0.075	0.082	0.077	0.000	0.000	0.071	0.082	0.080	0.075	0.080	0.073	0.030	1 009
Builder07	Raw	4	3	4	4	<u>0.321</u> <u>4</u>	3	3	4	3	3	<u>0.402</u>	3	<u>0.224</u>	4.002
Dunderoy	Weight	0.088	0.075	0.082	0 077	0.080	0.080	0.071	0.082	0.080	0.075	0.080	0.073	0.056	
	Weighted Score	0.351	0.224	0.329	0.307	0.321	0.241	0.213	0.329	0.241	0.224	0.321	0.219	0.224	3.546
Builder08	Raw	3	3	3	3	3	3	3	3	3	3	3	3	3	
	Weight	0.088	0.075	0.082	0.077	0.080	0.080	0.071	0.082	0.080	0.075	0.080	0.073	0.056	
	Weighted Score	0.264	0.224	0.247	0.230	0.241	0.241	0.213	0.247	0.241	0.224	0.241	0.219	0.168	3.000
Builder09	Raw	2	3	4	3	3	2	3	2	4	4	3	2	4	
	Weight	0.088	0.075	0.082	0.077	0.080	0.080	0.071	0.082	0.080	0.075	0.080	0.073	0.056	
	Weighted Score	0.176	0.224	0.329	0.230	0.241	0.161	0.213	0.164	0.321	0.299	0.241	0.146	0.224	2.970
User01	Raw	4	3	4	3	4	4	4	3	3	3	3	3	3	
	Weight	0.086	0.084	0.083	0.081	0.077	0.075	0.073	0.075	0.073	0.073	0.073	0.068	0.077	2 20 4
Llaam02	Weighted Score	0.345	0.253	0.330	0.242	0.308	0.301	0.294	0.226	0.220	0.220	0.220	0.204	0.231	3.394
User02	Kaw Woight	0.096	0.094	4	4	4	4	4	4	0.072	0.072	0.072	3	4	
	Weighted Score	0.080	0.064	0.085	0.081	0.077	0.075	0.075	0.075	0.073	0.073	0.075	0.008	0.077	3 688
User03	Raw	3	3	3	3	3	3	3	2	3	2	3	2	3	5.000
0.50105	Weight	0.086	0.084	0.083	0.081	0.077	0.075	0.073	0.075	0.073	0.073	0.073	0.068	0.077	
	Weighted Score	0.259	0.253	0.248	0.242	0.231	0.226	0.220	0.150	0.220	0.147	0.220	0.136	0.231	2.783
User04	Raw	4	4	4	4	4	4	4	3	4	4	4	4	4	
	Weight	0.086	0.084	0.083	0.081	0.077	0.075	0.073	0.075	0.073	0.073	0.073	0.068	0.077	
	Weighted Score	0.345	0.338	0.330	0.323	0.308	0.301	0.294	0.226	0.294	0.294	0.294	0.272	0.308	3.925
User05	Raw	4	3	3	3	4	3	3	4	4	4	4	4	4	
	Weight	0.086	0.084	0.083	0.081	0.077	0.075	0.073	0.075	0.073	0.073	0.073	0.068	0.077	
LL OC	Weighted Score	0.345	0.253	0.248	0.242	0.308	0.226	0.220	0.301	0.294	0.294	0.294	0.272	0.308	3.604
User06	Kaw	3	3	3	3	2	2	3	4	2	2	4	3	2	
	Weight	0.086	0.084	0.083	0.081	0.0//	0.075	0.073	0.075	0.0/3	0.0/3	0.073	0.068	0.0//	2 772
Licer07	weighted Score	0.239	0.253	0.248	2 0.242	2 0.154	0.150	0.220	0.301	2	0.14/	0.294	0.204	0.154	2.772
User0/	Kaw Wajaht	0.086	4	0.092	0.091	0 077	0.075	0.072	0.075	0.072	$\frac{2}{0.072}$	0.072	0.068	0.077	
	Weighted Score	0.000	0.004	0.083	0.081	0.077	0.075	0.073	0.075	0.073	0.073	0.073	0.008	0.077	3 3 3 0
User08	Raw	3	4	3	2	2	2	2	3	2	2	3	1	3	5.557
030100	Weight	0.086	0.084	0.083	0.081	0.077	0.075	0.073	0.075	0.073	0.073	0.073	0.068	0.077	
	Weighted Score	0.259	0.338	0.248	0.161	0.154	0.150	0.147	0.226	0.147	0.147	0.220	0.068	0.231	2 495

			Int	egrate	d Proj	ect De	livery	(IPD)	Self	Intere	est				
		Owner/User Satisfaction	Clear & Realistic Objectives	Effective Communication	Collaboration of Project Team	Utility & Functionality	Trust & Respect	Alignment of Objectives	Owner/User Participation	Production of Specified Quality	Long-term Lifecycle Value	Time Performance	Owner's Vision	Constructability	Weighted Sum Aggregated Relatie Effectiveness
Owner01	Raw Weight Weighted Score	5 0.082 0.411	5 0.084 0.418	5 0.082 0.411	4 0.082 0.329	4 0.076 0.306	3 0.069 0.208	5 0.078 0.390	5 0.074 0.368	4 0.075 0.300	3 0.079 0.238	4 0.068 0.272	4 0.081 0.323	4 0.069 0.278	4.251
Owner02	Raw Weight Weighted Score	4 0.082 0.329	4 0.084 0.334	4 0.082 0.329	4 0.082 0.329	4 0.076 0.306	4 0.069 0.278	4 0.078 0.312	4 0.074 0.295	4 0.075 0.300	4 0.079 0.317	4 0.068 0.272	4 0.081 0.323	4 0.069 0.278	4.000
Owner03	Raw Weight Weighted Score	2 0.082 0.164	2 0.084 0.167	2 0.082 0.164	3 0.082 0.246	3 0.076 0.229	3 0.069 0.208	3 0.078 0.234	2 0.074 0.147	2 0.075 0.150	3 0.079 0.238	3 0.068 0.204	2 0.081 0.161	3 0.069 0.208	2.523
Owner04	Raw Weight Weighted Score	4 0.082 0.329	4 0.084 0.334	4 0.082 0.329	4 0.082 0.329	4 0.076 0.306	4 0.069 0.278	3 0.078 0.234	3 0.074 0.221	4 0.075 0.300	4 0.079 0.317	3 0.068 0.204	4 0.081 0.323	4 0.069 0.278	3.780
Owner05	Raw Weight Weighted Score	$ \begin{array}{r} 2 \\ 0.082 \\ 0.164 \end{array} $	2 0.084 0.167	3 0.082 0.246	2 0.082 0.164	2 0.076 0.153	2 0.069 0.139	2 0.078 0.156	3 0.074 0.221	3 0.075 0.225	3 0.079 0.238	4 0.068 0.272	2 0.081 0.161	5 0.069 0.347	2.654
Owner06	Raw Weight Weighted Score	5 0.082 0.411	5 0.084 0.418	5 0.082 0.411	5 0.082 0.411	5 0.076 0.382	4 0.069 0.278	5 0.078 0.390	4 0.074 0.295	4 0.075 0.300	5 0.079 0.397	5 0.068 0.340	4 0.081 0.323	3 0.069 0.208	4.562
Owner07	Raw Weight Weighted Score	5 0.082 0.411	4 0.084 0.334	5 0.082 0.411	5 0.082 0.411	5 0.076 0.382	4 0.069 0.278	4 0.078 0.312	5 0.074 0.368	4 0.075 0.300	5 0.079 0.397	4 0.068 0.272	5 0.081 0.404	4 0.069 0.278	4.557
Owner08	Raw Weight Weighted Score	4 0.082 0.329	4 0.084 0.334	5 0.082 0.411	4 0.082 0.329	4 0.076 0.306	4 0.069 0.278	4 0.078 0.312	4 0.074 0.295	4 0.075 0.300	4 0.079 0.317	5 0.068 0.340	4 0.081 0.323	4 0.069 0.278	4.150
Owner09	Raw Weight Weighted Score	2 0.082 0.164	3 0.084 0.251	3 0.082 0.246	3 0.082 0.246	3 0.076 0.229	3 0.069 0.208	3 0.078 0.234	3 0.074 0.221	3 0.075 0.225	3 0.079 0.238	4 0.068 0.272	3 0.081 0.242	4 0.069 0.278	3.055
Owner10	Raw Weight Weighted Score	4 0.082 0.329	4 0.084 0.334	4 0.082 0.329	5 0.082 0.411	4 0.076 0.306	4 0.069 0.278	4 0.078 0.312	4 0.074 0.295	4 0.075 0.300	4 0.079 0.317	3 0.068 0.204	4 0.081 0.323	4 0.069 0.278	4.014
Owner11	Raw Weight Weighted Score	5 0.082 0.411	4 0.084 0.334	4 0.082 0.329	5 0.082 0.411	4 0.076 0.306	5 0.069 0.347	5 0.078 0.390	4 0.074 0.295	4 0.075 0.300	3 0.079 0.238	4 0.068 0.272	4 0.081 0.323	5 0.069 0.347	4.302
Owner12	Raw Weight Weighted Score	4 0.082 0.329	5 0.084 0.418	5 0.082 0.411	5 0.082 0.411	3 0.076 0.229	5 0.069 0.347	5 0.078 0.390	5 0.074 0.368	5 0.075 0.375	5 0.079 0.397	4 0.068 0.272	5 0.081 0.404	5 0.069 0.347	4.697
Owner13	Raw Weight Weighted Score	5 0.082 0.411	5 0.084 0.418	3 0.082 0.246	5 0.082 0.411	5 0.076 0.382	5 0.069 0.347	5 0.078 0.390	3 0.074 0.221	5 0.075 0.375	5 0.079 0.397	5 0.068 0.340	3 0.081 0.242	5 0.069 0.347	4.527
Owner14	Raw Weight Weighted Score	3 0.082 0.246	4 0.084 0.334	5 0.082 0.411	5 0.082 0.411	4 0.076 0.306	5 0.069 0.347	5 0.078 0.390	5 0.074 0.368	4 0.075 0.300	4 0.079 0.317	5 0.068 0.340	3 0.081 0.242	5 0.069 0.347	4.360
Designer01	Raw Weight Weighted Score	3 0.086 0.259	3 0.083 0.248	5 0.075 0.373	4 0.081 0.322	5 0.079 0.393	5 0.077 0.383	4 0.079 0.314	5 0.069 0.344	5 0.075 0.373	3 0.075 0.224	4 0.069 0.275	5 0.079 0.393	3 0.077 0.230	4.132
Designer02	Raw Weight Weighted Score	4 0.086 0.346	5 0.083 0.413	5 0.075 0.373	4 0.081 0.322	4 0.079 0.314	5 0.077 0.383	5 0.079 0.393	5 0.069 0.344	5 0.075 0.373	5 0.075 0.373	5 0.069 0.344	5 0.079 0.393	5 0.077 0.383	4.754
Designer03	Raw Weight Weighted Score	4 0.086 0.346	4 0.083 0.330	4 0.075 0.299	4 0.081 0.322	4 0.079 0.314	4 0.077 0.306	4 0.079 0.314	5 0.069 0.344	5 0.075 0.373	5 0.075 0.373	4 0.069 0.275	5 0.079 0.393	4 0.077 0.306	4.297
Designer04	Raw Weight Weighted Score	4 0.086 0.346	3 0.083 0.248	4 0.075 0.299	5 0.081 0.403	5 0.079 0.393	5 0.077 0.383	5 0.079 0.393	4 0.069 0.275	4 0.075 0.299	5 0.075 0.373	5 0.069 0.344	5 0.079 0.393	4 0.077 0.306	4.454
Designer05	Raw Weight Weighted Score	4 0.086 0.346	2 0.083 0.165	4 0.075 0.299	5 0.081 0.403	3 0.079 0.236	4 0.077 0.306	4 0.079 0.314	3 0.069 0.206	2 0.075 0.149	4 0.075 0.299	4 0.069 0.275	3 0.079 0.236	3 0.077 0.230	3.464
Designer06	Raw Weight Weighted Score	2 0.086 0.173	2 0.083 0.165	4 0.075 0.299	4 0.081 0.322	2 0.079 0.157	1 0.077 0.077	1 0.079 0.079	1 0.069 0.069	1 0.075 0.075	1 0.075 0.075	1 0.069 0.069	1 0.079 0.079	2 0.077 0.153	1.790
Designer07	Raw Weight Weighted Score	2 0.086 0.173	1 0.083 0.083	1 0.075 0.075	1 0.081 0.081	2 0.079 0.157	1 0.077 0.077	1 0.079 0.079	1 0.069 0.069	1 0.075 0.075	1 0.075 0.075	3 0.069 0.206	2 0.079 0.157	1 0.077 0.077	1.381
Designer08	Raw Weight Weighted Score	5 0.086 0.432	5 0.083 0.413	4 0.075 0.299	4 0.081 0.322	4 0.079 0.314	4 0.077 0.306	4 0.079 0.314	4 0.069 0.275	4 0.075 0.299	4 0.075 0.299	4 0.069 0.275	5 0.079 0.393	4 0.077 0.306	4.248
Designer09	Raw Weight Weighted Score	4 0.086 0.346	4 0.083 0.330	5 0.075 0.373	5 0.081 0.403	3 0.079 0.236	3 0.077 0.230	4 0.079 0.314	5 0.069 0.344	4 0.075 0.299	3 0.075 0.224	4 0.069 0.275	4 0.079 0.314	4 0.077 0.306	3.994

			Int	egrate	d Proj	ect De	livery	(IPD)	Self	Intere	est				
		Owner/User Satisfaction	Clear & Realistic Objectives	Effective Communication	Collaboration of Project Team	Utility & Functionality	Trust & Respect	Alignment of Objectives	Owner/User Participation	Production of Specified Quality	Long-term Lifecycle Value	Time Performance	Owner's Vision	Constructability	Weighted Sum Aggregated Relatie Effectiveness
Builder01	Raw Weight Weighted Score	4 0.089 0.357	4 0.076 0.305	3 0.082 0.245	4 0.067 0.268	4 0.076 0.305	4 0.082 0.327	4 0.078 0.312	4 0.084 0.335	4 0.076 0.305	4 0.069 0.275	4 0.080 0.320	3 0.072 0.217	4 0.069 0.275	3.846
Builder02	Raw Weight Weighted Score	3 0.089 0.268	3 0.076 0.229	$\frac{3}{0.082}$	3 0.067 0.201	3 0.076 0.229	$\frac{3}{0.082}$	$\frac{3}{0.078}$	3 0.084 0.251	3 0.076 0.229	3 0.069 0.206	$\frac{3}{0.080}$	3 0.072 0.217	3 0.069 0.206	3,000
Builder03	Raw Weight Weighted Score	5 0.089 0.446	5 0.076 0.381	5 0.082 0.409	5 0.067 0.335	5 0.076 0.381	4 0.082 0.327	4 0.078 0.312	4 0.084 0.335	3 0.076 0.229	4 0.069 0.275	4 0.080 0.320	4 0.072 0.290	4 0.069 0.275	4 314
Builder04	Raw Weight	<u>3</u> 0.089 0.268	5 0.076 0.381	5 0.082 0.409	5 0.067 0.335	5 0.076 0.381	5 0.082 0.409	5 0.078 0.390	5 0.084 0.418	5 0.076 0.381	5 0.069 0.344	3 0.080 0.240	5 0.072 0.362	<u>3</u> 0.069	4,524
Builder05	Raw Weight	4 0.089 0.357	4 0.076 0.305	4 0.082 0.327	5 0.067 0.335	4 0.076 0.305	3 0.082 0.245	4 0.078 0.312	5 0.084 0.418	4 0.076 0.305	3 0.069 0.206	4 0.080 0.320	4 0.072 0.290	3 0.069 0.206	3.931
Builder06	Raw Weight Weighted Score	5 0.089 0.446	2 0.076 0.152	4 0.082 0.327	5 0.067 0.335	5 0.076 0.381	5 0.082 0.409	3 0.078 0.234	0.410 5 0.084 0.418	4 0.076 0.305	4 0.069 0.275	4 0.080 0.320	4 0.072 0.290	5 0.069 0.344	4 236
Builder07	Raw Weight Weighted Score	0.440 5 0.089 0.446	4 0.076 0.305	5 0.082 0.409	0.0055 0.067 0.335	5 0.076 0.381	4 0.082 0.327	4 0.078 0.312	0.410 5 0.084 0.418	4 0.076 0.305	5 0.069 0.344	5 0.080 0.400	5 0.072 0.362	3 0.069 0.206	4,550
Builder08	Raw Weight Weighted Score	3 0.089 0.268	3 0.076 0.229	$\frac{3}{0.082}$	3 0.067 0.201	3 0.076 0.229	$\frac{3}{0.082}$	$\frac{3}{0.078}$	3 0.084 0.251	3 0.076 0.229	3 0.069 0.206	$\frac{3}{0.080}$	$\frac{3}{0.072}$	3 0.069 0.206	3.000
Builder09	Raw Weight Weighted Score	4 0.089 0.357	4 0.076 0.305	5 0.082 0.409	5 0.067 0.335	4 0.076 0.305	4 0.082 0.327	5 0.078 0.390	4 0.084 0.335	5 0.076 0.381	5 0.069 0.344	5 0.080 0.400	5 0.072 0.362	5 0.069 0.344	4 593
User01	Raw Weight	<u>3</u> 0.087	<u>3</u> 0.079	<u>3</u> 0.081	<u>3</u> 0.091	<u>3</u> 0.072 0.215	<u>3</u> <u>0.077</u>	<u>3</u> 0.083	0.333 2 0.074	<u>3</u> 0.077	<u>3</u> 0.075	3 0.066 0.108	<u>3</u> 0.074	<u>3</u> 0.064	2 926
User02	Raw Weight	4 0.087	4 0.079	5 0.081	5 0.091	5 0.072	0.232 5 0.077	5 0.083	0.147 5 0.074	0.232 5 0.077	5 0.075	4 0.066 0.264	5 0.074	0.192 5 0.064	4 769
User03	Raw Weighted Score	0.347 5 0.087	0.317 5 0.079	0.400 5 0.081	0.455 5 0.091 0.453	0.338 5 0.072 0.358	0.387 5 0.077 0.387	0.413 5 0.083	0.308 5 0.074 0.368	0.387 5 0.077 0.387	0.377 5 0.075 0.377	0.204 5 0.066 0.330	0.308 5 0.074 0.368	0.321 5 0.064	5.000
User04	Raw Weight	4 0.087 0.347	4 0.079 0.317	3 0.081 0.243	4 0.091 0.362	4 0.072 0.287	4 0.077 0.309	4 0.083 0.332	5 0.074 0.368	4 0.077 0.309	5 0.075 0.377	5 0.066 0.330	5 0.074 0.368	4 0.064 0.257	4.208
User05	Raw Weight	5 0.087 0.434	5 0.079 0.396	5 0.081 0.406	5 0.091 0.453	5 0.072 0.358	5 0.077 0.387	5 0.083 0.415	5 0.074 0.368	5 0.077 0.387	5 0.075 0.377	5 0.066 0.330	4 0.074 0.294	5 0.064 0.321	4 926
User06	Raw Weighted Score	$\frac{0.454}{4}$ 0.087	4	<u>3</u> 0.081	4	<u>3</u> 0.072	2 0.077 0.155	<u>3</u> 0.083	4	<u>3</u> 0.077	<u>3</u> 0.075	<u>3</u> 0.066	<u>3</u> 0.074	<u>3</u> 0.064	3 252
User07	Raw Weight	0.047 5 0.087	0.079	0.243 5 0.081	0.302 5 0.091	0.215 5 0.072	0.155 5 0.077	0.249 5 0.083	0.294 5 0.074	0.252 5 0.077	4	2 0.066	0.221 5 0.074	<u>3</u> 0.064	
User08	Weighted Score Raw Weight Weighted Score	<u>0.454</u> <u>3</u> 0.087 0.260	<u>3</u> 0.079 0.238	<u>3</u> 0.081 0.243	0.453 3 0.091 0.272	0.358 3 0.072 0.215	<u>0.387</u> <u>3</u> 0.077 0.232	<u> </u>	0.368 3 0.074 0.221	<u> </u>	<u>0.302</u> <u>3</u> 0.075 0.226	0.152 1 0.066 0.066	0.368 3 0.074 0.221	1 0.064 0.064	2.740

			Integ	rated	Projec	t Deli	very (I	PD)	Proje	ct-Inte	rest				
		Owner/User Satisfaction	Clear & Realistic Objectives	Effective Communication	Collaboration of Project Team	Utility & Functionality	Trust & Respect	Alignment of Objectives	Owner/User Participation	Production of Specified Quality	Long-term Lifecycle Value	Time Performance	Owner's Vision	Constructability	Weighted Sum Aggregated Relatie Effectiveness
Owner01	Raw Weight Weighted Score	5 0.078 0.390	5 0.084 0.418	5 0.081 0.404	4 0.078 0.312	4 0.081 0.323	3 0.078 0.234	5 0.074 0.368	5 0.079 0.397	4 0.079 0.317	3 0.076 0.229	4 0.071 0.283	4 0.065 0.260	4 0.076 0.306	4.241
Owner02	Raw Weight Weighted Score	4 0.078 0.312	4 0.084 0.335	4 0.081 0.323	4 0.078 0.312	4 0.081 0.323	4 0.078 0.312	4 0.074 0.294	4 0.079 0.317	4 0.079 0.317	4 0.076 0.306	4 0.071 0.283	4 0.065 0.260	4 0.076 0.306	4.000
Owner03	Raw Weight Weighted Score	2 0.078 0.156	2 0.084 0.167	3 0.081 0.242	4 0.078 0.312	3 0.081 0.242	3 0.078 0.234	3 0.074 0.221	3 0.079 0.238	3 0.079 0.238	3 0.076 0.229	3 0.071 0.212	3 0.065 0.195	3 0.076 0.229	2.916
Owner04	Raw Weight Weighted Score	4 0.078 0.312	4 0.084 0.335	4 0.081 0.323	4 0.078 0.312	4 0.081 0.323	4 0.078 0.312	2 0.074 0.147	3 0.079 0.238	4 0.079 0.317	4 0.076 0.306	3 0.071 0.212	4 0.065 0.260	4 0.076 0.306	3.703
Owner05	Raw Weight Weighted Score	2 0.078 0.156	2 0.084 0.167	3 0.081 0.242	2 0.078 0.156	2 0.081 0.162	2 0.078 0.156	2 0.074 0.147	3 0.079 0.238	3 0.079 0.238	3 0.076 0.229	4 0.071 0.283	2 0.065 0.130	5 0.076 0.382	2.687
Owner06	Raw Weight Weighted Score	5 0.078 0.390	5 0.084 0.418	5 0.081 0.404	5 0.078 0.390	4 0.081 0.323	5 0.078 0.390	5 0.074 0.368	4 0.079 0.317	4 0.079 0.317	4 0.076 0.306	5 0.071 0.354	4 0.065 0.260	$\frac{3}{0.076}$	4.466
Owner07	Raw Weight Weighted Score	5 0.078 0.390	4 0.084 0.335	5 0.081 0.404	5 0.078 0.390	5 0.081 0.404	4 0.078 0.312	4 0.074 0.294	4 0.079 0.317	4 0.079 0.317	5 0.076 0.382	4 0.071 0.283	5 0.065 0.325	4 0.076 0.306	4.459
Owner08	Raw Weighted Score	4 0.078 0.312	4 0.084 0.335	5 0.081 0.404	4 0.078 0.312	$\frac{3}{0.081}$	3 0.078 0.234	4 0.074 0.294	4 0.079 0.317	4 0.079 0.317	4 0.076 0.306	5 0.071 0.354	4 0.065 0.260	4 0.076 0.306	3 993
Owner09	Raw Weighted Score	4 0.078 0.312	2 0.084 0.167	3 0.081 0.242	3 0.078 0.234	3 0.081 0.242	3 0.078 0.234	3 0.074 0.221	4 0.079 0.317	4 0.079 0.317	4 0.076 0.306	4 0.071 0.283	3 0.065 0.195	4 0.076 0.306	3 377
Owner10	Raw Weight	4 0.078 0.312	$4 \\ 0.084 \\ 0.335$	4 0.081 0.323	4 0.078 0.312	4 0.081 0.323	4 0.078 0.312	4 0.074 0.294	4 0.079 0.317	4 0.079 0.317	4 0.076 0.306	3 0.071 0.212	4 0.065 0.260	4 0.076 0.306	3.929
Owner11	Raw Weight Weighted Score	5 0.078 0.390	4 0.084 0.335	4 0.081 0.323	5 0.078 0.390	4 0.081 0.323	5 0.078 0.390	5 0.074 0.368	4 0.079 0.317	4 0.079 0.317	3 0.076 0.229	4 0.071 0.283	4 0.065 0.260	5 0.076 0.382	4.307
Owner12	Raw Weight	$\frac{3}{0.078}$	5 0.084 0.418	5 0.081 0.404	5 0.078 0.390	3 0.081 0.242	5 0.078 0.390	5 0.074 0.368	5 0.079 0.397	5 0.079 0.397	5 0.076 0.382	4 0.071 0.283	5 0.065 0.325	5 0.076 0.382	4.612
Owner13	Raw Weight	5 0.078 0.390	5 0.084 0.418	5 0.081 0.404	5 0.078 0.390	5 0.081 0.404	5 0.078 0.390	5 0.074 0.368	3 0.079 0.238	5 0.079 0.397	5 0.076 0.382	3 0.071 0.212	3 0.065 0.195	$\frac{3}{0.076}$	4.417
Owner14	Raw Weight	5 0.078 0.390	5 0.084 0.418	5 0.081 0.404	5 0.078 0.390	5 0.081 0.404	5 0.078 0.390	4 0.074 0.294	5 0.079 0.397	4 0.079 0.317	4 0.076 0.306	5 0.071 0.354	5 0.065 0.325	5 0.076 0.382	4,771
Designer01	Raw Weight Weighted Score	4 0.089 0.354	2 0.082 0.165	5 0.080 0.402	5 0.072 0.362	5 0.087 0.433	5 0.078 0.392	4 0.074 0.298	5 0.078 0.392	5 0.070 0.352	3 0.066 0.199	5 0.078 0.392	5 0.064 0.322	4 0.078 0.314	4.378
Designer02	Raw Weight Weighted Score	5 0.089 0.443	5 0.082 0.412	5 0.080 0.402	5 0.072 0.362	5 0.087 0.433	5 0.078 0.392	5 0.074 0.372	5 0.078 0.392	5 0.070 0.352	5 0.066 0.332	5 0.078 0.392	5 0.064 0.322	5 0.078 0.392	5.000
Designer03	Raw Weight Weighted Score	5 0.089 0.443	4 0.082 0.330	5 0.080 0.402	4 0.072 0.290	5 0.087 0.433	5 0.078 0.392	5 0.074 0.372	5 0.078 0.392	4 0.070 0.282	5 0.066 0.332	4 0.078 0.314	5 0.064 0.322	4 0.078 0.314	4.618
Designer04	Raw Weight Weighted Score	5 0.089 0.443	5 0.082 0.412	5 0.080 0.402	5 0.072 0.362	5 0.087 0.433	5 0.078 0.392	5 0.074 0.372	5 0.078 0.392	4 0.070 0.282	5 0.066 0.332	4 0.078 0.314	5 0.064 0.322	4 0.078 0.314	4.773
Designer05	Raw Weight Weighted Score	4 0.089 0.354	4 0.082 0.330	5 0.080 0.402	5 0.072 0.362	4 0.087 0.346	5 0.078 0.392	4 0.074 0.298	3 0.078 0.235	3 0.070 0.211	3 0.066 0.199	5 0.078 0.392	4 0.064 0.258	4 0.078 0.314	4.095
Designer06	Raw Weight Weighted Score	2 0.089 0.177	2 0.082 0.165	4 0.080 0.322	4 0.072 0.290	2 0.087 0.173	1 0.078 0.078	1 0.074 0.074	1 0.078 0.078	1 0.070 0.070	1 0.066 0.066	1 0.078 0.078	1 0.064 0.064	2 0.078 0.157	1.795
Designer07	Raw Weight Weighted Score	2 0.089 0.177	1 0.082 0.082	1 0.080 0.080	1 0.072 0.072	1 0.087 0.087	1 0.078 0.078	1 0.074 0.074	1 0.078 0.078	1 0.070 0.070	1 0.066 0.066	4 0.078 0.314	1 0.064 0.064	1 0.078 0.078	1.324
Designer08	Raw Weight Weighted Score	5 0.089 0.443	5 0.082 0.412	5 0.080 0.402	4 0.072 0.290	4 0.087 0.346	4 0.078 0.314	4 0.074 0.298	4 0.078 0.314	4 0.070 0.282	4 0.066 0.266	4 0.078 0.314	5 0.064 0.322	4 0.078 0.314	4.316
Designer09	Raw Weight Weighted Score	4 0.089 0.354	4 0.082 0.330	5 0.080 0.402	5 0.072 0.362	3 0.087 0.260	4 0.078 0.314	4 0.074 0.298	4 0.078 0.314	4 0.070 0.282	3 0.066 0.199	5 0.078 0.392	4 0.064 0.258	4 0.078 0.314	4.078

			Integ	rated	Projec	t Deliv	very (I	PD)	Proje	ct-Inte	rest				
		Owner/User Satisfaction	Clear & Realistic Objectives	Effective Communication	Collaboration of Project Team	Utility & Functionality	Trust & Respect	Alignment of Objectives	Owner/User Participation	Production of Specified Quality	Long-term Lifecycle Value	Time Performance	Owner's Vision	Constructability	Weighted Sum Aggregated Relatie Effectiveness
Builder01	Raw Weight	4	4	3 0.082	4	4	4	4	4 0.082	4	4	4	3 0.073	4 0.056	2.945
Builder02	Weighted Score Raw Weight Weighted Score	0.351 3 0.088 0.264	0.299 3 0.075 0.224	0.247 3 0.082 0.247	0.307 3 0.077 0.230	0.321 3 0.080 0.241	0.321 3 0.080 0.241	0.284 3 0.071 0.213	0.329 3 0.082 0.247	0.321 3 0.080 0.241	0.299 3 0.075 0.224	0.321 3 0.080 0.241	0.219 3 0.073 0.219	0.224 4 0.056 0.224	3.845
Builder03	Raw Weight Weighted Score	5 0.088 0.439	5 0.075 0.374	5 0.082 0.411	5 0.077 0.383	5 0.080 0.402	4 0.080 0.321	4 0.071 0.284	4 0.082 0.329	3 0.080 0.241	4 0.075 0.299	4 0.080 0.321	4 0.073 0.292	4 0.056 0.224	4.321
Builder04	Raw Weight Weighted Score	5 0.088 0.439	5 0.075 0.374	5 0.082 0.411	5 0.077 0.383	5 0.080 0.402	5 0.080 0.402	5 0.071 0.355	5 0.082 0.411	5 0.080 0.402	5 0.075 0.374	3 0.080 0.241	5 0.073 0.364	3 0.056 0.168	4.727
Builder05	Raw Weight Weighted Score	4 0.088 0.351	4 0.075 0.299	4 0.082 0.329	5 0.077 0.383	4 0.080 0.321	3 0.080 0.241	4 0.071 0.284	5 0.082 0.411	4 0.080 0.321	3 0.075 0.224	4 0.080 0.321	4 0.073 0.292	3 0.056 0.168	3.948
Builder06	Raw Weight Weighted Score	5 0.088 0.439	4 0.075 0.299	5 0.082 0.411	5 0.077 0.383	5 0.080 0.402	5 0.080 0.402	3 0.071 0.213	5 0.082 0.411	4 0.080 0.321	4 0.075 0.299	4 0.080 0.321	4 0.073 0.292	5 0.056 0.280	4.475
Builder07	Raw Weight Weighted Score	5 0.088 0.439	4 0.075 0.299	5 0.082 0.411	5 0.077 0.383	5 0.080 0.402	4 0.080 0.321	4 0.071 0.284	5 0.082 0.411	4 0.080 0.321	4 0.075 0.299	5 0.080 0.402	5 0.073 0.364	3 0.056 0.168	4.507
Builder08	Raw Weight Weighted Score	3 0.088 0.264	3 0.075 0.224	3 0.082 0.247	3 0.077 0.230	3 0.080 0.241	3 0.080 0.241	3 0.071 0.213	3 0.082 0.247	3 0.080 0.241	3 0.075 0.224	3 0.080 0.241	3 0.073 0.219	3 0.056 0.168	3.000
Builder09	Raw Weight Weighted Score	4 0.088 0.351	4 0.075 0.299	5 0.082 0.411	5 0.077 0.383	4 0.080 0.321	4 0.080 0.321	5 0.071 0.355	4 0.082 0.329	5 0.080 0.402	5 0.075 0.374	5 0.080 0.402	5 0.073 0.364	5 0.056 0.280	4.594
User01	Raw Weight Weighted Score	4 0.086 0.345	3 0.084 0.253	3 0.083 0.248	4 0.081 0.323	4 0.077 0.308	4 0.075 0.301	4 0.073 0.294	3 0.075 0.226	3 0.073 0.220	3 0.073 0.220	3 0.073 0.220	3 0.068 0.204	3 0.077 0.231	3.393
User02	Raw Weight Weighted Score	4 0.086 0.345	5 0.084 0.422	5 0.083 0.413	5 0.081 0.404	5 0.077 0.385	4 0.075 0.301	5 0.073 0.367	5 0.075 0.376	5 0.073 0.367	5 0.073 0.367	4 0.073 0.294	4 0.068 0.272	5 0.077 0.385	4.69
User03	Raw Weight Weighted Score	5 0.086 0.431	5 0.084 0.422	5 0.083 0.413	5 0.081 0.404	5 0.077 0.385	5 0.075 0.376	5 0.073 0.367	5 0.075 0.376	5 0.073 0.367	5 0.073 0.367	5 0.073 0.367	5 0.068 0.339	5 0.077 0.385	5.000
User04	Raw Weight Weighted Score	3 0.086 0.259	4 0.084 0.338	3 0.083 0.248	4 0.081 0.323	4 0.077 0.308	4 0.075 0.301	4 0.073 0.294	4 0.075 0.301	4 0.073 0.294	4 0.073 0.294	4 0.073 0.294	4 0.068 0.272	4 0.077 0.308	3.83
User05	Raw Weight Weighted Score	5 0.086 0.431	5 0.084 0.422	5 0.083 0.413	5 0.081 0.404	5 0.077 0.385	5 0.075 0.376	5 0.073 0.367	5 0.075 0.376	5 0.073 0.367	5 0.073 0.367	5 0.073 0.367	4 0.068 0.272	5 0.077 0.385	4.932
User06	Raw Weight Weighted Score	4 0.086 0.345	4 0.084 0.338	3 0.083 0.248	3 0.081 0.242	3 0.077 0.231	2 0.075 0.150	3 0.073 0.220	4 0.075 0.301	2 0.073 0.147	2 0.073 0.147	3 0.073 0.220	3 0.068 0.204	3 0.077 0.231	3.024
User07	Raw Weight Weighted Score	2 0.086 0.172	2 0.084 0.169	5 0.083 0.413	5 0.081 0.404	5 0.077 0.385	5 0.075 0.376	5 0.073 0.367	5 0.075 0.376	5 0.073 0.367	4 0.073 0.294	2 0.073 0.147	5 0.068 0.339	3 0.077 0.231	4.04
User08	Raw Weight	2 0.086	3 0.084	2 0.083	3 0.081	2 0.077	3 0.075	3 0.073	2 0.075	3 0.073	3 0.073	1 0.073	3 0.068	1 0.077	2 2 7 9

BIBLIOGRAPHY

The Buy American Act United States Code § 41 USC Sec. 10a-10d (2009).

- Agresti, Alan, and Barbara Finlay. *Statistical Methods for the Social Sciences*. Upper Saddle River, NJ: Prentice Hall, 2009.
- Alberti, Leon Battista. *On the Art of Building in Ten Books*. Translated by Josheph Rykwert, Neil Leach, and Robert Travernor. Cambridge, MA: MIT Press, 1402-1472.
- Allison, Markku. "The Future is Integrated Project Delivery." *Archi-Tech* March 2009. Stamats Business Media. http://www.architechweb.com/DesktopModules/BB_ArticleMax/ ArticleDetail/BBArticleDetail. Accessed 18 July 2009.
- American Institute of Architects, California Council. Integrated Project Delivery -- A Working Definition. AIA California Council. Sacramento, CA: AIA California, 2007.
 ——. Integrated Project Delivery Frequently Asked Questions, edited by California Council American Institute of Architects. Sacramento, CA: AIA California Council, August 2006.
- Baker, Norma J., and Lawrence S. Wrightsman. "Nature of Mixed-Motive Games: Introduction." In *Cooperation and Competition: Readings on Mixed-Motive Games*, edited by Lawrence S. Wrightsman, John O'Connor, and Norma J. Baker, 1-7. Belmont, CA: Brooks/Cole Publishing Company, 1972.
- Bedrisk, Jim, and Tony Rinella. "8 Technology, Process Improvement, and Cultural Change." In *Report on Integrated Practice*. Webcor Builders and Anshen+Allen Architects (Respectively). San Mateo and San Francisco, CA: American Institute of Architects, 2007.
- Bennett, John. Construction: The Third Way. Oxford: Butterworth-Heinemann, 2000.
- Bernstein, Phillip G., and Martin Hague. "Integrated Project Delivery (IPD): Why Owners Choose Multi-Party." Webinar. Washington, DC: AGC of America (The Associated General Contractors of America), 2009.
- Bresciani, J. Marilee et al. "Examining Design and Inter-Rater Reliability of a Rubric Measuring Research Quality Across Multiple Disciplines." *Practical Assessment, Research & Evaluation* 14, no. 12 (May 2009).

- Brewer, Graham, and Thayaparan Gajendran. "Emerging ICT Trends in Construction Project Teams: A Delphi Study." *Journal of Information Technology in Construction* 14 March 2009: 80-97. ITcon. www.itcon.org. Accessed 6 August 2010.
- Broshar, Michael, AIA Vice President 2005-2007 -- Board Knowledge Committee Chair. "0
 Preface." In *Report on Integrated Practice*. American Institute of Architects, 1. Waterloo, Iowa: American Institute of Architects, 2007. Special Report American Institute of Architects.
- Brown, Dale Ray. "A Designing Image: Integrating Design, Planning, and Decision Theories with Cognitive Processes." Ph. D. diss., Architecture, University of California, Berkley, 1992.
- Bryson, Barbara White, and Canan Yetmen. "Why Owners Make The Difference." *ENR: Engineering News-Record* (New York), 02 August 2010, 80.
- Burns, Joseph. "7 Applications in Engineering." In *Report on Integrated Practice*. Thornton-Tomasetti Group. Chicago, IL: American Institute of Architects, 2007.
- Chatterjee, Kalyan, and William F. Samuelson. *Game Theory and Business Applications*. Boston, MA: Klumer Academic Publishers, 2001.
- Cheng, Renee. "5 Suggestions for an Integrative Education." In *Report on Integrated Practice*. University of Minnesota. American Institute of Architects, 2007.
- Chua, D.K.H, Y.C. Kog, and P.K. Loh. "Critical Success Factors for Different Project Objectives." *Journal of Construction Engineering and Management* 125, no. 3 (May/June 1999): 142-50.
- Chung, Raymond Waiman. "The Role and Function Opportunities of Architect in the Design/Build Method of Building Construction." Ph. D. diss., University of Toronto (Canada), 1992. *Masters of Architecture* (1992).
- Colledge, Barbara. "Relational Contracting Creating Value Beyond the Project." *Lean Construction Journal* 2, no. 1 (April 2005): 30-45.
- Constructeck magazine. "Integrated Project Delivery." Specialty Publishing Co. 18 July 2009. http://www.constructech.com/news/articles/article.aspx?article_id=7619. Accessed 18 July 2009.

- Corgel, John B., David C. Ling, and Halbert C. Smith. *Real Estate Perspectives, An Introduction* to Real Estate, Fourth Edition. New York: McGraw-Hill/Irwin, 2001.
- Coulton, J.J. Ancient Greek Architects at Work, Problems of Structure and Design. Ithaca, NY: Cornell University Press, 1977.
- Crawford, Lynn H., and Jane Helm. "Government and Governance: The Value of Project Management in the Public Sector." *Project Management Journal* 40, no. 1 (March 2009): 73-87.
- Cuhls, Kerstin. "Delphi Method." Edited by United Nations Industrial Development Organization. 2010. United Nations Industrial Development Organization. http://www.unido.org/fileadmin/import/16959_DelphiMethod.pdf. Accessed 16 August 2010.
- Curiel, Imma. Cooperative Game Theory and Applications: Cooperative Game Arising from Combinatorial Optimization Problems. Boston, MA: Klumer Academic Publishers, 1997.
- Baiden, B.K., Price A.D.F., and A.R.J. Dainty. "The Extent of Team Integration Within Construction Projects." *International Journal of Project Management* 24 (2006): 13-23.
- Decision Making with the Analytic Network Process: Economic, Political, Social and Technological Applications with Benefits, Opportunities, Costs and Risks. Edited by Thomas L. Saaty and Luis G. Vargas. New York: Springer Science+Business Medical, LLC, 2006.
- U.S. Department of Defense, Cong., Defense Federal Acquisition Regulation (DFAR) Supplement (1998) (Revised January 15, 2009).
- Delbecq, Andre L., Andrew H. Van de Ven, and David H. Gustafson. Group Techniques for Program Planning: A Guide to Nominal Group and Delphi Processes. Glenview, IL: Scott, Foresman, 1975.
- The Delphi Method: Techniques and Applications / Edited by Harold A. Linstone, Murray Turoff; with a Foreword by Olaf Helmer. Edited by Murray Turoff and Harold A. Linstone. Reading, MA: Addison-Wesley Pub. Co., Advanced Book Program, 2002.
- Deutsch, Morton. *The Handbook of Conflict Resolution: Theory and Practice*. Edited by Morton Deutsch and Peter T. Coleman. San Francisco, CA: Jossey-Bas Publishers, 2000.
- Eastman, Chuck, Contributing author. "2 University and Industry Research in Support of BIM." In *Report on Integrated Practice*. Georgia Institute of Technology. Atlanta, GA: American Institute of Architects, 2007.
- Eckblad, Stuart, Zigmund Rabel, and Jim Bedrick, The American Institute of Architects.
 "Integrated Project Delivery: Putting It All Together." In *AIA 150 Convention 2007: Growing Beyond Green.* Integrated Project Delivery: What, Why, and How. San Francisco, CA: The American Institute of Architects - California Council, 2 May, 2007.
- Elvin, George. Integrated Practice in Architecture: Mastering Design-Build, Fast-Track, and Building Information Modeling. Hoboken, New Jersey: John Wiley & Sons, 2007.
 ——. "A Process Model for Integrated Design and Construction." Ph. D. diss., University of California, Berkley, 1998.
- ENR. "The Top 400 Contractors Sourcebook." Special issue of *ENR: Engineering News-Record* 265, no. 7 (13 September 2010 2010). New York: The McGraw-Hill Companies.
 . "The Top 500 Design Firms Sourcebook." Special issue of *ENR: Engineering News-Record* 265, no. 1 (05 July 2010 2010). New York: The McGraw-Hill Companies.
- Eskerod, Pernile, and Eva Riis. "Value Creation by Building an Intraorganizational Common Frame of Reference Concerning Project Management." *Project Management Journal* 40, no. 3 (September 2009): 6-13.
- Fallon, Kristine K., and Stephen R. Hagen. "10 Information for the Facility Live Cycle." In *Report on Integrated Practice*. Kristine Fallon Associates, Inc; U.S. General Services Administration. Chicago, IL; Washington D.C.: American Institute of Architects, 2007.
- Federal Register, Department of Defense, Office of the Secretary. Base Closures and Realignments (BRAC); Notice, Vol. 70, No. 93, 28030-61. Washington, DC: National Archives and Records Administration, 2005.
- Federal Register, Presidential Documents, Title 3 The President. Executive Order 13423 of January 24, 2007 - Strengthening Federal Environmental, Energy, and Transportation Management, Vol. 72, No.17, 3979-23. Washington, D.C.: National Archives and Records Administration, 2007.
- Filarete (aka Antonio di Piero Averlino). *Filarete's Treatise on Architecture*. 1965. Translated by John R. Spencer. New Haven: Yale University Press, c1460.

- Fisher, Richard Barnard. "Partnering Public Sector Construction Contracts: A Conflict Avoidance Process." Ph. D. diss., Virginia Commonwealth University, 2004. *Ph.D.* (2004).
- Friedman, Daniel S. "Architectural Education and Practice on the Verge." In *Report on Integrated Practice*. School of Architecture, University of Illinois at Chicago. American Institute of Architects, 2006.
- Fuller, David N. "Value Creation: Theory and Practice." 2001. http://www.valueinc.com/Press/files/ValueCreation-TheoryandPractice.pdf. Accessed 2 November 2010.
- *Game Practice: Contributions from Applied Game Theory*. Edited by Fioravante Patrone, Ignacio Garcia-Jurado, and Stef Tijs. Boston, MA: Klumer Academic Publishers, 2000.
- Garcia, Salvador, and Francisco Herrera. "An Extension on "Statistical Comparisons of Classifiers Over Multiple Data Sets" for All Pairwise Comparisons." *Journal of Machine Learning Research* 9 (August 2008).
- Geckil, Ilhan Kubilay, and Patrick L. Anderson. *Applied Game Theory and Strategic Behavior*. Boca Raton, FL: CRC Press, 2010.
- Geltner, David M. et al. *Commercial Real Estate Analysis and Investments, Second Edition.* Mason, OH: Thomson Higher Education, 2001.
- Gibson, G. Edward, Giovanni C. Migliaccio, and James T. O'Connor. "Changing Project Delivery Strategy: An Implementation Framework." *Public Works Management and Policy* 12, no. 3 (January 2008): 483-502.
- Griffith, Andrew F. et al. "Project Success Index for Capital Facility Construction Projects." *Journal of Performance of Constructed Facilities* 13, no. 1 (February 1999): 39-45.
- Haskell, Preston H. "Construction Industry Megatrends." In *Haskell White Papers*. Http://Www.Thehaskellco.Com/en/News/WhitePapers/, edited by Haskell. Jacksonville, FL: Haskell, 18 June 2009.
- ——. Construction Industry Productivity.
 - Http://Www.Thehaskellco.Com/en/News/WhitePapers/, edited by Haskell. Jacksonville, FL: Haskell, 18 June 2009.
- *Risk Reduction in Design-Build.* Http://Www.Thehaskellco.Com/en/News/WhitePapers/, edited by Haskell. Jacksonville, FL: Haskell.

- Hinze, Jimmie W. *Construction Planning and Scheduling*. Upper Saddle River, NJ: Prentice Hall, 1998.
- Ho, S. Ping, and Liang Y. Liu. "Analytical Model for Analyzing Construction Claims and Opportunistic Bidding." *Journal of Construction Engineering and Management* 130, no. 1 (1 February 2004): 94-104.
- Hogg, Robert V., and Elliot A. Tanis. *Probability and Statistical Inference, Seventh Edition*. Upper Saddle River, NJ: Pearson Prentice Hall, 2006.
- Howell, Greg, and Will Lichtig. "Special Report: LCI Response to the May 5th, 2010 ENR Article, "Integrated-Project-Delivery Boosters Ignore Many Flashing Red Lights."" 21 May 2010. Lean Construction Institute. http://www.leanconstruction.org/pdf/ SpecialReportENR.pdf. Accessed 25 May 2010.
- Howell, Ian. "9 International Developments." In *Report on Integrated Practice*. Newforma, Inc. Manchester, NH: American Institute of Architects, 2007.
- Hsu, Chia-Chien, and Brian A. Sandford. "The Delphi Technique: Making Sense of Consensus." *Practical Assessment, Research & Evaluation* 125, no. 10 August 2007. http://pareonline.net/pdf/v12n10.pdf. Accessed 09 September 2010.
- Hyun, ChangTaek et al. "Effect of Delivery Methods on Design Performance in Multifamily Housing Projects." *Journal of Construction Engineering and Management* 134, no. 7 1 July 2008. www.acselibrargy.org. Accessed 5 August 2010.
- Integrated Designs. "Integrated Project Delivery (IPD): IDI Project Delivery Tool." 2009. http://www.i-designs.ca. Accessed 19 June 2009.
- Interagency Sustainability Working Group, The Office of the Federal Environmental Executive. Federal Leadership in High Performance and Sustainable Buildings Memorandum of Understanding. Washington, DC, 2006.
- International Council for research and Innovation in Building and Construction (CIB). *CIB White Paper on IDDS: Integrated Design and Delivery Solutions*. Edited by Robert Owen. Rotterdam, The Netherlands: CIB General Secretariat, 2009.
- Jackson, Barbara J. Construction Management Jump Start. Indianapolis, IN: Wiley Publishing, Inc., 2004.

- Jonassen, James O. "3 Changing Business Models in BIM-Driven Integrated Practice." In *Report on Integrated Practice*. NBBJ. Seattle, Washington: American Institute of Architects, 2007.
- Jones, Barry Kenneth. "A Model for Collaborative Engineering in the Construction Industry." Ph. D. diss., University of Southampton (United Kingdom), 1998. *Ph.D.* (1998).
- Keeney, Ralph L., Raiffa. *Decisions with Multiple Objectives: Preferences and Value Tradeoffs.* New York: John Wiley & Sons, 1976.
- Kim, Michael Kyong-il. "Countermodeling as a Strategy for Decision Making: Epistemological Problems in Design." Ph. D. diss., University of California, Berkley, 1980.
 - ———. ARCH 599 Report Feedback. Mkkim1@illinois.edu. University of Illinois at Urbana-Champaign, July 01, 2009.
- ———. 2011. File: Manuscript in Progress. Integrative Design of Buildings: Principles and Strategies. University of Illinois at Urbana-Champaign, Unpublished.
- ———. "Pedagogic Principles for Comprehensive Design Integration." Presented at the ConnectEd 2007 - The International Conference on Design Education. Sydney, Australia, 8-12 July, 2007.
- ———. "What Would You Say Now, Mr. Vitruvius? Building Design Teleology, Then and Now." Presented at the ConnectEd 2010 - 2nd International Conference on Design Education. University of New South Wales, Sydney, Australia, 28 June - 1 July, 2010.
- King, Ross. Brunelleschi's Dome: How a Renaissance Genius Reinvented Architecture. New York: Penguin Books, 2000.
- Kizer, Kenneth W., Merrily McGowan, and Sheila Bowman. Achieving World Class: An Independent Review of the Design Plans for the Walter Reed National Military Medical Center and the Fort Belvoir Community Hospital. National Capital Region Base Realignment and Closure Health Systems Advisory Subcommittee of the Defense Health Board. Office of the Assistant Secretary of Defense for Health Affairs, 2009.
- Kliniotou, Maria. "Identifying, Measuring, and Monitoring Value During Project Development." *European Journal of Engineering Education* 29, no. 3 (September 2004): 367-76.

- Kurmel, Thomas David. "Projecting Building Technology for Hospitals: A Study of Growth and Change in Diagnostic Imaging." D. Des. diss., Graduate School of Design, Harvard University, 1991.
- Lesniewski, Luara, Eddy Krygiel, and Bob Berkebile. "4 Roadmap for Integration." In *Report on Integrated Practice*. BNIM Architects. Kansas City, MO: American Institute of Architects, 2007.
- Levy, Sidney M. Design-Build Project Delivery: Managing the Building Process from Proposal Through Construction. New York: McGraw-Hill, 2006.
- Lichtig, William. "Ten Key Decision to A Successful Construction Project (Choosing Something New: The Integrated Agreement for Lean Project Delivery)." Presented at the American Bar Association Forum on the Construction Industry. The Intercontinental Toronto Center: American Bar Association, September 29-30, 2005.
- Lim, C.S., and M. Zain Mohamed. "Criteria of Project Success: An Exploratory Re-Examination." *International Journal of Project Management* 17, no. 4 (August 1999): 243-48.
- Lo, W., C. L. Lin, and M. R. Yan. "Contractor's Opportunistic Bidding Behavior and Equilibrium Price Level in the Construction Market." *Journal of Construction Engineering and Management* 133, no. 6 (1 June 2007): 409-16.
- Matthews, Owen, and Gregory A. Howell. "Integrated Project Delivery An Example of Relational Contracting." *Lean Construction Journal* 2, no. 1 (April 2005): 46-61.
- Mayne, Thom, Remarks on building information modeling at the 2005 AIA Convention, Las Vegas. "1 - Change or Perish." In *Report on Integrated Practice*. Morphosis, Santa Monica, CA. Las Vegas, NV: American Institute of Architects, 2007.
- McKew, Howard. "Integrated Project Delivery: Moving Both Parties to the Same Side of the Table, It's the Long-Awaited Answer to D-B Project Delivery Problems." *Engineered Systems* January 2009: 108.
- McNall, Todd Weston. "Bridging: An Alternative Project Delivery Method." Ph. D. diss., Texas Tech University, 1998. *Masters of Architecture* (1998).

- Menches, Cindy L., and Awad S. Hanna. "Quantitative Measurement of Successful Performance from the Project Manager's Perspective." *Journal of Construction Engineering & Management* 132, no. 12 (December 2006): 1284-93.
- Mendelson, Elliott. *Introducing Game Theory and Its Applications*. Boca Raton, FL: Chapman & Hall/CRC, 2004.
- Michaud, Pierre. "The True Rule of the Marquis de Condorcet." In *Compromise, Negotiation and Group Decision*, edited by Bertrand R. Munier and Melvin F. Shakun, 83-100.Boston, MA: D. Reidel Publishing Company, 1988.
- Murvin, H.L. *The Architect's Responsibilities: In the Project Delivery Process*. Oakland, CA: American Institute of Architects, 1982.
- Novitski, B.J. "New AIA Agreements Support Integrated Project Delivery." *Architectural Record* 196, no. 7 (July 2008): 59-59.
- Nunnally, S. W. Construction Methods and Management. Upper Saddle River, NJ: Prentice Hall, 2001.
- Onuma, Kimon G. "6 The Twenty-First Century Practitioner." In *Report on Integrated Practice*. Onuma & Associates. Tokyo and Pasadena, CA: American Institute of Architects, 2007.
- O'Connor, Jr., Patrick J. Integrated Project Delivery: Collaboration Through New Contract Forms. Faegre & Benson, LLP. Minneapolis, MN: The Associated General Contractors of America, 2009.
- Palladio, Andrea. *The Four Books on Architecture*. 1997. Translated by Robert Tavernor and Richard Schofield. Cambridge, MA: The MIT Press, c1570.
- Pena-Mora, Feniosky, Carlos E. Sosa, and D. Sean McCone. *Introduction to Construction Dispute Resolution*. Upper Saddle River, NJ: Prentice Hall, 2003.
- Peniwati, Kirti. "Chapter 13: Criteria for Evaluating Group Decision-Making Methods." In Decision Making with the Analytic Network Process: Economic, Political, Social and Technological Applications with Benefits, Opportunities, Costs and Risks, edited by Thomas L. Saaty and Luis G. Vargas, 251-73. New York: Springer Science+Business Medical, LLC, 2005.

- Phua, Florence T.T., and Steve Rowlinson. "How Important is Cooperation to Construction Project Success? A Grounded Empirical Quantification." *Engineering, Construction and Architectural Management* 11, no. 1 (2004): 45-54.
- Pinto, J.K, and J.G. Covin. "Critical Factors in Project Implementation, a Comparison of Construction and R&D Projects." *Technovation* 9, no. 1 (1989): 49-62.
- Pocock, James Bryant. "The Relationship Between Alternate Project Approaches, Integration, and Performance." Ph. D. diss., Department of Civil and Environmental Engineering, University of Illinois at Urbana-Champaign, 1996. Doctor of Philosophy in Civil Engineering (1996).
- Pocock, James Bryant, Liang Y. Liu, and Michael K. Kim. "Impact of Management Approach on Project Interaction and Performance." *Journal of Construction Engineering and Management* December 1997: 411-18.
- Porter, Michael E., and Elizabeth Olmsted Teisberg. "Using Competition to Reform Healthcare." *Harvard Business School Working Knowledge* 5 June 2006. President and Fellows of Harvard College. http://hbswk.edu/cgi-bin/print?id=5369. Accessed 2 November 2010.
- Post, Nadine M. "Job Collaboration Raises Many Issues." *ENR: Engineering News-Record* 260, no. 18 (2 June 2008): 10-11.
- The Brooks Act: Federal Government Selection of Architects and Engineers *Public Law 92-582* (1972) (92nd Congress, H.R. 12807, October 27, 1972).
- Putri, Nilda Tri, and S.M. Yusof. "Critical Success Factors for Implementing Quality Engineering Tools and Techniques in Malaysian's and Indonesian's Automotive Industries: An Exploratory Study." Presented at the International MultiConference of Engineers and Computer Scientists. Hong Kong: IMECS, 18 - 20 March 2009, 2009.
- Raiffa, Howard, John Richardson, and David Metcalfe. *Negotiation Analysis: The Science and Art of Collaborative Decision Making*. Cambridge, MA: The Belknap Press of Harvard University Press, 2002.
- Roberson, Paula K. et al. "Analysis of Paired Likert Data: How to Evaluate Change and Preference Questions." *Family Medicine Research Series* 27, no. 10 (November-December 1995): 1-7.

- Romp, Graham. *Game Theory: Introduction and Applications*. New York: Oxford University Press, 1997.
- Rose, Casey. "Beyond Blueprints: 3-D Saves Time, Money." Article. *Boston Globe, Globe Newspaper Company* (Boston, MA), April 25 2011.
- Rothenberg, Jerome. "Problems in Behavior Under Risk: Individual Vs. Group Decision Making." In *Compromise, Negotiation and Group Decision*, edited by Bertrand R. Munier and Melvin F. Shakun, 9-26. Boston, MA: D. Reidel Publishing Company, 1988.
- Shapley, L.S., and L.K. Raut. *Theory of Games and Its Applications to Economics and Politics*. Delhi: Macmillan India Limited, 1981.
- Skulmoski, Gregory J., Francis T. Hartman, and Jennifer Krahn. "The Delphi Method for Graduate Research." *Journal of Information Technology Education* 6 (2007). Journal of Information Technology Education. http://informationscience.org/jite/documents/ Vol6/JITEv6p001-021Skulmoski212.pdf. Accessed 6 August 2010.
- Smith, Valerie Rose Riecke. "Impact to Alternative Contracting Methods Using Multivariate Analysis in the Regulatory Environment." Ph. D. diss., Georgia Institute of Technology, 2008. *Ph.D.* (2008).
- Strong, Norman, AIA Vice President 2005-2007 -- Chair, Integrated Practice Discussion Group.
 "0 Introduction." In *Report on Integrated Practice*. American Institute of Architects, 1.
 Seattle, Washington: American Institute of Architects, 2007. Special Report.
- Stumpf, Annette, U.S. Army Corps of Engineers. "Army Sustainability Policy." In Infrastructure Systems Conference: Building National Technical Competency. April 16, 2010. Cleveland, OH, July 21, 2009.
- Sweet, Justin, and Mark M. Schneier. *Legal Aspects of Architecture, Engineering, and the Construction Process.* 1970. 8th ed. Stamford, CT: Cengage Learning, 2008.
- The American Institute of Architects. *Integrated Project Delivery: A Guide*. AIA National & AIA California Council, 2007.
- The American Institute of Architects, California Council. *Handbook on Project Delivery*. Sacramento, CA: AIA California Council, 1996.

- The Department of Defense (DoD) Acquisition, Technology and Logistics (AT&L). *Defense Procurement and Acquisition Policy*. 3 Aug 2009. Office of the Secretary of Defense (OSD). http://www.acq.osd.mil/dpap/index.html.
- Department of Defense General Services Administration, and National Aeronautic and Space Administration, Federal Acquisition Regulation (FAR) VOLUME I—PARTS 1 TO 51 (, 3 Aug Sess.2005).
- Thomas, Janice L., and Mark Mullaly. "Explorations of Value: Perspectives of the Value of Project Management." *Project Management Journal* 40, no. 1 (March 2009): 2-3.
 ———. "Understanding the Value of Project Management: First Steps on an International Investigation in Search of Value." *Project Management Journal* 38, no. 3 (September 2007): 74-89.
- Thomsen, Chuck et al. *Managing Integrated Project Delivery*. Http://Cmaanet.Org/Files/Shared/ Ng_Integrated_Project_Delivery_11-19-09_2_.Pdf. McLean, VA: Construction Management Association of America, 19 November, 2009. June 15, 2010.
- Military Construction and Military Family Housing *Title 10 Armed Forces United States Code* § Title 10 USC Chapter 169 (2009).
- Department of Defense General Services Administration, and National Aeronautic and Space Administration, Cong., Federal Acquisition Regulation (FAR) VOLUME II -- Parts 52, 53, & Index (TITLE 48—FEDERAL ACQUISITION REGULATIONS SYSTEM 2005).
- U.S. Army Corps of Engineers. "Early Contractor Involvement (EIC)." In *Industry Workshop*. 7 March 2010. New Orleans, LA: U.S. Army Corps of Engineers, 28 January, 2009.
- U.S. Army Corps of Engineers, Norfolk District. Fort Belvoir Community Hospital Lessons Learned Report Phase I: Pre-Design, Design, and Construction. Special Report. Norfolk, VA, 2010.
- U.S. Army Corps of Engineers. *NGA New Campus East Project Lessons Learned 2007-2011*. Special Report. Fort Belvoir, VA: U.S. Army Corps of Engineers, 2011.
- U.S. Department of Defense. *Unified Facilities Criteria (UFC)* 4-510-1 Design: Medical Military Facilities. Department of Defense, 19 November, 2009.

- U.S. Department of Energy, Federal Energy Management Program, In Collaboration with the Interagency Sustainability Working Group. *The Business Case for Sustainable Design in Federal Facilities*, http://www1.eere.energy.gov/femp/program/ sustainable_businesscase.html, April 14, 2010. Washington, D.C.: U.S. Department of Energy, 2003.
- U.S. Department of Energy, Federal Energy Management Program, In Collaboration with the Interagency Sustainability Working Group. *The Business Case for Sustainable Design in Federal Facilities Resource Document*, http://www1.eere.energy.gov/femp/program/ sustainable_businesscase.html, April 14, 2010. Washington, D.C.: U.S. Department of Energy, 2003.
- U.S. Green Building Council. *Guiding Principles*. 2006. U.S. Green Building Council. http://communicate.usgbc.org/usgbc/2006/08.15.06_guiding_principles/guidingPrinciples /. Accessed 14 April 2010.
- "US. v. Spearin, 248 U.S. 132," 1918.
- Vega, Arturo Olvera. "Risk Allocation in Infrastructure Financing." *Journal of Project Finance* 3, no. 2 (Summer 1997): 38-42.
- Vitruvius. *The Ten Books on Architecture*. 1914. Translated by Morris Hickey Morgan. New York: Dover Publications, 1st Century B.C.
- Weirich, Pual. *Collective Rationality: Equilibrium in Cooperative Games*. New York: Oxford University Press, 2010.
 - ——. *Equilibrium and Rationality: Game Theory Revised by Decision Rules*. New York: Cambridge University Press, 1998.
- Westbrook. "Integrated Project Delivery IPD: More Than Design-Build."Www.leanconstruction.org presented at the Introduction to Lean Project Delivery Annual Lean Project Congress. Berkeley, CA, 2 August, 2002.
- White, John A., Marvin H. Agee, and Kenneth E. Case. *Principles of Engineering Economic Analysis, Third Edition.* New York: Wiley Publishing, Inc., 1989.
- White, John A. et al. *Principles of Engineering Economic Analysis, Fourth Edition*. New York: Wiley Publishing, Inc., 1998.

- Williams, Terry, and Knut Samset. "Issues in Front-End Decision Making on Projects." *Project Management Journal* 41, no. 2 (April 2010): 38-49.
- Yeung, John F. Y., Albert P. C. Chan, and Daniel W. M. Chan. "Developing a Performance Index for Relationship-Based Construction Projects in Australia: Delphi Study." *Journal* of Management in Engineering 25, no. 2 (April 2009): 59-68.
- Yousuf, Muhammad Imran. "Using Experts' Opinion Through Delphi Technique." *Practical Assessment, Research & Evaluation* 12, no. 4 May 2007. http://pareonline.net/pdf/v12n4.pdf. Accessed 09 September 2010.
- Zhai, Li, Yanfei Xin, and Chaosheng Cheng. "Understanding the Value of Project Management From a Stakeholder's Perspective: Case Study of Mega-Project Management." *Project Management Journal* 40, no. 1 (March 2009): 99-109.

AUTHOR'S BIOGRAPHY

Lieutenant Colonel Michael D. Brennan is an active duty Army officer. Brennan graduated from the University of Notre Dame with a Bachelor of Architecture degree and was commissioned through ROTC as a Medical Service officer in 1992. His first duty assignments were in the 3d Infantry Division serving as Medical Platoon Leader in the 1st Battalion, 15th Infantry in Schweinfurt, Germany, and as Division Medical Supply Officer and Executive Officer of Foxtrot Company, 703d Main Support Battalion in Kitzingen, Germany. In 1995 he was assigned as Architect in the Europe Region Medical Command's Health Facility Project Office in Heidelberg, Germany. In 1997 Brennan returned stateside and was reassigned to the Pentagon serving as Deputy Director, Medical Military Construction (MILCON), and later as Chief, Programs Evaluation and Analysis Branch at the U.S. Army Health Facility Planning Agency. In 1999 he deployed to Kosovo leading a team to design the hospital at Camp Bondsteel. In 2000 Brennan was reassigned overseas to Korea where he served as Commander, Charlie Company, 302d Forward Support Battalion, 2d Infantry Division at Camp Casey, and later as Chief Materiel Branch at the 121st General Hospital in Seoul.

In 2002 Brennan was selected for long-term civilian graduate studies at the University of Illinois, where in 2004 he graduated with both a Master of Architecture degree and a Master of Science degree in Civil Engineering. After completing Command and General Staff College, Lieutenant Colonel Brennan was assigned to duty at the Walter Reed Army Medical Center where as Chief of the Health Facility Project Office he oversaw completion of over \$100 million of construction projects to include the Military Amputee Training Center. He oversaw the Base Realignment and Closure (BRAC) planning and programming for the \$2.3 billion relocation of the Walter Reed Army Medical Center to Bethesda, Maryland and Fort Belvoir, Virginia. In

324

2006 Lieutenant Colonel Brennan deployed to Iraq serving as 3d MEDCOM Deputy G4, and Health Facility Planner responsible for theater-wide medical facility projects including construction of the Multi-National Forces Hospital at Al Asad. Following Iraq, he returned to the U.S. Army Health Facility Planning Agency as Program Manager for the \$1.2 billion, 1.3 million square foot Fort Belvoir Community Hospital project. In 2008 he returned to the University of Illinois to start a Ph.D. program in the School of Architecture.

He is married to Lieutenant Colonel (retired) Mia Brennan, and they have four young children. Upon completion of his Ph.D. Lieutenant Colonel Brennan will continue on activeduty with the U.S. Army Health Facility Planning Agency.